NHPC LIMITED (A GOVT. OF INDIA NAVRATNA ENTERPRISE)



REQUEST FOR SELECTION (RfS) DOCUMENT (DOMESTIC COMPETITIVE BIDDING-OPEN TENDER) UNDER TARIFF BASED COMPETITIVE BIDDING WITH E – REVERSE AUCTION

FOR

"Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding"



Tender ID: 2025_NHPC_842440_1

CONTRACTS (CIVIL) DIVISION REGD. OFFICE: NHPC OFFICE COMPLEX, SECTOR-33, FARIDABAD – 121003

CIN: L40101HR1975GOI032564

TABLE OF CONTENTS

| Volumes | DESCRIPTION | | | | |
|---------|---|--|--|--|--|
| | Check List | | | | |
| | Disclaimer | | | | |
| Vol.0 | NOTICE INVITING TENDER (NIT) | | | | |
| | (I) Bidding Data Sheet | | | | |
| | (II) Instructions to Bidders (ITB) | | | | |
| | Annexure -1: Form of Declaration | | | | |
| | Annexure -2: Format of Integrity Pact | | | | |
| | Annexure -3 - Qualification Information Sheets | | | | |
| | 3A: General Information | | | | |
| | 3B: Summary of Work Experience | | | | |
| | 3C: SELF CERTIFICATE FOR LOCAL CONTENT BY BIDDER | | | | |
| | Annexure-4: Form of Declaration of Ineligibility | | | | |
| | Annexure-5A: Deleted | | | | |
| | Annexure-5B: Deleted | | | | |
| | (III): PROCEDURE FOR E-REVERSE AUCTION (e-RA) Annexure -6 : Conditions of conducting e-RA after e-tendering Annexure -7: Business Rule for e-RA Annexure -8: Process Compliance form for e-RA Annexure -9: Price Confirmation form for e-RA | | | | |
| Vol.1 | Information for Bidders (IFB) | | | | |
| Vol.2 | Definition & Standard Copy of Power Purchase Agreement | | | | |
| Vol.3 | Special Conditions of Contract (SCC) alongwith Technical Specification | | | | |
| Vol.4 | Forms and Procedures (FP) | | | | |
| | Format of Covering Letter | | | | |
| | Bid Security Form (Bank Guarantee) | | | | |
| | Format for POI for Earnest Money Deposit | | | | |

| Volumes | DESCRIPTION | | | | |
|---------|--|--|--|--|--|
| | Format of Insurance Surety Bond for Earnest Money Deposit | | | | |
| | Performance Security Forms | | | | |
| | Format for POI for Performance Bank Guarantee | | | | |
| | FORMAT OF INSURANCE SURETY BOND FOR PERFORMANCE SECURITY | | | | |
| | Format for Disclosure alongwith Annexure | | | | |
| | CERTIFICATE FOR COMPLIANCE TO ALL PROVISIONS OF RfS DOCUMENT | | | | |
| | DECLARATION REGARDING INSOLVENCY PROCEEDINGS UNDER INSOLVENCY AND BANKRUPTCY CODE 2016 | | | | |
| | Self-Certificate for Cyber Security | | | | |
| | PRICE SCHEDULES | | | | |
| | ECS Form | | | | |
| | Proforma of certificate from the CEO/CFO of the holding company | | | | |
| | Proforma of undertaking | | | | |
| | Proforma of certificate from the CEO/CFO in accordance with item clause 3.2.B.3 of eligibility criteria | | | | |
| | Joint letter of undertaking between the CEO of the holding company / group company or subsidiary company & CEO of the bidder company | | | | |
| | Bidders undertaking (on Letter head of the Company) mentioning the requisite Category for which Bid(s) are submitted | | | | |
| | Undertaking from parent/ holding company | | | | |
| | Parent/ holding company agreement | | | | |
| | Parent/ Holding Company Agreement (In case of Joint Venture/Consortium) | | | | |

CHECK LIST

Bidders are requested to check the following documents in their submissions prior to submitting their bids:

| Sl. No. | Description | Check |
|---------|---|--------|
| 1 | Categorywise DD towards cost of RFS document as mentioned in NIT. | YES NO |
| 2 | Categorywise DD/BG/POI/Insurance Surety Bond towards Bid security as per ITB | YES NO |
| 3 | Certified copy of the partnership deed duly registered, as per ITB, if applicable. | YES NO |
| 4 | Power of Attorney along with Board resolution for authorizing the executants for issuing Power of Attorney as per ITB, if applicable. | YES NO |
| 5 | Form of Declaration (Annexure-1) duly filled in | YES NO |
| 6 | Duly signed Integrity Pact (Annexure-2) | YES NO |
| 7 | Duly filled Qualification Information Sheet (Annexure-3A, 3B& 3C) | YES NO |
| 8 | Form of Declaration of Ineligibility (Annexure-4) duly filled in | YES NO |
| 9 | Copies of successful completion certificate(s) of works/assignments | YES NO |
| 10 | Annual Reports for last three financial years prior to last date of bid submission as per ITB. | YES NO |
| 11 | Technical & Financial Details as required | YES NO |
| 12 | Forms & Formats duly completed and signed by the Bidder, as per ITB | YES NO |
| 13 | Duly Filled Annexure -8: Process Compliance form for e-RA | YES NO |

| 14 | <u>Duly Filled Annexure-9:</u> Price Confirmation form for e-RA | YES | NO |
|----|---|-----|----|
| 15 | Bidders undertaking (on Letter head of the Company) mentioning the requisite Category for which Bids are submitted. | YES | NO |
| 16 | <u>Applicable Formats</u> | YES | NO |

Note: The Checklist has been framed to facilitate the Bidders so that they may not miss the important inclusion in their bids as well as to facilitate the Employer to avoid delays. Bidders are advised in their own interest to go through entire document thoroughly so that documentation is complete in all respect. Employer shall not be responsible in case the Bidder fails to submit all requisite documents along with its bid.

DISCLAIMER

- Though adequate care has been taken while preparing the Request for Selection (RfS) document, the Bidder shall satisfy himself that the document is complete in all respects. Intimation of any discrepancy shall be given to this office immediately. If no intimation is received from any bidder within 20 (Twenty) days from the date of issuance of RfS documents, it shall be considered that the document is complete in all respect and has been received/ acknowledged by the bidder
- 2. NHPC Limited (NHPC) reserves the right to modify, amend or supplement this RfS document.
- 3. Though this RfS document has been prepared in good faith, neither NHPC nor its employees or advisors make any representation or warranty, expressed or implied, or accept any responsibility or liability, whatsoever, in respect of any statements or omissions herein, or the accuracy, completeness or reliability of information, and shall incur no liability under any law, statute, rules or regulations as to the accuracy, reliability or completeness of this RfS document, even if any loss or damage is caused by any act or omission on their part.
- 4. In case of any discrepancy in the documents uploaded on the websites of NHPC and CPP portal, the documents uploaded on the CPP Portal will prevail.

Authorized Person General Manager (Civil Contracts-I) Room No.218, 2nd Floor, NHPC OFFICE COMPLEX, Sector-33, Faridabad-121003, Haryana, India Tele No: +91 (129) 2270596 <u>Email: contcivil1-co@nhpc.nic.in</u>

VOLUME – 0

Notice Inviting Tender (NIT), Bidding Data Sheet

and

Instruction to Bidders (ITB)

| | TABLE OF CONTENTS | | | | |
|-------------|------------------------------|---|--|--|--|
| NOTICE INV | NOTICE INVITING TENDER (NIT) | | | | |
| (I) BIDDING | I) BIDDING DATA SHEET | | | | |
| (II) INSTRU | CTIONS | TO BIDDERS | | | |
| A. |] | ntroduction | | | |
| | 1. | Scope | | | |
| | 2. | Eligible Bidders | | | |
| | 3. | Qualification of the Bidder | | | |
| | 4. | Cost of Bidding | | | |
| | 5. | Site Visit | | | |
| В. |] | The RFS Documents | | | |
| | 6. | Content of RFS Documents (or Bid Documents) | | | |
| | 7. | Clarification of RFS Document | | | |
| | 8. | Amendment of RFS Documents | | | |
| C. | J | Preparation of Bids | | | |
| | 9. | Language of Bid | | | |
| | 10. | Documents Comprising the Bid | | | |
| | 11. | Bid Prices | | | |
| | 12. | Bid Currencies | | | |
| | 13. | Period of Bid Validity | | | |
| | 14. | Bid Security | | | |
| | 15. | Format and Signing of Bid | | | |
| D. | Submiss | sion of Bids | | | |
| | 16. | Sealing and Marking of Bids | | | |
| | 17. | Deadline for Submission of Bids | | | |
| | 18. | Late Bids | | | |
| | 19. | Modification and Withdrawal of Bids | | | |
| E. | | ening and Evaluation | | | |
| | 20. | Opening of Bids by Employer | | | |
| | 21. | Clarification of Bids | | | |
| | 22. | Preliminary Examination of Bids | | | |
| | 23. | Techno-commercial Evaluation | | | |
| | 24. | Price Bid Evaluation | | | |
| | 25. | Deleted. | | | |

| F. | Award o | Award of Contract | | | | |
|------------------|--|---|--|---------|--|--|
| | 26. | Contacting the Employer | | | | |
| | 27. | Award Criteria | | | | |
| | 28. | Employer's Right to Accept Any Bid and to Reject Any or All Bids | | | | |
| | 29. | Notification | Notification of Award | | | |
| | 30. | Signing of C | ontract Agreement | | | |
| | 31. | Performance | Security | | | |
| | 32. | Corrupt or F | raudulent Practices | | | |
| | 33. | Integrity Pac | t | | | |
| | 34. | Procurement | from countries sharing land border wit | h India | | |
| | 35. | Public Procurement (Preference to Make in India) Order | | | | |
| | 36. | Planning and Designing in purview of Vulnerability Atlas of India | | | | |
| | 37. | Cyber Security for Power System Equipments | | | | |
| G. | | | | | | |
| | | re 1. Form of Declaration | | | | |
| | | ure 2. Format of Integrity Pact | | | | |
| | Annexure | e 3. Qualificati | on Information Sheets | | | |
| | Annexure | e 3A | General Information | | | |
| | Annexure | e 3B | Summary of Work Experience | | | |
| | Annexure | re 3C SELF CERTIFICATE FOR LOCAL CONTENT BY BIDDER | | | | |
| | Annexur | ure 4. Form of Declaration of Ineligibility | | | | |
| | Annexur | are 5A. Deleted | | | | |
| | Annexure 5B: Deleted | | | | | |
| (III): PROCE | DURE FOR E-REVERSE AUCTION (e-RA) Annexure -6 : Conditions of conducting e-RA after e-tendering Annexure -7: Business Rule for e-RA Annexure -8: Process Compliance form for e-RA | | | | | |
| \triangleright | Annexur | Annexure -9: Price Confirmation form for e-RA | | | | |



NHPC Office Complex, Sector-33, Faridabad-121003

NOTICE INVITING TENDER Request for Selection (RfS) (Domestic Competitive Bidding-Open Tender) under Tariff Based Competitive Bidding with e – Reverse Auction

Tender ID: 2025_NHPC_842440_1

 Online RfS are invited through Domestic Tariff Based Competitive Bidding in Single Stage-Two Part Bidding Basis (i.e. Part-I:QR+Technical-Bid and Part-II:Financial Bid) with <u>e-Reverse Auction (e-RA)</u> by NHPC Limited on behalf of NHPC Renewable Energy Limited (NHPC REL)- a wholly owned subsidiary of NHPC Limited {herein after referred to as Project Management Consultant (PMC) for the RESCO Project set up by Roof Top Developer (RTD)} from eligible Bidders for "Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding"

Complete RfS Documents/Tender Document can be viewed and downloaded from Central Public Procurement (CPP) Portal http://eprocure.gov.in/eprocure/app. The site can also be viewed through e-procurement corner of NHPC website www.nhpcindia.com and CPP Portal. Any Bidder who wishes to quote for this Tender can download the Tender Document from aforesaid portal after online Bidder registration for e-tendering. However, the bid is to be submitted online only on <u>http://eprocure.gov.in/eprocure/app</u> up to last date and time of submission of tender. Sale of hard copy of tender document is not applicable.

E-Reverse Auction (e-RA) shall be conducted after e-tendering. The decision of NHPC regarding adoption of e-RA shall be final.

| Sl. No. | Item | Description | |
|---------|------------------|--|--|
| i) | Mode of | e-procurement System | |
| | tendering | Cover-I: Online Techno-Commercial Bid | |
| | | Cover-II: Price Bid | |
| ii) | Tender ID No. | 2025_NHPC_842440_1 | |
| iii) | Tender reference | NH/CCW/CC-I/CO-361/PR61324/7 Dt. 01.01.2025 | |
| | No. | | |
| iv) | Cost of bid | Rs.5,000/- (Rupees Five thousand only) - for Each Category | |
| | document | seperatly | |
| | | Cost of bid document shall be in the form of demand draft from a Nationalized / Scheduled bank in favour of NHPC Ltd. Payable | |

THE BRIEF DETAILS OF THE TENDER ARE AS UNDER:

| | | at Faridabad. | |
|-------|---------------------------|--|--|
| v) | Bid Security (EMD) | Rs 67,00,000/-(Rupees Sixty Seven Lakhs only) – for Category-1 | |
| | × , | Rs 89,00,000/-(Rupees Eighty Nine Lakhs only)- for Category-2 | |
| | | Rs 1,02,00,000/-(Rupees One Crore Two Lakhs only)- for Category-3 | |
| | | Bid Security shall be in the form of DD/BG/POI/Insurance Surety Bond as per clause No. 14 of ITB. | |
| vi) | Period of Bid Validity | 120 days from the last date of submission of online Bids | |
| vii) | Commissioning | Commissioning Period of "Selection of Roof Top Developer | |
| | Period | for Setting up Solar Roof Top Plants at Various | |
| | | Departments of Law, Justice & Parliamentary Affairs, | |
| | | Jal Shakti, Irrigation & Flood Control, Forest, Ecology | |
| | | & Environment and Home in State of J&K through | |
| | | JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding" | |
| | | shall be 09 (Nine) months from the date of signing of PPA | |
| viii) | Tender inviting | General Manager (Civil Contracts-I) | |
| | Authority | Room No218, 2nd Floor, Jyoti Sadan, | |
| | | NHPC OFFICE COMPLEX, | |
| | | Sector-33, Faridabad-121003, Haryana, India | |
| | | Tele No :+91 (129) 2270596 | |
| | | Email: <u>contcivil1-co@nhpc.nic.in</u> | |
| I | | | |

THE CRITICAL DATES OF TENDER ARE AS UNDER:

| S. No. | Particulars | Date & Time |
|--------|--|---|
| i) | Publishing Date & Time | 17.01.2025 (18.30 Hrs) |
| ii) | Document Download Start Date & Time | 17.01.2025 (18.30 Hrs) |
| iii) | Last date of Reciept of queries /clarification on bid. | 28.01.2025 (upto 11:00 Hrs) |
| iv) | Pre bid meeting Date & Time | 30.01.2025 (15:00 Hrs) |
| v) | Online Bid Submission Start Date & Time | 12.02.2025 (11:00 Hrs) |
| vi) | Online Bid Submission Closing Date & Time | 21.02.2025 (17:30 Hrs) |
| vii) | Last date of Offline submission (date, time and address) | 24.02.2025 (upto 17:00 Hrs) <u>Address</u> : General Manager (Civil Contracts-I) Room No.218, 2nd Floor, Jyoti Sadan, NHPC OFFICE COMPLEX, Sector-33, Faridabad-121003, Haryana, India Tele No :+91 (129) 2270596 Email: contcivil1-co@nhpc.nic.in |
| viii) | Opening: Date, Time & Venue of Online/ Offline Bid (Technical Bid (Part-I) submission | 25.02.2025 (15.00 Hrs) <u>Venue</u> : General Manager (Civil Contracts-I) Room No.214, 2nd Floor, Jyoti Sadan, |

| ix) | Price Bid Opening (Part-II) – Financial Bid | Venue, Date & time to be intimated later to the bidders whose Techno-commercial Bids will be |
|-----|--|---|
| | | found responsive |
| x) | Date & Time of Start of e- | Shall be intimated separately by NHPC |
| | Reverse Auction for each | |
| | Category | |

2. ELIGIBLE BIDDERS:

- 2.1 This Invitation for Bid is open in respect of bidders who meet the following conditions:
 - a) The bidder must be an Indian incorporated legal Entity operating under commercial law of their respective jurisdiction.
 - b) all bidders meeting the qualification criteria as defined in ITB clause 3.2
- 2.2 Bidders shall not be under a declaration of ineligibility for corrupt and fraudulent practices by the Employer, any Government institution or Public Sector Undertaking in India in accordance with ITB clause 32.1(a), as on bid submission date.
- 2.3 The Bidders whose contract(s) have been terminated due to poor performance by the Employer, shall not be allowed to participate in the bidding process for next 5 years w.e.f the date of notification of termination.
- 2.4 Bidders shall not have been banned/ de-listed/ black listed/ debarred from business on the ground mentioned in para 6 of Guidelines on Banning of Business dealings (Annexure-2A) to Integrity Pact (Annexure-2 of ITB) of Tender Document at the time of submitting the bid and shall intimate immediately of any such banning/de-listing/ black- listed/debarring after submission of bid till the award of the tender. Self-Declaration in this regard is to be submitted as per the proforma enclosed in tender document.
- 2.5 To qualify for Contract for which bids are invited in the Notice Inviting Tender, the bidder must demonstrate having work experience, financial capability and resources sufficient to meet the aggregate of the qualifying criteria. Failure to produce the certificates and documents in this regard shall make the bid non-responsive.

Even though the bidders meet the qualifying criteria, they are subject to be disqualified if they have made misleading or false representations in the forms, statements, declarations and attachments submitted in proof of the qualification requirements.

3. QUALIFICATION OF THE BIDDER

- **3.1** All bidders shall include the following information and documents with their bids in Qualification Information unless otherwise stated in the ITB:
- a) Written power of attorney of the signatory of the Bid to commit the Bidder. Additional information as sought in the Annexure-3A (Qualification Information Sheets of ITB) shall be provided.
- b) Work experience to demonstrate meeting the criteria stipulated in clause 3.2.A shall be provided in Annexure-3 (3B: Qualification Information Sheets).The work experience shown shall be supported with certificate(s) from the Engineer-in-charge/Project head of the concerned work. In case of experience certificate produced by the Contractors for having works for Private Organizations, TDS Certificates shall also be produced along with experience certificate.

c) Information on financial criteria stipulated in clause 3.2 B shall be furnished along with Copy of affidavit/Certificate of CA mentioning Financial Turnover of immediately preceding 3 (three) years. Printed Annual reports or financial statements of the Bidder, such as balance sheet, profit and loss statements and auditor's reports as the case may be for the immediately preceding 3 (Three) years shall be submitted to ascertain bidder's meeting the financial criteria.

3.2. QUALIFICATION CRITERIA

The qualification will be subject to Bidder's fulfillment of the Qualification Criteria set and stipulated hereunder, substantiated by authentic and relevant information and details. Additional information in support of their claims of achievements may be furnished in any form of their device and design. The Bidder(s) who wish to apply for Tender for the aforesaid work must satisfy the following qualifying criteria:

- **3.2. A** The bidder should fulfill Technical Qualifying criteria as mentioned below:
- 3.2.A.1 The Bidder should have designed, supplied, erected and commissioned Solar Photo Voltaic (SPV) based grid connected power plant(s) of cumulative installed capacity of 400 kWp or higher, out of which at least one plant should have been of 40 kWp or more.

OR

3.2.A.2 The Bidder should be a developer of Solar Photo Voltaic (SPV) based grid connected power plant(s) of cumulative installed capacity of 400 kWp or higher, out of which at least one plant should have been of 40kWp or more.

OR

3.2.A.3(a) The Bidder should have executed in the last ten (10) years either as developer or as EPC Contractor in process industry or in infrastructure development of single work having minimum value of Rs. 4.0 Crores.

AND

(b) The Bidder should have executed at least one (1) Electrical Sub-station of 11 kV or above voltage level, consisting of equipment such as 11 kV or above voltage level circuit breakers and Power transformer, either as developer or as EPC Contractor which should be commissioned up to the last date of submission of bid.

The works referred to clause 3.2.A.3 (a) &3.2.A.3 (b) can be in same or different projects.

Notes:

- a. The reference SPV based grid connected power plant of 40 kWp or higher capacity should be at a single location developed by Bidder for itself or any other client.
- b. SPV based Roof-top/Ground mounted/Floating solar power projects, which are grid connected, shall be considered eligible for QR purposes.
- c. Bidder shall submit certificate of successful completion and operation from the Owner.
- d. In case the award for the reference works has been received by the Bidder either directly from owner of plant or any other intermediary organization, a certificate from such

owner of plant or the intermediary organization shall be required to be furnished by the Bidder along with its bid document in support of its claim of meeting requirement stipulated above.

- e. Developer means an entity who has either executed or got executed the work/ project as owner of industrial projects.
- f. The qualification criteria in Clause 3.2.A.1 or 3.2.A.2 or 3.2.A.3 are mutually exclusive. The bidder has to fulfil any one of the above-mentioned technical eligibility criteria which are individually exclusive.
- g. Bidder can also claim the experience of a Group Company/Holding company /Subsidiary Company for meeting the requirement (s) of clause 3.2.A.1 or 3.2.A.2 or 3.2.A.3. In such a case, the Bidder shall furnish an Undertaking jointly executed by the Firm qualified as per Clause 3.2.A.1 or 3.2.A.2 or 3.2.A.3 and the Bidder along with its bid document for complete performance of the Contract (in case of award) jointly or severally, as per the format enclosed in the bid document, failing which, the Bidder's bid document is liable to be rejected.
- h. Experience of bidders as EPC Contractor for supply of Balance of System (BOS) items and erection, testing, commissioning of entire solar plant shall be considered.
- i. To ascertain the executed value of work for Cl. 3.2.A.3(a), the value of Work experience in INR shall be enhanced on pro-rata basis by 7% simple rate per annum up to the date of Bid Opening from the completion date mentioned in the completion certificate.
- j. Capacity below 10kWp at single installation shall not be considered in cumulative capacity.

3.2.B Financial Criteria

| 3.2.B.1 Turnover: A | Average Annual | Turnover in the | preceding 3 years | shall be not less than | as under: |
|---------------------|----------------|-----------------|-------------------|------------------------|-----------|
| | | | | | |

| Category | Turnover |
|--|-----------------|
| Category-1 for (>= $5kW$ to $< 20kW$) | Rs. 17.83 crore |
| Category-2 for (≥ 20 kW to < 100 kW) | Rs. 23.69 crore |
| Category-3 for (>= 100kW and above) | Rs. 27.13 crore |

- **3.2.B.2 Networth:** The bidder should have positive "Net Worth" in at least 02 financial years out of the last 03 financial years, with the condition of positive net worth in immediately preceding financial year.
 - 'Net Worth' means the aggregate value of the paid-up share capital and all reserves created out of the profits, securities premium account and debit or credit balance of profit and loss account, after deducting the aggregate value of the accumulated losses, deferred expenditure and miscellaneous expenditure not written off, as per the audited balance sheet, but does not include reserves created out of revaluation of assets, write-back of depreciation and amalgamation.

For assessment, relevant pages of Annual Report containing Net Worth figure duly certified by Authorized representative of Bidder, otherwise, figure duly authenticated by the Statutory Auditors should be submitted.

| 3.2.B.3 | Working | Capital: | The | working | capital | (current | assets | minus | current | liabilities) | shall I | be as |
|---------|---------|----------|-----|---------|---------|----------|--------|-------|---------|--------------|---------|-------|
| | under | | | | | | | | | | | |

| Category | Working |
|--|-----------------|
| | Capital |
| Category-1 for ($\geq 5kW$ to $< 20kW$) | Rs. 9.91 crore |
| Category-2 for (≥ 20 kW to < 100 kW) | Rs. 13.16 crore |
| Category-3 for (>= 100kW and above) | Rs. 15.07 crore |

For this purpose current assets and current liabilities will be considered as classified in the audited balance sheet for the year immediately preceding the date of opening of bids. In case current assets and/or current liabilities are not classified separately in audited Balance Sheet, a certificate from Statutory Auditor/CPA carrying out the Statutory Audit, for current assets and/or current liabilities, as the case may be, clearly defining the items considered for the same, should be enclosed. However, trade payables shall be a part of current liabilities. If the Working Capital calculated from the audited Balance Sheets is negative, then such working capital shall be treated as zero.

Notes:

- 1. Other income shall not be considered for arriving at Annual Turnover.
- 2. The Bidder's financial evaluation vis-à-vis the requirement as stipulated above shall be done on the basis of duly printed (offset) Annual Report/Equivalent Financial Statement for the immediately preceding Three years submitted by the Bidder along with the Bid. In case, Bid submission date falls upto 30th September, the annual reports/ equivalent financial statement upto previous year of immediate preceding Financial Year shall be considered and for the Tenders having Bid submission date after September, the annual reports/ equivalent Financial statement upto immediate preceding Financial year shall be considered. Further, standalone audited Annual Financial Statement of Bidder shall be forming part of the Annual Report/ equivalent financial statement.

In case, if bidder has not submitted the above Annual Report/ Equivalent Financial Statement along with Bid, then a certificate from CEO/CFO of the Bidder shall be submitted along with Bid mentioning that the requirement of Annual Report as per governing law of country is not mandatory in case of such bidders Equivalent Financial Statement shall be considered. In such cases, duly notarized copies of Audited Printed Annual Financial Statement (Balance Sheet, Profit & Loss Statement, cash flow statement, Auditor 's Report thereon including all relevant Schedules/ annexures etc.) for

the immediately preceding 3(Three) years be submitted by the bidder along with the Bid. The Annual Financial Statement shall be considered for evaluation in same manner as defined in above para for consideration of Annual Reports/ Equivalent Financial Statement.

- 3. Wherever, the Annual Report/ duly self-certified/Attested copies of Audited Printed Annual Financial Statement are in language other than English, then copy duly translated & printed into English language and certified by approved / recognized English translator shall be submitted with the Bid.
- 4. For the Purpose of compliance to the stipulated turnover criteria given at Financial Criteria, the construction turnover from Joint Ventures(s) as declared in the Joint Venture Agreement shall also be considered.

The proportionate JV Turnover shall be certified by their Chartered Accountant in the case of the same not appearing in Audited Financial Statement.

- 5. The qualification given by Statutory Auditors on the Financial Statements which have impact on Net Worth, Turnover, Working Capital or any other financial item shall be taken into account for calculating the above financial parameters of Qualifying Criteria.
- 6. In case where Audited financial results for the immediately preceding year are not available, then an statement of account as on closing date of the immediately preceding financial year depicting the Turnover and Net worth (calculated as per laid down criteria) duly certified by their Statutory Auditor/Certified Public Accountant carrying out the statutory audit shall be enclosed with the Bid along with copy of appointment letter of the statutory auditor. The Certificate should be issued with Unique Document Identification Number (UDIN) as per Gazette Notification No. 1-CA(7)/192/2019 dated 02.08.2019 or any other document identification number issued to Certified Public Accountants by their respective Institute, if applicable.

7. The bidders who intend to quote for more than one Category, have to fulfil the total financial criteria required for each Category cumulatively.

3.2. B.4 Bids of any insolvent firm:

The Bidder or partner of JV against whom an application for initiating corporate insolvency resolution process has been admitted by the Adjudicating Authority under the Insolvency and Bankruptcy Code 2016, or as amended from time to time, shall not be eligible for bidding. The same shall also be applicable to the Bidder company, who has taken unconditional technical and/or financial support from their Parent/Holding Company; against whom an application for initiating corporate insolvency resolution process has been admitted by the Adjudicating Authority under the Insolvency and Bankruptcy Code 2016 (IBC 2016), or as amended from time to time.

In case, bidder or JV partner in respect of whom any application for initiating corporate insolvency resolution process was not admitted at the time of submission of bid, but

subsequently during the period of evaluation of bids or any time before the work is awarded, any such application is admitted by the Adjudicating Authority under the IBC 2016 (equivalent laws of the country where the firm has been incorporated/registered shall be applicable), the bidder shall be considered as ineligible and his bid shall be rejected.

An Undertaking that "no insolvency proceedings are admitted by the Adjudicating Authority against bidder or partner of JV under the IBC 2016 (equivalent laws of the country where the firm has been incorporated / registered shall be applicable)" shall be submitted on letter head duly signed by the Authorized representative of bidder or all the partners of JV.

Further, the Bidder, after submitting the bid till the time of award of work, shall inform NHPC regarding any admission of application for corporate insolvency resolution process by the Adjudicating Authority under the IBC 2016 against bidder or partner of JV and any suppression of such fact shall render the bidder liable for rejection of his bid and banning of business dealing as per terms and conditions of Guidelines of banning of business dealings under Integrity Pact.

3.3 **Nature of bidders**

The Bidder should be a Sole bidder for participating in the tendering for this Project.

For evaluation of specific construction experience of various bidders, following criteria shall be considered: -

- 1. The evaluation of Bidder's Specific Construction experience for a work executed on JV basis shall be based on his role and scope of work in such joint ventures. To establish his role and scope of work in such Joint Venture contracts, the Bidder shall provide documentary evidence such as JV agreement/contract of the work to the extent of experience claimed by him. In case the quantum of work as per experience certificate does not match with the scope of work as defined in the JV/consortium agreement, then the experience credentials shall be considered as per the experience certificate issued by the employer / client.
- 2. In case scope of work of individual partners is not clearly defined in the JV/consortium agreement then credential of bidder being partner in a JV/consortium shall be decided in the following manner:
 - a. Where specific experience certificate in respect of individual JV partners is available, the same shall be considered to the extent of work executed by such JV partner as per the experience certificate.
 - b. Where specific experience certificate is not available, or experience certificate is issued in the name of JV/consortium, the evaluation shall be done in the following manner:-
 - 1. In case the participation / profit sharing percentage of bidder as per JV agreement is at least 35%, full credit of the work executed by the JV/consortium shall be given to such bidder.
 - 2. In case the participation / profit sharing percentage of bidder in JV agreement is less than 35%, credit of work executed under such JV shall be given to such bidder on pro-rata basis.

3. Notwithstanding (i) and (ii) above, in case of availability of specific experience certificate, (a) above shall apply.

In case both participation share and profit sharing ratio are available in the JV Agreement, participation share shall prevail.

3. For the portion of work executed through a sub-contractor, full experience for that portion of work can be claimed by the Contractor as well as sub-contractor for fulfilling criteria for evaluation of bidder's specific experience.

3.4 Joint Venture/Consortium Bidders: Deleted

3.5 Deleted.

3.6 Bids by Merged/ Acquired/ Subsidiary Companies:

In case of a Bidder Company, formed after merger and/or acquisition of other companies, past experience and other antecedents of the merged/acquired companies will be considered for qualification of such Bidder Company provided such Bidder Company continues to own the requisite assets and resources of the merged/acquired companies needed for execution and successful implementation of the work package put to tender.

If the Bidder Company is a Subsidiary Company and applies for tender on the unconditional technical and financial strength of the Parent/Holding Company, the same shall be considered provided the Parent/Holding Company commits to sign a Separate Agreement with NHPC Limited, confirming full support for the Technical and Financial requirements of the Subsidiary Company and commits to take up the works itself in case of non-performance by the Subsidiary Company in the event of award of the works to the Bidder Subsidiary Company. An undertaking by the Parent/ Holding Company to this effect shall be submitted- along with the tender. A Subsidiary Company intending to qualify on the strength of Parent/ Holding Company shall not be allowed to participate as a 'Sub-Contractor'.

For the purpose stated herein above in this clause, 'Parent Company' shall mean the 'Holding Company' owning majority (more than 50%) shares of such Bidder (Subsidiary) Company. Similarly by extensions of this interpretation, if "A" is owned by a 'Holding Company' "B" which in turn is owned by another 'Holding Company' "C", then "C" is construed as the 'Parent Company' of "A" as well as "B" and so on. An apex 'Parent Company' may own number of independent Subsidiary/Group Companies and if any of these Subsidiary/Group Company commits assured support and unhindered access to its assets and resources to another Subsidiary/Group Company (Bidder in this case) under the same apex 'Parent Company' then experience and other credentials of such Subsidiary/Group Company shall be considered for qualification of the Bidder Subsidiary Company provided such commitment is evidenced/ authorized and guaranteed by the apex 'Parent Company'.

The experience of subsidiary companies of the Parent/ Holding Company will be considered experience of the Parent/ Holding Company.

However, for fulfilment of financial criteria, the Bidder's financial evaluation vis-à-vis the requirement as stipulated above shall be done on the basis of consolidated Annual Report for the

immediately preceeding 3 (three) years of the Parent Company/ Apex Parent Company submitted by the Bidder along with the Bid/Application.

- **3.7** Each bidder must also produce with their Bid:
 - i) PAN No. and EPF Registration No.
 - ii) GST Registration No.
 - iii) A declaration that the information furnished with the bid documents is correct in all respects in Annexure-1: Form of declaration
 - iv) Such other certificates, if any, as defined in the ITB.
- **3.8** To qualify for Contract for which bids are invited in the Notice Inviting Tender, the bidder must demonstrate having work experience, financial capability and resources sufficient to meet the aggregate of the qualifying criteria. Failure to produce the certificates and documents as required under clause 3.2 shall make the bid non-responsive.
- **3.9** Even though the bidders meet the above qualifying criteria, they are subject to be disqualified if they have made misleading or false representations in the forms, statements, declarations and attachments submitted in proof of the qualification requirements.

4. TIME FOR COMPLETION

The successful bidder shall complete the entire work within the time specified under S.No.-1(vii) of NIT.

5. Tenders must be accompanied by the Bid Security. The bids not accompanied by the requisite bid security shall be rejected as non-responsive.

6. Deleted

7. <u>PRE-BID MEETING</u>

a) A Pre-bid meeting open to all the prospective bidders will be held on the date mentioned at para 1 above at NHPC Office Complex, Sector-33, Faridabad, wherein they shall be given an opportunity to obtain clarifications, if any, regarding the work and bid conditions.

The bidders may also attend Pre-bid meeting through video conferencing on scheduled date and time or as amended. The bidders who wish to join the meeting through video-conferencing shall intimate Tender Inviting Authority (TIA) at least 02 (two) days prior to pre-bid meeting their details viz. name of participant & designation, mobile no., e-mail address, name of firm or any other information required for video-conferencing. The participant should have good internet connectivity, as TIA shall not be held responsible for any disruption due to internet or any technical issues.

Meeting ID alongwith password shall be shared 01 (one) hour prior to scheduled time to the participant's e-mail/ mobile no. of prospective bidder.

b) Prospective bidders may submit their queries, if any, by email / courier at address of Tender Inviting Authority at least three days before the pre-bid meeting so that the same can be replied during the meeting.

8. **BID SUBMISSION**

- i) Online Bid Submission –Technical Bid (Cover-I) (Refer ITB) and Category wise Price Bid (Cover-II) electronic format complete in all respect must be uploaded at the aforesaid portal by the date & time as per SI. No.1 above.
- ii) Offline Bid Submission (Refer ITB) complete in all respect must be delivered in sealed envelopes to the address upto the last date & timeof submission.

In the event of the specified date or amendment, if any, for the submission of bids being declared a holiday for the Employer, the hard copy of the documents will be received up to the specified time on the next working day. Similarly, in the event of the specified date or amendment if any for the opening of bids being declared a holiday for the Employer, the opening shall be carried out at the specified time on the next working day. However, the date and time for online submission of the Bids shall continue to be the date and time specified or amendment if any.

- 9. The currency for the Bid shall be Indian Rupee only.
- 10. Bids shall be valid for a period as mentioned in S No. 1 after the deadline for online Bid submission. If any Bidder withdraws his Bid before the said period or makes any modification in his Bid, the bid of such bidder shall be rejected and Bid Security submitted by the Bidder shall be forfeited.
- 11. The Techno-commercial Bid shall be opened online at venue on the specified date & time as per S.No.-1. The time and date for opening of Financial Bid of bidders qualified in the technical bid shall be communicated to them at later date after evaluation of technical bids. The NHPC/ NHPC REL/Tender inviting Authority at his discretion may open Technical and Financial Bid simultaneously and evaluate the Bid completely.

12. E-tendering: Instruction for online bid submission

The Techno-commercial Bid and Price Bid to be submitted on-line at Central Public Procurement e-Portal http://eprocure.gov.in/eprocure/app. The bidders are required to submit soft copies of their bids electronically on the Central Public Procurement (CPP) Portal, using valid Digital Signature Certificates. The instructions given below are meant to assist the bidders in registering on the CPP Portal, prepare their bids in accordance with the requirements and submitting their bids online on the CPP Portal.

12.1 Registration:

- i) Bidders are required to enrol on the e-Procurement module of the Central Public Procurement Portal by using the "Online Bidder Enrolment" option available on the home page. Enrolment on the CPP Portal is free of charge.
- ii) As part of the enrolment process, the bidders will be required to choose a unique username and assign a password for their accounts.
- iii) During enrolment/ registration, the bidders should provide the correct/true information including valid email-id & mobile no. All the correspondence shall be made directly with the Consultants/ bidders through email-id provided.
- iv) For e-tendering possession of valid Digital Signature Certificate (Class II or Class III Certificates with signing key usage) is mandatory which can be obtained from SIFY/TCS/ nCode/eMudra or any Certifying Authority recognized by CCA India one Token/Smart Card.
- v) Upon enrolment on CPP Portal for e-tendering, the bidders shall register their valid Digital Signature Certificate with their profile.

- vi) Only one valid DSC should be registered by a bidder. Bidders are responsible to ensure that they do not lend their DSCs to others which may lead to misuse and should ensure safety of the same.
- vii) Bidders can then log into the site through the secured login by entering their userID/password and the password of the DSC/ eToken.

12.2 Searching for Tender documents:

- a) There are various search options built in the CPP Portal to facilitate bidders to search active tenders by several parameters. These parameters could include Tender ID, Organization Name, Location, Date, Value, etc. There is also an option of advanced search for tenders, wherein the bidders may combine a number of search parameters such as Organization Name, Form of Contract, Location, Date, Other keywords etc. to search for a tender published on the CPP Portal.
- b) Once the bidders have selected the tenders they are interested in, they may download the required documents/tender schedules. These tenders can be moved to the respective 'My Tenders' folder. This would enable the CPP Portal to intimate the bidders through SMS/e-mail in case there is any corrigendum issued to the tender document.
- c) The bidder should make a note of the unique Tender ID assigned to each tender, in case they want to obtain any clarification/ help from the Helpdesk.

12.3 Preparation of Bids:

- a) For preparation of bid, Bidders shall search the tender from published tender list available on site and download the complete tender document and should take into account corrigendum, if any, published before submitting their bids. After selecting the tender document same shall be moved to the '**My Favourite**' folder of bidders account from where bidder can view all the details of the tender document.
- b) Bidder shall go through the tender document carefully to understand the documents required to be submitted as part of the bid. Bidder shall note the number of covers in which the bid documents have to be submitted, the number of documents, including the names and content of each of the document that need to be submitted. Any deviations from these may lead to rejection of the bid.
- c) Any clarifications if required then same may be obtained online through the tender site or through the contact details given in the tender document.
- d) Bidders should get ready in advance the bid documents to be submitted as indicated in the tender document/ schedule in PDF/ xls/ rar/ zip/ dwf formats. If there is more than one document, they can be clubbed together using zip format. Bid documents may be scanned with 100 dpi with black and white option which helps in reducing size of scanned documents.
- e) To avoid the time and effort required in uploading the same set of standard documents which are required to be submitted as a part of every bid, a provisionof uploading such standard documents (e.g. PAN card copy, Annual Reports, Auditor Certificates etc.) has been provided to the bidders. Bidders can use "MySpace" or "Other Important Documents" area available to them to upload such documents. These documents may be directly submitted from the "My Space" or "Other Important Documents" area as per tender requirements while submitting the bid and need not be uploaded again and again. This will lead to reduction in the time required for bid submission process.

12.4 Submission of Bids:

- i) Bidder should log into the site well in advance for bid submission so that he/ she uploads the bid in time i.e. on or before the bid submission time.
- ii) Bidder should prepare the respective zonewise Tender Fee and EMD as per the instructions specified in the NIT/tender document. The originals should be submitted to

the Tender Inviting Authority, on or before the last date & time of offline bid submission. The details of the DD/BC/BG/Insurance Surety Bond, physically sent, should tally with the details available in the scanned copy and the data entered during bid submission time. Otherwise the uploaded bid will be rejected.

- iii) While submitting the bids online, the bidder shall read the terms & conditions (of CPP portal) and accepts the same in order to proceed further to submit their bid.
- iv) Deleted
- v) Bidder shall digitally sign and upload the required bid documents one by one as indicated in the tender document.
- vi) Bidders shall note that the very act of using DSC for downloading the tender document and uploading their offers is deemed to be a confirmation that they have read all sections and pages of the tender document without any exception and have understood the complete tender document and are clear about the requirements of the tender document.
- vii) Bidder shall note that each document to be uploaded for the tender should be less than 2 MB. If any document is more than 2MB, it can be reduced through zip/rar and the same can be uploaded. For the file size of less than 1 MB, the transaction uploading time will be very fast.
- viii) Utmost care shall be taken for uploading Schedule of Quantity & Price and any change/modification of the price schedule shall render it unfit for bidding. Bidders shall download the Schedule of Quantities & Prices in XLS format and save it without changing the name of the file. Bidder shall quote their rates in figures in white background cells, thereafter save and upload the file in financial bid cover (Price bid) only. If the template of "Schedule of Quantities & Prices" file is found to be modified/ tampered by the bidder which tantamount to fraudulent practices and the bid shall be rejected and further dealt as per provision of clause no 32 of ITB including forfeiture of EMD/ BG/Insurance Surity Bond. The bidders are cautioned that uploading of financial bid elsewhere i.e. other than in cover 2 will result in rejection of the tender.
- Bidders shall submit their bids through online e-tendering system to the Tender Inviting Authority (TIA) well before the bid submission end date & time (as per Server System Clock). The TIA will not be held responsible for any sort of delay or the difficulties faced during the submission of bids online by the bidders at the eleventh hour.
- x) After the bid submission (i.e. after Clicking "Freeze Bid Submission" in the portal), the bidders shall take print out of system generated acknowledgement number and keep it as a record of evidence for online submission of bid, which will also act as an entry pass to participate in the bid opening.
- xi) Bidder should follow the server time being displayed on bidder's dashboard at the top of the tender site, which shall be considered valid for all actions of requesting bid submission, bid opening etc., in the e-tender system.
- xii) All the documents being submitted by the bidders would be encrypted using PKI (Public Key Infrastructure) encryption techniques to ensure the secrecy of the data. The data entered cannot be viewed by unauthorized persons until the time of bid opening. The confidentiality of the bids is maintained using the secured Socket Layer 128 bit encryption technology.

Any queries relating to the process of online bid submission or queries relating to CPP Portal in general may be directed to 24x7 CPP Portal Helpdesk.Toll Free Number 0120-4001-002, 0120-4001-005 and 0120-6277-787.

13.0 Any corrigendum, subsequent amendments and/or extension of dates, if any, for submission of Bids shall be posted on the portal https://eprocure.gov.in/eprocure/app. Bidder(s) are advised to visit the portal regularly before the deadline for submission of Bids.

- **14.0** The employer reserves the right to accept or reject any Bid and to cancel the Bidding process and reject all Bids, at any time prior to the award of Contract, without thereby incurring any liability to the affected Bidder or Bidder(s). The Bidder shall bear all costs associated with the preparation and submission of its bid, and the Employer will in no case be responsible or liable for these costs, regardless of the conduct or outcome of the bidding process. However, the Bidder(s) who wish to seek reasons for such decision of cancellation/rejection shall be informed of the same by Employer unless its disclosure reasonably could be expected to affect the sovereignty and integrity of India, the security, strategic, scientific or economic interest of the state or lead to incitement of an offence.
- **15.0** In case of any difference between wordings of English and Hindi version of 'Notice Inviting Tender', English version shall prevail.

(For & on behalf of NHPC Ltd.) General Manager (Civil Contracts-I) Room No.218, ^{2nd} Floor, Jyoti Sadan, NHPC OFFICE COMPLEX, Sector-33, Faridabad-121003, Haryana, India Tele No: +91 (129) 2270596 Email: contcivil1-co@nhpc.nic.in

(I) <u>BIDDING DATA</u>

The following specific data for the Works to be procured shall complement, amend, or supplement the provisions in the Instructions to Bidders (ITB). Whenever there is a conflict, the provisions herein shall prevail over those in Instructions to Bidder:

| S. No. | Reference Clause No. | Description | Particulars |
|-----------|-------------------------|--|--|
| 1 | 1 | Name of Work | Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding |
| 2 | 7 | Pre-bid meeting: Venue, Date & Time | Venue: NHPC Ltd., NHPC Office Complex, Sector- 33, Faridabad (Haryana). The bidders may also attend Pre-bid meeting through video conferencing on scheduled date and time or as amended. Date: 30.01.2025 Time: 15:00 Hrs |
| 3 | 13 | Bid Validity | 120 days from the last date of submission of online Bid. |
| 4 | 14 | Earnest Money Deposit | (i) Rs. 67,00,000/-(Rupees Sixty Seven Lakhs only) – for Category-1 (ii) Rs 89,00,000/-(Rupees Eighty Nine Lakhs only)- for Category-2 (iii) Rs 1,02,00,000/-(Rupees One Crore Two Lakhs only)- for Category-3 |
| 5 | 17 | Deadline for online submission of bids | 21.02.2025 , Time : 17:30 Hrs |
| 6 | 17 | Deadline for offline submission of bids | 24.02.2025, Time : 17:00 Hrs |
| 7 | 20 | Time and Date for opening Technical Bids (online and offline) | 25.02.2025, Time: 15:00 Hrs |
| 8 | 24 | Time and Date of opening Financial Bids | Shall be intimated separately to Bidders whose Bids are found to be Technically responsive. |
| 9 | 24 | Time and Date of e-RA for each Category | Shall be intimated separately after activity at S.No. 8 above. |
| 10 | 33 | Name of IEM | Dr. Vinod Aggarwal Sri Prabhash Singh |

INSTRUCTION TO BIDDERS A. INTRODUCTION

1. SCOPE:

Online bids are invited through **Domestic Competitive Bidding in Single Stage-Two Part Bidding** Basis (i.e.Part-I: QR+Technical-Bid and Part-II:Financial Bid) with <u>e-Reverse Auction (e-RA)</u> by NHPC Ltd. on behalf of NHPC Renewable Energy Limited (NHPC REL)-a wholly owned subsidiary of NHPC Limited {herein after referred to as Project Management Consultant (PMC) for the RESCO Project set up by Roof Top Developer (RTD)} from eligible Bidders for "Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding"

The bid will be received, opened and evaluated online in electronic form through NHPC's Etendering portal i.e. http://eprocure.gov.in/eprocure/app [Link to reach at site is also available at NHPC website i.e., <u>www.nhpcindia.com</u> \rightarrow e-Procurement \rightarrow Govt. e-Procurement System of NIC (GePNIC) under Central Public Procurement Portal (CPPP)].

After e-tendering, Category wise e-Reverse Auction (e-RA) shall be conducted at application service provider's portal. Bid shall be prepared and submitted in accordance with instructions contained in this Section.

This section of the RFS document provides the information necessary for Bidders to prepare online responsive bids, in accordance with the requirements of the Employer. It also provides information on online bid submission, opening, evaluation, e-Reverse Auction and Contract award.

The bidders may bid for any/all Categories. The evaluated lowest bidder(L1) will be selected for each Category.

2. ELIGIBLE BIDDERS:

- i. The bidder must be an Indian incorporated legal Entity operating under commercial law of their respective jurisdiction.
- ii. Bidders shall not be under a declaration of ineligibility for corrupt and fraudulent practices by the NHPC/ NHPC REL, any Government institution or Public Sector Undertaking in India in accordance with ITB Clause 32.1 (a) of tender document, as on bid submission date.
- iii. The Bidders whose contract(s) have been terminated due to poor performance by the Employer, shall not be allowed to participate in the bidding process for next 5 years w.e.f. the date of notification of termination.
- iv. Bidders shall not have been banned/ de-listed/ black listed/ debarred from business on the ground mentioned in para 6 of Guidelines on Banning of Business dealings (Annexure-2A) to Integrity (Annexure-2 of ITB) Pact of tender document at the time of submitting the bid and shall intimate immediately of any such banning/de-listing/ black- listed/debarring after submission of bid till the award of the tender. Self-Declaration in this regard is to be submitted as per the proforma enclosed in tender document.
- v. Bidders shall meet the qualification criteria as defined in Clause 3.2.
- vi. To qualify for Contract for which bids are invited in the Notice Inviting Tender, the bidder must demonstrate having work experience, financial capability and resources sufficient to meet the aggregate of the qualifying criteria. Failure to produce the certificates and documents in this regard shall make the bid non-responsive.

vii. Even though the bidders meet the qualifying criteria, they are subject to be disqualified if they have made misleading or false representations in the forms, statements, declarations and attachments submitted in proof of the qualification requirements.

3. QUALIFICATION OF THE BIDDER

- **3.1** All bidders shall include the following information and documents with their bids in Qualification Information unless otherwise stated in the ITB:
- a) Written power of attorney of the signatory of the Bid to commit the Bidder. Additional information as sought in the Annexure-3A (Qualification Information Sheets of ITB) shall be provided.
- b) Work experience to demonstrate meeting the criteria stipulated in clause 3.2.A shall be provided in Annexure-3 (3B: Qualification Information Sheets).The work experience shown shall be supported with certificate(s) from the Engineer-in-charge/Project head of the concerned work. In case of experience certificate produced by the Contractors for having works for Private Organizations, TDS Certificates shall also be produced along with experience certificate.
- c) Information on financial criteria stipulated in clause 3.2 B shall be furnished along with Copy of affidavit/Certificate of CA mentioning Financial Turnover of immediately preceding 3 (three) years. Printed Annual reports or financial statements of the Bidder, such as balance sheet, profit and loss statements and auditor's reports as the case may be for the immediately preceding 3 (three) years shall be submitted to ascertain bidder's meeting the financial criteria.

3.2. QUALIFICATION CRITERIA

The qualification will be subject to Bidder's fulfillment of the Qualification Criteria set and stipulated hereunder, substantiated by authentic and relevant information and details. Additional information in support of their claims of achievements may be furnished in any form of their device and design. The Bidder(s) who wish to apply for Tender for the aforesaid work must satisfy the following qualifying criteria:

- **3.2.A** The Bidder(s) must satisfy the following technical qualifying criteria:
- **3.2.A.1** The Bidder should have designed, supplied, erected and commissioned Solar Photo Voltaic (SPV) based grid connected power plant(s) of cumulative installed capacity of 400 kWp or higher, out of which at least one plant should have been of 40 kWp or more.

OR

3.2.A.2 The Bidder should be a developer of Solar Photo Voltaic (SPV) based grid connected power plant(s) of cumulative installed capacity of 400 kWp or higher, out of which at least one plant should have been of 40 kWp or more.

OR

3.2.A.3(a) The Bidder should have executed in the last ten (10) years either as developer or as EPC Contractor in process industry or in infrastructure development of single work having minimum value of Rs. 4.0 Crores.

AND

(b) The Bidder should have executed at least one (1) Electrical Sub-station of 11 kV or above voltage level, consisting of equipment such as 11kV or above voltage level circuit breakers

and Power transformer, either as developer or as EPC Contractor which should be in successful operation for at least one(1) year prior to the last date of submission of bid.

The works referred to clause 3.2.A.3 (a) & 3.2.A.3 (b) can be in same or different projects.

Notes:

- a) The reference SPV based grid connected power plant of 40 kWp or higher capacity should be at a single location developed by Bidder for itself or any other client.
- b) SPV based Roof-top/Ground mounted/Floating solar power projects, which are grid connected, shall be considered eligible for QR purposes.
- c) Bidder shall submit certificate of successful completion and operation from the Owner.
- d) In case the award for the reference works has been received by the Bidder either directly from owner of plant or any other intermediary organization, a certificate from such owner of plant or the intermediary organization shall be required to be furnished by the Bidder along with its bid document in support of its claim of meeting requirement stipulated above.
- e) Developer means an entity who has either executed or got executed the work/ project as owner of industrial projects.
- f) The qualification criteria in Clause 3.2.A.1 or 3.2.A.2 or 3.2.A.3 are mutually exclusive. The bidder has to fulfil any one of the above-mentioned technical eligibility criteria which are individually exclusive.
- g) Bidder can also claim the experience of a Group Company/Holding company /Subsidiary Company for meeting the requirement (s) of clause 3.2.A.1 or 3.2.A.2 or 3.2.A.3. In such a case, the Bidder shall furnish an Undertaking jointly executed by the Firm qualified as per Clause 3.2.A.1 or 3.2.A.2 or 3.2.A.3 and the Bidder along with its bid document for complete performance of the Contract (in case of award) jointly or severally, as per the format enclosed in the bid document, failing which, the Bidder's bid document is liable to be rejected.
- h) Experience of bidders as EPC Contractor for supply of Balance of System (BOS) items and erection, testing, commissioning of entire solar plant shall be considered.
- To ascertain the executed value of work for Cl. 3.2.A.3(a), the value of Work experience in INR shall be enhanced on pro-rata basis by 7% simple rate per annum up to the date of Bid Opening from the completion date mentioned in the completion certificate.
- j) Capacity below 10kWp at single installation shall not be considered in cumulative capacity.

3.2. B Financial Criteria

3.2.B.1 Turnover: Average Annual Turnover in the preceding 3 years shall be not less than as under:

| Category | Turnover |
|--|-----------------|
| Category-1 for (>= $5kW$ to $< 20kW$) | Rs. 17.83 crore |
| Category-2 for (≥ 20 kW to < 100 kW) | Rs. 23.69 crore |
| Category-3 for (>= 100kW and above) | Rs. 27.13 crore |

3.2.B.2 Networth: The bidder should have positive "Net Worth" in at least 02 financial years out of the last 03 financial years, with the condition of positive net worth in immediately preceding financial year.

'Net Worth' means the aggregate value of the paid-up share capital and all reserves created out of the profits, securities premium account and debit or credit balance of profit and loss account, after deducting the aggregate value of the accumulated losses, deferred expenditure and miscellaneous expenditure not written off, as per the audited balance sheet, but does not include reserves created out of revaluation of assets, write-back of depreciation and amalgamation.

For assessment, relevant pages of Annual Report containing Net Worth figure duly certified by Authorized representative of Bidder, otherwise, figure duly authenticated by the Statutory Auditors should be submitted.

3.2.B.3 Working Capital: The working capital (current assets minus current liabilities) shall be as under

| Category | Working Capital |
|--|-----------------|
| Category-1 for ($\geq 5kW$ to $< 20kW$) | Rs. 9.91 crore |
| Category-2 for (≥ 20 kW to < 100 kW) | Rs. 13.16 crore |
| Category-3 for (>= 100kW and above) | Rs. 15.07 crore |

For this purpose current assets and current liabilities will be considered as classified in the audited balance sheet for the year immediately preceding the date of opening of bids. In case current assets and/or current liabilities are not classified separately in audited Balance Sheet, a certificate from Statutory Auditor/CPA carrying out the Statutory Audit, for current assets and/or current liabilities, as the case may be, clearly defining the items considered for the same, should be enclosed. However, trade payables shall be a part of current liabilities. If the Working Capital calculated from the audited Balance Sheets is negative, then such working capital shall be treated as zero.

Notes:

1. Other income shall not be considered for arriving at Annual Turnover.

2.

The Bidder's financial evaluation vis-à-vis the requirement as stipulated above shall be done on the basis of duly printed (offset) Annual Report/Equivalent Financial Statement for the immediately preceding Three years submitted by the Bidder along with the Bid. In case, Bid submission date falls upto 30th September, the annual reports/ equivalent financial statement upto previous year of immediate preceding Financial Year shall be considered and for the Tenders having Bid submission date after September, the annual reports/ equivalent Financial statement upto immediate preceding Financial year shall be considered. Further, standalone audited Annual Financial Statement of Bidder shall be forming part of the Annual Report/ equivalent financial statement.

In case, if bidder has not submitted the above Annual Report/ Equivalent Financial Statement along with Bid, then a certificate from CEO/CFO of the Bidder shall be submitted along with Bid mentioning that the requirement of Annual Report as per governing law of country is not mandatory in case of such bidders Equivalent Financial Statement shall be considered. In such

cases, duly notarized copies of Audited Printed Annual Financial Statement (Balance Sheet, Profit & Loss Statement, cash flow statement, Auditor 's Report thereon including all relevant Schedules/ annexures etc.) for the immediately preceding 3(Three) years be submitted by the bidder along with the Bid. The Annual Financial Statement shall be considered for evaluation in same manner as defined in above para for consideration of Annual Reports/ Equivalent Financial Statement.

- 3. Wherever, the Annual Report/ duly Self-certified/Attested copies of Audited Printed Annual Financial Statement are in language other than English, then copy duly translated & printed into English language and certified by approved / recognized English translator shall be submitted with the Bid.
- 4. For the Purpose of compliance to the stipulated turnover criteria given at Financial Criteria, the construction turnover from Joint Ventures(s) as declared in the Joint Venture Agreement shall also be considered.

The proportionate JV Turnover shall be certified by their Chartered Accountant in the case of the same not appearing in the Audited Financial Statement.

- 5. The qualification given by Statutory Auditors on the Financial Statements which have impact on Net Worth, Turnover, Working Capital or any other financial item shall be taken into account for calculating the above financial parameters of Qualifying Criteria.
- 6. In case where Audited financial results for the immediately preceding year are not available, then an statement of account as on closing date of the immediately preceding financial year depicting the Turnover and Net worth (calculated as per laid down criteria) duly certified by their Statutory Auditor/Certified Public Accountant carrying out the statutory audit shall be enclosed with the Bid along with copy of appointment letter of the statutory auditor. The Certificate should be issued with Unique Document Identification Number (UDIN) as per Gazette Notification No. 1-CA(7)/192/2019 dated 02.08.2019 or any other document identification number issued to Certified Public Accountants by their respective Institute, if applicable..
- 7. The bidders who intend to quote for more than one Category, have to fulfil the total financial criteria required for each Category cumulatively.

3.2. B.4 Bids of any insolvent firm:

The Bidder or partner of JV against whom an application for initiating corporate insolvency resolution process has been admitted by the Adjudicating Authority under the Insolvency and Bankruptcy Code 2016, or as amended from time to time, shall not be eligible for bidding. The same shall also be applicable to the Bidder company, who has taken unconditional technical and/or financial support from their Parent/Holding Company; against whom an application for initiating corporate insolvency resolution process has been admitted by the Adjudicating Authority under the Insolvency and Bankruptcy Code 2016 (IBC 2016), or as amended from time to time.

In case, bidder or JV partner in respect of whom any application for initiating corporate insolvency resolution process was not admitted at the time of submission of bid, but subsequently during the period of evaluation of bids or any time before the work is awarded, any such application is admitted by the Adjudicating Authority under the IBC 2016 (equivalent laws of the country where the firm has been incorporated/registered shall be applicable), the bidder shall be considered as ineligible and his bid shall be rejected.

An Undertaking that "no insolvency proceedings are admitted by the Adjudicating Authority against bidder or partner of JV under the IBC 2016 (equivalent laws of the country where the firm has been incorporated / registered shall be applicable)" shall be submitted on letter head duly signed by the Authorized representative of bidder or all the partners of JV.

Further, the Bidder, after submitting the bid till the time of award of work, shall inform NHPC regarding any admission of application for corporate insolvency resolution process by the Adjudicating Authority under the IBC 2016 against bidder or partner of JV and any suppression of such fact shall render the bidder liable for rejection of his bid and banning of business dealing as per terms and conditions of Guidelines of banning of business dealings under Integrity Pact.

3.3 Nature of Bidders.

The Bidder should be a Sole bidder for participating in the tendering for this Project.

For evaluation of specific construction experience of various bidders, following criteria shall be considered: -

- 1. The evaluation of Bidder's Specific Construction experience for a work executed on JV basis shall be based on his role and scope of work in such joint ventures. To establish his role and scope of work in such Joint Venture contracts, the Bidder shall provide documentary evidence such as JV agreement/contract of the work to the extent of experience claimed by him. In case the quantum of work as per experience certificate does not match with the scope of work as defined in the JV/consortium agreement, then the experience credentials shall be considered as per the experience certificate issued by the employer / client.
- 2. In case scope of work of individual partners is not clearly defined in the JV/consortium agreement then credential of bidder being partner in a JV/consortium shall be decided in the following manner:-

(a)Where specific experience certificate in respect of individual JV partners is available, the same shall be considered to the extent of work executed by such JV partner as per the experience certificate.

(b)Where specific experience certificate is not available, or experience certificate is issued in the name of JV/consortium, the evaluation shall be done in the following manner:-

i. In case the participation / profit sharing percentage of bidder as per JV agreement is at least 35%, full credit of the work executed by the JV/consortium shall be given to such bidder.

- ii. In case the participation / profit sharing percentage of bidder in JV agreement is less than 35%, credit of work executed under such JV shall be given to such bidder on pro-rata basis.
- iii. Notwithstanding (i) and (ii) above, in case of availability of specific experience certificate, (a) above shall apply.

In case both participation share and profit sharing ratio are available in the JV Agreement, participation share shall prevail.

3. For the portion of work executed through a sub-contractor, full experience for that portion of work can be claimed by the Contractor as well as sub-contractor for fulfilling criteria for evaluation of bidder's specific experience.

3.4 JOINT VENTURE BIDDERS/CONSORTIUM BIDDERS: Deleted

3.5 Deleted

3.6 Bids by Merged/ Acquired/ Subsidiary Companies:

In case of an Bidder Company, formed after merger and / or acquisition of other companies, past experience and other antecedents of the merged / acquired companies will be considered for qualification of such Bidder Company provided such Bidder Company continues to own the requisite assets and resources of the merged / acquired companies needed for execution and successful implementation of the work package put to tender.

If the Bidder Company is a Subsidiary Company and applies for tender on the unconditional technical and financial strength of the Parent / Holding Company, the same shall be considered provided the Parent / Holding Company commits to sign a Separate Agreement with NHPC Ltd confirming full support for the technical and financial requirements of the Subsidiary Company and commits to take up the work itself in case of non-performance by the Subsidiary Company in the event of award of the work to the Bidder Subsidiary Company. An undertaking by the Parent / Holding Company to this effect shall be submitted along with the tender. A subsidiary company intending to tender on the strength of Parent / Holding Company shall not be allowed to participate as a 'Sub-Contractor'.

For the purpose stated herein above in this clause, 'Parent Company' shall mean the 'Holding Company' owning majority (more than 50%) shares of such Bidder (Subsidiary) Company. Similarly by extensions of this interpretation, if "A" is owned by a 'Holding Company' "B" which in turn is owned by another 'Holding Company' "C", then "C" is construed as the 'Parent Company' of "A" as well as and so on. An apex 'Parent Company' may own number of independent Subsidiary / Group Companies and if any of these Subsidiary / Group Company commits assured support and unhindered access to its assets and resources to another Subsidiary

/ Group Company (Bidder in this case) under the same apex 'Parent Company' then experience and other credentials of such Subsidiary / Group Company shall be considered for tender of the Bidder Subsidiary Company provided such commitment is evidenced / authorized and guaranteed by the apex 'Parent Company'.

The experience of subsidiary companies of the Parent / Holding Company will be considered experience of the Parent / Holding Company.

However, for fulfillment of financial criteria, financial evaluation vis-à-vis the requirement as stipulated above shall be done on the basis of consolidated printed Annual Report for the immediately preceding 3 (three) years of the Parent Company/Apex Parent Company submitted by the Bidder along with the Bid/Application.

- **3.7** Each bidder must also produce with their Bid:
 - i) PAN No. and EPF Registration No.;
 - ii) GST Registration No.
 - iii) A declaration that the information furnished with the bid documents is correct in all respects in Annexure-1: Form of declaration
 - iv) Such other certificates, if any, as defined in the ITB.
- **3.8** To qualify for Contract for which bids are invited in the Notice Inviting Tender, the bidder must demonstrate having work experience, financial capability and resources sufficient to meet the aggregate of the qualifying criteria. Failure to produce the certificates and documents as required under clause 3.2 shall make the bid non-responsive.
- **3.9** Even though the bidders meet the above qualifying criteria, they are subject to be disqualified if they have made misleading or false representations in the forms, statements, declarations and attachments submitted in proof of the qualification requirements.
- **3.10** To improve transparency and fairness in tendering process the Employer is implementing Integrity Pact as per Clause 33 of this ITB. The bidder must submit the Integrity Pact as per Annexure- 2 duly signed as per clause 33 of ITB.

Pre- contract Integrity Pact is to be executed on plain paper at the time of submission of Bid. The successful bidder (contractor) shall submit duly executed Integrity Pact on non-judicial stamp paper of appropriate value prior to signing of Notification of Award.

To oversee the compliance of obligation under the Integrity Pact, **Sh. Prabhash Singh** and **Dr. Vinod Aggarwal** has been appointed as Independent External Monitor by the Owner.

The Contact Address of IEMs is as under:-

| Sh. Prabhash Singh, | Dr.Vinod Aggarwal, | |
|-------------------------------------|----------------------------------|--|
| E7/M702 Housing Board Colony, Arere | B-103, Sarvodaya Enclave, | |
| Colony, Bhopal, MP, 462016 | 2 nd Floor, | |
| e-mail – srgmhrbpl@gmail.com | New Delhi – 110 017 | |
| | e-mail – <u>arsv50@gmail.com</u> | |

4. COST OF BIDDING

4.1 Complete bid document can be viewed and downloaded from NHPC Limited website **www.nhpcindia.com** (For reference only) and Central Public Procurement (CPP) Portal

http://eprocure.gov.in/eprocure/app/nhpc. The bidder will be required to submit a nonrefundable fee as detailed below including GST, if applicable, in the form of Crossed Demand Draft in favour of "NHPC Limited" payable at 'Faridabad' towards the cost of Tender fee:

Rs.5,000/- (Rupees Five thousand only) – for Category 1 Rs.5,000/- (Rupees Five thousand only) – for Category 2 Rs.5,000/- (Rupees Five thousand only) – for Category 3

The bidder shall fill the tender fee details online and submit same in terms of Clause 16 of ITB on or before the date as specified in "Notice Inviting e-Tender"

Bidders willing to submit the bids for more than one Category have to submit the cost of bid documents for individual Category separately.

4.2 Deleted

Note: Mere downloading of Bid document by prospective Bidder shall not be construed that such a bidder automatically fulfills the prescribed eligibility criteria. Whether the bidder meets the specific eligibility criteria or not, shall be checked or ascertained, on opening their bids by scrutinizing documentary evidences furnished by them along with their bid.

The Bidder shall bear all costs associated with the preparation and submission of its bid, and the Employer will in no case be responsible or liable for these costs, regardless of the conduct or outcome of the bidding process.

5 SITE VISIT

5.1 The tenderers, in their own interest, should inspect and examine the site and its surroundings and satisfy themselves, before submitting their tender, in respect of the site conditions including but not restricted to the following which may influence or affect the work or cost thereof under the Contract:

- a) Site conditions including access to the site, existing and required roads and other means of transport/communication for use by the in connection with the works;
- b) Requirement and availability of rooftop area and other facilities for their enabling works, colonies, stores and workshops etc.
- c) Ground conditions including those bearing upon transportation, disposal, handling and storage of materials required for the work or obtained there from;
- d) Source and extent of availability of suitable materials including water etc., construction power and labour (skilled and un-skilled) required for work and Laws and Regulations governing their use and employment;
- e) Geological, meteorological, topographical, Hydrological and other general features of the site and its Surroundings as are pertaining to and needed for the performance of the work;
- f) The type `of equipment and facilities needed, preliminary to, for and in the performance of the work; and
- g) All other information pertaining to and needed for the work including information as to the risks, contingencies and other circumstances which may influence or affect the work or the cost thereof under this contract.
- 5.2 The tenderers should note that information, if any, in regard to the site and local conditions, in these tender documents, except for the material agreed to be supplied by NHPC/ NHPC REL, if any, has been given merely to assist the tenderers and is not warranted to be complete.

5.3 The tenderers should note and bear in mind that NHPC/ NHPC REL shall bear no responsibility for the lack of acquaintance of the site and other conditions or any information relating thereto, on their part. The consequences of the lack of any knowledge, as aforesaid, on the part of the tenderers shall be at their risk and cost and no charges or claims whatsoever consequent upon the lack of any information, knowledge or understanding shall be entertained or payable by the NHPC/ NHPC REL

5.4 ELIGIBLE PLANT, EQUIPMENT AND SERVICES

For the purposes of these RFS documents, the word "Facilities" means the plant and equipment to be supplied and installed, together with the services to be carried out by the RTD under the contract including design.

B. THE RFS DOCUMENTS

6 CONTENT OF RFS DOCUMENTS (OR BID DOCUMENTS)

- 6.1 The Facilities required, bidding procedures, contract terms and technical requirements are prescribed in the RFS documents. The RFS documents include the following sections:
 - Vol.0 Notice Inviting Tender (NIT), Bidding Data Sheets & Instructions to Bidders (ITB)
 - Vol.1 Information for Bidders (IFB)
 - Vol.2 Definitions & Standard Copy of Power Purchase Agreement (PPA)
 - Vol.3 Special Conditions of Contract (SCC) with Techical Specifications

Vol.4 Forms & Procedure (FP)

6.2 The Bidder is expected to examine all instructions, forms, terms, technical specifications and other information in the RFS documents. Failure to furnish all information required by the RFS documents or submission of a bid not substantially responsive to the RFS documents in every respect will be at the Bidder's risk and may result in rejection of its bid.

7 CLARIFICATION OF RFS DOCUMENTS & PRE-BID MEETING

- 7.1 A prospective Bidder requiring any clarification of the RFS documents may notify the NHPC/NHPC REL in writing or by cable (hereinafter, the term cable is deemed to include telefax or email at the NHPC/NHPC REL mailing address indicated in the Invitation for Bids.) The NHPC/NHPC REL will respond during Pre-bid Conference to any request for clarification or modification of the RFS documents, if deemed fit, NHPC/NHPC REL response will be published on the portal <u>https://eprocure.gov.in/eprocure/app</u>, including a description of the inquiry, but without identifying its source. The NHPC/NHPC REL shall not be obliged to respond to any request for clarification received later than the above said period. Further, the mere request for clarification from the bidder(s) shall not be a ground for seeking extension in the deadline for submission of bids.
- 7.2 The Bidder is advised to visit and examine the site where the Facilities are to be installed and its surroundings and obtain for itself on its own responsibility of all information that may be necessary for preparing the bid and entering into a contract for supply and installation of the Facilities. The costs of visiting the site shall be at the Bidder's own expense.
- 7.3 The Bidder and any of its personnel will have to take permission by the respective Building Owner or the Concerned Authority to enter upon its premises and lands for the purpose of such inspection as per 7.2 above, but only upon the express condition that the Bidder and its personnel will release and indemnify the Building Owner/ the Concerned Authority or NHPC/NHPC REL and its personnel from and against all liability in respect thereof and will be responsible for death or personal injury, loss of or damage to property and any other loss, damage, costs and expenses incurred as a result of the inspection.

7.4 **PRE-BID CONFERENCE:**

Pre bid conference will be held as per details specified in 'NIT' where intending bidders may obtain necessary clarifications to their queries, if any, from NHPC/NHPC REL. All such queries seeking clarification on the bid documents shall be made in writing or by e-mail or facsimile to Tender Inviting Authority at the address indicated in the Bidding Data so as to reach the NHPC/NHPC REL not later than 3 days before the pre-bid meeting so that the same can be replied during the meeting.

The prospective bidder or his authorized representative may attend the pre-bid meeting as indicated above. The bidders who wish to join the meeting on due date & time through video-conferencing shall intimate Tender Inviting Authority (TIA) at least 02 (two) days prior to prebid meeting their details viz. name of participant & designation, mobile no., e-mail address, name of firm or any other information required for video-conferencing. The participant should have good internet connectivity as TIA shall not be held responsible for any disruption due to internet or any technical issues.

Meeting ID alongwith password shall be shared 01 (one) hour prior to scheduled time to the participant's e-mail/ mobile no. of prospective bidder. The purpose of the meeting will be to clarify issues and to answer questions on any matter related to the bid that may be raised at that stage.

The NHPC/NHPC REL will respond during Pre-bid Conference to any request for clarification or modification of the RFS documents. If deemed fit, written copies of the NHPC/NHPC REL response (including an explanation of the query) will be sent after Pre-bid Conference to the respective Bidders who have sought clarification.

Any modifications in the Tender documents listed in Clause 6.1 of ITB, which may become necessary as a result of the pre-bid meeting shall be made by the NHPC/NHPC REL exclusively through the issue of an Addendum pursuant to Clause 8 of ITB.

Non-attendance at the pre-bid meeting will not be a cause for disqualification of a bidder.

8 AMENDMENT OF RFS DOCUMENTS

- 8.1 At any time prior to the deadline for submission of bids, the NHPC/NHPC REL may, for any reason, whether at its own initiative, or in response to a clarification requested by a prospective Bidder, amend the RFS documents by issuing Addendum / Corrigendum.
- 8.2 Any Addendum / Corrigendum thus issued shall be part of the bid documents and shall be posted on E-procurement portal of CPP (<u>http://eprocure.gov.in/eprocure/app</u>) / NHPCwebsite. Bidders are advised to visit the portal regularly before the deadline for submission of bids. The amendments to the bid documents will be binding on the prospective bidders and the notification of the amendment communicated through portal, shall be deemed to be construed that such amendment(s) to the bid documents have been taken into account by the bidder in its application. The responsibility of downloading the related Addendum / Corrigendum, if any, will be that of the bidder. No separate intimation in respect of Addendum / Corrigendum (if any) will be sent to bidder(s)
- 8.3 In order to afford prospective Bidders reasonable time in which to take the amendment into account in preparing their bid, the NHPC/NHPC REL may, at its discretion, suitably extend the deadline for the submission of bids and notify throughportal <u>http://eprocure.gov.in/eprocure/app</u> where all prospective bidders may see the extended deadline.

C. PREPARATION OF BIDS

9 LANGUAGE OF BID

The bid prepared by the Bidder and all correspondence and documents related to the bid exchanged by the Bidder and the NHPC/NHPC REL shall be written in English language, provided that any printed literature furnished by the Bidder may be written in another language, as long as such literature is accompanied by a translation of its pertinent passages in English language, in which case, for purposes of interpretation of the bid, the translation in English shall govern.

10 DOCUMENTS COMPRISING THE BID

- 10.1 The bid submitted by the Bidder shall comprise the following documents
 - (a) Covering Letter duly completed and signed by the Bidder (without including the quoted price) as provided in Vol. 4 (Form & Procedures) together with all Attachments identified in ITB Sub-Clause 10.2 below.
 - (b) Category Wise Price Schedules (as applicable) duly completed by the Bidder in online.
- 10.2 Each Bidder shall submit with its bid the following attachments:

(a) Attachment 1: Bid Security

A bid security furnished in accordance with ITB Clause 14.

(b) Attachment 2: Power of Attorney

A power of attorney along with Board resolution for authorizing the executants for issuing Power of Attorney, duly authenticated by a Notary Public, indicating that the person(s) signing the bid has the authority to sign the bid and the bid is binding upon the Bidder during the full period of its validity in accordance with ITB Clause 13.

(c) Attachment 3: Bidders Eligibility and Qualifications

The Bidder shall furnish requisite documents / information in support of meeting the qualification requirement as stipulated under ITB clause 2 &3.

- (i) General Information Sheet as per Annexures -3 (A) ,3(B) & 3(C)
- (ii) Notarized copy of Letter of Award (LOA), Agreements and Completion Certificates etc.

In case of certificates being issued by Private Employers, the bidders are required to submit notarized TDS certificate(s) or Form 26AS in respect of works claimed to be executed by the respective bidder(s). The Bidder's financial evaluation as per clause 3.2 B above shall be done on the basis of duly printed Annual Report for the years immediately preceding 3 (Three) years submitted by the Bidder along with the Bid. Further, standalone audited Annual Financial Statement of Bidder shall be forming part of the Annual Report.

(d) Attachment 4: Eligibility and Conformity of the Facilities and RFS Documents.

The documentary evidence of the eligibility of the Facilities shall consist of a statement on origin of the plant and equipment offered, which shall be confirmed by a certificate of origin issued at the time of shipment.

The documentary evidence of the conformity of the Facilities to the RFS Documents may be in the form of literature, drawings & data, and shall furnish:

- i) All the RFS Documents as per ITB Clause 6.1 duly signed and stamped on each page as a proof of its acceptance.
- ii) Deleted

(iii) Bidders shall note that standards for workmanship, materials and equipment designated by the NHPC/NHPC REL in the RFS documents are intended to be descriptive (establishing standards of quality and performance) only and not restrictive. The Bidder may substitute alternative standards, brand names and/or catalogue numbers, during currency of contract, provided that it demonstrates to the NHPC/NHPC REL satisfaction that the substitutions are substantially equivalent or superior to the standards designated in the Technical Specifications

- (e) Attachment 5: Deleted
- (f) Attachment 6: Deleted
- (g) Attachment 7: PERT Network (L2 Schedule). The bidder shall submit the Category Wise brief integrated PERT Network (L2 Schedule) for the work matching the Time schedule.
- (h) Attachment 8: Deleted
- (i) Attachment 9: Deleted.
- (j) Attachment 10: Deleted
- (k) Attachment 11: Deleted
- (1) Attachment 12: Insolvency and Bankruptcy Code 2016
- (m) Attachment 13: Self Cerificate for Cyber Security as per ITB clause No. 37.
- (n) Form of Disclosure alongwith Annexures as per ITB Cl. 34.
- (o) Certificate of Compliance to all provisions of Rfs Document
- (p) Bidders undertaking (on Letter head of the Company) mentioning the requisite Category for which Rfs(s) are submitted

10.3 **BID FORM AND PRICE SCHEDULES**

The Bidder shall complete the Bid Form and the appropriate Category Wise Price Schedules furnished in the RFS Documents as indicated therein and in the sub-section "Bid Form and Price Schedules" of the RFS Documents, following the requirements of ITB Clauses 11 and 12.

11 **BID PRICES**

- 11.1 Unless otherwise specified in the Technical Specification, Bidders shall Category Wise quote for the entire Facilities on a "single responsibility" basis such that the total Category wise bid price (Tariff) covers all the Contractor's obligations mentioned in or to be reasonably inferred from the RFS documents in respect of the design, manufacture, including procurement and subcontracting (if any), delivery, construction, installation and completion of the Facilities including supply of Tools & instruments of that category. This includes all requirements under the RTD's responsibilities for testing and commissioning of the Facilities and, where so required by the RFS documents, the acquisition of all permits, approvals and licenses, etc.; the operation, maintenance and training services and such other items and services as may be specified in the RFS documents, all in accordance with the requirements of the PPA and Special Conditions of Contract.
- 11.2 Bidders are required to quote the Tariff for the commercial, contractual and technical obligations outlined in the RFS documents.
- 11.3 Bidders shall give Category wise Tariff in the manner and detail called for in the Category wise Price Schedules or BOQ.

Category Wise Tariffshall be submitted in electronic form on the portal.

- 11.4 In the Category Wise Schedules, Bidders shall quote the requisite tariff inclusive of supply, erection, installation, O&M, taxes, necessary permissions etc. of his scope as mentioned in this RFS Document.:
- 11.5 Deleted.
- 11.6 The Category Wise tariff emerged after e-RA shall remain firm in accordance with the provisions of RFS Documents.
- 11.7 Bidders shall apprise themselves with "Conditions for conducting e-RA after e-tendering" and "Business Rules for e-Reverse Auction" enclosed as Annexure-6&7 respectively with this ITB before quoting their prices.
- 11.8 Deleted

12 BID CURRENCIES

Bids shall be furnished in Indian Rupee only.

13 PERIOD OF BID VALIDITY

- 13.1 Bids shall remain valid for a period as mentioned in Bidding Data after the deadline date for online bid submission specified in Clause 17 of ITB or amendment thereof. A bid valid for a shorter period may be rejected by the NHPC/NHPC REL as being non responsive.
- 13.2 In exceptional circumstances, the NHPC/NHPC REL may solicit the Bidders' consent to an extension of the bid validity period. The request and bidders responses thereto shall be made in writing or by email. If a Bidder accepts to prolong the period of validity, the bid security shall also be suitably extended. A Bidder may refuse the request without forfeiture of its bid security. A Bidder granting the request will not be required nor permitted to modify its bid.
- 13.3 All the provisions of bid including those regarding discharge and invokation of Bid Security shall continue to apply during the extended period of bid validity specified by the Bidder.

14. **BID SECURITY**

- 14.1 Bid shall be accompanied by bid security /EMD/BG/Payment on Order Instrument (POI)/Insurance Surety Bond of requisite amount applicable for the category in which bidder is willing to apply as stipulated in NIT / Bidding data. Bidders willing to submit the bids for more than one Category have to submit the Bid Security for individual category separately.
- 14.2 The bid security shall, at the Bidder's option, be in the form of a crossed Demand Draft in favour of NHPC Limited, Sector-33, Faridabad, Haryana (India) payable at Faridabad or a bank guarantee/POI/Insurance Surety Bond. The bid security in the form of a bank guarantee and for the amounts expressed in Indian Rupees shall be issued by an Indian Nationalized / Scheduled Commercial Bank. However, the issuing Bank shall submit an unstamped duplicate copy of Bank Guarantees directly to the NHPC/NHPC REL (Tender Inviting Authority) at the address mentioned at S. No. 1 (viii) of NIT with a forwarding letter.

The format of the bank guarantee shall be in accordance with the form of bid security included in the RFS documents; other formats may be permitted subject to prior approval of the NHPC/NHPC REL. Bid securities shall remain valid for a period of 90 days, beyond the original validity period of the bid or beyond any period of extension subsequently requested under sub-clause 13.2.

Bidder shall communicate the following bank details of NHPC to the issuing Bank for online confirmation of Bank Guarantee to be submitted in terms of this clause:

Name of the beneficiary: NHPC Limited Account No: 10813608692 IFSC Code: SBIN0017313 Credit Bank: State Bank Of India Account Type: CURRENT Branch: CAG-II New Delhi (17313) Address of the Bank : 5th Floor, Redfort Capital, Parsvnath Towers, Bhai Veer Singh Marg, Gole Market, New Delhi-110 001 Bidder is required to submit the SFMS receipt issued by the bank along with its bid for BG Confirmation.

Payment on Order Instrument (POI): As an alternative to submission of EMD as above, the Bidder also has an option to submit a letter of undertaking issued by either of the following three organizations, viz. (i) Indian Renewable Development Agency Limited (IREDA) or (ii) Power Finance Corporation Limited or (iii) REC Limited only. The instrument of 'Letter of Undertaking' as described above in any other form, from any other non-banking financial institutions or bank except (i) Indian Renewable Energy Development Agency Limited (IREDA) or (ii) Power Finance Corporation Limited or (iii) REC Limited shall not be acceptable. This Letter of Undertaking shall be issued as "Payment on Order Instrument" (POI), wherein the POI issuing organization undertakes to pay in all scenarios under which the EMD would be liable to be encashed by NHPC/NHPC REL within the provisions of RfS/PPA. This instrument would have to be furnished as per Format available in the RfS document. POI shall remain valid for a period of 90 days, beyond the original validity period of the bid or beyond any period of extension subsequently requested under sub-clause 13.2.

Insurance Surety Bond: As an alternative to submission of EMD as above, the Bidder also has an option to submit Insurance Surety Bond issued by Insurance Company registered in India under Insurance Act 1938 or as amended from time to time and approved by the Insurance Regulatory Development Authority (IRDAI) acceptable to employer in the prescribed proforma. This Insurance Surety Bond would have to be furnished as per Format available in the RfS document. Insurance Surety Bond shall remain valid for a period of 90 days, beyond the original validity period of the bid or beyond any period of extension subsequently requested under sub-clause 13.2.

The term "Bank Guarantee (BG) towards/ against EMD" occurring in the RfS shall be read as "Bank Guarantee (BG)/ Payment on Order Instrument (POI)/ Insurance Surety Bond towards/ against EMD".

However, in case Earnest Money has been submitted by the bidder in the form of Demand Draft, the information relating to particulars of the bidders' bank shall be submitted by the bidder along with the bid in the ECS form at Volume-IV. The Earnest Money of unsuccessful bidders shall be remitted by the NHPC/NHPC REL in the above account through ECS mode.

- 14.3 Any bid not accompanied by an acceptable bid security shall be rejected by the NHPC/NHPC REL as being non-responsive.
- 14.4 Bid Security of bidders who are not qualified in any category for opening of price bid shall be returned within 15 days after such notification for that category only. The Bid Security of all the unsuccessful bidders, whose price bid has been opened, will be returned within 15 days of notification of the award of Contract of that category to the successful bidder.

- 14.5 The bid security of the successful bidder of any category will be returned when the bidder has signed the PPA, pursuant to ITB Clause 30, and has furnished the required performance security, pursuant to ITB Clause 31. Interest or any other charges, whatsoever, will not be payable by the NHPC/NHPC REL on bid security amount.
- 14.6 The bid security may be forfeited under following conditions:
 - (a) if the Bidder withdraws its bid or varies any terms & conditions without the consent of the NHPC/NHPC REL, in regard thereto during the period of bid validity or
 - (b) if the Bidder adopts corrupt or fraudulent or collusive or coercive practices or defaults commitments under Integrity Pact or
 - (c) in the case of a successful Bidder, if the Bidder fails within the specified time limit
 - (i) to sign the PPA, in accordance with ITB Clause 30, or
 - (ii) to furnish the required performance security, in accordance with ITB Clause 31.
 - (d) In case, Government Client offers to execute the PPA with the Selected Roof Top Bidder and if the Selected Bidder does not submit the requisite documents or does not execute the PPA, within the stipulated time period.
 - (e) If the bidder refuses to withdraw, without any cost to NHPC/NHPCREL, any deviation, variation, additional condition or any other mention anywhere in the bid, contrary to the provisions of the RfS Document;
 - (f) If after issuance of LoA, it is found that the documents furnished by the Bidder as part of response to RfS are misleading or misrepresented in any way;
- 14.7 Deleted.
- 14.8 Interest or any other charges, whatsoever, will not be payable by the NHPC/NHPC REL on the Earnest Money Deposit.
- 14.9 Deleted.
- 14.10 Deleted

15. FORMAT AND SIGNING OF BID

The Bid shall be signed /digitally signed for hard copy / online submission, as the case may be by a person duly authorized to sign on behalf of the Bidder as below:

- i) If the tender is submitted by an individual, it shall be signed by the individual.
- ii) If the tender is submitted by a proprietary firm, it shall be signed by the proprietor.
- iii) If the tender is submitted by a firm in partnership, it shall be signed by a partner holding the power of attorney of the firm for signing the tender, in which case a certified copy of the power of attorney shall accompany the tender. A certified copy of the partnership deed duly registered and current business address of all the partners of the firm shall also accompany the tender.
- iv) If the tender is submitted by a limited company or a limited corporation, it shall be signed by a duly authorized person holding the power of attorney or any other legally valid document for signing the tender, in which case a certified copy of the power of attorney or any such legally valid document supported by copy of Board Resolution shall accompany the tender. Such limited company or corporation may be required to furnish satisfactory evidence of its existence before the contract is awarded.
- v) All witnesses and sureties shall be persons of status and their full names, occupations and addresses shall be stated below their signatures.

D. SUBMISSION OF BIDS

16. SEALING AND MARKING OF BIDS

Bids shall be submitted under single stage two envelope bid through e-tendering as per procedure detailed below. Bidders are to submit various details/ document along with copies of certificates in support of their claims for meeting the qualifying requirements.

In e-tendering Mode:

The bids shall be received through e-tendering Mode as detailed in **Important Instructions** to bidders for online bidding.

PART A – Techno-Commercial Bid

(I) Online Submissions:

The Techno-commercial bid submitted by the bidder in electronic form (online) shall comprise the following documents:

- Covering Letter duly completed and signed by the bidder (without indicating the quoted price) as provided Vol. 4 (Forms & Procedures) together with all Attachments as defined under clause 10 of ITB.
- ii) Category Wise Scanned copy of DD/Bank Guarantee/POI/Insurance Surety Bond shall be as per the format provided in RfS Document.
- iii) Category wise Scanned copy of DD towards cost of RFS document as mentioned in NIT.
- iv) Scanned copy of Power of Attorney along with requisite documents as specified under II-Offline Submissions sl. no .(iii).
- v) Scanned copy of duly filled Forms along with supporting documents meeting the qualifying requirements as per clause 3.2 of ITB.
- vi) Notarized Copies of successful completion certificate(s) of works/assignments. In addition, in case of these certificates being issued by Private Employers, the bidders are also required to submit notarized Letter of Award / Agreements / TDS certificate(s) etc. in respect of works claimed to be executed by the respective bidder(s).
- vii) Scanned copy of duly signed Integrity Pact as per Clause 33 of ITB.
- viii) Deleted
- ix) Deleted
- x) Annual Reports for last three financial years ending on FY 2023-24.
- xi) Process Compliance Form for e-RA as per ITB Clause 24.1.
- xii) Price Confirmation Form for e-RA as per ITB Clause 24.1.
- xiii) Such other certificates / documents as defined in the ITB.
- xiv) Scanned copy of undertaking on company letter head duly signed by the authorized person as per Attachment 12 in line with Cl. 3.2.B.4 of ITB, as specified under II-Offline Submissions sl. no .(iv).
- xv) Attachment -7- Category Wise Brief integrated PERT Network (L2 Schedule)
- xvi) Attachment 13: Self Cerificate for Cyber Security as per ITB clause No. 37.
- xvii) Form 13 (a): Bidders undertaking (on Letter head of the Company) mentioning the requisite Category for which RfS(s) are submitted.
- xviii) Form of Disclosure alongwith Annexures as per ITB Cl. 34.
- xix) Certificate of Compliance to all provisions of Rfs Document

The NHPC/NHPC REL reserves the right to request the bidder for submission of original documents in respect of any of the online submissions defined above.

(II) Offline Submissions (Hard Copies)

Bidders shall submit the following documents duly signed by Authorized signatory in an envelope by the date mentioned in Bidding Data. The envelope received late or after the prescribed due date and time of receipt as mentioned in Bidding Data will not be entertained. NHPC will not be responsible for any postal delay.

- i) Category Wise DD /banker's cheque payable at "Faridabad" (in original) towards cost of RFS document as mentioned in NIT.
- ii) Category Wise DD/BG/POI/Insurance Surety Bond (in original) towards Bid Security as mentioned in Clause No. 14 of ITB.
- iii) Power of Attorney along with Board resolution (original/notarized) for authorizing the executants for issuing Power of Attorney as per Clause 15(iv) of ITB, if applicable. In case of General Power of Attorney, a true copy of the POA shall be duly notarized by Notary and in case of specific Power of Attorney, the original POA.
- iv) An undertaking on company letter head duly signed by the authorized person as per Attachment 12 in line with Cl. 3.2.B.4 of ITB.
- v) Duly signed Integrity Pact along with its all annexures as per Clause 33 of ITB.
- vi) The envelope shall be superscribed as "Hard Copy of documents for E- tender No. 2025_NHPC_842440_1

Out of offline documents, category wise DD towards cost of bid document and category wise DD/BG/POI/Insurance Surety Bond towards Bid Security must be strictly as per Bid conditions and if any discrepancy is found between the hard copies of the above documents and scanned copy of the same uploaded online, then the Bid uploaded on the portal shall not be considered at all any further. However, NHPC/NHPC REL may ask for additional documents / clarifications on the other documents submitted online.

Part B) Price Bid

There is an excel file having price schedules of all the category available under different sheets. Bidder willing to submit their bid for a particular category shall carefully fill that particular category only.

To be uploaded for e-tender

- i) Category Wise "Price bid" quoting requisite Tariff is to be submitted in Electronic Form on the portal. Submission of the same by any other means shall not be accepted by the NHPC/NHPC REL in any circumstances.
- ii) Deleted
- iii) Deleted
- iv) The bidders in their own interest are advised to be very careful while filling up the price bid in Electronic Form.
- v) Bidders will not be allowed to revise the quoted tariff on their own, once "Price Bids" have been opened by the NHPC/NHPC REL.
- vi) The bidders will have to submit bids quoting a fixed levelised tariff for entire project duration of 25 years..

- vii) Bids not covering the entire scope shall be treated as incomplete and hence may be rejected.
- viii) Price bid of bidders, whose techno-commercial bids are not considered acceptable to the NHPC/NHPC REL, shall not be opened and will be archived unopened. The decision of the NHPC/NHPC REL is final and binding in this regard.
- ix) Category Wise Tariff shall be quoted in Rs./kWh. Tariff (in Rs/kWh) shall be quoted upto two places of decimal only. If tariff is offered with more than two digits after decimal, it shall be truncated and considered upto two places of decimal (in Rs/kWh).

17. DEADLINE FOR SUBMISSION OF BIDS

- a) Complete Bids must be uploaded at the portal and hard copies of the documents mentioned in Clause 16, Part A-II of ITB must be received by the NHPC/NHPC REL at the address specified in NIT and in bidding data sheet not later than the time and date stipulated in the Bidding Data Sheet or extension thereof. In the event of the specified date for the submission of bids being declared a holiday for the NHPC/NHPC REL, the hard copy of the documents will be received up to the specified time on the next working day. Similarly, in the event of the specified date or amendment if any for the opening of bids being declared a holiday for the NHPC/NHPC REL, the opening shall be carried out at the specified time on the next working day. However, the date and time for online submission of the Bids shall continue to be the date and time specified or amendment if any.
- b) The NHPC/NHPC REL may, in exceptional circumstances and at its discretion, extend the deadline for submission of bids by issuing an Addendum / Corrigendum in accordance with Clause 8.1 of ITB, in which case all rights and obligations of the NHPC/NHPC REL and the bidders previously subject to the original deadline will thereafter be subject to the deadline as extended.
- c) The NHPC/NHPC REL shall not be responsible if bid could not be opened within reasonable time for reasons attributable to the Bidder or for whatsoever reasons. In such a case, the bid shall be sent unopened and 'Archive' on the portal and shall not be considered at all any further and bid security of such bidder may be returned.

18. LATE BIDS

a) Online submission of the bid will not be permitted on the portal after expiry of submission time and the bidder shall not be permitted to submit the same by any other mode.

In such case, even if the bidder has submitted the specific documents in hard copy (viz., DD/BG/POI/Insurance Surety Bond towards bid security, Duly signed Integrity Pact, Certified copy of the duly registered partnership deed, Balance Sheets and Power of Attorney) within the stipulated deadline, its bid shall be considered as late and shall not be considered at all any further.

Similarly hard copy (ies) of the documents, if received by the NHPC/NHPC REL after the deadline for submission of Bids prescribed in NIT, then it will be considered as late bid even if the bidder has uploaded the bid within the stipulated deadline. In such a case, the bid uploaded on the portal shall be considered as non-responsive and 'Archived'. The hard copies shall be returned to bidder unopened.

19. MODIFICATION AND WITHDRAWAL OF BIDS

19.1 After submission of the bid the contractor can re-submit revised /modify/substitute or withdraw its bid any number of times but before stipulated deadline for submission of bid.

- 19.2 In case NHPC/NHPC REL desired through amendment/corrigendum to submit revised financial bid then it shall be mandatory to submit revised Price bid. In such case the bid submitted before amendment/corrigendum shall become invalid
- 19.3 The server time (which is displayed on the bidders' dashboard) will be considered as the standard time for referencing the deadlines for submission of the bids by the bidders, opening of bids etc. The bidders should follow this time during bid submission.
- 19.4 It should be strictly noted that as per portal provisions, the bid once withdrawn by the bidder, the bidder cannot participate in the same tender again.
- 19.5 No bid can be withdrawn or modified in the interval between the bid submission deadline and the expiration of the bid validity period specified in bid documents. Withdrawal or modification of a bid during this interval will result in forfeiture of its bid security and bid shall be considered non-responsive.

E. BID OPENING AND EVALUATION

20. OPENING OF BIDS BY NHPC/NHPC REL

The NHPC/NHPC REL will open all bids (except the categorywise price bid part, which shall be governed as per ITB Clause 16, Part B) in the presence of bidders designated representatives who choose to attend, at the time, date, and location stipulated in the bid document. The bidders' representatives who are present shall sign a register evidencing their Bid opening attendance. Bidder(s) can also view online on the portal https://eprocure.gov.in/eprocure/appat their end.

In the event of the specified date for the opening of bids being declared a holiday for the NHPC/NHPC REL, the bids will be opened at the appointed time on the next working day.

Initially, the Techno-commercial bid shall be opened and the Price Bid of only those bidders whose Techno-commercial bid is acceptable to the NHPC/NHPC REL shall be opened online subsequently.

All important information and any such other details as the NHPC/NHPC REL at his discretion may consider appropriate, will be announced by the NHPC/NHPC REL at the opening. This shall include but may not be limited to the Bidders' names, the Bid Prices including deviations, withdrawals, bid modifications, and the presence (or absence) of bid security.

21. CLARIFICATION OF BIDS

21.1 During bid evaluation, the NHPC/NHPC REL may, at its discretion, ask the Bidder for a clarification of its bid. The request for clarification and the response shall be in writing or email or through the portal, and no change in the price or substance of the bid shall be sought, offered or permitted. Reply shall be submitted by Bidder within a stated reasonable period of time. If Bidder does not provide clarifications of the information requested by the date and time set in the NHPC/NHPC REL request for clarification, its Bids may be rejected.

22 PRELIMINARY EXAMINATION OF BIDS

22.1 The NHPC/NHPC REL will examine the bids to determine whether they are complete, whether any computational errors have been made, whether required sureties have been furnished, whether the documents have been properly signed, and whether the bids are generally in order. The NHPC/NHPC REL will, thereafter check and ascertain whether the bidder fulfils the Eligibility Criteria and other requirements specified under ITB clause 3. The Bids submitted by the Bidders who meet the Eligibility Criteria set under ITB clause 3 shall only qualify for consideration and further techno-commercial evaluation by the NHPC/NHPC REL.

- 22.2 Category wise Price Bid duly filled in excel sheet in conformity with the tender specification shall be uploaded on the e-procurement portal only. Unit of Measure (UOM) is indicated in the BOQ.
- 22.3 The NHPC/NHPC REL may waive any minor infirmity, nonconformity or irregularity in a bid that does not constitute a material deviation, and that does not prejudice or affect the relative ranking of any Bidder as a result of the technical and commercial evaluation, pursuant to ITB Clauses 23 and 24.

Also, if any discrepancy is found between the hard copies of the online documents viz. DD/BG/POI/Insurance Surety Bond towards Bid Security, Power of Attorney and scanned copy of same uploaded which tantamount to fraudulent practice by the bidder, then the online bid shall be liable for rejection.

- 22.4 Prior to the detailed evaluation, the NHPC/NHPC REL will determine whether each Bid is prima facie complete and is substantially responsive to the RFS Documents. For purposes of this determination, a substantially responsive Bid is one that conforms to all the terms, conditions and specifications of the RFS Documents without material deviations, objections, conditionality or reservations. A material deviation, objection, conditionality or reservation is one (i) that affects in any substantial way the scope, quality of performance of the Contract; (ii) that limits in any substantial way and/or is inconsistent with the Bid Documents or the NHPC/NHPC REL rights or the successful Bidder's obligations under the Contract; or (iii) whose rectification would unfairly affect the competitive position of other Bidders who are presenting substantially responsive Bids.
- 22.5 In particular, Bids with deviations from, objections to or reservations on provision such as those concerning Bid Security/EMD, Bid validity, if any, will be treated as non-responsive.

The NHPC/NHPC REL determination of a bid's responsiveness is to be based on the contents of the bid itself without recourse to extrinsic evidence.

- 22.6 If a bid is not substantially responsive, it will be rejected by the NHPC/NHPC REL, and may not subsequently be made responsive by the Bidder by correction of the nonconformity.
- 22.7 The Bid shall also be examined / evaluated on National Defence & Security considerations of the country where the site is located. The NHPC/NHPC REL reserves the right to set aside / reject the bid on these considerations.

23. TECHNO-COMMERCIAL EVALUATION

The NHPC/NHPC REL will carry out a detailed evaluation of the bids previously determined to be substantially responsive in order to determine whether the technical aspects are in accordance with the requirements set forth in the RFS documents. In order to reach such a determination, the NHPC/NHPC REL will examine and compare the technical aspects of the bids on the basis of the information supplied by the bidders, taking into account the

- (a) Overall completeness and compliance with NHPC/NHPC REL's requirements and quality, function and operation of any process control concept included in the bid. The bid that does not meet minimum acceptable standards of completeness, consistency and detail will be rejected for non-responsiveness.
- (b) Achievement of specified performance criteria by the Facilities
- (c) Deleted
- (d) Any other relevant factors, if any, listed in the RfS Document or that the NHPC/NHPC REL deems necessary or prudent to take into consideration.
- (e) If a bidder submits their bids for more than one category and during Techno-Commercial Evaluation it is observed that their bid is not responsive for all the submitted category, then NHPC/NHPC REL may, at its discretion, ask the bidder for their consent of consideration of their bids for that category in which bidder is Technocommercially responsive. Bidder has to give the necessary consent within the specified

time limit, failing which, the submitted bids will be treated as non responsive for all the applied category.

24. PRICE BID EVALUATION

- 24.1 (i) Bidders have to quote the requisite Tariff for the Category in which the bids are submitted. **The bidders will have to submit bids quoting a fixed levelised tariff for entire project duration of 25 years.** Price bids of only categorywise qualified and techno-commercially responsive bidders shall be opened.
 - (ii) Categorywise Price evaluation of the bids shall be carried out, subject to e-reverse auction:
 - a) Rebates (discounts) offered, in any form of bid proposal or on the forwarding letter shall not be taken cognizance under any circumstances.
 - iii). Categorywise Bid price of the bidder shall be calculated as stipulated in ITB Clause 24.1.(ii) against their individual categorywise price bid submission. Categorywise e-Reverse Auction shall be carried out as per the procedure defined in Part-III of ITB comprising the following:
 - i. Conditions of conducting e-RA after e-tendering (enclosed as Annexure-6)
 - ii. Business Rule for e-RA (enclosed as Annexure-7)
 - iii. Process Compliance form for e-RA (enclosed as Annexure-8)
 - iv. Price Confirmation form for e-RA (enclosed as Annexure-9)

During categorywise e-Reverse Auction, if no bid is received with in specified time, the NHPC/NHPC REL at its discretion may decide to close the e-Reverse Auction process of that category and proceed with results of e-tendering for that category. The decision regarding categorywise e-RA by NHPC shall be final and binding on all the bidders.

The Bidders are advised in their own interest to take utmost care while filling prices in the Price Bid .

The Price Bid of any category duly filled in spreadsheet in conformity with the tender specification shall be uploaded against the Price Schedule/BOQ of that category on the e-procurement portal only. Unit of Measure (UOM) is indicated in the BOQ. The BOQ is to be filled in by filling the Tariff in Rs./kWh to be filled in by the Bidder.

- 24.2 Technical and commercial evaluation of the Tenders will be carried out for the eligible Bidders on the basis of completeness and conformity with respect to the tender requirements. The Price bid of bidders for any category, whose Techno-commercial bids are not found eligible by the NHPC/NHPC REL for that category, shall not be opened/considered. The decision of the NHPC/NHPC REL is final and binding in this regard.
- 25. Deleted

26. CONTACTING THE NHPC/NHPC REL

- 26.1 Subject to ITB Clause 21, no Bidder shall contact the NHPC/NHPC REL on any matter relating to its bid, from the time of the opening of bids to the time the contract is awarded.
- 26.2 Information relating to the examination, clarification, evaluation, and comparison of bids and recommendations for the award of a contract shall not be disclosed to bidder(s) or any other persons not officially concerned with such process until the award to the successful Bidder has been announced. Any effort by a Bidder to influence the NHPC/NHPC REL in the NHPC/NHPC REL's bid evaluation, bid comparison or contract award decisions may result in rejection of the Bidder's bid.

F. AWARD OF CONTRACT

27. **AWARD CRITERIA**

- a) Subject to Clause 28, the NHPC/NHPC REL will award the contract on Categorywise basis to the successful bidder whose bid has been determined to be substantially responsive as per the "Conditions for conducting e-RA after e-tendering" which is enclosed as per Part III to the ITB. The NHPC/NHPC REL may request the substantially responsive Categorywise lowest evaluated bidders to attend the pre-award discussions / Post Tender Negotiations (PTN), if required. LOA (Letter of Award) will be issued only after JAKEDA/associated departments confirm acceptance of L1 bidder's quoted tariff.
 - b) The mode of contracting with the categorywise successful bidders will be as per stipulation outlined in the RfS and briefly indicated below:

Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding.

c) Entire responsibility with regard *to individual category for* Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding " will remain with RTD who decleared successful for that category and the RTD shall coordinate all activities of that category for smooth and timely completion of the project.

- 27.2 Deleted.
- 27.3 During categorywise e-Reverse Auction, if no bid is received within specified time, the NHPC/NHPC REL at its discretion may decide to close the e-Reverse Auction process and proceed with results of e-tendering. In such a situation, evaluated price shall be considered at the time of award of work to the lowest bidder and shall be considered to arrive at Contract Price.

28. NHPC/NHPC REL'S RIGHT TO ACCEPT ANY BID AND TO REJECT ANY OR ALL BIDS

The NHPC/NHPC REL reserves the right to accept or reject any bid, and to annul the bidding process and reject all bids at any time prior to award of contract for any reason including National Defence & Security conditions of the country where the site is located without thereby incurring any liability to the affected Bidder or bidders.

The Bidder shall bear all costs associated with the preparation and submission of its bid, and the NHPC/NHPC REL will in no case be responsible or liable for these costs, regardless of the conduct or outcome of the bidding process However, the bidders who wish to seek reasons for such decision of cancellation / rejection shall be informed of the same by NHPC/NHPC REL unless its disclosure reasonably could be expected to affect the sovereignty and integrity of India, the security, strategic, scientific or economic interest of the State, relation with foreign State or lead to incitement of an offence.

29. NOTIFICATION OF AWARD

- 29.1 Prior to the expiration of the period of bid validity, the NHPC/NHPC REL will notify the categorywise successful Bidders in writing by registered letter/ speed post / fax, that its bid has been accepted. The categorywise issued notification of awards will constitute the formation of the contract and will be considered for all purposes of execution of contract provisions till such time the signing of the PPA except as per provisions of Clause 30 of ITB.
- 29.2 Upon the categorywise successful Bidder's furnishing of the performance security pursuant to ITB Clause 31, the NHPC/NHPC REL will promptly notify the other bidders that their bids have been unsuccessful.
- 29.3 The categorywise unsuccessful Bidders shall also be informed simultaneously about their status of RfS on the tender portal.

29.4 SERVICE / PROJECT MANAGEMENT (PMC) CHARGES:

- 29.4.1 The evaluated lowest bidder (L1) shall have to pay Charges @ INR 1485/kW (Indian Rupees One thousand four hundred eighty five only/ kW) + Applicable taxes, to NHPC/NHPC REL towards project management charges (PMC).
- 29.4.2 The Service Charges will be calculated on the total Project capacity for which PPAs are signed by the RTD. These charges will be payable in two installments as follows:
 - (i) 50% to be paid within 30 days of issuance of LOAs by NHPC/NHPC REL, and,
 - (ii) Remaining 50% to be paid to NHPC/NHPC REL prior to signing of PPA. In case of any reduction/addition in the total Project capacity for which PPAs are executed, the second installment of 50% will be adjusted suitably. No interest shall be payable by NHPC/NHPC REL to RTD in case of refund of excess Service/PMC charges deposited by the RTD. Also, the tax component of the excess Service/PMC charges deposited by the RTD, shall not be refunded.

The due date for payment of PMC Charges shall be the date as intimated by NHPC/NHPC REL, which shall be at the most seven (07) days prior to the signing of PPA. The payment has to be made by the RTD in the form of DD/ Pay Order / NEFT/ RTGS mode to NHPC REL.

Any delay in depositing the said amount to NHPC REL as mentioned above within the stipulated time shall attract delay charges @18% per annum + applicable GST, levied on per day basis, on the amount corresponding to the 1st installment (i.e. 50% of PMC Charges), until (and including) the date of payment of the said amount, which shall not be later than 90 days after issuance of LOA. In case of delay in submission of the 1st installment beyond 210 days after issuance of LOA, the awarded capacity shall stand cancelled and BG against EMD will be encashed by NHPC/NHPC REL.

29.4.3 The 2nd installment shall be paid prior to signing of PPA, and the PPA shall be signed only after deposit of the second installment to NHPC REL.

30. SIGNING OF POWER PURCHASE AGREEMENT (PPA)

30.1 Power Purchase Agreement shall be signed between JAKEDA/its associated departments and successful Roof top bidder. A copy of Power Purchase Agreement to be executed between JAKEDA/its associated departments and the Selected Bidder is available at forms provided in the Volume-2 of RfS. The PPA shall be executed within 90 (Ninety) days of the date of issue of Letter of Award, if not extended by JAKEDA/its associated departments.

Separate PPA will be executed between Government client /organization and RTD for each Rooftop plant/ Project site or as per Government client decides.

Subsequent extension in this timeline shall be finalized as mutually agreed by the Client Organization and the RTD. PPA will be executed between Client Organization and selected bidder separately for each Category. The PPA shall be valid for a period of 25 years from date of commissioning of the Project.

Bidder has to make the complete and careful examinations of feasibility of government client Roof top plant/capacity estimation as per the load study of buildings and respective regulation of net metering/net billing before signing of PPA with client.

Any extension of the PPA period beyond 25 years shall be through mutual Agreement between the RTD and Government client /organization.

- 30.2 <u>Capacity identification and confirmation</u>: Subsequent to issuance of LoAs by NHPC REL, RTD will be allowed to carry out site visits and due diligence of the buildings offered by the Client Organization, and assess the Project sizing and estimated capacity installation under the cumulative capacity awarded. Within this period, the individual Project size to be set up on each building and the cumulative Project capacity for which PPAs are to be signed, will be finalized by both parties.
- 30.3 The Effective Date of PPA shall be the actual date of signing of PPA.
- 30.4 The PPAs shall be valid for a period of 25 years from the date of Project Commissioning, i.e. the Commercial Operation Date, or COD of the Project. Any extension of the PPA term beyond this shall be carried out through mutual agreement between the RTD and the JAKEDA/its associated departments.

30.5 **Limits on capacity modification:**

- 30.5.1 Capacity enhancement is also allowed with submission of additional revised BG amount as calculated. Such capacity enhancement will require consent from the concerned Department.
- 30.5.2 All such modifications in the awarded capacity are required to be concluded within 45 days from issuance of LoAs, unless extended by the Beneficiary Organization.
- Note: (i) The Performance Bank Guarantee, shall be submitted by the RTD prior to signing of PPA. Before signing of PPA, NHPC REL will verify the shareholding of the Project Company along with a copy of complete documentary evidence. If at this stage, it is found that the documents furnished by the RTDs are false/ misleading or misrepresented in any way, then the provisions contained in this RfS will be applicable.

(ii) NHPC REL will issue necessary go-ahead for signing of PPA, upon verification of compliance of pre-requisites by the RTD as brought out above. The PBG along with SFMS of the issued PBG will be submitted to NHPC REL for verification.

(iii) JAKEDA shall facilitate the PPA signing process and will bear no responsibility whatsoever on account of non-availability of indicated buildings and/or non-execution of PPA for any particular building.

31. PERFORMANCE SECURITY

31.1 Before signing of PPA, the successful Bidder shall furnish the Performance Security in the form of DD/BG/POI/Insurance Surety Bond of Rs. 2,475 / kW to NHPC REL as per format provided in the section "Forms and Procedures" of the RFS Documents or in any other format acceptable to the NHPC/NHPC REL.

The PBG should be valid for a period up to (& including) the date as on 12 months after the Scheduled Commencement of Supply Date of the Project (Commissioning of Project).

On receipt and after verification of the Total Performance Security in acceptable format and after signing of PPA, the earlier Bid Security towards EMD would be returned to the Bidder. It may be noted that PPA will be signed only upon successful verification of the PBG submitted by the RTD.

Any recovery from Performance Security shall be on Pro-Rata basis of the capacity of the respective category(s). The performance security shall not be released till liquidated damages, if any, is pending for recovery.

If the RTD does not submit the performance security within the stipulated period due to any valid reason, NHPC REL may grant time extension for submission of performance security based on the request of RTD.

In case, the RTD does not submit performance security without a valid reason, the NHPC/NHPC REL shall impose simple interest @12% per annum on the full amount of applicable performance security (alongwith applicable taxes, if any) for the period of delay in submission of performance security. The interest on delayed period shall be calculated on pro rata basis for number of delayed days.

The interest accrued shall be payable by the Contractor within 14 days from the date of intimation by NHPC/NHPC REL in form of Bank Demand Draft/ Banker Cheque in favour of 'NHPC REL', otherwise the same shall be recovered from any payment due or become due against bills / any other amount lying with NHPC/NHPC REL.

The delayed submission of Performance Security by the RTD shall be recorded in substantial completion and final completion certificates. Further, no claim for extension of time for completion period or any other type of claim on account of delayed submission of performance security shall be entertained.

- 31.2 If RTD fails to submit the Performance Security within 45 days from the date of Notification of Signing of Power Purchase Agreement (PPA), then following actions shall be taken against such RTD:
- (i) Award shall summarily be terminated
- ii) EMD/ Bid security shall be forfeited.
- iii) The bidder shall be debarred / banned to participate in the business dealings with NHPC/ NHPC REL for a period of one year.
- iv) The name of the Contractor shall be hosted on the NHPC/ NHPC REL website etc. as per existing norms of NHPC / Govt. of India.
- v) Such defaulted contractor shall not be eligible to participate in the bidding process of retender of this work.
- **Note:** (i) In case any extension is given to the Project, the corresponding extension needs to be made in the validity of Perforance Security. The Perforance Security are required to be submitted in the name NHPC REL.

(ii) The selected Bidder for the Project selected based on this RfS is required to sign PPA with JAKEDA/its associated departments within the timeline as stipulated in Clause 30 of the RfS.In case, JAKEDA/its associated departments offers to execute the PPA with the Selected Bidder and if the Selected Bidder does not submit the requisite documents as per the RfS, or does not meet eligibility criteria upon submission of documents or does not execute the PPA within the stipulated time period, then the Bank Guarantee equivalent to the amount of the EMD shall be encashed by NHPC REL from the Bank Guarantee available with NHPC REL (i.e. EMD or PBG) as liquidated damages not amounting to penalty, the selected Project shall stand cancelled and the selected Bidder expressly waives off its rights and objections, if any, in that respect.

(iii) In addition to the other remedies, this Performance Security can be encashed to recover any damages/dues of the Roof Top Developer in terms of the PPA.

(iv) The PBG of RTDs shall be returned to them, after successful commissioning of their projects as per Terms of PPA, after taking into account any liquidated damages due to delays in commissioning as per relevant Clauses of the RfS.

31.3 Bidders shall communicate the following bank details of NHPC REL to the issuing Bank for online confirmation of Bank Guarantee to be submitted in terms of this Clause:

Name of the beneficiary: NHPC Renewable Energy Limited. Account No.: 40852639765

IFSC Code: SBIN0017313

Address of the Bank: State Bank of India,

5th Floor, Redfort Capital, Parsvnath Towers, Bhai Veer Singh Marg, Gole Market, New Delhi-110001.

RTD is required to submit the SFMS receipt issued by the bank for PBG Confirmation.

32. CORRUPT OR FRAUDULENT OR COLLUSIVE OR COERCIVE PRACTICES

- 32.1 It is expected from the Bidders / suppliers / contractors that they will observe the highest standard of ethics during the procurement and execution of such contracts. In pursuance of this policy:
- (a) for the purposes of this provision, the terms set forth below shall mean as under:
 - i) "**Corrupt practice**" means the offering, giving, receiving or soliciting directly or indirectly of anything of value to influence the action of a public official in the procurement process or in contract execution; and
 - ii) "Fraudulent practice" means a misrepresentation/omission of facts in order to influence a procurement process or the execution of a contract to the detriment of the NHPC/NHPC REL, and includes collusive practice among Bidders (prior to or after bid submission) designed to establish bid prices at artificial non-competitive levels and to deprive the Borrower of the benefits of free and open competition;
 - iii) **"Collusive Practice**" means a scheme or arrangement between two or more bidders, with or without the knowledge of NHPC/NHPC REL, designed to establish bid prices at artificial, non-competitive levels.
 - iv) "**Coercive Practice**" means harming or threatening to harm, directly or indirectly, person or their property to influence their participation in a procurement process or affect the execution of Contract.
- (b) An agreement called Integrity Pact between the prospective bidders and the NHPC/NHPC REL shall be signed committing the persons / officials of both the parties, not to exercise any corrupt influence on any aspect of the Tender / Contract. The Independent External Monitor(s) (IEM) appointed by NHPC/NHPC REL shall oversee the compliance of obligation under the Integrity Pact.
- (c) A Bid may be rejected by the NHPC/NHPC REL if it is determined at any stage that the respective Bidder has engaged in corrupt or fraudulent or collusive or coercive or undesirable or restrictive practices or default commitment under Integrity Pact in competing for the contract in question.
- (d) The NHPC/NHPC REL may declare a firm ineligible, either indefinitely or for a stated period of time, if it at any time determines that the firm has engaged in corrupt or fraudulent or collusive or coercive practices in competing for or default commitment under Integrity Pact or in executing the contract.

- e) **Banning of Business Dealings:** It is not in the interest of NHPC to deal with Agencies who commit deception, fraud or other misconduct in the tendering process. The grounds on which Banning of Business Dealings can be initiated are as follows:
 - i) If the security consideration, including questions of loyalty of the Agency to NHPC so warrants;
 - ii) If the director/owner of the Agency, proprietor or partner of the firm, is convicted by a court of law for offences involving moral turpitude in relation to its business dealings with the Government or any other public sector enterprises, during last five years;
 - iii) If the Agency has resorted to Corrupt, Fraudulent, Collusive, Coercive practices including misrepresentation of facts and violation of the any provisions of the Integrity Pact provided in the Contract.
 - iv) If the Agency uses intimidation / threatening or brings undue outside pressure on NHPC or its official for acceptance / performances of the job under the contract:
 - v) If the Agency misuses the premises or facilities of the NHPC, forcefully occupies or damages the NHPC's properties including land, water resources, forests / trees or tampers with documents/records etc.
 - vi) If the Agency does not fulfill the obligations as required under the Contract and Violates terms & conditions of the contract which has serious affect for continuation of the Contract.
 - vii) If the work awarded to the agency has been terminated by NHPC due to poor performance of the contract in the preceding 5 years.
 - viii) If the Central Vigilance Commission, Central Bureau of Investigation or any other Central Government investigation Agency recommends such a course in respect of a case under investigation or improper conduct on agency's part in matters relating to the Company (NHPC) or even otherwise;
 - ix) On any other ground upon which business dealings with the Agency is not in the public interest.
 - x) If business dealings with the Agency have been banned by the Ministry of Power, Government of India OR any PSU/ any other authority under the MOP if intimated to NHPC or available on MOP Website, the business dealing with such agencies shall be banned with immediate effect for future business dealing except banning under Integrity Pact without any further investigation.
 - (Note: The examples given above are only illustrative and not exhaustive. The Competent Authority may decide to ban business dealing for any good and sufficient reason).

The procedure for banning of Business Dealings shall be governed as per NHPC's "Policy and Procedure for Banning Business Dealings". This policy is published in NHPC website under Integrity corner.

32.2 Deleted

33. INTEGRITY PACT

To improve transparency and fairness in the tendering process the NHPC/NHPC REL is implementing Integrity Pact.

The Integrity Pact, signed by all the prospective Bidders and the NHPC/NHPC REL, shall commit the persons/officials of both the parties, not to exercise any corrupt/

fraudulent/collusive/coercive practices in the Tendering process and also during implementation of the Contract. All Applicants shall enter into an Integrity Pact (to be executed on plain paper) with the NHPC/NHPC REL at the time of submission of their Bids. Only those Bidders who have entered into Integrity Pact with the NHPC/NHPC REL shall be eligible to participate in the bidding process. Entering into Integrity Pact as per Performa provided in the Section Forms & Procedure is a basic qualifying requirement.

The Integrity Pact digitally signed on behalf of the NHPC/NHPC REL is provided as per Annexure-2 in ITB. The Integrity Pact shall be downloaded, printed and signed by the Applicant and the hard copy shall be submitted.

Successful bidder shall submit duly executed Integrity pact on Non-Judicial Stamp paper of appropriate value prior to signing of Contract Agreement.

To oversee the compliance of obligation under the Integrity Pact, **Sh. Prabhash Singh** and **Dr. Vinod Aggarwal** has been appointed as Independent External Monitor (IEM) by the NHPC/NHPC REL. The Contact address of IEMs is as under:

| Sh. Prabhash Singh | Dr.Vinod Aggarwal, |
|--------------------------------|---------------------------------|
| E7/M702, Housing Board Colony, | B-103, Sarvodaya Enclave, |
| Arera Colony | 2 nd Floor, |
| Bhopal, | New Delhi – 110 017 |
| MP-462016 | Email : <u>arsv50@gmail.com</u> |
| Email : srgmhrbpl@gmail.com | |

34. Procurement from countries sharing land border with India

In line with the O.M. issued by the Department of Expenditure, Ministry of Finance, vide OM No. F.7/10/2021-PPD(1) dated 23.02.2023 and subsequent amendments and clarifications thereto, the Bidder shall meet the following criteria for its bid to be considered for evaluation under the RfS:

- (i) Any bidder from a country which shares a land border with India will be eligible to bid in this tender only if the bidder is registered with the Competent Authority (as defined in the OM as referred above).
- (ii) "Bidder" in this reference, means any person or firm or company, including any member of a onsortium, every artificial juridical person not falling in any of the descriptions of bidders stated herein before, including any agency branch or office controlled by such person, participating in this tender.
- (iii) "Bidder from a country which shares a land border with India" for the purpose of this clause, means:
 - a. An entity incorporated, established or registered in such a country; or
- **b.** A subsidiary of an entity incorporated, established or registered in such a country; or
- **c.** An entity substantially controlled through entities incorporated, established or registered in such a country; or
- d. An entity whose beneficial owner is situated in such a country; or
- e. An Indian (or other) agent of such an entity; or
- f. A natural person who is a citizen of such a country; or
- **g.** A consortium where any member of the consortium falls under any of the above.
- (iv) In support of the above, the Bidder shall be required to submit necessary Undertaking, as per Format available in the RfS.

Other provisions of the referred OM dated 23.02.2023, except Sl. 17 of the OM, will also be applicable for this tender. Any interpretation of the above clauses will be made in line with the referred OM, including subsequent amendments and clarifications thereto.

<u>35.</u> Public Procurement (Preference to Make in India) Order

Definitions:

Local Content: means the amount of value added in India which shall, unless otherwise prescribed by the Nodal Ministry, be the total value of the item procured (excluding net domestic indirect taxes) minus the value of imported content in the item (including all custom duties) as a proportion of the total value, in percent.

In case of items having sufficient local capacity and local competition, local content equal to or more than 50% shall be allowed only. The list of items having sufficient local capacity and local competition shall be as per Ministry of New & Renewable Energy order no. F. No. 283/22/2019-GRID SOLAR dated 23.09.2020 and its further revisions.

Class-I local supplier: means a supplier or service provider, whose Goods/ services/ Works offered for procurement, has local content equal to or more than 50%.

Class-II local supplier: means a supplier or service provider, whose Goods/ services/ Works offered for procurement, has local content equal to or more than 20% but less than 50%

Non Local Supplier: means a supplier or service provider, whose Goods/ services/ Works offered for procurement, has local content equal to or less than 20%.

Margin of purchase preference: means the maximum extent to which the price quoted by a 'Class-1 local supplier' may be above the L-1 for availing purchase preference. The margin of purchase preference shall be 20%.

Works: means all works as per Rule 130 of GFR-2017, and shall also include Turnkey works, Engineering, Procurement & Construction (EPC) Contracts and Services include System Integrator (SI) contracts.

Eligibility Criteria for Class I / II and Non Local Suppliers.

a) For procurement of Goods/Services/Works, irrespective of purchase value where there is sufficient local capacity and local competition, only Class-I local supplier shall be eligible to bid.

b) For procurement of Goods/Services/Works, not covered under a) above and having estimated value less than INR 200.00 Crore, Global tender enquiry shall not to be issued except with the approval of Competent Authority as designated by Department of Expenditure. Only Class-I and Class-II local suppliers shall be eligible to bid in procurements, except when Global tender enquiry has been issued. In Global tender enquiries, Non-local suppliers shall also be eligible to bid alongwith Class-I and Class-II local suppliers.

The above criteria shall be subject to compliance of following:

i) The bidder shall have to be an entity registered in India in accordance with law.

Indian subsidiaries of foreign bidders are eligible to participate in the bidding process provided they meet the qualifying criteria in terms of capability, competency, financial position, past performance etc. Further, Foreign bidders shall compulsorily set up their manufacturing units on a long term basis in India.

- ii) Foreign bidders can also participate in the bidding process provided they form Joint Venture with any bidder registered in India in accordance with law.
- iii) Country of origin of the equipment/material shall be provided in the bid.
- iv) The bids shall be in Indian National Rupees (INR) only in respect of local content.
- v) The bidder shall follow Indian laws, regulations and standards.
- vi) For supply of equipment / material from the country of origin other than India, the bidder shall submit performance certificate in support of satisfactory operation in India or a country other than the country of origin having climatic and operational conditions including ambient temperature similar to that of India for minimum one year.
- vii) The manufacturer/ supplier shall list out the products and components producing Toxic E-waste and other waste. It shall have an Extended Producers Responsibility (EPR) so that after the completion of the lifecycle, the materials are safely recycled / disposed of by the Manufacturer/ Supplier and for this, the Manufacturer/Supplier along with procurer has to establish recycling / disposal unit or as may be specified.
- viii) The bidder shall have to furnish a certificate regarding cyber security/safety of the equipment / process to be supplied/services to be rendered as safe to connect.
- ix) Wherever required, the Foreign supplier shall establish fully functional service centers in India and shall keep spares/material locally for future needs of Utilities.
- x) Arbitration proceedings shall be instituted in India only and all disputes shall be settled as per applicable Indian Laws.

Procedure for Purchase Preference

- a) Subject to the provisions of this Order and to any specific instructions issued by the Nodal Ministry or in pursuance of this Order, purchase preference shall be given to "Class-I local supplier" only in procurement undertaken by procuring entities in the manner specified here under.
- b) In the case of procurement of **goods** / **works**, which are **divisible in nature**, the 'Class-I local supplier' shall get purchase preference over 'Class-II local supplier' as well as 'Non-local supplier', as per following procedure(not applicable in this tender):
 - (i) Among all qualified bids, the lowest bid will be termed as L1. If L1 is 'Class-I local supplier', the contract for full quantity will be awarded to L1.

- (ii) If L1 bid is not a 'Class-I local supplier', 50% of the order quantity shall be awarded to L1. Thereafter, the lowest bidder among the 'Class-I local supplier' will be invited to match the L1 price for the remaining 50% quantity subject to the Class-I local supplier's quoted price falling within the margin of purchase preference, and contract for that quantity shall be awarded to such 'Class-I local supplier' subject to matching the L1 price. In case such lowest eligible 'Class-I local supplier' fails to match the L1 price or accepts less than the offered quantity, the next higher 'Class-I local supplier' within the margin of purchase preference shall be invited to match the L1 price for remaining quantity and so on, and contract shall be awarded accordingly. In case some quantity is still left uncovered by Class-I local suppliers, then such balance quantity shall be ordered to the L1 bidder.
- c) For procurement of goods / works, which are not divisible in nature, and in procurement of services where the bid is evaluated on price alone, the 'Class-I local supplier' shall get purchase preference over 'Class-II local supplier' as well as 'Non-local supplier', as per following procedure(applicable for this tender):
 - (i) Among all qualified bids, the lowest bid will be termed as L1. If L1 is 'Class-I local supplier', the contract will be awarded to L1.
 - (ii) If L1 is not 'Class-I local supplier', the lowest bidder among the 'Class-I local supplier', will be invited to match the L1 price subject to Class-I local supplier's quoted price falling within the margin of purchase preference, and the contract shall be awarded to such 'Class-I local supplier' subject to matching the L1 price.
 - (iii) In case such lowest eligible 'Class-I local supplier' fails to match the L1 price, the 'Class-I local supplier' with the next higher bid within the margin of purchase preference shall be invited to match the L1 price and so on and contract shall be awarded accordingly. In case, none of the 'Class-I local supplier' within the margin of purchase preference matches the L1price, the contract may be awarded to the L1 bidder.
 - (d) "Class-II local supplier" and "Non-local supplier" will not get purchase preference in any procurement, undertaken by procuring entities.

Verification of local content:

- (a) The 'Class-I local supplier'/ 'Class-II local supplier' at the time of tender, bidding or solicitation shall be required to indicate percentage of local content and provide selfcertification that the item offered meets the local content requirement for 'Class- I local supplier'/ 'Class-II local supplier', as the case may be. They shall also give details of the location(s) at which the local value addition is made.
- (b) In cases of procurement for a value in excess of INR 10.00 Crores, the 'Class-I local supplier'/ 'Class-II local supplier' shall be required to provide a certificate from the statutory auditor or cost auditor of the company (in the case of companies) or from a practicing cost accountant or practicing chartered accountant (in respect of suppliers other than companies) giving the percentage of local content.
- (c) Decision on complaints relating to implementation of this order shall be taken by the competent authority which is empowered to look in procurement related complaints relating to the procuring entity.

- (d) Nodal Ministries may constitute committees with internal and external experts for independent verification of self declarations and auditors / accountants certificate on random basis and in the case of complaints.
- (e) Nodal Ministries and procuring entities may prescribe fees for such complaints.
- (f) False <u>declarations</u> will be in breach of the <u>Integrity Pact</u> for which a bidder or its successors can be debarred for up to two years along with such other actions as may be permissible under law. A supplier who has been debarred by any procuring entity for violation of this Order shall not be eligible for preference under this Order for procurement by any other procuring entity for the duration of the debarment. The debarment for such other procuring entities shall take effect prospectively from the date on which it comes to the notice of other procurement entities.

Reciprocity Clause

- (i) When a Nodal Ministry / department identifies that Indian suppliers of an items are not allowed to participate and / or compete in procurement by any foreign government, due to restrictive tender conditions which have direct / indirect effect of barring Indian companies such as registration in the procuring country, execution of project of specific value in the procuring county etc., it shall provide such details to all its procuring entities including CMDs / CEOs of PSEs / PSUs, State Government and other procurement agencies under their administrative control and GeM for appropriate reciprocal action.
- (ii) Entities of countries which have been identified by the nodal ministry / department as not allowing Indian companies to participate in their Government procurement for any item related to that Nodal Ministry shall not be allowed to participate in Government procurement in India for all items related to that Nodal Ministry / Department, except for the list of items published by the Ministry / Department permitting their participation.

36. Planning and Designing in purview of Vulnerability Atlas of India

Vulnerability Atlas of India is a comprehensive document which provides existing hazard scenario for the entire country and presents the digitized State/ UT-wise hazard, maps with respect to earthquakes, winds and floods for district wise identification of vulnerable areas. It also includes additional digitized maps for thunderstorms, cyclones and landslides. The main purpose of this atlas is its use for disaster preparedness and mitigation at policy planning and project formulation stage.

This atlas is one of its kind single point source for various stake holders including policy makers, administrators, municipal commissioners, urban managers, engineers, architects, planners, public etc. to ascertain proneness of any city/location/site to multi-hazard which includes earthquakes, winds, floods, thunderstorms, cyclones and landslides. While project formulation, approvals and implementation of various housing, building and infrastructure schemes, this atlas provides for necessary information for risk analysis and hazard assessment.

The Vulnerability Atlas of India has been prepared by Building Materials and Technology Promotion Council under Ministry of Housing and Urban Affairs, Government of India and available at their website <u>www.bmtpc.org</u>.

It is mandatory for the bidders to refer Vulnerability Atlas of India for multi-hazard risk assessment and include the relevant hazard proneness specific to project location while planning and designing the project in terms of

- i) Seismic Zone (II to V) for earthquakes)
- ii) Wind Velocity (Basic Wind Velocity 55, 50, 47, 44, 39 & 33 m/s)
- iii) Area liable to floods and Probable max. Surge height
- iv) Thunderstorms history
- v) Number of cyclonic storms/ severe cyclonic storms and max sustained wind specific to coastal region.
- vi) Landslides incidences and Annual rainfall normal
- vii) District wise Probable Max. Precipitation.

37. Cyber Security for Power System Equipments

Ministry of Power (Government of India) has issued Circular no. No.25-1 1/6/2018-PG Dated 02.07.2020 on the vulnerabilities in the Power Supply System & Network mainly arise out of the possibilities of cyber attacks through malware / Trojans etc. embedded in imported equipment & Circular No. 12/34/2020-T&R dated 08.06.2021 on Testing power system equipment for use in the Supply System and Network in the country for Cyber Security. Bidders are advised in their own interest to follow these guidelines and circular issued by the Ministry from time to time.

.In order to protect the security, integrity and reliability of the strategically important and critical power supply system & network from cyber-attacks through malware / trozens etc. embedded in imported equipments:

- a. All equipments, components, and parts imported for use in Power Supply System and Network shall be tested in the country to check for any kind of embedded malware / Trojans / cyber threat and for adherence to Indian standards.
- b. All such testing shall be done in certified laboratories that will be designated by the Ministry of Power (MOP).
- c. Any import of equipment / components / parts from "prior reference" countries, as specified or by persons owned by, controlled by, or subject to the jurisdiction or the direction of these prior reference countries will require prior permission of the Government of India.
- d. Where the equipment / components / parts are imported from prior reference countries with special permission, the protocol for testing in certified and designated laboratories shall be approved by the Ministry of Power (MOP).

Any cost implication on account of above shall be borne by the contractor. Bidder shall submit the Self-Certificate for Cyber Security as per Attachment-13.

FOUR KEY INSTRUCTIONS FOR BIDDERS

Note The following 'FOUR KEY INSTRUCTIONS for BIDDERS' must be assiduously adhered to:-

- i) <u>Obtain individual Digital Signature Certificate (DSC or DC) well in advance of tender</u> <u>submission deadline.</u>
- ii) <u>Register your organization on CPPP well in advance of tender submission deadline.</u>
- iii) <u>Get your organization's concerned executives trained on CPPP well in advance of tender deadline.</u>
- iv) Submit your bids well in advance of tender submission deadline on portal (There could be last minute problems due to internet timeout, breakdown, etc.) While the first three instructions mentioned above are especially relevant to first-time users of portal, the fourth instruction is relevant at all times.
- Note:Electronic procurement system will not allow any Bidder to place their RFS after the
expiry of scheduled date & time. NHPC/ NIC-CPPP shall not be responsible for any
delays/ problems related to bandwidth, connectivity etc., which are beyond the control
of the NHPC/NIC-CPPPP.

Annexure – 1

FORM OF DECLARATION

DO HEREBY DECLARE THAT:

- 1. We have investigated the site and satisfied ourselves regarding the character of the work and local conditions that may affect the work or its performance and confirm that the work can be performed and completed as required in the contract. Further, we accept all risks directly or indirectly connected with the performance of the Contract.
- 2. We are familiar with all the requirements of the Contract.
- 3. We have not been influenced by any statement or promise of any person of the NHPC/NHPC REL but only the RFS Documents.
- 4. We are financially solvent.
- 5. We are experienced and competent to perform the Contract to the satisfaction of NHPC/NHPC REL.
- 6. We are familiar with all general and special laws, acts, ordinances, rules and regulations of the Municipalities, District, State and Central Government of India that may affect the work, its performance or personnel employed therein.
- 7. We have not been banned / de-listed / black listed / debarred from business on the grounds mentioned in para 6 of Guidelines on Banning of Business dealings (Annex-2) to Integrity Pact.
- 8. We hereby authorize the NHPC/NHPC REL to seek reference from the bankers of us for its financial position.
- 9. We undertake to abide by all labour welfare legislations.
- 10. It is hereby certified that proceedings for insolvency under the Insolvency and Bankruptcy code 2016, or as amended from time to time, have not started against us.
- 11. The statement submitted by us is true and correct.

For and on behalf of the Bidder

Dated:

Annexure- 2

(To be executed on plain paper at the time of submission of bid/ and on Non-judicial stamp paper of appropriate value by successful Bidder (Contractor) prior to signing of Contract Agreement) (Format of Integrity Pact)

PRE CONTRACT INTEGRITY PACT

Between

NHPC Limited, a company incorporated under the Companies Act 1956 and having its registered office at NHPC Office Complex, Sector-33, Faridabad-121003 (Haryana), hereinafter referred to as " The NHPC/NHPC REL" which expression shall mean and include, unless the context otherwise requires, his successors in office and assigns of the **First Part**.

And

M/s ______, a company/ firm/ individual (status of the company) and having its registered office at represented by Shri ______, hereinafter referred to as "The Bidder/Contractor" which expression shall mean and include, unless the context otherwise requires, his successors and permitted assigns of the **Second Part**.

WHEREAS the NHPC/NHPC REL proposes to procure under laid down organizational procedures, contract/s for ------(Name of the work/ goods/ services) and the Bidder/Contractor is willing to offer against NIT No. ------

WHEREAS the Bidder/Contractor is a private company / public company/Government undertaking/partnership/consortium/joint venture constituted in accordance with the relevant law in the matter and the NHPC/NHPC REL is a Public Sector Enterprise.

NOW, THEREFORE,

To avoid all forms of corruption by following a system that is fair, transparent and free from any influence/prejudiced dealings prior to, during and subsequent to the currency of the contract to be entered into with a view to:-

Enabling the NHPC/NHPC REL to obtain the desired said (work/ goods/ services) at a competitive price in conformity with the defined specifications by avoiding the high cost and the distortionary impact of corruption on public procurement, and

Enabling the Bidder(s)/Contractor(s) to abstain from bribing or indulging in any corrupt practice in order to secure the contract by providing assurance to them that their competitors will also abstain from bribing and other corrupt practices and the NHPC/NHPC REL will commit to prevent corruption, in any form, by its officials by following transparent procedures.

1.0 <u>Commitments of the NHPC/NHPC REL</u>

- 1.1 The NHPC/NHPC RELundertakes that no official of the NHPC/NHPC REL, connected directly or indirectly with the contract, will demand, take a promise for or accept, directly or through intermediaries, any bribe, consideration, gift, reward, favour or any material or immaterial benefit or any other advantage from the Bidder/Contractor, either for themselves or for any person, organization or third party related to the contract in exchange for an advantage in the bidding process, bid evaluation, contracting or implementation process related to the contact.
- 1.2. The NHPC/NHPC REL will, during the pre-contract stage, treat all the Bidders/Contractors alike, and will provide to all the Bidders/Contractors the same information and will not provide

any such information to any particular Bidder/Contractor which could afford an advantage to that particular Bidder/Contractor in comparison to other Bidders/Contractors.

- 1.3. All the officials of the NHPC/NHPC REL will report to the appropriate Authority any attempted or completed breaches of the above commitments as well as any substantial suspicion of such a breach.
- **2.0** In case any such preceding misconduct on the part of such official(s) is reported by the Bidder to the NHPC/NHPC REL with full and verifiable facts and the same is prima facie found to be correct by the NHPC/NHPC REL, necessary disciplinary proceedings, or any other action as deemed fit, including criminal proceedings may be initiated by the NHPC/NHPC REL or Independent External Monitor and such a person shall be debarred from further dealings related to the contract process. In such a case while an enquiry is being conducted by the NHPC/NHPC REL the proceedings under the contract would not be stalled.

3.0 <u>Commitments of the Bidder(s)/Contractor(s)</u>

The Bidder(s)/Contractor(s) commits themselves to take all measures necessary to prevent corrupt practices, unfair means and illegal activities during any stage of its bid or during any pre-contract or post-contract stage in order to secure the contract or in furtherance to secure it and in particular commit themselves to observe the following principles during participation in the tender process and during contract execution:-

- 3.1. The Bidder(s)/Contractor(s) will not offer, directly or through intermediaries, any bribe, gift, consideration, reward, favour, any material or immaterial benefit or other advantage, commission, fees, brokerage or inducement to any official of the NHPC/NHPC REL, connected directly or indirectly with the bidding process, or to any person, organization or third party related to the contract in exchange for any advantage in the bidding, evaluation, contracting and implementation of the contract.
- 3.2 The Bidder/Contractor further undertakes that it has not given, offered or promised to give, directly or indirectly any bribe, gift consideration, reward, favour, any material or immaterial benefit or other advantage, commission, fees, brokerage or inducement to any official of the NHPC/NHPC REL or otherwise in procuring the Contract or forbearing to do or having done any act in relation to the obtaining or execution of the contract or any other contract with NHPC/NHPC REL for showing or forbearing to show favour or disfavour to any person in relation to the contract or any other contract with NHPC/NHPC REL.
- 3.3 The Bidder(s)/Contractor(s) shall disclose the name and address of agents and representatives and Indian Bidder(s)/Contractor(s) shall disclose their foreign principals or associates.
- 3.4 The Bidder(s)/Contractor(s) shall, when presenting their bid, disclose any /all payments made, is committed to or intends to make to agents, brokers or any other intermediaries, in connection with this bid/award of the contract
- 3.5 Deleted.
- 3.6 The Bidder, either while presenting their bid or during pre-contract negotiations or before signing the contract, shall disclose any payments made, is committed to or intends to make to officials of the NHPC/NHPC REL or their family members, agents, brokers or any other intermediaries in connection with the contract and the details of services agreed upon for such payments.
- 3.7 The Bidder/Contractor will not collude with other parties interested in the contract to impair the transparency, fairness and progress of the bidding process, bid evaluation, contracting and implementation of the contract.

- 3.8 The Bidder/Contractor will not accept any advantage in exchange for any corrupt practice, unfair means and illegal activities.
- 3.9 The Bidder/Contractor shall not use improperly, for purposes of competition or personal gain, or pass on to others, any information provided by the NHPC/NHPC REL as part of the business relationship, regarding plans, technical proposals and business details, including information contained in electronic data carrier. The Bidder/Contractor also undertakes to exercise due and adequate care lest any such information is divulged.
- 3.10 The Bidder(s)/Contractor(s) commits to refrain from giving any complaint directly or through any other manner without supporting it with full and verifiable facts.
- 3.11 The Bidder(s)/Contractor(s) shall not instigate or cause to instigate any third person to commit any of the actions mentioned above.
- 3.12 If the Bidder/Contractor or any employee of the Bidder/Contractor or any person acting on behalf of the Bidder/Contractor, either directly or indirectly, is a relative of any of the officers of the NHPC/NHPC REL, or alternatively, if any relative of an officer of the NHPC/NHPC REL has financial interest/stake in the Bidder(s)/Contractor(s) firm(excluding Public Ltd. Company listed on Stock Exchange), the same shall be disclosed by the Bidder/Contractor at the time of filling of tender.

The term 'relative' for this purpose would be as defined in Section 2(77) of the Companies Act 2013.

- 3.13 The Bidder(s)/Contractor(s) shall not lend to or borrow any money from or enter into any monetary dealings or transactions, directly or indirectly, with any employee of the NHPC/NHPC REL.
- 3.14 The representative of the Bidders/Contractors signing Integrity Pact shall not approach the Courts while representing the matters to IEMs and he/she will wait their decision in the matter.
- 3.15 In case of Sub-Contracting, the bidder shall take the responsibility of the adoption of Integrity Pact by the Sub-Contractor.

4.0 <u>Previous Transgression</u>

4.1 The Bidder(s)/Contractor(s) declares that no previous transgression occurred in the last three years immediately before signing of this Integrity Pact, with any other company in any country in respect on any corrupt practices envisaged hereunder or with any Public Sector Enterprise in India or any Government Department in India.

5.0 <u>Earnest Money (Security Deposit)</u>

The provision regarding Earnest Money/Security Deposit as detailed in the Notice Inviting Tender (NIT) and Instruction to Bidders (ITB) section of the Bid Document is to be referred.

6.0 <u>Sanctions for Violations</u>

- 6.1 Any breach of the aforesaid provisions, before award or during execution by the Bidder/Contractor or any one employed by it or acting on its behalf such as to put their reliability or credibility in question shall entitle the NHPC/NHPC REL to take action as per the procedure mentioned in the "<u>Guidelines on Banning of Business Dealings</u>" attached as Annexure-2A and initiate all or any one of the following actions, wherever required:-
 - (i) To immediately call off the pre contract negotiations without assigning any reason or giving any compensation to the Bidder/Contractor. However, the proceedings with the other Bidder(s)/Contractor(s) would continue.

- (ii) The Earnest Money Deposit (in pre-contract stage) and/or Security Deposit/Performance Bond (after the contract is Signed) shall stand forfeited either fully or partially, as decided by the NHPC/NHPC REL and the NHPC/NHPC REL shall not be required to assign any reason thereof.
- (iii) To immediately cancel the contract, if already signed, without giving any compensation to the Contractor. The Bidder/Contractor shall be liable to pay compensation for any loss or damage to the NHPC/NHPC REL resulting from such cancellation/rescission and the NHPC/NHPC REL shall be entitled to deduct the amount so payable from the money(s) due to the Bidder/Contractor.
- (iv) Deleted.
- (v) To encash the Bank guarantee, in order to recover the dues if any by the NHPC/NHPC REL, along with interest as per the provision of contract.
- (vi) Deleted .
- (vii) To debar the Bidder/Contractor from participating in future bidding processes of NHPC Ltd., as per provisions of "Guidelines on Banning of Business Dealings" of NHPC Ltd. (Annexure-2A), which may be further extended at the discretion of the NHPC/NHPC REL.
- (viii) To recover all sums paid in violation of this Pact by Bidder(s)/Contractor(s) to any middleman or agent or broker with a view to securing the contract.
- (ix) In cases where irrevocable Letters of Credit have been received in respect of any contract signed by the NHPC/NHPC REL with the Bidder/ Contractor, the same shall not be opened/operated.
- (x) Forfeiture of Performance Security in case of a decision by the NHPC/NHPC REL to forfeit the same without assigning any reason for imposing sanction for violation of this Pact.
- 6.2 The NHPC/NHPC REL will be entitled to take all or any of the actions mentioned at para 6.1 (i) to (x) of this Pact also on the Commission by the Bidder/Contractor or any one employed by it or acting on its behalf (whether with or without the knowledge of the Bidder/Contractor), of an offence as defined in Chapter IX of the Indian Penal Code, 1860 or Prevention of Corruption Act, 1988 or any other statute enacted for prevention of corruption.
- 6.3 The decision of the NHPC/NHPC REL to the effect that a breach of the provisions of this Pact has been committed by the Bidder/Contractor shall be final and conclusive on the Bidder/Contractor. However, the Bidder/Contractor can approach the Independent External Monitor(s) appointed for the purposes of this Pact.

7.0 <u>Independent External Monitor(s)</u>

- 7.1 The NHPC/NHPC REL has appointed Independent External Monitors (hereinafter referred to as monitors) for this Pact after approval by the Central Vigilance Commission.
- 7.2 The task of the Monitors shall be to review independently and objectively, whether and to what extent the parties comply with the obligations under this Pact.
- 7.3 The Monitors shall not be subject to instructions by the representatives of the parties and perform their functions neutrally and independently. The Monitors report to CMD, NHPC.
- 7.4 Both the parties accept that the Monitors have the right to access all the Contract documents relating to the project/procurement, including minutes of meetings, whenever required. The right to access records should only be limited to the extent absolutely necessary to investigate the issue related to the subject tender/contract.

- 7.5 As soon as the Monitor notices, or has reason to believe, a violation of this Pact, he / she will so inform CMD, NHPC and request NHPC Ltd. to discontinue or take corrective action, or to take other relevant action. The Monitor can in this regard submit non-binding recommendations. Beyond this the Monitor has no right to demand from the parties that they act in a specific manner, refrain from action or tolerate action.
- 7.6 The Bidder(s)/Contractor(s) accepts that the Monitor has the right to access without restriction, to all Project documentation of the NHPC/NHPC REL including that provided by the Bidder/Contractor. The Bidder/Contractor will also grant the Monitor, upon his / her request and demonstration of a valid interest, unrestricted and unconditional access to their project documentation. The same is applicable to Subcontractor(s). The Monitor shall be under obligation treat the information contractual to and documents of the Bidder/Contractor/Subcontractor(s) with confidentiality. The Monitor has also signed declaration on 'Non-Disclosure Agreement' and of 'Absence of Conflict of Interest'. In case of any conflict of interest arising at an later date, the IEM shall inform CMD, NHPC and recuse himself / herself from that case.
- 7.7 The NHPC/NHPC REL will provide to the Monitor sufficient information about all meetings among the parties related to the project provided such meetings could have an impact on the contractual relations between the parties. The parties will offer to the Monitor the option to participate in such meetings as and when required.
- 7.8 The Monitor will submit a written report to the CMD, NHPC Ltd., within 8 to 10 weeks from the date of reference or intimation to him by the NHPC/NHPC REL/Bidder and should the occasion arise, submit proposals for correcting problematic situations.
- 7.9 The word 'Monitor' would include both singular and plural.

8.0 Facilitation of Investigation

In case of any allegation of violation of any provisions of this Pact or payment of commission, the NHPC/NHPC REL or its agencies shall be entitled to examine all the documents including the Books of Accounts of the Bidder/Contractor and the Bidder/Contractor shall provide necessary information and documents in English and shall extend all possible help for the purpose of such examination.

9.0 Law and Place of Jurisdiction

This Pact is subject to Indian Law. The place of performance and jurisdiction is the Registered Office of the NHPC, i.e. Faridabad (Haryana). The arbitration clause provided in the tender document/contract shall not be applicable for any issue/dispute arising under Integrity Pact.

10.0 Other Legal Actions

- 10.1 The actions stipulated in this Integrity Pact are without prejudice to any other legal action that may follow in accordance with the provisions of the extant law in force relating to any civil or criminal proceedings.
- 10.2 Changes and supplements as well as termination notice need to be made in writing.
- 10.3 If the Contractor is a partnership or a consortium or a joint venture, this pact must be signed by all partners of the consortium/joint venture.

11.0 Validity

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

- 11.1 The validity of this Integrity Pact shall be from date of its signing and expires for the Contractor 12 months after the last payment under the contract, and for all other Bidders 6 months after the contract has been awarded. Any violation of the same would entail disqualification of the bidders and exclusion from business dealings.
- 11.2 Should one or several provisions of this Pact turn out to be invalid, the remainder of this Pact shall remain valid. In this case, the parties will strive to come to an agreement to their original intention.
- 11.3 Issues like Warranty / Guarantee etc. shall be outside the purview of IEMs.
- 11.4 In the event of any contradiction between Integrity Pact and its Annexure, the clause in the Integrity Pact will prevail.

For & On behalf of the NHPC/NHPC REL

For & On behalf of the Bidder / Contractor

| Office Seal) | |
|-------------------|--|
| Place | |
| Date | |
| | |
| Witness1 | |
| Name and address) | |
| 2 | |

(Name and address)

Witness1.__

(Office Seal)

Place-----

2._____

(Name and address)

(Name and address)

Guidelines on Banning Business Dealings

1. Introduction

NHPC Limited (NHPC) deals with Agencies viz. parties/ contractors/ suppliers/ bidders, who are expected to adopt ethics of highest standards and a very high degree of integrity, commitments and sincerity towards the work undertaken. It is not in the interest of NHPC to deal with Agencies who commit deception, fraud or other misconduct in the tendering process and/or during execution of work undertaken. NHPC is committed for timely completion of the Projects within the awarded value without compromising on quality.

1.2 Since banning of business dealings involves civil consequences for an Agency concerned, it is incumbent that adequate opportunity of hearing is provided and the explanation, if tendered, is considered before passing any order in this regard keeping in view the facts and circumstances of the case.

2.0 Scope

- 2.1 NHPC reserves its rights to remove from list of approved suppliers / contractors (if such list exists) or to Suspend/Ban Business Dealings if any Agency has been found to be non / poor performing or have committed misconduct or fraud or anything unethical or any of its action(s) which falls into any such grounds as laid down in this policy.
- 2.2 The procedure of (i) Removal of Agency from the List of approved suppliers / contractors; (ii) Suspension and (iii) Banning of Business Dealing with Agencies, has been laid down in these guidelines.
- 2.3 These guidelines shall apply to all the units of NHPC.
- 2.4 These guidelines shall not be applicable in Joint Venture, Subsidiary Companies of NHPC unless they are assignees, successors or executor.
- 2.5 The suspension/banning except suspension/banning under Integrity Pact shall be with prospective effect, i.e. future business dealings.

3.0 Definitions

In these Guidelines, unless the context otherwise requires:

- i) "Agency /Party / Contractor / Supplier / Bidders/Vendors" shall mean and include a public limited company or a private limited company, a joint Venture, Consortium, HUF, a firm whether registered or not, an individual, cooperative society or an association or a group of persons engaged in any commerce, trade, industry, etc. "Party / Contractor/ Supplier / Bidder/Vendor" in the context of these guidelines is indicated as 'Agency'.
- *"Unit"* shall mean the Corporate Office, Project/ Power Station/ Regional Office/Liaison Office or any other office of NHPC.
- *iii) "Competent Authority"* and *'Appellate Authority'* shall mean the following:
 - a) For works awarded/under Tendering from corporate office (falling in the competency of CMD /Board of Directors).
 - Competent Authority: CMD
 - > Appellate Authority : Board of Directors
 - b) For works awarded/under Tendering from Corporate Office/Projects/ Power Stations/ Regional Offices/Liaison Offices (falling in the competency of Director /Executive Director)

- Competent Authority: Concerned Director/Executive Director as the case may be
- Appellate Authority: CMD /Concerned Director as the case may be
- c) For works awarded/under tendering from <u>Corporate Office</u> / Regional Offices / Projects/ Power Stations/ Liaison Offices (falling in the competency of CGM and below).
 - Competent Authority in case of works awarded/under tendering from Corporate Office/Regional office shall be CGM or GM of the concerned division as the case may be.
 - Competent Authority: Head of the Unit not below the rank of General Manager
 - > Appellate Authority: Next higher authority
- *iv)* **"Investigating Committee"** shall mean a Committee appointed by Competent Authority to conduct investigation.

4.0 Initiation of Suspension /Banning

Action for Suspension / Banning Business Dealings with any Agency shall be initiated by the department responsible for invitation of Bids/Engineer-in-charge after noticing the irregularities or misconduct on the part of Agency concerned. Besides the concerned department, Vigilance Department of each Unit/ Corporate Vigilance may also be competent to initiate such action.

5.0 Suspension of Business Dealings.

- 5.1 If the conduct of any Agency dealing with NHPC is under investigation, the Competent Authority may consider whether the allegations (under investigation) are of a serious nature and whether pending investigation, it would be advisable to continue business dealing with the Agency. If the Competent Authority, decides that it would not be in the interest to continue business dealings pending investigation, it may suspend business dealings with the Agency. The order of suspension would operate for a period not more than six months and may be communicated to the Agency as also to the Investigating Committee. The Investigating Committee may ensure that their investigation is completed and whole process of final order is over within such period. However if the investigations are not completed within six months ,the investigation committee shall put up the proposal to the competent authority for approval of extension of time maximum up to further three months with in which the committee shall conclude the proceedings.
- 5.2 The order of suspension shall be effective throughout NHPC in case of work falling in the Competency of CMD/ Board of Directors/Directors, in case of work falling in the competency of Executive Director suspension shall be effective throughout Region /Corporate office (in case the works awarded/under Tendering from Corporate office) .In case of work falling in the competency of HOP or below suspension shall be effective throughout the Project/Power Station and attached liaison offices/units and in case of work falling under the competency of CGM and below at Corporate

office suspension shall be effective at Corporate office. During the period of suspension, no business dealing shall be held with the Agency.

- 5.3 If the Agency concerned asks for detailed reasons of suspension, the Agency may be informed that its conduct is under investigation. It is not necessary to enter into correspondence or argument with the Agency at this stage.
- 5.4 It is not necessary to give any show-cause notice or personal hearing to the Agency before issuing the order of suspension.
- 5.5 The format for intimation of suspension of business dealing is placed at **Annexure-2B.**

6.0 Ground on which Banning of Business Dealings can be initiated

- 6.1 If the security consideration, including questions of loyalty of the Agency to NHPC so warrants;
- 6.2 If the director /owner of the Agency, proprietor or partner of the firm, is convicted by a Court of Law for offences involving moral turpitude in relation to its business dealings with the Government or any other public sector enterprises, during the last five years;
- 6.3 If the Agency has resorted to Corrupt, Fraudulent, Collusive, Coercive practices including misrepresentation of facts and violation of the any provisions of the Integrity Pact provided in the Contract.
- 6.4 If the Agency uses intimidation / threatening or brings undue outside pressure on NHPC or its official for acceptance / performances of the job under the contract;
- 6.5 If the Agency misuses the premises or facilities of NHPC, forcefully occupies or damages the NHPC's properties including land, water resources, forests / trees or tampers with documents/records etc.
- 6.6 If the Agency does not fulfil the obligations as required under the Contract and Violates terms & conditions of the contract which has serious affect for continuation of the Contract.
- 6.7 If the work awarded to the agency has been terminated by NHPC due to poor performance of the contract in the preceding 5 years.
- 6.8 If the Central Vigilance Commission, Central Bureau of Investigation or any other Central Government investigation Agency recommends such a course in respect of a case under investigation or improper conduct on agency's part in matters relating to the Company (NHPC) or even otherwise;
- 6.9 On any other ground upon which business dealings with the Agency is not in the public interest.
- 6.10 If business dealings with the Agency have been banned by the Ministry of Power, Government of India OR any PSU/ any other authority under the MOP if intimated to NHPC or available on MOP Website, the business dealing with such agencies shall be banned with immediate effect for future business dealing except banning under Integrity Pact without any further investigation.

(Note: The examples given above are only illustrative and not exhaustive. The Competent Authority may decide to ban business dealing for any good and sufficient reason).

7.0 **Procedure for Banning of Business Dealings**

- 7.1 An Investigating Committee shall be constituted by the authority competent to Ban the dealing comprising members from Engineering/Indenting department (convener), Finance, Law and Contract. The level of the committee members shall be CGM and above for works falling in the competency of CMD/Board of Directors, General Manager and above for the works falling in the competency of Director/ Executive Director and DGM/SM with at least one member of the level of General Manager for works falling in the competency of CGM and below.
- 7.2 The order of Banning of Business Dealings shall be effective throughout the NHPC. During the period of Banning of Business Dealings, no Business Dealing shall be held with the Agency.
- 7.3 The functions of Investigating Committee in each Unit to be appointed by the competent authority in line with para 3 (iii) shall, inter-alia include:
 - a) To study the report of the department responsible for invitation of bids and decide if a prima-facie case for banning exists, if not, send back the case to the Competent Authority.
 - b) To recommend for issue of show-cause notice (after vetting by legal deptt.) to the Agency by the concerned department as per clause 7.4 "Show Cause Notice".
 - c) To examine the reply to show-cause notice and call the Agency for personal hearing, if required.
 - d) To submit final recommendations to the Competent Authority for banning or otherwise including the period for which the ban would be operative considering the implications for NHPC on account of the act/omission on the part of the agency, intention of the agency as established from the circumstances of the case etc.

7.4 Show Cause Notice

Once the proposal for issuance of Show Cause Notice is approved by the Competent Authority, a 'Show Cause Notice' (as per format at **Annexure -2C** of these guidelines) shall be issued to the delinquent Agency by the Competent Authority or by a person authorized by the Competent Authority for the said purpose. The Agency shall be asked to submit the reply to the Show Cause Notice within 15 days of its issuance. Further, the Agency shall be given an opportunity for Oral hearing to present its case in person, if it so desires, and the date for Oral Hearing shall necessarily be indicated in the Show Cause Notice.

The purpose of issuing the Show Cause Notice is only that the Agency concerned shall be given an opportunity to explain their stand before any action is taken. The grounds on which action is proposed to be taken shall be disclosed to the Agency inviting representation and after considering that representation, orders may be passed. Such orders require only the satisfaction of the authority that passed the final orders.

If the Agency requests for inspection of any relevant document in possession of NHPC, necessary facility for inspection of documents may be provided.

During the conductance of oral hearing, only the regular duly authorized employees of Agency will be permitted to represent the Agency and no outsider shall be allowed to represent the Agency on its behalf.

Reply to the Show Cause Notice given by the Agency and their submissions in oral hearing, if any, will be processed by the Committee for obtaining final decision of the Competent Authority in the matter.

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

7.5 In case, no reply to Show Cause Notice is received from the Agency within stipulated time, further reminder shall be given with further period of 10 days thereafter if no reply is received action for processing ex-parte against the concerned Agency shall be initiated.

Speaking Order

The speaking order (reasoned order) for banning the business dealing with the Agency shall be issued by the Competent Authority or by a person authorized for the said purpose.

The decision regarding banning of business dealings taken after the issue of a Show Cause Notice and consideration of representation, if any, in reply thereto, shall be communicated to the Agency concerned along with a reasoned order. The fact that the representation has been considered shall invariably be mentioned in the communication. Also the fact that if no reply was received to the Show Cause Notice shall invariably be indicated in the final communication to the Agency. The format for intimation of banning of business dealing is placed at **Annexure-2D**.

7.6 Period of banning

In case banning is processed for violation of provisions of Integrity pact or due to corrupt or fraudulent practices, the Competent Authority shall decide on the period of banning on case to case basis depending on the gravity of the case and considering the implications for NHPC on account of the act/omission on the part of the Agency, intention of the Agency as established from the circumstances of the case etc. The period of banning shall not be less than 6 months and shall not exceed 2 years and in case termination of contract due to poor performance the period of banning shall be for 5 years. For contracts awarded to JV/Consortium, a constituent of the JV shall be permitted to participate in the bidding process if it has not been banned on grounds of its role and responsibility in the tendering process for which the Joint Venture has been banned in Past. In case if the Joint Venture which has been banned does not indicate the roles and responsibility of individual Partner(s) then, the partner of the banned Joint Venture shall only be allowed to participate in the bidding process if its participate in the partner of the banned Joint Venture shall only be allowed to participate in the bidding process if its participation share is less than 35%.

In case the information/documents submitted by Agency in competing for the tender found to be false/forged then NHPC, without prejudice to any other rights or remedies it may possess, shall recover from Agency the cost incurred in carrying out physical assessment for establishing veracity of such information/documents. In case Agency refuses to reimburse such cost to NHPC then banning period of Agency shall be extended by another one year.

7.7 Effect of Banning

As far as possible, the existing ongoing contract(s) with the Agency may continue unless the Competent Authority, having regard to the circumstances of the case, decides otherwise, keeping in view contractual and legal issues which may arise thereof. In case the existing Contracts are allowed to continue, the suspension/Banning of Business Dealing along with default of the Contractor shall be recorded in the experience certificate issued for the work.

The Agency, (after issue of the order of banning of business dealings) would not be allowed to participate in any future tender enquiry and if the Agency has already participated in tender process as stand-alone OR constituent of joint venture and the price bids are not opened, his techno-commercial bid will be rejected and price bid will be returned unopened. However, where the price bids of Agency have been opened prior to order of banning, bids of Agency shall not be rejected and tendering process shall be continued unless Competent Authority having regard to the circumstances of the case decides otherwise keeping in view the Contractual, Legal issues which may arise thereafter. However, in case the suspension /Banning is due to default of an Agency for the provisions under Integrity Pact and the Agency happens to be Lowest Bidder (L1), the tendering process shall be annulled and fresh tenders shall be invited.

During the Suspension/ Banning period, if it is found at any stage that Agency has participated in tender enquiry under a different name then such Agency would immediately be debarred from the tender/contract and its Bid Security/Performance Security would be forfeited. Payment, if any, made shall also be recovered.

After Suspension/ Banning order, the Suspended/ Banned Agency shall not be allowed to participate as Sub-Vendor/Sub-Contractor in the tenders.

Further, if the Suspended/ Banned agency is an approved Sub-Vendor under any Contract for such equipment/component/service, the Main Contractor shall not be permitted to place work order/Purchase order/Contract on the Suspended/ Banned agency as a Sub-Vendor /Sub-Contractor after the date of Suspension/ Banning even though the name of the party has been approved as a Sub-Vendor /Sub-Contractor earlier.

There would be no bar on procuring the spares and awarding Contracts towards Annual Maintenance (AMC)/ O&M/ Repair works on Agencies pertaining to the packages for which they have been banned provided the Equipment has been supplied by such Agency.

Banning of business dealing shall not be applicable to the Subsidiary Company of the Banned agency provided subsidiary company has not participated on the strength of the Banned agency. However, in case of a default by a Sub-Contractor, the banning shall be applicable to the Sub-Contractor as well as the Lead Partner of the concerned JV or the Sole bidder as the case may be.

7.8 Hosting at NHPC website

The concerned unit shall forward the name and details of the Agency (ies) banned along with period and reasons of banning to IT&C Division of Corporate Office for displaying the same on the NHPC website.

8.0 Appeal against the Decision of the Competent Authority.

The Agency may file an appeal against the order of the Competent Authority banning of business dealing before Appellate Authority. Such an appeal shall be preferred within 30 (Thirty) Days from the date of receipt of the order of banning of business dealing. Appellate Authority would consider the appeal if convinced may constitute another committee for further investigation. The investigation Committee constituted by the Appellate Authority shall study the report of the previous investigating committee and reply submitted by the Agency while filing its case for appeal and call the Agency for personal hearing , if requested by the Agency. Based on the recommendation of the committee Appellate Authority shall pass appropriate Speaking (Reasoned) order in line with Sub-Para 7.5 above which shall be

communicated to the Agency as well as the Competent Authority (as per format enclosed as **Annexure-2E** with these guidelines).

9.0 Circulation of the names of Agencies with whom Business Dealings have been banned

The name of the concerned banned agency shall also be shared with MOP and other PSU in the sector and all the units of NHPC.

The provisions of this policy supersede and will have overriding effect on all earlier guidelines, procedures & system issued for the similar purpose.

Annexure-2B

(Format for Intimation of Suspension of business dealing)

BY REGD. POST/SPEED POST/COURIER

No.....

Date.....

То

M/s

Attn.: Shri

Sub: Intimation of Suspension of Business Dealings

Dear Sir,

Whereas the work of was awarded to your firm vide letter of award no ...dt..... amounting to Rs. OR In response to NHPC NIT (e-tender / physical tender) nodt. you have submitted your bid. (strike out whichever is not applicable) Whereas the conduct of your firm in respect of the following is under investigation:

Brief of the default

"Whereas the Competent Authority prima facie considered the allegations (under investigation) are of a serious nature and decided pending investigation, it is not in the interest of the corporation to continue business dealing with your firm

This order shall have the following effects:

- (i) Further business dealings with your firm is Suspended within Region/Project/Unit/wide NHPC. The order of Suspension is effective with immediate effect and would operate for a period of six months or till the investigation is completed and whole process of final order is over within such period. However, if investigations are not completed in six months' time, the Competent Authority may extend the period of Suspension
- (ii) During the period of Suspension, no business dealing shall be held with your firm. No enquiry / bid / tender shall be issued to your firm nor will the bids submitted by your firm be entertained.
- (iii) In cases where tenders have already been issued to you and price bids are yet to be opened, the Price Bid submitted by you shall not be opened and BG/EMD, if any, submitted by you shall be returned.
- (iv) In cases where tenders have already been issued to you and Price Bids have already been opened, the tendering process shall be continued
- (v) In case of ongoing contracts between you & NHPC, (including cases were contract has already been awarded before the issue of Suspension order) you will be required to continue with the execution and perform as per terms of the contract.
- (vi) (a) In case the Firm is in Joint Venture the following would also be applicable:

i) Participation of Agency in Joint Venture

Tenders in which your firm has been proposed as Joint Venture Partner by any of the bidders and price bids have been opened prior to Suspension of your firm in such

cases the tendering process shall not be annulled on this ground and the Agency shall be permitted to continue as Partner in the Joint Venture for such bidding. However where event of Price Bid opening has not taken place prior to Suspension/Banning of Agency then in such case Agency shall not be permitted to participate as Partner in the Joint Venture.

ii) Banning of joint Venture:

As the Joint Venture is Banned, your firm intends to bid as Partner(s) of Joint Venture in bidding process then it shall be permitted to participate in the bidding process if it has not been Banned on grounds of its role and responsibility in the tendering process for which the Joint Venture has been Banned in Past. In case if the Joint Venture which has been Banned does not indicate the roles and responsibility of individual Partner(s) then, the partner of the Banned Joint Venture shall only be allowed to participate in the bidding process if its participation share is less than 35%.

(b) Your firm shall not be allowed to participate as Sub-Vendor/Sub-Contractor in the tenders.

Further if your firm is an approved Sub-vendor under any Contract for such equipment/component/service, the Main Contractor shall not be permitted to place work order/Purchase order/Contract on your agency as a Sub-Vendor/Sub-Contractor after the date of Suspension/ Banning even though the name of the party has been approved as a Sub-Vendor/Sub-Contractor earlier.

- (c) There would be no bar on procuring the spares and awarding Contracts towards Annual Maintenance (AMC)/ O&M/ Repair works on Agencies pertaining to the packages for which they have been Banned provided the Equipment has been supplied by such Agency.
- (d) Banning of business dealing shall not be applicable to the Subsidiary company of the Banned agency provided subsidiary company has not participated on the strength of the Banned agency.

On expiry of the above period of Suspension/Banning, you may approach...

(Indicate the concerned procurement department), with request for revocation of the order mentioning inter-alia the steps taken by you to avoid recurrence of misconduct which has led to Suspension.

Yours faithfully,

For & On behalf of NHPC.

Note: Strike out whichever is not applicable

Annexure-2C

(Format of Show Cause Notice)

BY REGD. POST/SPEED POST/COURIER

| No |
|-------------|
| Date |
| То |
| M/s |
| |
| Attn.: Shri |

Sub: Show Cause Notice

Ref :

Dear Sir,

You are hereby required to Show Cause in writing within 15 days from the date hereof why Business Dealing with your firm should not be banned / your firm is placed in the Banning List (as the case may be) and be debarred from entering into any contracts with NHPC for the following reasons:

(Give Reasons)

Your reply (if any) should be supported by documents and documentary evidence which you wish to rely in support of your reply. In case, you desire to present your case in person to NHPC, a personal hearing shall be conducted onat hours for which prior intimation be furnished to this office. Should you fail to reply to this Show Cause Notice within the time and manner aforesaid, it will be presumed that you have nothing to say and we shall proceed accordingly.

Your reply, if any, and the documents / documentary evidence given in support shall be taken into consideration prior to arriving at a decision.

Yours faithfully,

For & On behalf of NHPC.

Annexure-2D

(Format for Intimation of Banning of Business Dealing)

BY REGD. POST/SPEED POST/COURIER

No.....

То

 $M\!/\!s \ \ldots \ldots$

Attn.: Shri

Sub: Intimation of Suspension of Business Dealings

Dear Sir,

Whereas the work of was awarded to your firm vide letter of award no ...dtd..... amounting to Rs. OR In response to NHPC NIT (e-tender / physical tender) nodt. you have submitted your bid .(strike whichever is not applicable)

Whereas the Competent Authority had prima facie considered the allegations as detailed below are of a serious nature and decided to conduct investigation.

"Brief of the Default may be mentioned"

- (In order to make the Intimation of Banning of Business Dealing Speaking Order (reasoned order), the issue of a Show Cause Notice and consideration of representation in reply to show cause notice , opportunity of personal hearing, if any, shall be communicated to the Agency concerned along with a reasoned order. The order for Banning should also contain the reasons with detailed justification for conclusion of decision to Ban the Agency. Also the fact that if no reply to the Show Cause Notice was received or request for personal hearing was not made shall invariably be indicated in the communication to the Agency. The above order shall mention the grounds considering violation of any provision of Integrity Pact, any ground mentioned in Guidelines of Banning of Business Dealings, default by the agency under fraudulent practice / or any unethical practice and/or violation of any provision of Tender/Contract Condition having serious implications.)

This order shall have the following effects:

- i) Further business dealings with your firm is banned with immediate effect. The order of Banning would operate for a period ofyears/month Competent Authority may extend the period of Banning.
- ii) During the period of Banning, no Business Dealing shall be held with your firm. No Enquiry / Bid / Tender shall be issued to your firm nor will the bids submitted by your firm be entertained.

Date.....

- iii) In cases where tenders have already been issued to you and Price Bids are yet to be opened, the Price Bid submitted by you shall not be opened and BG/EMD, if any, submitted by you shall be returned.
- iv) In cases where tenders have already been issued to you and price bids have already been opened, the Tendering Process shall be continued.
- (v) In case of ongoing contracts between you & NHPC, (including cases where contract has already been awarded before the issue of Banning order) you will be required to continue with the execution and perform as per terms of the contract.
- (vi) (a) In case the Firm is in Joint Venture the following would also be applicable:

i) Participation of Agency in Joint Venture

Tenders in which your firm has been proposed as Joint Venture Partner by any of the bidders and price bids have been opened prior to Banning of your firm in such cases the tendering process shall not be annulled on this ground and the Agency shall be permitted to continue as Partner in the Joint Venture for such bidding. However where event of price bid opening has not taken place prior to Banning of Agency then in such case Agency shall not be permitted to participate as Partner in the Joint Venture.

ii) Banning of joint Venture:

As the Joint Venture is banned, your firm intends to bid as Partner(s) of Joint Venture in bidding process then it shall be permitted to participate in the bidding process if it has not been banned on grounds of its role and responsibility in the tendering process for which the Joint Venture has been banned in Past. In case if the Joint Venture which has been banned does not indicate the roles and responsibility of individual Partner(s) then, the partner of the banned Joint Venture shall only be allowed to participate in the bidding process if its participation share is less than 35%.

(b) Your firm shall not be allowed to participate as Sub-Vendor/Sub-Contractor in the tenders.

Further if your firm is an approved Sub-vendor under any Contract for such equipment/component/service, the Main Contractor shall not be permitted to place work Order/Purchase Order/Contract on your agency as a Sub-Vendor/Sub-Contractor after the date of Banning even though the name of the party has been approved as a Sub-Vendor/Sub-Contractor earlier.

- (c) There would be no bar on procuring the spares and awarding Contracts towards Annual Maintenance (AMC)/ O&M/ Repair works on Agencies pertaining to the packages for which they have been banned provided the Equipment has been supplied by such Agency.
- (d) Banning of Business Dealing shall not be applicable to the Subsidiary company of the Banned Agency provided subsidiary company has not participated on the strength of the Banned Agency. However, in case of a default by a Sub-Contractor, the banning shall be applicable to the Sub-Contractor as well as the Lead Partner of the concerned JV or the Sole bidder as the case may be.

On expiry of the above period of Banning, you may approach..... (Indicate the concerned procurement department), with request for revocation of the order mentioning inter-alia the steps taken by you to avoid recurrence of misconduct which has led to Banning.

Further if you desire to appeal against this order you may do so within 30 days from the date of issue of this order to the appellate authority as here under:

Appellate Authority: Designation: Address:.... Ph . no. e-mail :

Yours faithfully,

For & On behalf of NHPC.

Note: Strikeout whichever is not applicable

Annexure-2E

(Format for communication of Appellate Decision on Suspension/Banning Order)

BY REGD. POST/SPEED POST/COURIER

No.....

Attn.: Shri

Sub: Suspension Banning of Business Dealings - Intimation of decision of Appellate Authority

Ref: 1. Order dated Placing M/s on Suspension/Banning List by NHPC;

2. Your Appeal reference Dt......

Dear Sir,

[

This has reference to the order dt..... placing you on Suspension/Banning List and your appeal petition reference dt.... on the same.

After considering the findings of the Original Authority in order and submissions made by you in your appeal, and the documents/documentary evidences available on record, it has been decided finally that :

- * There is no infirmity in the order of the Original Authority, and the allegations stand substantiated and the Suspension/Banning for the period of years/month from the date of order, as ordered by the original Authority is upheld,
- * Considering your submissions and the evidence available on record, there is enough justification to annul the order of the original authority.

(*** Incorporate any one of the above as applicable

^{##}In order to make the Communication of Appellate Authority on Banning of Business Dealing Speaking Order (reasoned order), the fact that the representation of the agency has been considered and reference of grounds brought forward by the Agency in his defense and if any opportunity of personal hearing has been given to the Agency as a part of principle of natural justice shall invariably be mentioned in the communication, The order for Banning should also contain the reasons with detailed justification for conclusion of decision to Ban the Agency. Also the fact that if no sufficient ground has been furnished shall invariably be indicated in the final communication to the Agency. In case the option for Banning of Agency or reduction of Time Period for Banning of Agency is exercised then the above order shall mention the grounds considering violation of any provision of Integrity Pact, any ground mentioned in Guidelines of Banning of Business Dealings, default by the agency under fraudulent or any unethical practices and/or violation of any provision of Tender/Contract Condition having serious implications.)

Yours faithfully,

For & On behalf of NHPC

ANNEXURE-3

QUALIFICATION INFORMATION SHEETS

Annexure-3A

General Information

Bidders are requested to complete the information

| 1. | Name of firm | | | | |
|----|--|-------------------------|-----------|-------|-------------|
| 2. | Head office address | Country: | | | |
| 3. | Telephone | Contact I | Person(s) | | |
| | Fax | Name | | | |
| | Email | Title/Pos | ition | | |
| 4. | Place of incorporation / registration Date | | | | |
| 5. | Legal status of firm Field of speciality in business | | | | |
| 6. | Number of present permanent employees: | | | | - |
| | | Detail | Civil | Other | Non |
| | | Nos. | Engineers | Engrs | Engineering |
| | | | | | |
| 7. | Quality assurance system in head office | Certified | by: | | |
| 8. | Agent or representative in INDIA Name Address Telephone | (if exists) Fax & Ei | | | |
| | | | | | |

Date

Signature

Annexure -3B

SUMMARY OF WORK EXPERIENCE

I. SUMMARY OF LIST OF WORKS COMPLETED AS ON DATE OF SUBMISSION OF TENDER

| S. No | Description of the Work & contract No. | Contract Awarded by | Contract Value | Date of contract awarded | Date of completion as per agreement | Date of commencement of work | Actual completion Date | Final value of Contract |
|----------|---|---------------------------|----------------|-----------------------------|--|------------------------------------|---------------------------|----------------------------|
| | | | | | | | | |
| | | | | | | | | |
| | | | | | | | | |
| | | | | | | | | |

Note:

i.) Supportive documents / certificates from the Organizations' with whom they worked/ are working should be enclosed.

Signature of Bidder Seal of the Company

Annexure -3C

SELF CERTIFICATE FOR LOCAL CONTENT BY BIDDER

We [name of Bidder] hereby confirm, in respect of item(s) quoted in Schedules of Quantities & Prices of the subject tender, that Local Content is equal to (.....% of local content to be mentioned by bidder) as below:

| BOQ | Item | (% of local | Locations at which local value addition |
|-----|------|-------------|---|
| No. | | content) | is made |
| | | | |
| | | | |
| | | | |

Note:

i.) The Bidder shall be required to provide a certificate from the statutory auditor or cost auditor of the company (in the case of companies) giving the percentage of local content.

> Signature of Bidder Seal of the Company

{Ref. Sub-Article No.2(ii) of ITB}

<u>Annexure – 4</u>

FORM OF SELF DECLARATION OF INELIGIBILITY UNDERTAKING

I/ We, M/s(Name of Bidder) hereby certify that I/we are not under the declaration of ineligibility for banning/ de-listing/ black listing/ debarment from business on the ground mentioned in Para 6 of Guidelines on banning of Business dealing (Annexure-2) to integrity pact, (ITB Clause 2 of RFS document) at the time of submitting of bid and shall intimate immediately of any such banning/de-listing/black listing/ debarring after submission of bid till the award of Tender.

(Seal & Signature of the Bidder)

Annexure -5A

DELETED

Annexure-5B

Deleted

(III): PROCEDURE FOR E-REVERSE AUCTION (e-RA)

- > <u>ANNEXURE -6 : Conditions of conducting e-RA after e-tendering</u>
- > <u>ANNEXURE -7: Business Rule for e-RA</u>
- ANNEXURE -8: Process Compliance form for e-RA
- > <u>ANNEXURE -9: Price Confirmation form for e-RA</u>

ANNEXURE - 6

1.0 Conditions for conducting e-RA after e-tendering

- 1.1 Categorywise E-RA shall be implemented at Application Service Provider's Portal selected by NHPC.
- 1.2 Categorywise E-RA shall be followed after e-tender, if number of eligible bidders at categorywise price bid stage is at least 2 (two). e-RA shall be carried out on the basis of "**Tariff**" quoted by bidder in Rs./kWh. Tariff (in Rs/kWh) shall be quoted upto two places of decimal only. If tariff is offered with more than two digits after decimal, it shall be truncated and considered upto two places of decimal (in Rs/kWh) in evaluation. The H1 bidder of each category (whose derived "**Tariff**" as detailed in this paragraph is highest) will not be allowed to participate in further Reverse Auction process provided minimum three bidders are left in that category after removal of H1 bidder.
- 1.3 All those bidders who are found to be technically qualified, will be ranked according to the quoted tariff with highest ranking (Rank L1) to that bidder who has quoted the lowest tariff bid and so on.

If the tariff bid is same for two or more bidders, then the bidder with higher net worth during FY-2023-24 shall be given higher ranking. In case the net worth of bidders are also same, then relative ranking shall be decided through draw of lottery.1.4 The Base Bid Price for e-Reverse Auction of a category shall be the lowest "**Tariff Bid**" of that category as derived in Clause 1.2 & 1.3 as described above. The minimum decrement value for Reverse Auction is 0.01 INR i.e. one paisa.

In cases where no bidder accepts to quote lower than the base "**Tariff Bid**," e-RA of that category is to be treated as invalid and the NHPC/NHPC REL proceeds for fresh e-RA for that category or tenders shall be evaluated based on "**Tariff Bid**" as derived in Clause 1.2 & 1.3 in already opened e-Tender. The decision of the NHPC/NHPC REL in this regard shall be final and binding on the bidders.

- 1.5 After categorywise e-Reverse Auction and evaluating the categorywise lowest bid of the bidder based on lowest **"Tariff Bid"**, the categorywise award shall be processed as per the procedure defined in the RFS document.
- 1.6 In case of failure of cateorywise e-RA process for any reasons, e-tender results will be processed for award. Further, NHPC/ NHPC REL shall reserve the right to call the categorywise *L1 bidder for further process/negotiation or to cancel the e-reverse auction process/ re-tender at any time.
- *L1= Successful Bidder after e-Reverse Auction (as the case may be)

ANNEXURE-7

Business Rules for e-Reverse Auction

- 1) NHPC have presently made arrangements with Application Service Provider, who will be NHPC's authorized Service Provider for e-Reverse Auction (e-RA). Please go through the guidelines given below and submit your acceptance to the same.
- 2) Bidders who are found technically and commercially acceptable and whose Price Bid is as per the Bid Condition after e-tender are eligible to participate in e-RA. The intimation of schedule date & time to conduct categorywise e-RA shall be sent to eligible bidders in advance. Categorywise E-Reverse Auction (e-RA) shall be conducted by Application Service Provider on scheduled date and time. Eligible bidders shall participate from their own offices / place of their choice. Internet connectivity shall be ensured by the respective bidders themselves. Designated officials i.e. Tender Inviting Authority of NHPC shall have access to portal and shall witness e-RA process.
- 3) Application Service Provider shall arrange to demonstrate / train (if not trained earlier) bidders' nominated person(s), without any cost. They will also explain all the Rules related to e-Reverse Auction/Business Rules Document to be adopted along with training manual.

The Application Service Provider (ASP) shall arrange to demonstrate/ train (if not trained earlier) bidders' nominated person(s) through ONLINE training by <u>ppts</u> / Audio Visual file without any cost. These training materials shall remain available on the portal all the time whenever bidders access the portal for such purposes. If bidders require any explanation in regard to the Rules related to e-Reverse Auction / Business Rules Document to be adopted along with bid manual, the same shall be explained to the bidder through any means of communication i.e. telephonic/email.

4) Before start of e-RA process, NHPC will provide the template calculation sheet (Spreadsheet) which will help bidders to arrive at "**Tariff Bid**" for the total scope of work.

Rank of the bidders would be displayed as per the "**Tariff Bid**" quoted by the Bidder in **Price Schedule.**

5) The Start Bid Price (SBP) for e-Reverse Auction in a category shall be Lowest Bidder's derived "Tariff Bid" of that category **as** above.

6) **Procedure of e-Reverse Auctioning:**

Dynamic Template Bidding (Rank Disclosed)

- i. In 'Dynamic Template Bidding (Rank Disclosed)', bidders shall enter the "Tariff Bid" in line with clause 4 above for the total scope for subject Package. The value of decrement must be 0.01 INR i.e. one paisa and its multiple thereof of "Lowest Tariff Bid" after e-tender.
- ii. In the "Dynamic Bidding" tie bid is not allowed on Lowest "Tariff Bid" at e-RA Portal, i.e. if any higher price ranked vendor places bid equal to Lowest "Tariff Bid" in e- reverse auction, their bid will be rejected. The bidder is required to click on "Rejected" to know the reason of bid rejection. In that case, the bidder will have to instantly reduce their price within 5 minutes to submit lower quoted "Tariff Bid" otherwise the "Tariff Bid" will be considered as higher bid or rejected bid (system does not support tie bids in this type of auction).

- iii. The "**Tariff Bid**" received under e-RA shall be evaluated on the basis of evaluation method specified in the RFS document and the final evaluated Tariff thus obtained shall be used for comparison of bids
- 7) Within one (1) hour after conclusion of e-RA, successful bidder shall furnish confirmation through his registered e-mail id to the Tender Inviting Authority for present tender the following:

Final "**Tariff Bid**" quoted in Reverse Auction In case, there is any variation between the final "**Tariff Bid**" quoted in Reverse Auction and the document received after Auction, the first i.e. "**Tariff Bid**" in Auction will be taken as final offered "**Tariff**" by the bidder.

8) Auction shall be for a period of **60 minutes**. The bidding continues with an auto extension of **10 min** time if any Bidder quotes a further lower price within final 10 minutes of specified closing time of auction.

In final countdown and auto extension time, the Lowest "**Tariff Bid**" at e-RA Portal would be visible to all the vendors who have placed valid bids. However, real identity of the bidder would not be disclosed during entire e-reverse auction process. The bidding could continue even after the final run till the time there are no changes in prices within aforesaid final 10 minutes.

- 9) During Auction, if no bid is received within the specified time, NHPC/ NHPC REL, at its discretion, may decide to close the reverse auction process and proceed with results of e-tendering.
- 10) Each Bidder shall be assigned **Unique User Name & Password** by Application Service Provider. Bidders are advised to change the Password and edit the information in the Registration Page after the receipt of initial Password from Application Service Provider to ensure confidentiality. All bids made from the Login ID given to the bidder will be deemed to have been made by the bidder.
- 11) Bidders shall be able to view the following on their screens along with the necessary fields during the auction.
 - a. Rank of the respective bidder[@]
 - b. Bid placed by the respective bidder.
 - [@] "Tariff Bid" of Lowest bidder at e-RA Portal will be displayed in the last 10 minutes and in auto extension time.
- 12) Consequent upon completion of e-Reverse Auction, NHPC's/ NHPC REL decision on categorywise award of contract shall be final and binding on all the bidders.
- 13) NHPC/ NHPC REL shall be at liberty to call the *categorywise* *L1 bidder for further process/ negotiation/ cancel the e-reverse auction process/ re-tender at any time, without assigning any reason thereof.
- 14) NHPC/NHPC REL/ Service Provider shall not have any liability to bidders for any interruption or delay in access to the site irrespective of the cause.
- 15) NHPC/ NHPC REL along with Service Provider can decide to reschedule or cancel any Auction; the bidders shall be informed accordingly.
- 16) The bidder shall not involve himself or any of his representatives in price manipulation of any kind directly or indirectly by communicating with other bidders. The bidder shall also not divulge either his bid or any other exclusive details of NHPC/ NHPC REL to any other party. If it has come to notice of NHPC/ NHPC REL/Service Provider that bidders

have colluded while submitting price in e-RA, the same be dealt as per provision of Integrity Pact.

- 17) Any updation/ modification in the e-RA process brought out above shall be conveyed in advance to the bidders before e-RA.
- 18) Other terms and conditions shall be as per the RFS documents.

*L1 = Lowest Evaluated Bidder for that category after e-Reverse Auction (as the case may be)

ANNEXURE - 8

Process Compliance Form for e-Reverse Auction

(Bidders are required to submit this on their Company's Letter Head and sign & stamp)

To,

___Name, Designation, Address of Tender Inviting Authority____

| Contact Person: | ••••• |
|------------------------|-------|
| E-mail: | ••••• |

Sub: Agreement to the Process related Terms and Conditions

Dear Sir,

This has reference to the Terms & Conditions for the e-Reverse Auction mentioned in the Business Rules for <.....Name of tender Package & Tender Specification No.:>

This letter is to confirm that:

- 1. The undersigned is authorized representative* of the Company.
- 2. We have studied the e-Reverse Auction Terms & Conditions and the Business Rules governing the e-Reverse Auction as mentioned in the RFS document and confirm our agreement to them.
- 3. We also confirm that we shall take the training on the auction tool and shall understand the functionality of the same thoroughly, before start of e-RA.
- 4. We also confirm that we will submit Tariff confirmation within an hour of conclusion of e-Reverse Auction through company email ID.
- 5. Our registered email id is _____

We hereby confirm that we will honor the Bids placed by us during the auction process.

With regards

Signature with company seal

Name –

Company / Organization -

Designation within Company / Organization -

Address of Company / Organization -

*Person having power of attorney for the subject package

ANNEXURE-9

Format for submitting Price Confirmation for e-Reverse Auction

(To be submitted on Company Letter Head duly signed and stamped)

REF No. To,

Date:

____Name, Designation, Address of Tender Inviting Authority____

Contact Person: E-mail:

Re: Price Confirmation for thee-Reverse Auction of

Auction Notice Dated:

For Procurement of... [...Name of package.....].

For M/s NHPC LIMITED E-Tender Id no.....

Reference above, we here by confirm that we have given our final offer in the e-Reverse Auction System as per the template, which is attached herewith duly signed and stamped from our end:

(i) Final TARIFF: .

Signature: Name: Designation :

Seal of the Company:

VOLUME 1

INFORMATION FOR BIDDERS

INFORMATION FOR BIDDERS (IFB)

1.0 INTRODUCTION

- **1.0** NHPC Renewable Energy Limited (NHPC REL) is a wholly owned subsidiary of NHPC Limited. Incorporated in February 2022, NHPC REL is mandated for development of Renewable Energy, Small Hydro and Green hydrogen Projects. The Company is actively pursuing its entry into Solar, Wind, Green Hydrogen, Small Hydro and other Renewable Generation.
- 1.1 The scheme 'PM SURYA GHAR: MUFT BIJLI YOJANA' was launched by the Government on 13.02.2024 to promote adoption of rooftop solar in the residential sector, providing Central Financial Assistance for consumers. Operational guidelines of the scheme were issued by MNRE on 03.07.2024.
- 1.2 NHPC Limited has been allocated of 8 nos. of Central Ministries/ Departments by MNRE and five nos. States viz. Haryana, UT of J&K, Sikkim, Manipur and Nagaland (communicated by MNRE O.M. dated 25.01.2024, for taking up installation and coverage of roof top solar capacity in mission mode (under PM Surya Ghar Yojana) by Dec'2025 as per details below:

| Α. | Central Ministries allocated to NHPC REL: |
|----|---|
| 1 | Ministry of Food Processing Industries |
| 2 | Ministry of Heavy Industries and Public Enterprises |
| 3 | Ministry of Science and Technology |
| 4 | Ministry of Water Resources |
| 5 | Ministry of Drinking Water & Sanitation |
| 6 | Ministry of Environment, Forests & Climate Change |
| 7 | Ministry of Law and Justice |
| 8 | Ministry of Micro, Small & Medium Enterprises |
| | |
| В. | States allocated to NHPC REL: |
| 1 | Haryana |
| 2 | Sikkim |
| 3 | UT of J&K |
| 4 | Nagaland |
| 5 | Manipur |

Bidders are advised to collect all necessary information under the scheme as mentioned in the MNRE operational guidelines for government buildings No. 318/17/2024-Grid Connected Rooftop (Part-2) dated 03rd July 2024 and

must apprise themselves of all the relevant guidelines as and when notified and NO ADDITIONAL CLAIM shall be entertained in this regard.

- 1.3 In view of above, NHPC Renewable Energy Limited (hereinafter called NHPC REL) intends to invite proposal for setting up Grid connected rooftop projects for Departments of Law Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in UT of J&K through JAKEDA (hereinafter called Government Client / Institution / Organization), at various locations under RESCO mode. Bidding shall be Tariff based Competitive Based Process. After conclusion of bidding process, Jammu & Kashmir Energy Development Agency and aforesaid departments (Annexure-B) shall enter into Power Purchase Agreement (PPA) with the selected Bidder, based on this RfS for purchase of Solar power for a period of 25 years from the Scheduled Commencement of Supply Date of the project based on the terms, conditions and provisions of the RfS and PPA.
- 1.4 NHPC REL shall be Project Management Consultant (PMC) for the RESCO project set up by RTD.

2.0 OVERVIEW OF RFS:

- 2.1 Roof Top Developers (hereafter referred to as RTDs) selected by NHPC REL based on this RfS, shall set up roof top Solar plants on RESCO Mode in accordance with the provisions of this RfS document and standard Power Purchase Agreement (PPA) attached in this RfS.
- 2.2 Government Institutions shall enter into PPA with successful RTD for a period of 25 years from the date as per the provisions of PPA. The Bidders will be free to avail fiscal incentives like Accelerated Depreciation, Concessional Customs and Excise Duties, Tax Holidays etc. as available for such projects. The same will not have any bearing on comparison of bids for selection. As equal opportunity is being provided to all bidders at the time of tendering itself, it is up to the Bidders to avail various tax and other benefits. No claim shall arise on NHPC REL for any liability if bidders are not able to avail fiscal incentives and this will not have any bearing on the applicable tariff.
- 2.3 Bidder shall quote a single tariff for entire of its projects' capacity offered, which shall be applicable for all the 25 years.
- 2.4 No separate Central Financial Assistance is envisaged for implementation of the Projects selected under this RfS.

3.0 GUIDELINES FOR IMPLEMETATION OF THE RFS

3.1 Ministry of New & Renewable Energy (MNRE) vide its Office Memorandum No. 318/17/2024- Grid Connected Rooftop (part-2) dated 03.07.2024 issued Guidelines for Saturation of Government Buildings under PM Surya Ghar: Muft Bijli Yojana. Under the scheme, all Government rooftops under the administrative control of Central Government Ministries/Departments, including autonomous bodies, subordinate offices etc. shall be saturated with rooftop solar to the extent that is technically feasible by 31st December, 2025.

4.0 BRIEF SCOPE OF WORK

The Solar Photo Voltaic (PV) installation at Various locations of Departments of Law Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in UT of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding as mentioned in Annexure-A on shadow free area in such a way that the generation is maximized. The final plant capacity shall be as per the detailed engineering and approved design.

Tentative Capacity is: (a) Roof Top Solar Capacity: 34.67 MWac

(b) No of categories : 3 Nos.

Detailed scope of work is provided in the RFS document.

Tentative list of buildings/locations where rooftop solar is proposed is placed at **Annexure-A** of IFB.

The Departentwise contact details for any assistance related to site visit is enclosed as **Annexure-B** of IFB.

Annexure-A of IFB

| Sl. No. | Category | Estimated Rooftop Potential (kW) | No. Buildings |
|------------|-------------------------------------|-------------------------------------|------------------|
| 1. | Category-1 for (>= 5kW to < 20kW) | 9005.37 | 980 |
| 2. | Category-2 for (>= 20kW to < 100kW) | 11965.31 | 327 |
| 3. | Category-3 for (>= 100kW and above) | 13701.91 | 64 |
| | TOTAL= | 34672.60 | 1371 |

of

The combined identified capacity are as under:

Capacity Sr. No. Department **Sub Department** District Ministry (kW) J.E PHE Office Talli 1 Jal Shakti Department PHE, Jammu Jammu/R.S Pura 19.23 Morh, near power Pump Room Sub 2 PHE, Jammu 19.23 Jal Shakti Department Ramban/Gool **Division Gool** PHE Division 3 Jal Shakti Department PHE, Kashmir Anantnag/Anantnag 19.23 Bijbehara 4 Department of Home Department of Home pP Reshiwa Awantipwa 19.00 5 DPO Doda 19.00 Department of Home Department of Home Doda **RPHQ Bulding AT** 19.00 6 Department of Home Department of Home Doda Doda Department of Forest Doda/ 7 PCCF & HoFF BO Hut (Roat) 18.98 and Ecology Doda 8 Department of Home Department of Home PS Lakhanpur Kathua/ 18.75 Addu SP Office 9 Department of Home Rojouri Nowshera 18.50 Department of Home Residence Forest Protection Department of Forest Staff Quarter at 10 Shopian Shopian 18.25 Sindho Shirmal and Ecology Force Department of Forest Dogra Hall Forest 11 PCCF & HoFF Jammu/Jammu 18.13 and Ecology complex Jammu 12 Department of Home Department of Home North Srinagar 18.00 HeadQuarter Directorate office 13 18.00 Department of Home HomeGuard,SDRF,Ci Srinagar I&FC Division Irrigation & Flood 14 Jal Shakti Department Srinagar/South 18.00 Control, Kashmir Building, Batmalo Department of Forest Samba/ Bari JK Pollution Control **Divisional Office** 15 18.00 and Ecology Samba North J&K Committee Brahamana Department of Forest Forest Protection Armory Room 16 18.00 Ganderbal and Ecology Force Fatehpora Department of Forest Forest Protection 17 18.00 Munshiyat Anantnag and Ecology Force ROs residential Department of Forest Ramban/ 18 PCCF & HoFF 18.00 and Ecology quarter (Batote) Batote Department of Forest Ramban/ 19 PCCF & HoFF Guard Hut (Kriya) 18.00 and Ecology Ramban Department of Forest Inspection Hut Ramban/ 20 PCCF & HoFF 18.00 and Ecology (Gandhri) Ramban Department of Forest Forest Rest House Doda/ 21 PCCF & HoFF 18.00 and Ecology (Goha) Goha Department of Forest Range Office Dachhan 22 PCCF & HoFF Kishtwar/Dachhan 18.00 and Ecology at Sirshi Dacchan Department of Forest Log Hut Bungus 23 PCCF & HoFF Kupwara/ Qalamabad 18.00 and Ecology Department of Forest 24 PCCF & HoFF Range office Mattan Anantnag 18.00 and Ecology 25 Jal Shakti Department PHE, Kashmir Water Works Srinagar Srinagar 17.69 Fire and Emergency 26 Department of Home F&ES Hiranagar 17.50 Service Department-

Category-1 for (>= 5kW to < 20kW)

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|--|---------------------|------------------|
| 27 | Department of Home | HeadQuarter HomeGuard,SDRF,Ci | HG/CD & SDRF Rajouri | | 17.50 |
| 28 | Jal Shakti Department | PHE, Jammu | Sub division office Surankote | Poonch/Surankote | 17.31 |
| 29 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House (Patnitop) | Ramban/ Batote | 17.00 |
| 30 | Department of Forest and Ecology | PCCF & HoFF | Shanti Hut Paahalgam | Anantnag | 17.00 |
| 31 | Department of Forest and Ecology | PCCF & HoFF | Forest Information Centre Bhadarwah | Doda/ Bhadarwah | 16.67 |
| 32 | Department of Home | Department of Home | Adm.BLock Equipement Store | jammu/Bhalwal | 16.66 |
| 33 | Department of Forest and Ecology | PCCF & HoFF | Divisional Office Kulgam | Kulgam / Kulgam | 16.50 |
| 34 | Department of Forest and Ecology | Wildlife Protection | Check Post Kiyar | Kishtwar / Dachhan | 16.20 |
| 35 | Department of Home | Department of Home | PP Sangam | Anantnag | 16.16 |
| 36 | Jal Shakti Department | PHE, Kashmir | PHE Groundwater Division Srinagar | Srinagar/Srinagar | 16.15 |
| 37 | Department of Home | Department of Home | Family Quarter 'B' Block | Kathua | 16.00 |
| 38 | Department of Home | Department of Home | Family Quarter 'C' Block | Kathua | 16.00 |
| 39 | Department of Home | Department of Home | Family Quarter Block | Kathua | 16.00 |
| 40 | Department of Home | Department of Home | Four Storey Building | Kathua | 16.00 |
| 41 | Department of Home | HeadQuarter HomeGuard,SDRF,Ci | SDRF Doda | | 16.00 |
| 42 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | SSD Tangdar PHE Residential Quarter at | Kupwara/Karnah | 16.00 |
| 43 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut, Sananghat | Kathua/Basohli | 16.00 |
| 44 | Department of Forest and Ecology | PCCF & HoFF | Check Post Kuh Nallah | Udhampur / Ramnagar | 16.00 |
| 45 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Chishoti | Kishtwar/Paddar | 16.00 |
| 46 | Department of Forest and Ecology | PCCF & HoFF | Division Office Marwah at Sarkoot | Kishtwar/Kishtwar | 16.00 |
| 47 | Department of Forest and Ecology | PCCF & HoFF | New Horse Stable TTR PC DEPOT | Srinagar | 16.00 |
| 48 | Department of Forest and Ecology | PCCF & HoFF | Recently Established Forest Hut Badrakali | Kupwara/Zachaldara | 16.00 |
| 49 | Department of Forest and Ecology | PCCF & HoFF | CP Dhobiwan | Pulwama/ Tral | 16.00 |
| 50 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Pr. District & Sessions Judge,Jammu(Main | Jammu | 16.00 |
| 51 | Department of Forest and Ecology | PCCF & HoFF | Inspection Hut | Doda/ Bhadarwah | 15.87 |
| 52 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Add. Spl. Mmob. Magistrate | Poonch | 15.80 |
| 53 | Department of Forest and Ecology | PCCF & HoFF | Chowkidar Hut (Tanger) | Ramban/ Ramban | 15.75 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|--------------------|-------------------------------------|----------------------|------------------|
| 54 | Jal Shakti Department | PHE, Jammu | Filtration Plant Maitra | Ramban/Ramban | 15.38 |
| 55 | Jal Shakti Department | PHE, Jammu | Lab office building Mandi | Poonch/Haveli | 15.38 |
| 56 | Jal Shakti Department | PHE, Kashmir | PHE Division Chadoora | Budgam/Chadoora | 15.38 |
| 57 | Jal Shakti Department | PHE, Kashmir | PHE Division Sopore | Baramulla/Dangerpora | 15.38 |
| 58 | Jal Shakti Department | PHE, Kashmir | SSD Tangdar | Kupwara/Karnah | 15.38 |
| 59 | Jal Shakti Department | PHE, Kashmir | PHE Mechanical Division Srinagar | Srinagar | 15.38 |
| 60 | Department of Forest and Ecology | PCCF & HoFF | BO Hut (Dedni) | Doda/ Doda | 15.09 |
| 61 | Department of Home | Department of Home | Hazratbal | Srinagar | 15.00 |
| 62 | Department of Home | Department of Home | PS Lal Bazar | Srinagar | 15.00 |
| 63 | Department of Home | Department of Home | PS Zadibal | Srinagar | 15.00 |
| 64 | Department of Home | Department of Home | Police Station Soura | Srinagar | 15.00 |
| 65 | Department of Home | Department of Home | KOT/MHC/DAR Office Building | Anantnag | 15.00 |
| 66 | Department of Home | Department of Home | Canteen Welfare Building | Anantnag | 15.00 |
| 67 | Department of Home | Department of Home | Janwans Barrack | Anantnag | 15.00 |
| 68 | Department of Home | Department of Home | P/S Aishmuqam | Anantnag | 15.00 |
| 69 | Department of Home | Department of Home | P/S Dooru | Anantnag | 15.00 |
| 70 | Department of Home | Department of Home | P/S mattan | Anantnag | 15.00 |
| 71 | Department of Home | Department of Home | Police Station Rajpora | Pulwama | 15.00 |
| 72 | Department of Home | Department of Home | PCR Building DPO Kulgam | Kulgam | 15.00 |
| 73 | Department of Home | Department of Home | PCR Building DPO Kulgam | Kulgam | 15.00 |
| 74 | Department of Home | Department of Home | DySP (DAR)OFice | Kulgam | 15.00 |
| 75 | Department of Home | Department of Home | Bloxk-E DPI. Kulgam | Kulgam | 15.00 |
| 76 | Department of Home | Department of Home | HlgkGDP1 K ulgam | Kulgam | 15.00 |
| 77 | Department of Home | Department of Home | Block-H DP1. Kulgam | Kulgam | 15.00 |
| 78 | Department of Home | Department of Home | New Kot DPI. Kulgam | Kulgam | 15.00 |
| 79 | Department of Home | Department of Home | P/S Kund | Kulgam | 15.00 |
| 80 | Department of Home | Department of Home | P/S Yaripora | Kulgam | 15.00 |
| | | | | | |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--------------------|---|--|----------------------|------------------|
| 81 | Department of Home | Department of Home | P/S Bchibag | Kulgam | 15.00 |
| 82 | Department of Home | Department of Home | P/S Manzgam | Kulgam | 15.00 |
| 83 | Department of Home | Department of Home | PP Frisa | Kulgam | 15.00 |
| 84 | Department of Home | Department of Home | SSP Residence | Shopian | 15.00 |
| 85 | Department of Home | Department of Home | PCR Building DYSP DAR | Shopian | 15.00 |
| 86 | Department of Home | Department of Home | PIS Shopian Main Building | Shopian | 15.00 |
| 87 | Department of Home | Department of Home | P/S lmamsghib Main Building | Barbughlmamsahib | 15.00 |
| 88 | Department of Home | Department of Home | P/S Hcerpora Main Building | Shopian | 15.00 |
| 89 | Department of Home | Department of Home | P/S Zainapora Main Building | Zainapora | 15.00 |
| 90 | Department of Home | Department of Home | P/S Keller Main Building | Keller | 15.00 |
| 91 | Department of Home | Department of Home | SDPO Pampore Residence/PC Pampore | Pampore | 15.00 |
| 92 | Department of Home | Department of Home | SHO office/Sarishta | Baramulla/Kunrer | 15.00 |
| 93 | Department of Home | Department of Home | PP Fidarpora | PD Sopore/Dangrwacha | 15.00 |
| 94 | Department of Home | Department of Home | PCR Building DPO Kulgam | Kulgam | 15.00 |
| 95 | Department of Home | Department of Home | Main office Vijaypur | Samba | 15.00 |
| 96 | Department of Home | Department of Home | Administrative Block. | Ganderbal /Kangan. | 15.00 |
| 97 | Department of Home | Department of Home | Trainees Double Storey (Barrack) | Kathua | 15.00 |
| 98 | Department of Home | Department of Home | Adm, mock of IRROK Zewan | Srinagar | 15.00 |
| 99 | Department of Home | Department of Home | C Block Barrack/4 story Building | Anantnag/Mattan | 15.00 |
| 100 | Department of Home | Department of Home | Velecomm block triple story building. | Anantnag/Mattan | 15.00 |
| 101 | Department of Home | Department of Home | IRP-19th Bn. Hqrs Doda | Doda | 15.00 |
| 102 | Department of Home | Department of Home | Quarter Guard Building of IRP-12 Bn. | Samba | 15.00 |
| 103 | Department of Home | Department of Home | "1RP.19h Bn.Hars Kathua" | kathua | 15.00 |
| 104 | Department of Home | Department of Home | "Directorate office Srinagar" | srinagar | 15.00 |
| 105 | Department of Home | Fire and Emergency Service Department- | F&ES Shangus | | 15.00 |
| 106 | Department of Home | Fire and Emergency Service Department- | F&ES Hqrs Pulwama chatapora | | 15.00 |
| 107 | Department of Home | HeadQuarter HomeGuard,SDRF,Ci | HG/CD & SDRF | | 15.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|----------------------------|---|--------------------------|------------------|
| 108 | Jal Shakti Department | PHE, Kashmir | Hydraulic Circle Kupwara | Kupwara/Handwara | 15.00 |
| 109 | Department of Forest and Ecology | Forest Protection Force | Asstt Direcor Residence | Budgam, Khansahib | 15.00 |
| 110 | Department of Forest and Ecology | Forest Protection Force | Guard Barrack Single story Hardupanzoo | Budgam, Khansahib | 15.00 |
| 111 | Department of Forest and Ecology | Wildlife Protection | Range Officer Building Tangmarg | Baramulla | 15.00 |
| 112 | Department of Forest and Ecology | JK Forest Development | Divisional Office, Workshop Division, | Jammu/Bahu | 15.00 |
| 113 | Department of Forest and Ecology | JK Forest Development | Range Office, Sale Depot Auction | Jammu/Jammu | 15.00 |
| 114 | Department of Forest and Ecology | PCCF & HoFF | Range Forest Office, Basohli | Kathua/Basohli | 15.00 |
| 115 | Department of Forest and Ecology | PCCF & HoFF | Forest Check post old Bridge Akhnoor | Jammu/Akhnoor | 15.00 |
| 116 | Department of Forest and Ecology | PCCF & HoFF | Range Forest Office Sunderbani and R.o | Rajouri/Sunderbani | 15.00 |
| 117 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut Sujaligala | Rajouri/teryath | 15.00 |
| 118 | Department of Forest and Ecology | PCCF & HoFF | Mali Hut Nadpur Nursery | Rajouri/Nowshera | 15.00 |
| 119 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Sunderbani | Rajouri/Sunderbani | 15.00 |
| 120 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Sunderbani and | Rajouri/Sunderbani | 15.00 |
| 121 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Sunarka | Rajouri/Sunderbani | 15.00 |
| 122 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Makol | Rajouri/Sunderbani | 15.00 |
| 123 | Department of Forest and Ecology | PCCF & HoFF | Mali Hut Brevi Nursery | Rajouri/Kalakote | 15.00 |
| 124 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Dabber | Rajouri/Nowshera | 15.00 |
| 125 | Department of Forest and Ecology | PCCF & HoFF | Mali Hut Bagla Nursery | Rajouri/Qila Darhal | 15.00 |
| 126 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut Tori | Udhampur / Ramnagar | 15.00 |
| 127 | Department of Forest and Ecology | PCCF & HoFF | Community Hall Basantgarh | Udhampur / Basantgarh | 15.00 |
| 128 | Department of Forest and Ecology | PCCF & HoFF | Range Office Panchari | Udhampur/ Panchari | 15.00 |
| 129 | Department of Forest and Ecology | PCCF & HoFF | Range Quarter Panchari | Udhampur/ Panchari | 15.00 |
| 130 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter Panchari | Udhampur/ Panchari | 15.00 |
| 131 | Department of Forest and Ecology | PCCF & HoFF | Kainthgali Check Post | Udhampur/ Moungri | 15.00 |
| 132 | Department of Forest and Ecology | PCCF & HoFF | Mali Hut Thathli Nursery | Udhampur/ Panchari | 15.00 |
| 133 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House, Kashal (Pattan) - | Latti-Marothi | 15.00 |
| 134 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut Darhal | Rajouri | 15.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|------------------------------------|--------------------------|------------------|
| 135 | Department of Forest and Ecology | PCCF & HoFF | Check Post Dool | Kishtwar/Nagseni | 15.00 |
| 136 | Department of Forest and Ecology | PCCF & HoFF | Division Office Kralpora | Kupwara | 15.00 |
| 137 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Karmulla | Pulwama/ Tral | 15.00 |
| 138 | Department of Forest and Ecology | PCCF & HoFF | Rest House Kanjinag | Pulwama/ Awantipora | 15.00 |
| 139 | Department of Forest and Ecology | PCCF & HoFF | D.F.O's Residence | Anantnag/ Bijbehara | 15.00 |
| 140 | Department of Forest and Ecology | PCCF & HoFF | C.F's Residence | Anantnag/ Bijbehara | 15.00 |
| 141 | Department of Forest and Ecology | PCCF & HoFF | Range Office Pulwama | Pulwama/ Pulwama | 15.00 |
| 142 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Sub Judge Sarrankote | Poonch | 15.00 |
| 143 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff JMIC Mendhar | Poonch | 15.00 |
| 144 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Chenani | Udhampur | 15.00 |
| 145 | Jal Shakti Department | PHE, Jammu | F.P No. 1 | Jammu | 14.77 |
| 146 | Jal Shakti Department | PHE, Jammu | F.P No. 3 | Jammu | 14.77 |
| 147 | Jal Shakti Department | PHE, Jammu | F.P No. 4 | Jammu | 14.77 |
| 148 | Department of Forest and Ecology | PCCF & HoFF | Check Post Arizal | Budgam/Khansahib | 14.50 |
| 149 | Jal Shakti Department | PHE, Kashmir | PHE Division Handwara | Kupwara/Handwara | 14.46 |
| 150 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff/JMMIC Sogrammm | Kupwara | 14.40 |
| 151 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | IB hut Yousmarg | Budgam/ Charisbharief | 14.30 |
| 152 | Department of Home | Fire and Emergency Service Department- | F&ES Gowkadal | | 14.04 |
| 153 | Department of Home | Fire and Emergency Service Department- | F&ES Sopore | | 14.00 |
| 154 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Sitlee | Jammu | 14.00 |
| 155 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | I&FC Guest House New plot Jammu | Jammu/South | 14.00 |
| 156 | Department of Forest and Ecology | Wildlife Protection | Regional WLW office Manda | Jammu/Jammu | 14.00 |
| 157 | Department of Forest and Ecology | PCCF & HoFF | Timber, Firewood Shed Ramnagar | Udhampur / Ramnagar | 14.00 |
| 158 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut at Plassian | Udhampur / Ramnagar | 14.00 |
| 159 | Department of Home | Fire and Emergency Service Department- | F&ES Ganderbal | | 13.87 |
| 160 | Jal Shakti Department | PHE, Kashmir | PHE Division Kupwara | Kupwara | 13.85 |
| 161 | Jal Shakti Department | PHE, Kashmir | QCPD Srinagar | Srnagar/Shalteng | 13.85 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|---|--------------------------|------------------|
| 162 | Department of Home | HeadQuarter HomeGuard,SDRF,Ci | SDRF 1st Bn Srinagar | | 13.85 |
| 163 | Department of Forest and Ecology | PCCF & HoFF | BO's Hut Mundaji (under Contrr. | Baramulla/ Dangiwacha | 13.50 |
| 164 | Department of Forest and Ecology | PCCF & HoFF | Check Post Seer | Anantnag | 13.50 |
| 165 | Department of Forest and Ecology | PCCF & HoFF | Drivers quarter complex(4 Quarters) | Srinagar | 13.20 |
| 166 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut, Fresnar, Cheshmashahi Block | Srinagar | 13.20 |
| 167 | Jal Shakti Department | PHE, Kashmir | PHE Division Handwara | Kupwara/Handwara | 13.08 |
| 168 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | SSD Tangdar PHE Residential Quarter | Kupwara/Karnah | 13.00 |
| 169 | Department of Forest and Ecology | Wildlife Protection | Wildlife Warden office Manda | Jammu/Jammu | 13.00 |
| 170 | Department of Forest and Ecology | Wildlife Protection | Range Office Building near Mini Sectt. | Baramulla | 13.00 |
| 171 | Department of Forest and Ecology | Wildlife Protection | Range office Hokersar | Srinagar | 13.00 |
| 172 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut, Banj | Kathua/Basohli | 13.00 |
| 173 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Traffic Court, Complex | Jammu | 13.00 |
| 174 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Lawyers Chamber Building | Srinagar | 12.96 |
| 175 | Jal Shakti Department | PHE, Jammu | J.E Quarter | Jammu | 12.92 |
| 176 | Department of Forest and Ecology | PCCF & HoFF | Guard hut Seer | Anantnag | 12.90 |
| 177 | Department of Home | HeadQuarter HomeGuard,SDRF,Ci | SDRF Poonch | | 12.80 |
| 178 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | High Court of J&K and Ladakh, janipur, | Jammu | 12.75 |
| 179 | Department of Home | Fire and Emergency Service Department- | F&ES SH PORA | | 12.50 |
| 180 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Spill Channel Sub-Division | Srinagar /Srinagar | 12.50 |
| 181 | Department of Forest and Ecology | PCCF & HoFF | CP Kakapora | Pulwama/ Kakapora | 12.50 |
| 182 | Department of Forest and Ecology | PCCF & HoFF | CP Kehlil | Pulwama/Tral | 12.50 |
| 183 | Jal Shakti Department | PHE, Kashmir | SSD Tangdar | Kupwara/Karnah | 12.31 |
| 184 | Department of Forest and Ecology | Forest Protection Force | Inspector's Barrack at Sindho Shirmal | Shopian Shopian | 12.25 |
| 185 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff JMIC Mahanpur | Mahanpur | 12.24 |
| 186 | Department of Home | Department of Home | Fort View Guest house NHPC Reasi | Reasi | 12.22 |
| 187 | Department of Forest and Ecology | JK Forest Development | Divisional Office, Workshop Division, | Srinagar/Shalteng | 12.16 |
| 188 | Department of Home | Department of Home | SP South Office | Srinagar | 12.00 |
| | • | | | | |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|--|---|-------------------------|------------------|
| 189 | Department of Home | Department of Home | PS Khrew | Pampore | 12.00 |
| 190 | Department of Home | Department of Home | DP)Bandipora Nathpora Bandipora | Bandipora/Handipoa | 12.00 |
| 191 | Department of Home | Department of Home | ps GRP | Jammu | 12.00 |
| 192 | Department of Home | Department of Home | PS GRP Ramnagar | Udhampur | 12.00 |
| 193 | Department of Home | Department of Home | Family quarter-A | Ganderbal /Kangan. | 12.00 |
| 194 | Department of Home | Department of Home | Family quarter-B | Ganderbal /Kangan. | 12.00 |
| 195 | Department of Home | Department of Home | Family quarter-C | Ganderbal /Kangan. | 12.00 |
| 196 | Department of Home | Department of Home | Family quarter-D | Ganderbal /Kangan. | 12.00 |
| 197 | Department of Home | Department of Home | Family quarter-E | Ganderbal /Kangan. | 12.00 |
| 198 | Department of Home | Department of Home | Family Quarter 'E' Block | Kathua | 12.00 |
| 199 | Department of Home | Department of Home | Administrative Block | Kathua | 12.00 |
| 200 | Department of Home | Department of Home | Staff Mess/ Simulator Hall | Kathua | 12.00 |
| 201 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub-divisional Building Pakherpora | Budgam/ Charisharief | 12.00 |
| 202 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | I&FC Division Sopore & ISD Sopore (at | Baramulla/ Sopore | 12.00 |
| 203 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Ziladar Office Zainpora | Shopian/Chitragam | 12.00 |
| 204 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional stores Chanpora Tangmarg | Baramulla/Tangmarg | 12.00 |
| 205 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisonal office building MID | Baramulla/Baramulla | 12.00 |
| 206 | Department of Forest and Ecology | Forest Protection Force | Seizure Shed Fatehpora | Ganderbal | 12.00 |
| 207 | Department of Forest and Ecology | Forest Protection Force | Reporting Centre Urpash | Ganderbal | 12.00 |
| 208 | Department of Forest and Ecology | Wildlife Protection | Wildlife Warden office Manda | Jammu/Jammu | 12.00 |
| 209 | Department of Forest and Ecology | Wildlife Protection | Range office Jammu Manda | Jammu/Jammu | 12.00 |
| 210 | Department of Forest and Ecology | Wildlife Protection | Range Office Building near Mini Sectt. | Baramulla | 12.00 |
| 211 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Showara | Jammu/Chowki Choura | 12.00 |
| 212 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Juthal Gala | Jammu/Chowki Choura | 12.00 |
| 213 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Jadh | Jammu/Jourian | 12.00 |
| 214 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Jalara | Jammu/Chowki Choura | 12.00 |
| 215 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Chanota | Jammu/Khour | 12.00 |
| 216 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Balli | Jammu/Kharah Balli | 12.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|--|---|------------------------|------------------|
| 217 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Tungi | Jammu/Akhnoor | 12.00 |
| 218 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Choura | Jammu/Chowki Choura | 12.00 |
| 219 | Department of Forest and Ecology | PCCF & HoFF | Guest House Sidhra | Jammu | 12.00 |
| 220 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Sunjwan | Jammu | 12.00 |
| 221 | Department of Forest and Ecology | PCCF & HoFF | Depot Barmeen | Udhampur / Ramnagar | 12.00 |
| 222 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Surukote | Reasi/Arnas | 12.00 |
| 223 | Department of Forest and Ecology | PCCF & HoFF | Range Office cum residence Kotranka | Rajouri/KOtranka | 12.00 |
| 224 | Department of Forest and Ecology | PCCF & HoFF | Check Post Kotranka | Rajouri | 12.00 |
| 225 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Kesar Gala | Rajouri | 12.00 |
| 226 | Department of Forest and Ecology | PCCF & HoFF | Range Office Kalakote | Rajouri | 12.00 |
| 227 | Department of Forest and Ecology | PCCF & HoFF | Range Officer residence Kalakote | Rajouri | 12.00 |
| 228 | Department of Forest and Ecology | PCCF & HoFF | Solki Check Post | Rajouri | 12.00 |
| 229 | Department of Forest and Ecology | PCCF & HoFF | Range Office cum Qtr. Thakrakote HQ Pouni | Reasi/Pouni | 12.00 |
| 230 | Department of Forest and Ecology | PCCF & HoFF | Check Post Budgam | Budgam/Budgam | 12.00 |
| 231 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Cherhar | Baramulla/Dangerpora | 12.00 |
| 232 | Department of Forest and Ecology | PCCF & HoFF | Checkpost Yanner 1 | Anantnag | 12.00 |
| 233 | Department of Forest and Ecology | PCCF & HoFF | Old Range office Bijbehara | Anantnag/ Bijbehara | 12.00 |
| 234 | Department of Forest and Ecology | PCCF & HoFF | Divisional Complex Padawan | Shopian/ Shopian | 12.00 |
| 235 | Department of Forest and Ecology | PCCF & HoFF | Dogra Hall Forest complex Jammu | Jammu/Jammu | 12.00 |
| 236 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Lift Irrigation Scheme Marvel Stage 3rd | Budgam/B.K.Pora | 11.88 |
| 237 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Stage 3rd Budgam | Budgam/B.K.Pora | 11.88 |
| 238 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Irrigation Division Tral building | Pulwama/Tral | 11.84 |
| 239 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Irrigation and Flood Control Sub divisions | Bandipora | 11.57 |
| 240 | Jal Shakti Department | PHE, Kashmir | Water Works Srinagar | Ganderbal | 11.54 |
| 241 | Jal Shakti Department | PHE, Kashmir | PHE Division Handwara | Kupwara/Handwara | 11.54 |
| 242 | Department of Forest and Ecology | JK Forest Development | Divisional Office, Extraction Division, | Kupwara/Kupwara | 11.50 |
| 243 | Department of Forest and Ecology | PCCF & HoFF | Eco-tourism Hut No. 05 Panagar | Kathua/Basohli | 11.50 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|---|-----------------------------|------------------|
| 244 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Pump House-II (Hari) | Pulwama/ Awantipora | 11.40 |
| 245 | Department of Forest and Ecology | PCCF & HoFF | Hostel Block-B | Jammu/R.S.Pura | 11.40 |
| 246 | Department of Home | Department of Home | PS Safapora | Lar | 11.11 |
| 247 | Department of Home | Department of Home | SHO office Quarter Sanshta Old Quarter Old | Baramulla Tangmarg | 11.11 |
| 248 | Department of Home | Department of Home | PS Bani | KathuaiBani | 11.11 |
| 249 | Department of Forest and Ecology | PCCF & HoFF | DFO Residence OB Post Budgam | Budgam/Budgam | 11.08 |
| 250 | Department of Forest and Ecology | PCCF & HoFF | BO's Hut Yousmarg | Budgam/ Chrar-i- Sharief | 11.02 |
| 251 | Department of Home | Department of Home | J&K Police Building Sector 44 Plot No.125 | Gurugram/Wazirahad | 11.00 |
| 252 | Department of Home | Department of Home | SDRF 1"Bn Srinagar | Srinagar | 11.00 |
| 253 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisonal Building Amnoo Kulgam | Kulgam | 11.00 |
| 254 | Department of Forest and Ecology | Forest Protection Force | Watch Tower Fatehpora | Ganderbal | 11.00 |
| 255 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Kulwanta | Udhampur / Ramnagar | 11.00 |
| 256 | Department of Forest and Ecology | PCCF & HoFF | Purmandal Nursery Building | Samba/Bari Brahmana | 11.00 |
| 257 | Department of Forest and Ecology | PCCF & HoFF | Forest Hut Reshwari Nowgam | Kupwara/ Qalamabad | 11.00 |
| 258 | Department of Forest and Ecology | PCCF & HoFF | Forest Range Office Verinag | Anantnag | 11.00 |
| 259 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Additional District & Sessions Court | Kupwara | 11.00 |
| 260 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Office Dawar | Bandipora/Gurez | 10.86 |
| 261 | Department of Forest and Ecology | PCCF & HoFF | Machanpora Checkpost | Baramulla/ Dangiwacha | 10.65 |
| 262 | Department of Home | Fire and Emergency Service Department- | F&ES Station Kathua/Command | | 10.50 |
| 263 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut at Hanzal Dachhnan | Kishtwar/Dachhan | 10.50 |
| 264 | Department of Forest and Ecology | PCCF & HoFF | Check Post Sumbal | Bandipora/ | 10.50 |
| 265 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Contol Sub Division Kulgam | Kulgam/ Kulgam | 10.44 |
| 266 | Department of Forest and Ecology | PCCF & HoFF | BO Hut Ageli | Doda/ Bhadarwah | 10.33 |
| 267 | Department of Home | Fire and Emergency Service Department- | F&ES Command office Gandarbal | | 10.07 |
| 268 | Department of Home | Department of Home | RPHQ Srinagar - Police Public School | Srinagar | 10.00 |
| 269 | Department of Home | Department of Home | Youth Developemnt Rehabilitation Centre | Srinagar | 10.00 |
| | | | | | |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--------------------|--------------------|--|---------------------|------------------|
| 271 | Department of Home | Department of Home | Family Quarter Wahidpora | Tulmulla | 10.00 |
| 272 | Department of Home | Department of Home | LSQ Block 1 | Anantnag | 10.00 |
| 273 | Department of Home | Department of Home | LSQ Block 2 | Anantnag | 10.00 |
| 274 | Department of Home | Department of Home | PCR Building | Anantnag | 10.00 |
| 275 | Department of Home | Department of Home | FSL Building | Anantnag | 10.00 |
| 276 | Department of Home | Department of Home | O5 Storied Building (CRPF Accomodation) | Pulwama | 10.00 |
| 277 | Department of Home | Department of Home | pP Mirbaza | Kulgam | 10.00 |
| 278 | Department of Home | Department of Home | PC Aamnoo | Kulgam | 10.00 |
| 279 | Department of Home | Department of Home | DYSP Hqrs OFice | Shopian | 10.00 |
| 280 | Department of Home | Department of Home |)YSP DAR Oics | Shopian | 10.00 |
| 281 | Department of Home | Department of Home | ASP Residence l | Shopian | 10.00 |
| 282 | Department of Home | Department of Home | DYSP Hqrs /SHO Residencs | Shopian | 10.00 |
| 283 | Department of Home | Department of Home | SDPO OTice Zainapora/Residence | Zainapora | 10.00 |
| 284 | Department of Home | Department of Home | PP Kecgam Main Building | Kcegam | 10.00 |
| 285 | Department of Home | Department of Home | PP Wachi Main Building | Wach | 10.00 |
| 286 | Department of Home | Department of Home | pp Turkwangam Main Building | Zainapora | 10.00 |
| 287 | Department of Home | Department of Home | YSPUS Ke ResideneeHlee | Shupuan | 10.00 |
| 288 | Department of Home | Department of Home | DYSP PC Imamsahh()hee | Barbughlmamsahab | 10.00 |
| 289 | Department of Home | Department of Home | P/S Tral 05 Buildings | Tral | 10.00 |
| 290 | Department of Home | Department of Home | PP Anpal 01 Building | Tral | 10.00 |
| 291 | Department of Home | Department of Home | P/S Pampore 03 Buildings | Pamporc | 10.00 |
| 292 | Department of Home | Department of Home | DPO Complex | Kupwara | 10.00 |
| 293 | Department of Home | Department of Home | Police Station | Sogam | 10.00 |
| 294 | Department of Home | Department of Home | Police Station | Kupwara | 10.00 |
| 295 | Department of Home | Department of Home |)PL.Bandipora | Bandipora Bandipora | 10.00 |
| 296 | Department of Home | Department of Home | SDPO Sumba | Bandipora Sumha | 10.00 |
| 297 | Department of Home | Department of Home | Ps Sumbal Industral | Bandipora Sumha | 10.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|--|---|----------------------------------|------------------|
| 298 | Department of Home | Department of Home | PS Hand:porz | Bandipora | 10.00 |
| 299 | Department of Home | Department of Home | PS Pethkoo | Bandipora | 10.00 |
| 300 | Department of Home | Department of Home | PS Aragam SK | Bandiporx | 10.00 |
| 301 | Department of Home | Department of Home | PP Ajas | Bandipora/Sumbal | 10.00 |
| 302 | Department of Home | Department of Home | PP Nowgam | Bandipora/Sumha | 10.00 |
| 303 | Department of Home | Department of Home | PC Handwara | JIC Handwara | 10.00 |
| 304 | Department of Home | Department of Home | SSP PCR main OfTic | Srinagar | 10.00 |
| 305 | Department of Home | Department of Home | Old MjiC Building at Doda | Doda | 10.00 |
| 306 | Department of Home | Department of Home | Women P/S Doda /PCR Doda | Doda | 10.00 |
| 307 | Department of Home | Department of Home | State Investigation Agency (SIA).Mirar | "Tehisal Miran Sahib District | 10.00 |
| 308 | Department of Home | Department of Home | PP Sainik Colony | Bahu | 10.00 |
| 309 | Department of Home | Department of Home | PS GAndhi Nagar | Jammu South | 10.00 |
| 310 | Department of Home | Department of Home | PS Satwari | Jammu South | 10.00 |
| 311 | Department of Home | Department of Home | PS Channi Himmat | Bahu | 10.00 |
| 312 | Department of Home | Department of Home | LSQ Block Vij aypur | Samba | 10.00 |
| 313 | Department of Home | Department of Home | L-Block (Training Barrack) | Samba | 10.00 |
| 314 | Department of Home | Department of Home | Training Mess Vijaypur | Samba | 10.00 |
| 315 | Department of Home | Department of Home | DTHQ Ramban/PCR Ramban, DTHQ | Ramban | 10.00 |
| 316 | Department of Home | Department of Home | Trainees Single Storey | Kathua | 10.00 |
| 317 | Department of Home | Department of Home | Adminisarative Block IRP.3rd | Baramulaf Pattan | 10.00 |
| 318 | Department of Home | Department of Home | HQ Block Barrack double story building | Anantnag/Mattan | 10.00 |
| 319 | Department of Home | Department of Home | Main office | jammu | 10.00 |
| 320 | Department of Home | Department of Home | "Adm.Block IRP- 22nd Bn.Kishtwa" | kishtwar | 10.00 |
| 321 | Jal Shakti Department | PHE, Kashmir | SSD Tangdar | Kupwara/Karnah | 10.00 |
| 322 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Irrigation Sub- Divisional Building | Kupwara | 10.00 |
| 323 | Department of Forest and Ecology | Forest Protection Force | Reporting Centre Hayan | Kangan | 10.00 |
| 324 | Department of Forest and Ecology | Forest Protection Force | DD office Building Kehmil | Kupwara | 10.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|---------------------|--|---------------------|------------------|
| 325 | Department of Forest and Ecology | Wildlife Protection | Range office Jammu Manda | Jammu/Jammu | 10.00 |
| 326 | Department of Forest and Ecology | Wildlife Protection | Sarball Stafff Qtr | | 10.00 |
| 327 | Department of Forest and Ecology | Wildlife Protection | Staff Quarter/NIC Tawrian Limber | Baramulla | 10.00 |
| 328 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Kulwanta | Udhampur / Ramnagar | 10.00 |
| 329 | Department of Forest and Ecology | PCCF & HoFF | TSD Nud | Samba/Samba | 10.00 |
| 330 | Department of Forest and Ecology | PCCF & HoFF | Chowkidar Room Chulna | Udhampur/ Panchari | 10.00 |
| 331 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Dubigali | Udhampur/ Panchari | 10.00 |
| 332 | Department of Forest and Ecology | PCCF & HoFF | BO Hut Sankri Devta | Udhampur/ Panchari | 10.00 |
| 333 | Department of Forest and Ecology | PCCF & HoFF | Range Office building, Chenani | Chenani | 10.00 |
| 334 | Department of Forest and Ecology | PCCF & HoFF | Range Quarter, Chenani | Chenani | 10.00 |
| 335 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House, Chenani | Chenani | 10.00 |
| 336 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Chenani | Chenani | 10.00 |
| 337 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House, Kassal (Patnitop) | Chenani | 10.00 |
| 338 | Department of Forest and Ecology | PCCF & HoFF | Forest Guard Hut, Karlah | Chenani | 10.00 |
| 339 | Department of Forest and Ecology | PCCF & HoFF | Forest Guard Hut, Kud | Chenani | 10.00 |
| 340 | Department of Forest and Ecology | PCCF & HoFF | Forest Guard Hut, Margana | Chenani | 10.00 |
| 341 | Department of Forest and Ecology | PCCF & HoFF | Forest Guard Hut, Margana Nursery | Chenani | 10.00 |
| 342 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House, Sudhmahadev (under | Chenani | 10.00 |
| 343 | Department of Forest and Ecology | PCCF & HoFF | Old Gowdon | Udhampur | 10.00 |
| 344 | Department of Forest and Ecology | PCCF & HoFF | Mali Hut, Mand | Udhampur | 10.00 |
| 345 | Department of Forest and Ecology | PCCF & HoFF | FRH, Mand | Udhampur | 10.00 |
| 346 | Department of Forest and Ecology | PCCF & HoFF | Check Post Jakhani | Udhampur | 10.00 |
| 347 | Department of Forest and Ecology | PCCF & HoFF | Check Post Laddan | Udhampur | 10.00 |
| 348 | Department of Forest and Ecology | PCCF & HoFF | Resin Depot Kaldi | Udhampur | 10.00 |
| 349 | Department of Forest and Ecology | PCCF & HoFF | Check Post Mansar | Udhampur | 10.00 |
| 350 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut Madanchowra | Udhampur | 10.00 |
| 351 | Department of Forest and Ecology | PCCF & HoFF | Interpretation Hall Gangera | Udhampur | 10.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|-------------|---|----------|------------------|
| 352 | Department of Forest and Ecology | PCCF & HoFF | Bathrooms at Gangera | Udhampur | 10.00 |
| 353 | Department of Forest and Ecology | PCCF & HoFF | Resin Depot Pakhlai | Udhampur | 10.00 |
| 354 | Department of Forest and Ecology | PCCF & HoFF | Staff Quarter 1 | Udhampur | 10.00 |
| 355 | Department of Forest and Ecology | PCCF & HoFF | Staff Quarter 2 | Udhampur | 10.00 |
| 356 | Department of Forest and Ecology | PCCF & HoFF | Staff Quarter 3 | Udhampur | 10.00 |
| 357 | Department of Forest and Ecology | PCCF & HoFF | Staff Quarter 4 | Udhampur | 10.00 |
| 358 | Department of Forest and Ecology | PCCF & HoFF | Staff Quarter 5 | Udhampur | 10.00 |
| 359 | Department of Forest and Ecology | PCCF & HoFF | Staff Quarter 6 | Udhampur | 10.00 |
| 360 | Department of Forest and Ecology | PCCF & HoFF | Staff Quarter 7 | Udhampur | 10.00 |
| 361 | Department of Forest and Ecology | PCCF & HoFF | ACF Quarter | Udhampur | 10.00 |
| 362 | Department of Forest and Ecology | PCCF & HoFF | FRH Udhampur | Udhampur | 10.00 |
| 363 | Department of Forest and Ecology | PCCF & HoFF | Firewood Depot Devika | Udhampur | 10.00 |
| 364 | Department of Forest and Ecology | PCCF & HoFF | Soil Range Office Udhampur | Udhampur | 10.00 |
| 365 | Department of Forest and Ecology | PCCF & HoFF | Divisional Forest Office, Udhampur | Udhampur | 10.00 |
| 366 | Department of Forest and Ecology | PCCF & HoFF | Divisional Forest Officer, Residence | Udhampur | 10.00 |
| 367 | Department of Forest and Ecology | PCCF & HoFF | Old DFO Office | Udhampur | 10.00 |
| 368 | Department of Forest and Ecology | PCCF & HoFF | Urban Forestry Division office Dogra | Jammu | 10.00 |
| 369 | Department of Forest and Ecology | PCCF & HoFF | Van Sadan Talab Tillo | Jammu | 10.00 |
| 370 | Department of Forest and Ecology | PCCF & HoFF | FRMC Complex Narwal Jammu | Jammu | 10.00 |
| 371 | Department of Forest and Ecology | PCCF & HoFF | IWDP Traning Centre Gladni | Jammu | 10.00 |
| 372 | Department of Forest and Ecology | PCCF & HoFF | Forest Information Centre Bikram Chowk | Jammu | 10.00 |
| 373 | Department of Forest and Ecology | PCCF & HoFF | Range Office Gladni Range | Jammu | 10.00 |
| 374 | Department of Forest and Ecology | PCCF & HoFF | Forest officers institute building | Jammu | 10.00 |
| 375 | Department of Forest and Ecology | PCCF & HoFF | Range office Talab Tillo Range | Jammu | 10.00 |
| 376 | Department of Forest and Ecology | PCCF & HoFF | TTO Bahu Office | Jammu | 10.00 |
| 377 | Department of Forest and Ecology | PCCF & HoFF | TTO Toph Office | Jammu | 10.00 |
| 378 | Department of Forest and Ecology | PCCF & HoFF | Range Office Estate Range Dogra Hall | Jammu | 10.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|---|-------------|---|------------------|------------------|
| 379 | Department of Forest and Ecology | PCCF & HoFF | Residential Quarters Canal Road | Jammu | 10.00 |
| 380 | Department of Forest and Ecology | PCCF & HoFF | Residential Quarters Bikram Chowk | Jammu | 10.00 |
| 381 | Department of Forest and Ecology | PCCF & HoFF | Residential Quarters Dogra Hall | Jammu | 10.00 |
| 382 | Department of Forest and Ecology | PCCF & HoFF | Residential Quarters Gladni Narwal | Jammu | 10.00 |
| 383 | Department of Forest and Ecology | PCCF & HoFF | R. O. Office Surankote | Poonch/Surankote | 10.00 |
| 384 | Department of Forest and Ecology | PCCF & HoFF | Forester Quarter Poonch | Poonch/Haveli | 10.00 |
| 385 | Department of Forest and Ecology | PCCF & HoFF | Range Office Haveli | Poonch/Haveli | 10.00 |
| 386 | Department of Forest and Ecology | PCCF & HoFF | Inspection Hut Poonch | Poonch/Haveli | 10.00 |
| 387 | Department of Forest and Ecology | PCCF & HoFF | Check Post Building Chandak | Poonch/Haveli | 10.00 |
| 388 | Department of Forest and Ecology | PCCF & HoFF | Check Post Building Surankote | Poonch/Surankote | 10.00 |
| 389 | Department of Forest and Ecology | PCCF & HoFF | R.O. Surankote Quarter | Poonch/Surankote | 10.00 |
| 390 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter | Poonch/Surankote | 10.00 |
| 391 | Department of Forest and Ecology | PCCF & HoFF | DFO Quarter (Working Plan) | Poonch/Haveli | 10.00 |
| 392 | Department of Forest and Ecology | PCCF & HoFF | Forest Hut at Kalali | Poonch/Surankote | 10.00 |
| 393 | Department of Forest and Ecology | PCCF & HoFF | Chowkidar Hut Poonch | Poonch/Haveli | 10.00 |
| 394 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter Loran | Poonch/Haveli | 10.00 |
| 395 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter Gundi | Poonch/Surankote | 10.00 |
| 396 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter Sawjian | Poonch/Haveli | 10.00 |
| 397 | Department of Forest and Ecology Department of Forest | PCCF & HoFF | BO Quarter Khanater BO Quarter | Poonch/Haveli | 10.00 |
| 398 | and Ecology Department of Forest | PCCF & HoFF | Dharamsal | Poonch/ Mendher | 10.00 |
| 399 | and Ecology | PCCF & HoFF | BO Quarter Murrah | Poonch/Surankote | 10.00 |
| 400 | Department of Forest and Ecology | PCCF & HoFF | Divisional Office | Reasi/Thuroo | 10.00 |
| 401 | Department of Forest and Ecology | PCCF & HoFF | Clerical quarter | Reasi/Thuroo | 10.00 |
| 402 | Department of Forest and Ecology | PCCF & HoFF | Range Office Gulabgarh New Range Office | Reasi/Mahore | 10.00 |
| 403 | Department of Forest and Ecology | PCCF & HoFF | cum Residence of Block Officer | Reasi/Mahore | 10.00 |
| 404 | Department of Forest and Ecology | PCCF & HoFF | Qauarter Mahore | Reasi/Mahore | 10.00 |
| 405 | Department of Forest and Ecology | PCCF & HoFF | Guard hut Kouri | Reasi/Arnas | 10.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|-------------|---|----------------------------|------------------|
| 406 | Department of Forest and Ecology | PCCF & HoFF | Block Officer Quarter Sungri | Reasi/Chassana | 10.00 |
| 407 | Department of Forest and Ecology | PCCF & HoFF | Office of Range Officer Mahore | Reasi/Thuroo | 10.00 |
| 408 | Department of Forest and Ecology | PCCF & HoFF | D.F.O Residence | Reasi/Thuroo | 10.00 |
| 409 | Department of Forest and Ecology | PCCF & HoFF | Fire Control Room at Dharmari | Reasi/Thuroo | 10.00 |
| 410 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Thuru | Reasi/Thuroo | 10.00 |
| 411 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Kanthan | Reasi/Arnas | 10.00 |
| 412 | Department of Forest and Ecology | PCCF & HoFF | CF Clerical Quarter Panja | Rajouri | 10.00 |
| 413 | Department of Forest and Ecology | PCCF & HoFF | DFO Office, Rajouri | Rajouri | 10.00 |
| 414 | Department of Forest and Ecology | PCCF & HoFF | Range Officer Rajouri residence | Rajouri | 10.00 |
| 415 | Department of Forest and Ecology | PCCF & HoFF | Inspection Hut Rajouri | Rajouri | 10.00 |
| 416 | Department of Forest and Ecology | PCCF & HoFF | REst House Manjakote | Rajouri/Manjakote | 10.00 |
| 417 | Department of Forest and Ecology | PCCF & HoFF | DKG HUts | Rajouri | 10.00 |
| 418 | Department of Forest and Ecology | PCCF & HoFF | Check Post Thathari | Doda/ Thathari | 10.00 |
| 419 | Department of Forest and Ecology | PCCF & HoFF | Conservator of Chenab Circle Office | Doda/Doda | 10.00 |
| 420 | Department of Forest and Ecology | PCCF & HoFF | Range Office Thakrai in Kishtwar | Kishtwar/Kishtwar | 10.00 |
| 421 | Department of Forest and Ecology | PCCF & HoFF | Range Office Marwah at Yourdu Marwah | Kishtwar/Marwah | 10.00 |
| 422 | Department of Forest and Ecology | PCCF & HoFF | Check Post Braripathri | Budgam/ Chrar-i-Sharief | 10.00 |
| 423 | Department of Forest and Ecology | PCCF & HoFF | B.O's Hut Hajibal | Baramulla/Baramulla | 10.00 |
| 424 | Department of Forest and Ecology | PCCF & HoFF | Headquater of Bo Estate Zangli | Kupwara/Kupwara | 10.00 |
| 425 | Department of Forest and Ecology | PCCF & HoFF | Guest Hut Reddi | Kupwara | 10.00 |
| 426 | Department of Forest and Ecology | PCCF & HoFF | BO Hut Trathpora Hangnikoot | Kupwara | 10.00 |
| 427 | Department of Forest and Ecology | PCCF & HoFF | Mali Hut Malik Pora | Kupwara | 10.00 |
| 428 | Department of Forest and Ecology | PCCF & HoFF | Forest Checkpost Trathpora | Kupwara | 10.00 |
| 429 | Department of Forest and Ecology | PCCF & HoFF | Forest Checkpost Panzgam | Kupwara | 10.00 |
| 430 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Tee Pee Drangyari | Kupwara | 10.00 |
| 431 | Department of Forest and Ecology | PCCF & HoFF | BO HUt Drangyari | Kupwara | 10.00 |
| 432 | Department of Forest and Ecology | PCCF & HoFF | Shelter Shed Sadhna | Kupwara | 10.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|--|--------------------------|------------------|
| 433 | Department of Forest and Ecology | PCCF & HoFF | Range office Keran | Kupwara | 10.00 |
| 434 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Keran | Kupwara | 10.00 |
| 435 | Department of Forest and Ecology | PCCF & HoFF | Forest check post Kralpora | Kupwara | 10.00 |
| 436 | Department of Forest and Ecology | PCCF & HoFF | BO Hut Zurahama | Kupwara | 10.00 |
| 437 | Department of Forest and Ecology | PCCF & HoFF | Division Office Langate | Kupwara/ Langate | 10.00 |
| 438 | Department of Forest and Ecology | PCCF & HoFF | Aram Ghah Kishtwar | Kupwara/ Qalamabad | 10.00 |
| 439 | Department of Forest and Ecology | PCCF & HoFF | Rest House at Handwara | Kupwara/ Handwara | 10.00 |
| 440 | Department of Forest and Ecology | PCCF & HoFF | Newly Constructed Library Cum | Bandipora | 10.00 |
| 441 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | CIM Anantnag | Anantnag | 10.00 |
| 442 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Muniff Court Dooru | Dooru | 10.00 |
| 443 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Ciurt Complex Awantipora | Awantipora | 10.00 |
| 444 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Complext Gandoh | Doda | 10.00 |
| 445 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Pr. District and Sessions Court Reasi | Reaso | 10.00 |
| 446 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Complex Rajouri | Rajouri | 10.00 |
| 447 | Department of Forest and Ecology | PCCF & HoFF | Malies Hut Nagbal Shopian | Shopian/ Shopian | 9.90 |
| 448 | Department of Home | Fire and Emergency Service Department- | F&ES Family Qtr | | 9.75 |
| 449 | Department of Home | Fire and Emergency Service Department- | F&ES Family Qtr | | 9.75 |
| 450 | Department of Home | Sainik Welfare Department,UT of | Zila Sainik Welfare office Kathua | | 9.72 |
| 451 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut at Patimhala Palmar | Kishtwar/Kishtwar | 9.72 |
| 452 | Jal Shakti Department | PHE, Kashmir | PHE Division Sopore | Baramulla/Dangerpora | 9.62 |
| 453 | Department of Home | Sainik Welfare Department,UT of | Sainik Rest House Ramnagar | | 9.60 |
| 454 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut at Ghoomarshalli | Bhagwah/Doda | 9.60 |
| 455 | Department of Forest and Ecology | PCCF & HoFF | Hostel Block-C | Jammu/R.S.Pura | 9.25 |
| 456 | Jal Shakti Department | PHE, Jammu | Water Testing Lab. | Jammu | 9.23 |
| 457 | Jal Shakti Department | PHE, Kashmir | PHE Division Chadoora | Budgam/Chrari Sharief | 9.23 |
| 458 | Jal Shakti Department | PHE, Kashmir | PHE Division Budgam | Budgam/ Khansahib | 9.23 |
| 459 | Jal Shakti Department | PHE, Kashmir | PHE Division Budgam | Budgam/Beerwa | 9.23 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|---|--|---------------------|------------------|
| 460 | Jal Shakti Department | PHE, Kashmir | Water Works Srinagar | Srinagar | 9.23 |
| 461 | Jal Shakti Department | PHE, Kashmir | PHE Mechanical Division Srinagar | Srinagar | 9.23 |
| 462 | Jal Shakti Department | PHE, Kashmir | PHE Mechanical Division Srinagar | Ganderbal | 9.23 |
| 463 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Lift Irrigation Scheme Marvel Stage 4th | Budgam/B.K.Pora | 9.02 |
| 464 | Department of Home | Department of Home | PIS Kathua | Kathua/ | 9.00 |
| 465 | Department of Home | Fire and Emergency Service Department- | F&ES Gandoh | | 9.00 |
| 466 | Department of Home | Fire and Emergency Service Department- | F&ES Shopian | | 9.00 |
| 467 | Department of Home | Fire and Emergency Service Department- | F&ES Pampore | | 9.00 |
| 468 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | J.E,s Quarter at Sulkoot Kupwara | Kupwara | 9.00 |
| 469 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Office building | Bandipora/Sumbal | 9.00 |
| 470 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Office Building of Irr. Div. | Anantnag | 9.00 |
| 471 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub Division Building Zainapora | Shopian/Chitragam | 9.00 |
| 472 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | FBI Divison Tangmarg (Head | Baramulla/Tangmarg | 9.00 |
| 473 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Control Division Kakapora at | Pulwama/kakapora | 9.00 |
| 474 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub-divisional office/ stores | Baramulla/Baramulla | 9.00 |
| 475 | Department of Forest and Ecology | Forest Protection Force | Dy Director Residence Hardupanzoo | Budgam, Khansahib | 9.00 |
| 476 | Department of Forest and Ecology | Wildlife Protection | Inspection hut Nandini | Jammu/Nagrota | 9.00 |
| 477 | Department of Forest and Ecology | Wildlife Protection | Staff Quarter, Aru | Anantnag | 9.00 |
| 478 | Department of Forest and Ecology | PCCF & HoFF | Chowkidar Hut, (TSD Batote) | Ramban/ Batote | 9.00 |
| 479 | Department of Forest and Ecology | PCCF & HoFF | Gust House Sangrambhatta | Kishtwar/Kishtwar | 9.00 |
| 480 | Department of Forest and Ecology | PCCF & HoFF | D.F.O Residence at Forest Complex Doda | Doda/Doda | 9.00 |
| 481 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut at TSD Doda | Doda/Doda | 9.00 |
| 482 | Department of Forest and Ecology | PCCF & HoFF | Range Office Gurez | Bandipora/Gurez | 9.00 |
| 483 | Department of Forest and Ecology | PCCF & HoFF | Inspection Hut Dawar Gurez | Bandipora/Gurez | 9.00 |
| 484 | Department of Forest and Ecology | PCCF & HoFF | Forest Range Office Kuthar at Achabal | Anantnag | 9.00 |
| 485 | Department of Forest and Ecology | PCCF & HoFF | Range Office D.H.Pora | Kulgam / D.H.Pora | 9.00 |
| 486 | Department of Forest and Ecology | PCCF & HoFF | Instructor Quarter-I | Jammu/R.S.Pura | 9.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|--|--------------------------|------------------|
| 487 | Department of Forest and Ecology | PCCF & HoFF | New Foresters Hostel | Bandipora | 9.00 |
| 488 | Department of Forest and Ecology | PCCF & HoFF | Newly Constructed Instructors Residence | Bandipora | 9.00 |
| 489 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | munsiff Court Complex. Gutez | Gutez | 9.00 |
| 490 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Court Complex Tral/ | Tral | 9.00 |
| 491 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Sub Judge Court Chadoora | Budgam | 9.00 |
| 492 | Department of Home | Department of Home | PS Hiranagar | Kathuaf Hiranagar | 8.89 |
| 493 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Majalta | Udhampur | 8.72 |
| 494 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Mechnical Quarter (Bohu Aantipora) | Pulwama/ Awantipora | 8.61 |
| 495 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | I &FC Sub Division Harwun Srinagar | Srinagar/North | 8.50 |
| 496 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Tail Race sub division (Kunzer) | Baramulla/Tangmarg | 8.50 |
| 497 | Department of Forest and Ecology | Wildlife Protection | Chowkidar Quarter Aafti | Kishtwar / Warwan | 8.50 |
| 498 | Department of Forest and Ecology | PCCF & HoFF | B O 's Quarter Batkote | Anantnag | 8.50 |
| 499 | Jal Shakti Department | PHE, Kashmir | WSMP Srinagar | Srinagar / Rainawari | 8.46 |
| 500 | Department of Home | Sainik Welfare Department,UT of | ZSWO Baramulla | | 8.40 |
| 501 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Control Sub- Division Awantipora | Pulwama/Awantipora | 8.40 |
| 502 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Control Sub- Division Shopian at | Shopian/Shopian | 8.40 |
| 503 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Sallar | Anantnag | 8.40 |
| 504 | Department of Forest and Ecology | PCCF & HoFF | Residential Staff Quarters, Agrostology | Jammu/Nagrota | 8.40 |
| 505 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Office I&FC Handwara | Kupwara/Handwara | 8.32 |
| 506 | Department of Home | Fire and Emergency Service Department- | F&ES Baramulla | | 8.25 |
| 507 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | J.E,s Quarter Yousmarg | Budgam/ Charisbharief | 8.20 |
| 508 | Department of Forest and Ecology | PCCF & HoFF | BO's Hut Logripora | Anantnag | 8.16 |
| 509 | Department of Home | Department of Home | Dysp Hqrs. | Tulmulla | 8.00 |
| 510 | Department of Home | Department of Home | Addl. SP Residence | Tulmulla | 8.00 |
| 511 | Department of Home | Department of Home | PS Sonmarg | Gund | 8.00 |
| 512 | Department of Home | Department of Home | PP Bustand | PD Sopore/Sopor | 8.00 |
| 513 | Department of Home | Department of Home | MT Building | Srinagar | 8.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|---|--|--------------------------|------------------|
| 514 | Department of Home | Department of Home | Pies Rajbagh | Kathua/ Marheen | 8.00 |
| 515 | Department of Home | Department of Home | Police Petrol Pump Building at PC R | Srinagar | 8.00 |
| 516 | Department of Home | Department of Home | ZP UHQ Jammu PCR Jammu | Jammu | 8.00 |
| 517 | Department of Home | Department of Home | Family Quarter 'F' Block | Kathua | 8.00 |
| 518 | Department of Home | Department of Home | Family Quarter 'G' Block | Kathua | 8.00 |
| 519 | Department of Home | Department of Home | SDPO East | Jammu South | 8.00 |
| 520 | Department of Home | Department of Home | NGO Barrack/ 3 Story Building | Anantnag/Mattan | 8.00 |
| 521 | Department of Home | Fire and Emergency Service Department- | F&ES Pattan | | 8.00 |
| 522 | Department of Home | Fire and Emergency Service Department- | F&ES Tangmarg | | 8.00 |
| 523 | Department of Home | HeadQuarter HomeGuard,SDRF,Ci | HG/CD & SDRF Budgam | | 8.00 |
| 524 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Inspection Bunglow at Krusan Lolab | Kupwara/Lalpora | 8.00 |
| 525 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | I&FC Guest House Rammunshi bagh | Srinagar/South | 8.00 |
| 526 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Store Alyapora Shopian | Shopian/Shopian | 8.00 |
| 527 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Mechanical Irrigation sub-Divisional office, | Pulwama/Pampore | 8.00 |
| 528 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub division office building Pattan | Baramulla/Pattan | 8.00 |
| 529 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Store (Bijbehara near PHE | Anantnag/ Bijbehara | 8.00 |
| 530 | Department of Forest and Ecology | Forest Protection Force | Assistant Director Quarter/ FPF Gamma | Kathua/ Billawar | 8.00 |
| 531 | Department of Forest and Ecology | Forest Protection Force | Double Storey Staff Fatehpora | Ganderbal | 8.00 |
| 532 | Department of Forest and Ecology | Forest Protection Force | Genset Shed Fatehpora | Ganderbal | 8.00 |
| 533 | Department of Forest and Ecology | Forest Protection Force | Dy Dircetor Office Hardupanzoo | Budgam, Khansahib | 8.00 |
| 534 | Department of Forest and Ecology | Forest Protection Force | Assistant Directors residence | Anantnag | 8.00 |
| 535 | Department of Forest and Ecology | Wildlife Protection | Staff Quarter Gulmarg | Tangmarg | 8.00 |
| 536 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Chhallan | Kathua/Ramkot | 8.00 |
| 537 | Department of Forest and Ecology | PCCF & HoFF | | Udhampur / Ramnagar | 8.00 |
| 538 | Department of Forest and Ecology | PCCF & HoFF | Range Office Basantgarh | Udhampur / Basantgarh | 8.00 |
| 539 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Ghagwal | Samba/Ghagwal | 8.00 |
| 540 | Department of Forest and Ecology | PCCF & HoFF | Block Officer Quarter Budhan | Reasi/Thuroo | 8.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|---|--------------------------|------------------|
| 541 | Department of Forest and Ecology | PCCF & HoFF | Range Office Rajouri | Rajouri | 8.00 |
| 542 | Department of Forest and Ecology | PCCF & HoFF | Chowkidar Hut at Forest Complex Doda | Doda/Doda | 8.00 |
| 543 | Department of Forest and Ecology | PCCF & HoFF | DFO Residence, Ramban | Ramban/Ramban | 8.00 |
| 544 | Department of Forest and Ecology | PCCF & HoFF | Inspection Hut Teetwal | Kupwara | 8.00 |
| 545 | Department of Forest and Ecology | PCCF & HoFF | Common facility center Jumagand | Kupwara | 8.00 |
| 546 | Department of Forest and Ecology | PCCF & HoFF | BO's Hut Chatoosa (newly constructed) | Baramulla/ Dangiwacha | 8.00 |
| 547 | Department of Forest and Ecology | PCCF & HoFF | Forest Range Office Kokernag | Anantnag | 8.00 |
| 548 | Department of Forest and Ecology | PCCF & HoFF | Shanti Hut Kokernag | Anantnag | 8.00 |
| 549 | Department of Forest and Ecology | PCCF & HoFF | Range Office Tral | Pulwama/ Tral | 8.00 |
| 550 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Doodmarg | Pulwama/ Tral | 8.00 |
| 551 | Department of Forest and Ecology | PCCF & HoFF | Block Officers Hut at Zadender Kutmarg | Kulgam / D.H.Pora | 8.00 |
| 552 | Department of Forest and Ecology | PCCF & HoFF | Residential Quarter Mattan | Anantnag | 8.00 |
| 553 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Brad | Anantnag | 8.00 |
| 554 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Hangalpawa | Anantnag | 8.00 |
| 555 | Department of Forest and Ecology | PCCF & HoFF | Checkpost Yanner 2 | Anantnag | 8.00 |
| 556 | Department of Forest and Ecology | PCCF & HoFF | check post sangam | Anantnag/ Bijbehara | 8.00 |
| 557 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House- cum- Inspection Hut | Shopian/ Shopian | 8.00 |
| 558 | Department of Forest and Ecology | PCCF & HoFF | MaliHut | Jammu/R.S.Pura | 8.00 |
| 559 | Department of Forest and Ecology | PCCF & HoFF | Forest Guard Hut Poshkar Kangan | Ganderbal/Kangan | 8.00 |
| 560 | Department of Forest and Ecology | PCCF & HoFF | PI Division office, Dogra Hall | Jammu | 8.00 |
| 561 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Complex, Sumbal | Sumbal | 8.00 |
| 562 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Shangus | Anantnag | 8.00 |
| 563 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Complex, Kulgam | Kulgam | 8.00 |
| 564 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Sub-Judge Court Batote | Ramban | 8.00 |
| 565 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Banihal | Ramban | 8.00 |
| 566 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Complex | Kathua | 8.00 |
| 567 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | I&FC AEE'S Quarter Dawar | Bandipora/Gurez | 7.84 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|---|-------------------------------|------------------|
| 568 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | I&FC JE'S Quarter Kanzalwan | Bandipora/Gurez | 7.81 |
| 569 | Department of Home | Fire and Emergency Service Department- | F&ES Lar | | 7.70 |
| 570 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | I&FC JE'S Quarter Dawar | Bandipora/Gurez | 7.70 |
| 571 | Jal Shakti Department | PHE, Kashmir | Hydrualic Circle Anantnag | Anantnag/Bijbehara | 7.69 |
| 572 | Jal Shakti Department | PHE, Kashmir | PHE Division Bijbehara | Anantnag/Bijbehara | 7.69 |
| 573 | Jal Shakti Department | PHE, Kashmir | PHE Division Bijbehara | Anantnag/Shangus | 7.69 |
| 574 | Jal Shakti Department | PHE, Kashmir | PHE Division Shopian | Shopian | 7.69 |
| 575 | Jal Shakti Department | PHE, Kashmir | PHE Division Shopian | Shopian | 7.69 |
| 576 | Jal Shakti Department | PHE, Kashmir | PHE Division Chadoora | Budgam/Srinagar | 7.69 |
| 577 | Jal Shakti Department | PHE, Kashmir | PHE Division Budgam | Budgam/Budgam | 7.69 |
| 578 | Jal Shakti Department | PHE, Kashmir | PHE Division Budgam | Budgam/ Khansahib | 7.69 |
| 579 | Jal Shakti Department | PHE, Kashmir | PHE Mechanical Division North Sopore | Baramulla Tehsil Baramulla | 7.69 |
| 580 | Department of Home | Sainik Welfare Department,UT of | Zila Sainik Welfare office Akhnoor | | 7.68 |
| 581 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Pump House -I (Kai- Koot) | Pulwama/ Awantipora | 7.53 |
| 582 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional office building at Narbal | Budgam /Narbal | 7.50 |
| 583 | Department of Forest and Ecology | PCCF & HoFF | Range Office cum Check Post SK Bala | Bandipora/Ajas | 7.50 |
| 584 | Department of Forest and Ecology | PCCF & HoFF | Check Post Mader | Bandipora/Bandipora | 7.50 |
| 585 | Department of Forest and Ecology | PCCF & HoFF | Range Office Ningli Sopore | Baramulla/Sopore | 7.50 |
| 586 | Department of Forest and Ecology | PCCF & HoFF | Chowkidars Hut at Kralsangri | Srinagar | 7.50 |
| 587 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Himli Warnow. | Kupwara/Lalpora | 7.50 |
| 588 | Department of Forest and Ecology | PCCF & HoFF | Model Check post Handwara | Kupwara/ Handwara | 7.50 |
| 589 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Jetty Dock Yard cum Office | Baram ulla | 7.48 |
| 590 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court/ Court Complex Anantnag | Anantnag | 7.48 |
| 591 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Lawyers Chamber Building | Jammu | 7.46 |
| 592 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Bakshi Nagar Office | Jammu | 7.27 |
| 593 | Department of Forest and Ecology | Forest Protection Force | Deputy Directors Quarter | Kupwara | 7.21 |
| 594 | Department of Home | Fire and Emergency Service Department- | F&ES Safapora | | 7.20 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|--|--|---------------------|------------------|
| 595 | Department of Forest and Ecology | Wildlife Protection | NIC building Nandini | Jammu/Nagrota | 7.20 |
| 596 | Department of Forest and Ecology | PCCF & HoFF | Range Office Pahalgam | Anantnag | 7.20 |
| 597 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Residential Quarter Lift Irrigation Station | Baramulla/ Watergam | 7.16 |
| 598 | Jal Shakti Department | PHE, Kashmir | WSMP Srinagar | Srinagar/ Eidgah | 7.08 |
| 599 | Department of Forest and Ecology | PCCF & HoFF | Block Officers Hut Keller | Shopian/ Keller | 7.08 |
| 600 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut, Mahanpur | Kathua/Mahanpur | 7.05 |
| 601 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub-divisional Building Beerwah | Budgam / Beerwah | 7.04 |
| 602 | Department of Forest and Ecology | PCCF & HoFF | Central Resin Depott. Dewal | Kathua/Billawar | 7.01 |
| 603 | Department of Home | Department of Home | RPHQ Baramulla | Baramulla | 7.00 |
| 604 | Department of Home | Department of Home | Park/SDPO Office | Srinagar | 7.00 |
| 605 | Department of Home | Department of Home | P/S Kokernag | Anantnag | 7.00 |
| 606 | Department of Home | Department of Home | P/S Larnoo | Anantnag | 7.00 |
| 607 | Department of Home | Department of Home | SDPO Gurez PS | Bandipora (iure | 7.00 |
| 608 | Department of Home | Department of Home | PP Alea Paniba | Bandipora/Aloosa | 7.00 |
| 609 | Department of Home | Department of Home | PS Pacca Danga | Jammu north | 7.00 |
| 610 | Department of Home | Department of Home | PS GRP Udhampur | Udhampur | 7.00 |
| 611 | Department of Home | Department of Home | GOS residence triple story bulding | Anantnag/Mattan | 7.00 |
| 612 | Department of Home | Department of Home | CO Residence double story building. | Anantnag/Mattan | 7.00 |
| 613 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Office I&FC Div Kupwara | Kupwara | 7.00 |
| 614 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Residential Quarter at Halpal Kupwara near | Kupwara | 7.00 |
| 615 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Staff Residential Quarters Building | Anantnag | 7.00 |
| 616 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub Division Samboora | Pulwama/kakapora | 7.00 |
| 617 | Department of Forest and Ecology | Forest Protection Force | RC Ajas | Ajas Bandipora | 7.00 |
| 618 | Department of Forest and Ecology | Forest Protection Force | Guard Barrack Double Storey | Kupwara | 7.00 |
| 619 | Department of Forest and Ecology | Wildlife Protection | Check Post Kiyar | Kishtwar / Dachhan | 7.00 |
| 620 | Department of Forest and Ecology | Wildlife Protection | Mahadev Staff Quarter | Srinagar/Hazratbal | 7.00 |
| 621 | Department of Forest and Ecology | PCCF & HoFF | Eco-tourism Hut No 02 Panagar | Kathua/Basohli | 7.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|--|---|---------------------|------------------|
| 622 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Kardoh | Kathua/Bani | 7.00 |
| 623 | Department of Forest and Ecology | PCCF & HoFF | Mali Hutt at Hut Nursery | Kathua/Basohli | 7.00 |
| 624 | Department of Forest and Ecology | PCCF & HoFF | Range Office Complex, H.Q. | Kathua/Ramkot | 7.00 |
| 625 | Department of Forest and Ecology | PCCF & HoFF | DFO, Office Billawar | Kathua/Billawar | 7.00 |
| 626 | Department of Forest and Ecology | PCCF & HoFF | DFO, Residence Building, Dewal | Kathua/Billawar | 7.00 |
| 627 | Department of Forest and Ecology | PCCF & HoFF | Range Office Complex, Billawar | Kathua/Billawar | 7.00 |
| 628 | Department of Forest and Ecology | PCCF & HoFF | Resin Depott. Parnalla | Kathua/Billawar | 7.00 |
| 629 | Department of Forest and Ecology | PCCF & HoFF | Diivisional Forest Office Eco Task Force | Jammu | 7.00 |
| 630 | Department of Forest and Ecology | PCCF & HoFF | D.F. O Residence at Sarkoot Kishtwar | Kishtwar/Kishtwar | 7.00 |
| 631 | Department of Forest and Ecology | PCCF & HoFF | Range Office at Batote | Ramban/ Batote | 7.00 |
| 632 | Department of Forest and Ecology | PCCF & HoFF | Checkpost Raithan | Budgam / Khansahib | 7.00 |
| 633 | Department of Forest and Ecology | PCCF & HoFF | BO Hut Doursa | Kupwara /Sogam | 7.00 |
| 634 | Department of Forest and Ecology | PCCF & HoFF | BO Hut Lalpora | Kupwara /Lalpora | 7.00 |
| 635 | Department of Forest and Ecology | PCCF & HoFF | Forest Check post gushi | Kupwara | 7.00 |
| 636 | Department of Forest and Ecology | PCCF & HoFF | Rest House Zachaldara | Kupwara/ Zachaldara | 7.00 |
| 637 | Jal Shakti Department | PHE, Jammu | F.P No. 2 | Jammu | 6.92 |
| 638 | Jal Shakti Department | PHE, Kashmir | RWS Ganderbal | Ganderbal/ Kangan | 6.92 |
| 639 | Jal Shakti Department | PHE, Kashmir | RWS Ganderbal | Ganderbal/ Kangan | 6.92 |
| 640 | Jal Shakti Department | PHE, Kashmir | RWS Ganderbal | Ganderbal/ Kangan | 6.92 |
| 641 | Jal Shakti Department | PHE, Kashmir | Hydraulic Division Gurez | Bandipora/Gurez | 6.92 |
| 642 | Jal Shakti Department | PHE, Kashmir | PHE Division Baramulla | Baramulla/Baramulla | 6.92 |
| 643 | Jal Shakti Department | PHE, Kashmir | PHE Division Baramulla | Baramulla/Baramulla | 6.92 |
| 644 | Jal Shakti Department | PHE, Kashmir | PHE Division Baramulla | Baramulla/Baramulla | 6.92 |
| 645 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Chanwari Malikpora | Kupwara/ Qalamabad | 6.88 |
| 646 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Sogam | Kupwara /Sogam | 6.80 |
| 647 | Department of Forest and Ecology | PCCF & HoFF | Palma Check Post | Rajouri | 6.75 |
| 648 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | I&FC Weir Guest House Chattabal | Srinagar/South | 6.72 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|---|--|-------------------------------|------------------|
| 649 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Irrigation Sub division Dooru | Anantnag/Dooru | 6.72 |
| 650 | Department of Forest and Ecology | PCCF & HoFF | RO, Kupwara's Old residence, Zangli | Kupwara/Kupwara | 6.70 |
| 651 | Department of Home | Department of Home | PP Bhalra | Doda Bhaderwah | 6.67 |
| 652 | Jal Shakti Department | PHE, Kashmir | WSMP Srinagar | Budgam / Beerwah | 6.55 |
| 653 | Department of Home | Sainik Welfare Department,UT of | Sainik Rest House Basohli | | 6.50 |
| 654 | Department of Forest and Ecology | Soil & Water Conservation | District Soil Conservation Office | Anantnag / Anantnag | 6.50 |
| 655 | Department of Forest and Ecology | Ecology Environment & Remote Sensing | Office at Environment Park Bhawani Nagar, | Jammu | 6.50 |
| 656 | Department of Forest and Ecology | PCCF & HoFF | Range Office/ Residence Kuntwara | Doda/Doda | 6.50 |
| 657 | Department of Forest and Ecology | PCCF & HoFF | Divisional Office Building, Ramban | Ramban/Ramban | 6.50 |
| 658 | Department of Forest and Ecology | PCCF & HoFF | Range Office Banihal (SC) | Ramban/Banihal | 6.50 |
| 659 | Department of Forest and Ecology | PCCF & HoFF | Rest House, Matchil,Matchil | Kupwara/Matchil | 6.50 |
| 660 | Jal Shakti Department | PHE, Kashmir | QCPD Srinagar | Srnagar/Shalteng | 6.46 |
| 661 | Jal Shakti Department | PHE, Kashmir | FBI Division Tangmarg | Baramulla/Pattan | 6.31 |
| 662 | Department of Forest and Ecology | Wildlife Protection | Staff Quarter, Pahalgam | Anantnag | 6.20 |
| 663 | Jal Shakti Department | PHE, Kashmir | Hydrualic Circle Baramulla | Baramulla/Sopore | 6.15 |
| 664 | Jal Shakti Department | PHE, Kashmir | PHE Division Bijbehara | Anantnag/Pahalgam | 6.15 |
| 665 | Jal Shakti Department | PHE, Kashmir | PHE Division Shopian | Zainapora/Shopian | 6.15 |
| 666 | Jal Shakti Department | PHE, Kashmir | PHE Division Pulwama | Pulwama/Pulwama | 6.15 |
| 667 | Jal Shakti Department | PHE, Kashmir | PHE Division Awantipora | Pulwama/PamporeAw antipora | 6.15 |
| 668 | Jal Shakti Department | PHE, Kashmir | RWS Ganderbal | Ganderbal/ Kangan | 6.15 |
| 669 | Jal Shakti Department | PHE, Kashmir | RWS Ganderbal | Ganderbal/ Kangan | 6.15 |
| 670 | Jal Shakti Department | PHE, Kashmir | Hydraulic Division Gurez | Bandipora/Gurez | 6.15 |
| 671 | Jal Shakti Department | PHE, Kashmir | PHE Division Baramulla | Baramulla/Baramulla | 6.15 |
| 672 | Jal Shakti Department | PHE, Kashmir | PHE Division Baramulla | Baramulla/ Wagoora | 6.15 |
| 673 | Jal Shakti Department | PHE, Kashmir | FBI Division Tangmarg | Baramulla/Tangmarg | 6.15 |
| 674 | Jal Shakti Department | PHE, Kashmir | PHE Mechanical Division Srinagar | Srinagar | 6.15 |
| 675 | Jal Shakti Department | PHE, Jammu | Computer Section area | Jammu | 6.12 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-----------------------|--|--|-------------------------|------------------|
| 676 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Gang Hut (Bidhard Vailoo | Anantnag/ Larnoo | 6.02 |
| 677 | Department of Home | Department of Home | P/S Kralkhud | Srinagar | 6.00 |
| 678 | Department of Home | Department of Home | DPL Kishtwar | Kishtwar | 6.00 |
| 679 | Department of Home | Department of Home | PS Kishtwar | Kishtwar | 6.00 |
| 680 | Department of Home | Department of Home | PS Chairoo | Kishtwar | 6.00 |
| 681 | Department of Home | Department of Home | P/S Bhaderwah | Distt. Doda Bhaderwah | 6.00 |
| 682 | Department of Home | Department of Home | P/S Gandoh | Distt. Doda veh. Gandoh | 6.00 |
| 683 | Department of Home | Department of Home | JIC/PS Building/SHO Quarter Humhama | Budgam | 6.00 |
| 684 | Department of Home | Department of Home | JK Housing Colony Sector 43 Plot No.GH- | Gurugram/Wazirabad | 6.00 |
| 685 | Department of Home | Department of Home | NGO's Block-A | Ganderbal /Kangan. | 6.00 |
| 686 | Department of Home | Department of Home | NGO's Block-B | Ganderbal /Kangan. | 6.00 |
| 687 | Department of Home | Department of Home | Women Hostel | Ganderbal /Kangan. | 6.00 |
| 688 | Department of Home | Fire and Emergency Service Department- | F&ES Kokenag | | 6.00 |
| 689 | Department of Home | Fire and Emergency Service Department- | F&ES Larnoo | | 6.00 |
| 690 | Department of Home | Fire and Emergency Service Department- | F&ES Uri | | 6.00 |
| 691 | Department of Home | HeadQuarter HomeGuard,SDRF,Ci | Director HG Kashmir | | 6.00 |
| 692 | Department of Home | HeadQuarter HomeGuard,SDRF,Ci | HG/CD & SDRF Baramulla | | 6.00 |
| 693 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Office Building Ompora, | Budgam/ Budgam | 6.00 |
| 694 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Spill Channel Sub-Division | Srinagar /Narbal | 6.00 |
| 695 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Spill Channel Sub-Division | Budgam /Budgam | 6.00 |
| 696 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Irrigation Sub Division Suthsoo | Budgam/ B.K Pora | 6.00 |
| 697 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Stores | Anantnag | 6.00 |
| 698 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Residential quarter Building No. 2 | Anantnag | 6.00 |
| 699 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Residential quarters Building No.03 | Anantnag | 6.00 |
| 700 | Jal Shakti Department | Irrigation & Flood | Irrigation Sub Divison | Anantnag/Mattan | 6.00 |
| 701 | Jal Shakti Department | Control, Kashmir Irrigation & Flood | Bawan Exen Quarter | Shopian/Shopian | 6.00 |
| 702 | Jal Shakti Department | Control, Kashmir Irrigation & Flood Control, Kashmir | Alyapora Shopian Divisional Office Complex (Giraj) | Ganderbal/Ganderbal | 6.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|---|--------------------------|------------------|
| 703 | Jal Shakti Department | Irrigation & Flood | Sub division office | Baramulla/Sopore | 6.00 |
| 704 | Department of Forest | Control, Kashmir Soil & Water | building Sopore Soil & Water | Jammu/Bahu | 6.00 |
| | and Ecology Department of Forest | Conservation | Conservation Office Divisional Office, SF | | |
| 705 | and Ecology | Social Forestry | Division, Doda, Forest | Doda/ Doda | 6.00 |
| 706 | Department of Forest and Ecology | Social Forestry | Guard hut at Co. 12/V Naringwari | Anantnag/ Dooru | 6.00 |
| 707 | Department of Forest and Ecology | Ecology Environment & Remote Sensing | NIC at Environment Park, Jammu | Jammu | 6.00 |
| 708 | Department of Forest | Forest Protection | Assistant Director | Jammu/Jammu | 6.00 |
| 709 | and Ecology Department of Forest | Force Forest Protection | Quarter/Forest Guard Barrack No | Jammu/Jammu | 6.00 |
| | and Ecology Department of Forest | Force Forest Protection | 1/Forest Protection DD Residence | | |
| 710 | and Ecology | Force | Building | Kupwara | 6.00 |
| 711 | Department of Forest and Ecology | Wildlife Protection | Staff Qtr. Near Nic | Srinagar/Hazratbal | 6.00 |
| 712 | Department of Forest and Ecology | Wildlife Protection | Staff Quarter/NIC Bijhama Lachipora | Baramulla | 6.00 |
| 713 | Department of Forest and Ecology | Wildlife Protection | Bathnoor Staff Quarter | Pulwama/Tral | 6.00 |
| 714 | Department of Forest and Ecology | Wildlife Protection | Shikargah Staff Quarter | Pulwama/Tral | 6.00 |
| 715 | Department of Forest and Ecology | Wildlife Protection | Hangalgar Chowkidar Hut | Pulwama/Tral | 6.00 |
| 716 | Department of Forest and Ecology | Wildlife Protection | Khangund Range Office new | Pulwama/Aripal | 6.00 |
| 717 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Kaneri | Jammu/Chowki Choura | 6.00 |
| 718 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut at Khained | Udhampur / Basantgarh | 6.00 |
| 719 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut at Khained | Udhampur / Basantgarh | 6.00 |
| 720 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut Chowki | Udhampur / Ramnagar | 6.00 |
| 721 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut at Panjgrian | Udhampur / Ramnagar | 6.00 |
| 722 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut at Barmeen | Udhampur / Ramnagar | 6.00 |
| 723 | Department of Forest and Ecology | PCCF & HoFF | CF West Office,Rajouri | Rajouri | 6.00 |
| 724 | Department of Forest and Ecology | PCCF & HoFF | CF Residence | Rajouri | 6.00 |
| 725 | Department of Forest and Ecology | PCCF & HoFF | Inspection Hut, Panja Rajouri | Rajouri | 6.00 |
| 726 | Department of Forest | PCCF & HoFF | Chowkidar Quarter, | Rajouri | 6.00 |
| 727 | and Ecology Department of Forest | PCCF & HoFF | Panja Dehrilayote Check | Rajouri | 6.00 |
| 728 | and Ecology Department of Forest | PCCF & HoFF | Post Guard hut at Beoli | J Doda/Doda | 6.00 |
| 729 | and Ecology Department of Forest and Ecology | PCCF & HoFF | Wanda Inspection Hut at Forest Complex Doda | Doda/Doda | 6.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|---|-------------|---|----------------------------|------------------|
| 730 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut at Bangalnar Palmar | Kishtwar/Kishtwar | 6.00 |
| 731 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Yourdu at Yourdu | Kishtwar/Marwah | 6.00 |
| 732 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut at Singhpur | Kishtwar/Chatroo | 6.00 |
| 733 | Department of Forest and Ecology | PCCF & HoFF | Range Office Gool | Ramban/Gool | 6.00 |
| 734 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Gool | Ramban/Gool | 6.00 |
| 735 | Department of Forest and Ecology | PCCF & HoFF | Resin Hut Sangaldaan | Ramban/Gool | 6.00 |
| 736 | Department of Forest and Ecology | PCCF & HoFF | Range Officer Residence (New), | Ramban/Gool | 6.00 |
| 737 | Department of Forest and Ecology | PCCF & HoFF | Forest Control Room Banihal | Ramban/Banihal | 6.00 |
| 738 | Department of Forest and Ecology | PCCF & HoFF | Forest Control Room Ramban | Ramban/Ramban | 6.00 |
| 739 | Department of Forest and Ecology | PCCF & HoFF | Check Post at Kharpora | Ramban/Banihal | 6.00 |
| 740 | Department of Forest and Ecology | PCCF & HoFF | Chowkidar Hut Banihal | Ramban/Banihal | 6.00 |
| 741 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Digdole | Ramban/Ramban | 6.00 |
| 742 | Department of Forest and Ecology | PCCF & HoFF | Divisional Forest Office Bandipora | Bandipora/Bandipora | 6.00 |
| 743 | Department of Forest and Ecology | PCCF & HoFF | DFO Residence Chitternar Forest Rest House | Bandipora/Bandipora | 6.00 |
| 744 | Department of Forest and Ecology | PCCF & HoFF | Chitternar | Bandipora/Bandipora | 6.00 |
| 745 | Department of Forest and Ecology | PCCF & HoFF | Range Office Khuihama Forest Check Post | Bandipora/Bandipora | 6.00 |
| 746 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Nowpora Forest Check Post | Bandipora/Bandipora | 6.00 |
| 747 | Department of Forest and Ecology Department of Forest | PCCF & HoFF | Porest Check Post Pazalpora Forest Check Post | Bandipora/Bandipora | 6.00 |
| 748 | and Ecology Department of Forest | PCCF & HoFF | Arampora Forest Checkpost | Bandipora/Bandipora | 6.00 |
| 749 | and Ecology Department of Forest | PCCF & HoFF | Porest Checkpost Panzigam Forest Checkpost | Bandipora/Bandipora | 6.00 |
| 750 | and Ecology Department of Forest | PCCF & HoFF | Kanibathi Block Officers Hut | Bandipora/Bandipora | 6.00 |
| 751 | and Ecology Department of Forest | PCCF & HoFF | Athwatoo Range Office | Bandipora/Bandipora | 6.00 |
| 752 | and Ecology Department of Forest | PCCF & HoFF | Ajas(NEW) Check Post | Bandipora/Ajas | 6.00 |
| 753 | and Ecology Department of Forest | PCCF & HoFF | Kanzalwan Gurez Block Officers Hut | Bandipora/Gurez | 6.00 |
| 754 | and Ecology Department of Forest | PCCF & HoFF | Dawar Gurez BO's Hut Mono | Bandipora/Gurez Budgam/ | 6.00 |
| 755 | and Ecology Department of Forest | PCCF & HoFF | Zinpanchal | Chrar-i-Sharief | 6.00 |
| 756 | and Ecology | PCCF & HoFF | DFO Office Budgam | Budgam/ Budgam | 6.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|---|--------------------------|------------------|
| 757 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Bosian (Co.3/Khadinyar) | Baramulla/Baramulla | 6.00 |
| 758 | Department of Forest and Ecology | PCCF & HoFF | Chepost Shiva | Baramulla /Dangerpora | 6.00 |
| 759 | Department of Forest and Ecology | PCCF & HoFF | Rest House, Harwan | Baramulla/Zaingeer | 6.00 |
| 760 | Department of Forest and Ecology | PCCF & HoFF | Mali Hut Chandigam. | Kupwara /Sogam | 6.00 |
| 761 | Department of Forest and Ecology | PCCF & HoFF | Forest Check post Zangli | Kupwara/Kupwara | 6.00 |
| 762 | Department of Forest and Ecology | PCCF & HoFF | Forest Check post Sulkoote | Kupwara/Kupwara | 6.00 |
| 763 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest Thayan Kalarooe | Kupwara/Kupwara | 6.00 |
| 764 | Department of Forest and Ecology | PCCF & HoFF | Checkpost Baghballa | Kupwara | 6.00 |
| 765 | Department of Forest and Ecology | PCCF & HoFF | Range office Kakapora | Pulwama/ Kakapora | 6.00 |
| 766 | Department of Forest and Ecology | PCCF & HoFF | Block Officers Hut at Hallan | Kulgam / D.H.Pora | 6.00 |
| 767 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Ichnard | Anantnag | 6.00 |
| 768 | Department of Forest and Ecology | PCCF & HoFF | Principal Residene | Bandipora | 6.00 |
| 769 | Department of Forest and Ecology | PCCF & HoFF | PI Division office, CAMPA Building, | Srinagar | 6.00 |
| 770 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Aishmuqam | Anantnag | 6.00 |
| 771 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Complex Kangan | Ganderbal | 6.00 |
| 772 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Sub Judge Court Uri | Baramulla | 6.00 |
| 773 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Additonal Special Mobile magistrate | Kralpara | 6.00 |
| 774 | Jal Shakti Department | PHE, Kashmir | QCPD Srinagar | Srnagar/Shalteng | 5.92 |
| 775 | Department of Forest and Ecology | PCCF & HoFF | Forest Emergencies Management Centre | Srinagar | 5.90 |
| 776 | Department of Forest and Ecology | JK Forest Development | Divisional Office, TUD Division, J&K | Srinagar/Shalteng | 5.79 |
| 777 | Jal Shakti Department | PHE, Kashmir | RWS Ganderbal | Ganderbal/ Kangan | 5.77 |
| 778 | Jal Shakti Department | PHE, Kashmir | Hyd. Division Bandipora | Bandipora/Bandipora | 5.77 |
| 779 | Jal Shakti Department | PHE, Kashmir | PHE Division Baramulla | Baramulla/ Wagoora | 5.77 |
| 780 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Control Division Anantnag/ | Anantnag/ Anantnag | 5.76 |
| 781 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post, Pargana | Kathua/Basohli | 5.67 |
| 782 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | zildari office Beerwah | Budgam /Beerwah | 5.60 |
| 783 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Hydraulic Sub Division Mir Bazar | Anantnag/ Anantnag | 5.60 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|--|---|------------------|
| 784 | Department of Forest and Ecology | PCCF & HoFF | Office/ ACF Quarter | Ramban/ Batote | 5.60 |
| 785 | Department of Forest and Ecology | PCCF & HoFF | Instructor Quarter-II | Jammu/R.S.Pura | 5.60 |
| 786 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Advocate General (New Building) | Srinagar | 5.60 |
| 787 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Mechanical Quarter (Larkipora | Pulwama/ Awantipora | 5.59 |
| 788 | Department of Home | Department of Home | PS Qalamaha | Main Building Kupwara/ | 5.56 |
| 789 | Department of Home | Department of Home | PP Zachaldara | Main Building Kupwara/Handwara | 5.56 |
| 790 | Department of Home | Department of Home | PP Magam | Main Building Kupwara/Handwara | 5.56 |
| 791 | Jal Shakti Department | PHE, Jammu | Staff Qtr. 1 | Jammu | 5.54 |
| 792 | Jal Shakti Department | PHE, Jammu | Staff Qtr. 2 | Jammu | 5.54 |
| 793 | Jal Shakti Department | PHE, Jammu | Staff Qtr. 3 | Jammu | 5.54 |
| 794 | Jal Shakti Department | PHE, Jammu | Staff Qtr. 4 | Jammu | 5.54 |
| 795 | Jal Shakti Department | PHE, Jammu | Staff Qtr. 5 | Jammu | 5.54 |
| 796 | Jal Shakti Department | PHE, Jammu | Staff Qtr. 6 | Jammu | 5.54 |
| 797 | Department of Home | Department of Home | PS Qalamaha | Administrative Hlock Kupwara/ Handwara | 5.53 |
| 798 | Department of Home | Department of Home | GO's Mess Gulmarg | Baramulla | 5.50 |
| 799 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Doodpathri | Budgam / Khansahib | 5.50 |
| 800 | Department of Forest and Ecology | PCCF & HoFF | Two Room Hut at Kralsangri+ Nursery | Srinagar | 5.50 |
| 801 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Drangbal | Baramulla/Baramulla | 5.50 |
| 802 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Shahnagri | Kupwara/ Qalamabad | 5.50 |
| 803 | Department of Forest and Ecology | PCCF & HoFF | Check Post Tikna | Pulwama/ Pulwama | 5.50 |
| 804 | Department of Forest and Ecology | PCCF & HoFF | Check Post Domail | Shopian/ Shopian | 5.50 |
| 805 | Department of Forest and Ecology | PCCF & HoFF | Block Officers Hut Chotipora | Shopian/ Shopian | 5.40 |
| 806 | Jal Shakti Department | PHE, Kashmir | PHE Division Shopian | Shopian | 5.38 |
| 807 | Jal Shakti Department | PHE, Kashmir | PHE Division Shopian | Shopian | 5.38 |
| 808 | Jal Shakti Department | PHE, Kashmir | RWS Ganderbal | Ganderbal/ Gund | 5.38 |
| 809 | Jal Shakti Department | PHE, Kashmir | RWS Ganderbal | Ganderbal | 5.38 |
| 810 | Jal Shakti Department | PHE, Kashmir | PHE Division Baramulla | Baramulla/Baramulla | 5.38 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|--|--|----------------------------|------------------|
| 811 | Jal Shakti Department | PHE, Kashmir | PHE Mechanical Division Srinagar | Budgam/ Bkpora | 5.38 |
| 812 | Jal Shakti Department | PHE, Kashmir | Mechanical Rural Division Srinagar | Srinagar/Srinagar North | 5.38 |
| 813 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub Divisional Office I&FC Dawar | Bandipora/Gurez | 5.27 |
| 814 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Irrigation Sub Division Pattan | Baramulla/ Pattan | 5.27 |
| 815 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Exen Office, Technical Section and | Srinagar | 5.25 |
| 816 | Department of Forest and Ecology | JK Forest Development | Divisional Office, Extraction Division, | Rajouri/Rajouri | 5.25 |
| 817 | Department of Forest and Ecology | PCCF & HoFF | Range Office Mawar cum residence | Kupwara/ Langate | 5.25 |
| 818 | Department of Forest and Ecology | PCCF & HoFF | Check Post Mimendar | Shopian/ Shopian | 5.25 |
| 819 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Gulshanpora | Pulwama/ Tral | 5.12 |
| 820 | Department of Home | Department of Home | DIG(P) Residence | Baramulla | 5.00 |
| 821 | Department of Home | Department of Home | P/S Maisum | Srinagar | 5.00 |
| 822 | Department of Home | Department of Home | SDPO West Office | Srinagar | 5.00 |
| 823 | Department of Home | Department of Home | P/S Bemina | Srinagar | 5.00 |
| 824 | Department of Home | Department of Home | P/S Parimpora | Srinagar | 5.00 |
| 825 | Department of Home | Department of Home | Panthachowk | Srinagar | 5.00 |
| 826 | Department of Home | Department of Home | PS Rajbagh | Srinagar | 5.00 |
| 827 | Department of Home | Department of Home | PS Batamaloo | Srinagar | 5.00 |
| 828 | Department of Home | Department of Home | PS Karan Nagar | Srinagar | 5.00 |
| 829 | Department of Home | Department of Home | DPO Budgam | Budgam | 5.00 |
| 830 | Department of Home | Department of Home | PP Shadipora | Wakura | 5.00 |
| 831 | Department of Home | Department of Home | Telecom Building | Anantnag | 5.00 |
| 832 | Department of Home | Department of Home | P/S Utrasoo | Anantnag | 5.00 |
| 833 | Department of Home | Department of Home | P/S Achabal | Anantnag | 5.00 |
| 834 | Department of Home | Department of Home | PP Sheerbagh | Anantnag | 5.00 |
| 835 | Department of Home | Department of Home | PP General Bus Stand | Anantnag | 5.00 |
| 836 | Department of Home | Department of Home | ASP/DYSP (OPS) Office | Pulwama | 5.00 |
| 837 | Department of Home | Department of Home | Police Station Kakapora/ PS Building | Pulwama | 5.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--------------------|--------------------|-----------------------------------|--------------------------------------|------------------|
| 838 | Department of Home | Department of Home | DPO Kulgam | Kulgam | 5.00 |
| 839 | Department of Home | Department of Home | DPO Kulgam | Kulgam | 5.00 |
| 840 | Department of Home | Department of Home | PP Berthipora Main Building | Keegam | 5.00 |
| 841 | Department of Home | Department of Home | PP Saidapora Main Building | Shopian | 5.00 |
| 842 | Department of Home | Department of Home | PP Herman Main Building | Hermain | 5.00 |
| 843 | Department of Home | Department of Home | ASP/DySP Hqrs | Kupwara | 5.00 |
| 844 | Department of Home | Department of Home | Police Station | alpora | 5.00 |
| 845 | Department of Home | Department of Home | Police Station | Trchgam | 5.00 |
| 846 | Department of Home | Department of Home | Police Station | Kralpora | 5.00 |
| 847 | Department of Home | Department of Home | Womens Police | Kupwara | 5.00 |
| 848 | Department of Home | Department of Home | Police Post | Kupwara | 5.00 |
| 849 | Department of Home | Department of Home | Police Post | Kupwara | 5.00 |
| 850 | Department of Home | Department of Home | Police Post Hatmulla | Kupwara | 5.00 |
| 851 | Department of Home | Department of Home | PS Tulai | Bandipora/Tulai | 5.00 |
| 852 | Department of Home | Department of Home | PP Sumlar | Bandiporx/ | 5.00 |
| 853 | Department of Home | Department of Home | PPO Handwan | SPaffics Handwara Kunwaa/Handwara | 5.00 |
| 854 | Department of Home | Department of Home | Dv.SP DAR DPL Kathua | KathuwKathua | 5.00 |
| 855 | Department of Home | Department of Home | PCR Building Kathua | Kathua' | 5.00 |
| 856 | Department of Home | Department of Home | Administrative Block Kathua | Kathua' | 5.00 |
| 857 | Department of Home | Department of Home | Conference Hall DPL Kathua | Kathua' | 5.00 |
| 858 | Department of Home | Department of Home | Recreuon Ilall DPL Kathua | Kathua' | 5.00 |
| 859 | Department of Home | Department of Home | DPO R amban | R amban | 5.00 |
| 860 | Department of Home | Department of Home | Main Barrack Building DPL Doda | Doda | 5.00 |
| 861 | Department of Home | Department of Home | ASP (PCY)/DySP DAR office Doda | Doda | 5.00 |
| 862 | Department of Home | Department of Home | Welfare Complex DPI. Doda | Distt. Teh. Doda | 5.00 |
| 863 | Department of Home | Department of Home | NGO Block DPL Doda | Distt. Teh. Doda | 5.00 |
| 864 | Department of Home | Department of Home | ASP office Bhaderwah | Distt. Doda Teh. Bhaderwah | 5.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--------------------|--------------------|-------------------------------------|-------------------------------|------------------|
| 865 | Department of Home | Department of Home | SDPO office Bhaderah | Distt. Doda Teh. Bhaderwah | 5.00 |
| 866 | Department of Home | Department of Home | P/S Doda | Doda | 5.00 |
| 867 | Department of Home | Department of Home | P/S Thathri | Distt. Teh. Thathri | 5.00 |
| 868 | Department of Home | Department of Home | P/S Assar | Distt. Doda Teh. Assar | 5.00 |
| 869 | Department of Home | Department of Home | PS Dessa | Doda Bhagwah | 5.00 |
| 870 | Department of Home | Department of Home | PP Kastigarh | Doda Kastigarh | 5.00 |
| 871 | Department of Home | Department of Home | PP Karara | Doda Bhaderwah | 5.00 |
| 872 | Department of Home | Department of Home | PP Premnagar | Doda Thathri | 5.00 |
| 873 | Department of Home | Department of Home | cid ci Kathua | Kathua | 5.00 |
| 874 | Department of Home | Department of Home | DPO Kulgam | Kulgam | 5.00 |
| 875 | Department of Home | Department of Home | DPO Udampur | Udhampur | 5.00 |
| 876 | Department of Home | Department of Home | DPO Samba | Samba | 5.00 |
| 877 | Department of Home | Department of Home | DPO Kishtwar | Kishtwar | 5.00 |
| 878 | Department of Home | Department of Home | PS Bakshi Nagar | jammu north | 5.00 |
| 879 | Department of Home | Department of Home | PS Janipur | Jammu West | 5.00 |
| 880 | Department of Home | Department of Home | PS Nowanad | Jammu north | 5.00 |
| 881 | Department of Home | Department of Home | P/S Bahu Fort | Bahu | 5.00 |
| 882 | Department of Home | Department of Home | P/S Bagh-e-bahu | Bahu | 5.00 |
| 883 | Department of Home | Department of Home | PS Miran Sahib | R.S.Pura | 5.00 |
| 884 | Department of Home | Department of Home | PS Domana | Jammu north | 5.00 |
| 885 | Department of Home | Department of Home | PP Sarwal | jammu north | 5.00 |
| 886 | Department of Home | Department of Home | HG/CD & SDRF Budgam | Budgam | 5.00 |
| 887 | Department of Home | Department of Home | IIG/CD & SDRF Anantnag | Anantnag | 5.00 |
| 888 | Department of Home | Department of Home | B3/B4 Training B arrack | Samba | 5.00 |
| 889 | Department of Home | Department of Home | B7/B8 Training Barrack Vijaypur | Samba | 5.00 |
| 890 | Department of Home | Department of Home | District telecom I IqrsB udganÄ | Budgann | 5.00 |
| 891 | Department of Home | Department of Home | [District Telecom I iqrs.Shopian | Shopian | 5.00 |
| | | | | | |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-----------------------|--|---|--------------------|------------------|
| 892 | Department of Home | Department of Home | District Telecom I Iqrs Ganderbal | Ganderbal | 5.00 |
| 893 | Department of Home | Department of Home | District Telecom IqrsKums ara | Kupwara | 5.00 |
| 894 | Department of Home | Department of Home | District J elecom llqrs Anantnag | Anantnag | 5.00 |
| 895 | Department of Home | Department of Home | District Telecom liqrsBandipora | Bandipora | 5.00 |
| 896 | Department of Home | Department of Home | District Telecom IIqrsPulwama | Pulwama | 5.00 |
| 897 | Department of Home | Department of Home | Trainees Barrack (Tiger Hill) | Ganderbal /Kangan. | 5.00 |
| 898 | Department of Home | Department of Home | Trainees Barrack Chinab | Ganderbal /Kangan. | 5.00 |
| 899 | Department of Home | Department of Home | Trainees Barrack Tawi | Ganderbal /Kangan. | 5.00 |
| 900 | Department of Home | Department of Home | Trainees Barrack Sindh | Ganderbal /Kangan. | 5.00 |
| 901 | Department of Home | Department of Home | Principal's Residence | Kathua | 5.00 |
| 902 | Department of Home | Department of Home | SDPO West | Jammu north | 5.00 |
| 903 | Department of Home | Department of Home | Trade Man Double Story building | Anantnag/Mattan | 5.00 |
| 904 | Department of Home | Department of Home | M l. Room double story budding | Anantnag/Mattan | 5.00 |
| 905 | Department of Home | Department of Home | Quarter Guard signal story building. | Anantnag/Mattan | 5.00 |
| 906 | Department of Home | Department of Home | "HG/CD&SDRF Udhampur" | udhampur | 5.00 |
| 907 | Department of Home | Department of Home | SDRF Doda | doda | 5.00 |
| 908 | Jal Shakti Department | PHE, Kashmir | PHE Division Bijbehara | Anantnag/Bijbehara | 5.00 |
| 909 | Jal Shakti Department | PHE, Kashmir | RWS Ganderbal | Ganderbal/ Kangan | 5.00 |
| 910 | Jal Shakti Department | PHE, Kashmir | QCPD Srinagar | Srnagar/Shalteng | 5.00 |
| 911 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Spill Channel Sub-Division | Budgam /Chadoora | 5.00 |
| 912 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Irrigation Sub Division Sangrama | Baramulla/ Sopore | 5.00 |
| 913 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub Divisional Building (Quazigund) | kulgam/Qazigund | 5.00 |
| 914 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub Divisional Building (D.H. Pora) | kulgam/D.H.Pora | 5.00 |
| 915 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Sub divisional Office (ISD Sumbal) | Bandipora/Sumbal | 5.00 |
| 916 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Data Collection Sub Division Batmalo | Srinagar/South | 5.00 |
| 917 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | AEE Anantnag's Residential Quarter | Anantnag | 5.00 |
| 918 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Irrigation sub Division Bijbehara | Anantnag/Bijbehara | 5.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|--|---|--------------------------|------------------|
| 919 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Xen Office | Pulwama/Awantipora | 5.00 |
| 920 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional/Sub- Divisional Office | Pulwama/Awantipora | 5.00 |
| 921 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Flood Contol Sub Division DH Pora | Kulgam/ DH Pora | 5.00 |
| 922 | Department of Forest and Ecology | Forest Protection Force | Guard Barrack No 4/Forest Protection | Jammu/Jammu | 5.00 |
| 923 | Department of Forest and Ecology | Forest Protection Force | Tiolet Block Hardupanzoo | Budgam, Khansahib | 5.00 |
| 924 | Department of Forest and Ecology | Forest Protection Force | Chowkidar Hut/ Garage Hardupanzoo | Budgam, Khansahib | 5.00 |
| 925 | Department of Forest and Ecology | Wildlife Protection | Nambal Staff Quarter | Srinagar/Hazratbal | 5.00 |
| 926 | Department of Forest and Ecology | Wildlife Protection | Staff Quarter/NIC Tawrian Limber | Boniyar | 5.00 |
| 927 | Department of Forest and Ecology | Wildlife Protection | Staff Quarter/NIC Bijhama Lachipora | Boniyar | 5.00 |
| 928 | Department of Forest and Ecology | Wildlife Protection | Guard Hut Nagwadi Pinglish | Pulwama/Tral | 5.00 |
| 929 | Department of Forest and Ecology | Wildlife Protection | Khangund Range Office old | Pulwama/Aripal | 5.00 |
| 930 | Department of Forest and Ecology | Wildlife Protection | Daramgund Checkpost | Pulwama/Aripal | 5.00 |
| 931 | Department of Forest and Ecology | Wildlife Protection | Hajan Staff Quarter | Pulwama/Aripal | 5.00 |
| 932 | Department of Forest and Ecology | Wildlife Protection | Watkhour Satoora Staff Quarter | Pulwama/Aripal | 5.00 |
| 933 | Department of Forest and Ecology | Wildlife Protection | Staff Qtr. for anti- poaching Hokersar | Srinagar | 5.00 |
| 934 | Department of Forest and Ecology | Wildlife Protection | Guard Room /PA Gate Hokersar | Srinagar | 5.00 |
| 935 | Department of Forest and Ecology | Wildlife Protection | Two Story Garage cum store/Control | Srinagar | 5.00 |
| 936 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Ransoo | Reasi/Pouni | 5.00 |
| 937 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Thangriote | Rajouri/Sunderbani | 5.00 |
| 938 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Ranthal | Rajouri/Kalakote | 5.00 |
| 939 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Salheri | Rajouri/Sunderbani | 5.00 |
| 940 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut, Kathera | Kathua/Kathua | 5.00 |
| 941 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut, Malni | Kathua/kathua | 5.00 |
| 942 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Sundrikote | Kathua/kathua | 5.00 |
| 943 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Basantgarh (Old) | Udhampur / Basantgarh | 5.00 |
| 944 | Department of Forest and Ecology | PCCF & HoFF | Range Office Mohargarh | Samba/Samba | 5.00 |
| 945 | Department of Forest and Ecology | PCCF & HoFF | Chowkidar Hut Mansar Morh | Samba/Samba | 5.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|-------------|---|----------------------|------------------|
| 946 | Department of Forest and Ecology | PCCF & HoFF | Check Post Sansari | Kishtwar/Paddar | 5.00 |
| 947 | Department of Forest and Ecology | PCCF & HoFF | Range Office Cum Residence, Ramban | Ramban/Ramban | 5.00 |
| 948 | Department of Forest and Ecology | PCCF & HoFF | B.O's Hut Sitaharan | Budgam/Khansahib | 5.00 |
| 949 | Department of Forest and Ecology | PCCF & HoFF | Range Office Sindh Range Kangan | Ganderbal/Kangan | 5.00 |
| 950 | Department of Forest and Ecology | PCCF & HoFF | Range Office, TTR, PC DEPOT | Srinagar | 5.00 |
| 951 | Department of Forest and Ecology | PCCF & HoFF | BO Hut Rajpora | Baramulla/Dangerpora | 5.00 |
| 952 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut, Chountwari, | Kupwara/Matchil | 5.00 |
| 953 | Department of Forest and Ecology | PCCF & HoFF | Range Office Kupwara Zangli | Kupwara/Kupwara | 5.00 |
| 954 | Department of Forest and Ecology | PCCF & HoFF | Range office, North Lolab Zangli kupwara | Kupwara/Kupwara | 5.00 |
| 955 | Department of Forest and Ecology | PCCF & HoFF | Forest check post Branwari | Kupwara | 5.00 |
| 956 | Department of Forest and Ecology | PCCF & HoFF | Chokidar Hut Dedikoot | Kupwara | 5.00 |
| 957 | Department of Forest and Ecology | PCCF & HoFF | BO's Hut Ladwa (under Construction) | Baramulla/ Rohama | 5.00 |
| 958 | Department of Forest and Ecology | PCCF & HoFF | Bo's Hut Langate | Kupwara/ Langate | 5.00 |
| 959 | Department of Forest and Ecology | PCCF & HoFF | EAC Handwara (Meeting Hall) | Kupwara/ Handwara | 5.00 |
| 960 | Department of Forest and Ecology | PCCF & HoFF | Residential Quarter Forest Division | Anantnag | 5.00 |
| 961 | Department of Forest and Ecology | PCCF & HoFF | BOs Hut Daksum | Anantnag | 5.00 |
| 962 | Department of Forest and Ecology | PCCF & HoFF | BOs Hut Kund | Kulgam | 5.00 |
| 963 | Department of Forest and Ecology | PCCF & HoFF | Range office cum Division Office | Pulwama/ Awantipora | 5.00 |
| 964 | Department of Forest and Ecology | PCCF & HoFF | Range Office Pampore | Pulwama/ Pampore | 5.00 |
| 965 | Department of Forest and Ecology | PCCF & HoFF | CP Hurdmir | Pulwama/Tral | 5.00 |
| 966 | Department of Forest and Ecology | PCCF & HoFF | Forest Inspection Hut at Aharbal | Kulgam / D.H.Pora | 5.00 |
| 967 | Department of Forest and Ecology | PCCF & HoFF | Block Officers Hut Pulwama | Pulwama/ Pulwama | 5.00 |
| 968 | Department of Forest and Ecology | PCCF & HoFF | Block Officers Hut Sangerwani | Pulwama/ Rajpora | 5.00 |
| 969 | Department of Forest and Ecology | PCCF & HoFF | Block Officers Hut Zampathri | Shopian/ Keller | 5.00 |
| 970 | Department of Forest and Ecology | PCCF & HoFF | Block Officers Hut Sedow | Shopian/ Shopian | 5.00 |
| 971 | Department of Forest and Ecology | PCCF & HoFF | Block Officers Hut ladigasan | Shopian/ Shopian | 5.00 |
| 972 | Department of Forest and Ecology | PCCF & HoFF | Hostel Block-A | Jammu/R.S.Pura | 5.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|---|----------------|------------------|
| 973 | Department of Forest and Ecology | PCCF & HoFF | Store Room | Jammu/R.S.Pura | 5.00 |
| 974 | Department of Forest and Ecology | PCCF & HoFF | Oak Hostel | Bandipora | 5.00 |
| 975 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff/JMIC Court D.H. Pora | D.H. Pora | 5.00 |
| 976 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff/JMIC Court, Qazigund | Qazigund | 5.00 |
| 977 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Magam | Budgam | 5.00 |
| 978 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Chariesharief | Budgam | 5.00 |
| 979 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Bishnah | Jammu | 5.00 |
| 980 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Additional Special Mobile magistrate | Billawar | 5.00 |
| | | | | | 9005.37 |

Category-2 for (>= 20kW to < 100kW)

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|---|---------------|------------------|
| 1 | Department of Home | Department of Home | DPL Ganderbal | Tulmulla | 95.00 |
| 2 | Jal Shakti Department | PHE, Jammu | Jal Shakti Divisional Office Building, Reasi | Reasi/Reasi | 92.31 |
| 3 | Jal Shakti Department | PHE, Jammu | Jal Shakti Sub Division Mech. Office, Reasi | Reasi/Reasi | 92.31 |
| 4 | Jal Shakti Department | PHE, Jammu | Divisional Office Gandoh | Doda/Gandoh | 92.31 |
| 5 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Store | Baramulla | 90.00 |
| 6 | Department of Home | Department of Home | SHQ Sidhra three storey | Jammu | 83.00 |
| 7 | Department of Home | Department of Home | B-Block (24 Quarters) | Udhampur | 80.00 |
| 8 | Department of Home | Department of Home | Women Hostel (24 Rooms & 02 Hall) | Udhampur | 80.00 |
| 9 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Bhalwal | Jammu | 76.92 |
| 10 | Jal Shakti Department | PHE, Jammu | Executive Engineer PHE Mech. Division | Rajouri | 76.92 |
| 11 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Muthi Office Complex | Jammu | 76.92 |
| 12 | Jal Shakti Department | PHE, Jammu | Divisional Store Office | Ramban/Ramban | 76.92 |
| 13 | Jal Shakti Department | PHE, Kashmir | Water Works Srinagar | Srinagar | 76.92 |
| 14 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Dhounthly Office | Jammu | 76.08 |
| 15 | Department of Home | Department of Home | Accommodation Barrack- 04 Stories | Srinagar | 76.00 |
| 16 | Department of Home | Department of Home | DPL.Kupwara | Kupwara | 75.00 |
| 17 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Main Court High Court | Srinagar | 73.60 |
| 18 | Department of Home | Department of Home | SI WS Ground | Srinagar | 73.30 |
| 19 | Department of Home | Department of Home | SSP CID SB Kashmir Joint With CID Hqrs J&K | Srinagar | 72.00 |
| 20 | Department of Home | Department of Home | DPL Srinagar | Srinagar | 70.00 |
| 21 | Department of Home | Department of Home | Block E/F & ground floor Administrative B | Srinagar | 70.00 |
| 22 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Court Complex Pampore | Pampore | 70.00 |
| 23 | Jal Shakti Department | PHE, Jammu | PHE Sub-Division office Hiranagar | Hiranagar | 69.23 |
| 24 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Akhnoor | Jammu | 69.23 |
| 25 | Jal Shakti Department | PHE, Jammu | Sub Division Kandi-I Laboratory, Pouni Chak | Jammu/Marh | 69.23 |
| 26 | Jal Shakti Department | PHE, Jammu | Sub Division Bishnah Laboratory, near Tehsil | Jammu/Bishnah | 69.23 |
| 27 | Jal Shakti Department | PHE, Jammu | Divisional Store | Doda/Doda | 69.23 |

| 28 Jal Shakti Department PHE, Jammu Poonch Poonch Poonch 29 Jal Shakti Department PHE, Jammu Laboratory PHE Sub- Division office Billawar Billawar 30 Department of Home Department of Home Department of Home 05 No. ofDouble Storey Training Reasi 31 Department of Home Department of Home JICAnantnag Anantnag 32 Jal Shakti Department PHE, Jammu AEE quater Basohli Basohli 33 Jal Shakti Department PHE, Jammu Tawi filtration Plant Jammu 34 Jal Shakti Department PHE, Jammu Tawi filtration Plant Jammu 35 Department of Home Department of Home Department PD Sopor /Bomai 36 Department of Home Department of Home Department of Home PS Pan/alla PD Sopor/Rohama 37 Department of Home Department of Home Department of Home JKAP Sth Bn. Miran Jammu 40 Jal Shakti Department PHE, Jammu AEE PHE Sub Division Office Chenani Udhampur/Chenani <th>Capacity (kW)</th> <th>District</th> <th>Sub Department</th> <th>Department</th> <th>Ministry</th> <th>Sr. No.</th> | Capacity (kW) | District | Sub Department | Department | Ministry | Sr. No. |
|---|------------------|------------------|-------------------------|--|-----------------------|---------|
| 29 Jal Shakti Department PHE, Jammu Division office Billawar Billawar 30 Department of Home Department of Home 05 No. offDouble Storey Training Reasi 31 Department of Home Department of Home JICAnantnag Anantnag 32 Jal Shakti Department PHE, Jammu AEE quater Basohli Basohli 33 Jal Shakti Department PHE, Jammu AEE quater Basohli Basohli 34 Jal Shakti Department PHE, Jammu Tawi filtration Plant Jammu 35 Department of Home Department of Home ps Boma PD Sopor /Bomai 36 Department of Home Department of Home PS Pan/alla PD Sopor /Rohama 37 Department of Home Department of Home PC°Wadoora PD Sopor /Rohama 38 Department of Home Department of Home JKAP 13th Bn. Humhama Budgam 40 Jal Shakti Department PHE, Jammu AEE PHE Sub Division Office Chenani Udhampur/Chenani 41 Jal Shakti Department PHE, Jammu AEE PHE Sub Division Jammu 42 Jal Shakti Department PHE, Jammu Mataria Reasi/Katra 43 Jal Shakti Department PHE, Jammu Mataria Reasi/Kat | 69.23 | Poonch/Haveli | | PHE, Jammu | Jal Shakti Department | 28 |
| 30 Department of Home Department of Home Training Reasi 31 Department of Home Department of Home JICAnantnag Anantnag 32 Jal Shakti Department PHE, Jammu AEE quater Basohli Basohli 33 Jal Shakti Department PHE, Jammu PHE Sub-Division office Basohli Basohli 34 Jal Shakti Department PHE, Jammu Tawi filtration Plant Jammu 35 Department of Home Department of Home ps Boma PD Sopoc /Bomai 36 Department of Home Department of Home PS Pan/alla PD Sopor /Rohama 37 Department of Home Department of Home JKAP 13th Bn. Humhama Budgam 38 Department of Home Department of Home JKAP-8th Bn. Miran Jammu 40 Jal Shakti Department PHE, Jammu AEE PHE Sub Division Udhampur/Chenani 41 Jal Shakti Department PHE, Jammu Jal Shakti Department PHE, Jammu Katra 42 Jal Shakti Department PHE, Jammu Jal Shakti Department PHE, Jammu Katra 43 Jal Shakti Department </td <td>68.77</td> <td>Billawar</td> <td>2</td> <td>PHE, Jammu</td> <td>Jal Shakti Department</td> <td>29</td> | 68.77 | Billawar | 2 | PHE, Jammu | Jal Shakti Department | 29 |
| 32 Jal Shakti Department PHE, Jammu AEE quater Basohli Basohli 33 Jal Shakti Department PHE, Jammu PHE Sub-Division office Basohli Basohli 34 Jal Shakti Department PHE, Jammu Tawi filtration Plant Jammu 35 Department of Home Department of Home Department of PD Sopoc/Bomai 36 Department of Home Department of Home PS Pan/alla PD Sopor/Rohama 37 Department of Home Department of Home Department of Home PC°Wadoora PD Sopor/Bomai 38 Department of Home Department of Home Department Sahib Jammu 40 Jal Shakti Department PHE, Jammu JKAP 13th Bn. Humhama Budgam 41 Jal Shakti Department PHE, Jammu JKAP 13th Bn. Humhama Budgam 42 Jal Shakti Department PHE, Jammu Jal Shakti Department HE, Jammu Jal Shakti Dupartment HE, Jammu Jal Shakti Sub-Division Reasi/Katra 43 Jal Shakti Department PHE, Jammu Katra Katra Rajouri/Nowshera 44 Jal Shakti Department PHE, Jammu Su | 66.67 | Reasi | | Department of Home | Department of Home | 30 |
| 33 Jal Shakti Department PHE, Jammu PHE Sub-Division office Basohli Basohli 34 Jal Shakti Department PHE, Jammu Tawi filtration Plant Jammu 35 Department of Home Department of Home Department of Home PD Sopoc /Bomai 36 Department of Home Department of Home Department of Home PS Pan/alla PD Sopoc /Bomai 37 Department of Home Department of Home Department of Home PC*Wadoora PD Sopor /Rohama 38 Department of Home Department of Home Department of Home JKAP 13th Bn. Humhama Budgam 39 Department of Home Department of Home Department Het, Jammu AEE PHE Sub Division Udhampur/Chenani 40 Jal Shakti Department PHE, Jammu Jal Shakti Department PHE, Jammu Jal Shakti Department Udhampur 41 Jal Shakti Department PHE, Jammu Jal Shakti Department PHE, Jammu Katra Reasi/Katra 43 Jal Shakti Department PHE, Jammu Khour Khour Doda/Baderwah | 66.66 | Anantnag | JICAnantnag | Department of Home | Department of Home | 31 |
| 33Jal Shakti DepartmentPHE, Jammuoffice BasohliBasohli34Jal Shakti DepartmentPHE, JammuTawi filtration PlantJammu35Department of HomeDepartment of Homeps BomaPD Sopoc /Bomai36Department of HomeDepartment of HomePS Pan/allaPD Sopor /Rohama37Department of HomeDepartment of HomePC°WadooraPD Sopor /Rohama38Department of HomeDepartment of HomeJKAP 13th Bn. HumhamaBudgam39Department of HomeDepartment of HomeJKAP-8th Bn. Miran SahibJammu40Jal Shakti DepartmentPHE, JammuAEE PHE Sub Division Udhampur/Chenani Office ChenaniUdhampur/Chenani Hydraulic Circle Udhampur M.H ChowkUdhampur/Chenani Atara41Jal Shakti DepartmentPHE, JammuPHE Sub Division A KatraReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Division Building NowsheraRajouri/Nowshera44Jal Shakti DepartmentPHE, JammuSub Divisional Office Building NowsheraDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuRambanRamban/Ramban48Jal Shakti DepartmentPHE, JammuPHE Sub Division BhaderwahDoda/Bhaderwah49Department of HomeDepartment of HomePO's HarwanSrinagar50Department of HomeDepartment of HomeDO's Harwan <td>66.46</td> <td>Basohli</td> <td>AEE quater Basohli</td> <td>PHE, Jammu</td> <td>Jal Shakti Department</td> <td>32</td> | 66.46 | Basohli | AEE quater Basohli | PHE, Jammu | Jal Shakti Department | 32 |
| 34Jal Shakti DepartmentPHE, JammuTawi filtration PlantJammu35Department of HomeDepartment of Homeps BomaPD Sopoc/Bomai36Department of HomeDepartment of HomePS Pan/allaPD Sopor/Rohama37Department of HomeDepartment of HomePC°WadooraPD Sopor/Bomai38Department of HomeDepartment of HomeJKAP 13th Bn. HumhamaBudgam39Department of HomeDepartment of HomeJKAP 8th Bn. Miran SahibJammu40Jal Shakti DepartmentPHE, JammuAEE PHE Sub Division Office ChenaniUdhampur/Chenani Office Chenani41Jal Shakti DepartmentPHE, JammuJal Shakti Sub-Division Jal Shakti Sub-DivisionReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Divisional Lal Shakti DepartmentPHE, JammuReasi/Katra44Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah48Jal Shakti DepartmentPHE, JammuRambanRamban/Ramban49Department of HomeDepartment of HomePYC Doda EastSviangar50Department of HomeDepartment of HomeDPO Premses 09 BurlidingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional Office< | 64.62 | Basohli | | PHE, Jammu | Jal Shakti Department | 33 |
| 36Department of HomeDepartment of HomePS Pan/allaPD Sopor /Rohama37Department of HomeDepartment of HomePC°WadooraPD Sopore/Bomai38Department of HomeDepartment of HomeJKAP 13th Bn. HumhamaBudgam39Department of HomeDepartment of HomeJKAP 5th Bn. Miran SahibJammu40Jal Shakti DepartmentPHE, JammuAEE PHE Sub Division Office ChenaniUdhampur/Chenani41Jal Shakti DepartmentPHE, JammuHydraulic Circle Udhampur M.H ChowkUdhampur42Jal Shakti DepartmentPHE, JammuBHE Sub Division KatraReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Division KatraReasi/Katra44Jal Shakti DepartmentPHE, JammuPHE Sub Division Sub Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuRambanAmban/Ramban48Jal Shakti DepartmentPHE, JammuRambanRamban/Ramban49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDOP Premses 09 BurdingsAwantipora51Jal Shakti DepartmentIrrigation & FloodDivisional OfficeBardingra/Sumbal | 63.69 | Jammu | | PHE, Jammu | Jal Shakti Department | 34 |
| 37Department of HomeDepartment of HomePC°WadooraPD Sopore/Bomai38Department of HomeDepartment of HomeJKAP 13th Bn. HumhamaBudgam39Department of HomeDepartment of HomeJKAP-8th Bn. Miran SahibJammu40Jal Shakti DepartmentPHE, JammuAEE PHE Sub Division Office ChenaniUdhampur/Chenani Udhampur M.H Chowk41Jal Shakti DepartmentPHE, JammuHydraulic Circle Udhampur M.H ChowkUdhampur42Jal Shakti DepartmentPHE, JammuJal Shakti Sub-Division KatraReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Division KhourJammu44Jal Shakti DepartmentPHE, JammuPHE Sub Division Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division BhaderwahRamban/Ramban49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional Office BurldingsBandingara/Sumbal | 63.00 | PD Sopoc /Bomai | ps Boma | Department of Home | Department of Home | 35 |
| 38Department of HomeDepartment of HomeJKAP 13th Bn. HumhamaBudgam39Department of HomeDepartment of HomeJKAP-8th Bn. Miran SahibJammu40Jal Shakti DepartmentPHE, JammuAEE PHE Sub Division Office ChenaniUdhampur/Chenani41Jal Shakti DepartmentPHE, JammuHydraulic Circle Udhampur M.H ChowkUdhampur42Jal Shakti DepartmentPHE, JammuJal Shakti Sub-Division KatraReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Division KatraReasi/Katra44Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah48Jal Shakti DepartmentPHE, JammuPHE Sub Division BhaderwahDoda/Doda49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional Office BurldingsBardinora/Sumbal | 63.00 | PD Sopor /Rohama | PS Pan/alla | Department of Home | Department of Home | 36 |
| 39Department of HomeDepartment of HomeJKAP-8th Bn. Miran SahibJammu40Jal Shakti DepartmentPHE, JammuAEE PHE Sub Division Office ChenaniUdhampur/Chenani41Jal Shakti DepartmentPHE, JammuHydraulic Circle Udhampur M.H ChowkUdhampur42Jal Shakti DepartmentPHE, JammuJal Shakti Sub-Division KatraReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Division KatraReasi/Katra44Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division BhaderwahDoda/Doda49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Jal Shakti DepartmentIrrigation & FloodDivisional OfficeBandipora/Sumbal | 63.00 | PD Sopore/Bomai | PC°Wadoora | Department of Home | Department of Home | 37 |
| 39Department of HomeDepartment of HomeSahibJammu40Jal Shakti DepartmentPHE, JammuAEE PHE Sub Division Office ChenaniUdhampur/Chenani41Jal Shakti DepartmentPHE, JammuHydraulic Circle Udhampur M.H ChowkUdhampur42Jal Shakti DepartmentPHE, JammuJal Shakti Sub-Division KatraReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Division KatraReasi/Katra44Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division BhaderwahDoda/Doda49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Jal Shakti DepartmentIrrigation & FloodDivisional Office BurldingsBandinora/Sumbal | 63.00 | Budgam | JKAP 13th Bn. Humhama | Department of Home | Department of Home | 38 |
| 40Jal Shakti DepartmentPHE, JammuOffice ChenaniUdhampur/Chenani41Jal Shakti DepartmentPHE, JammuHydraulic Circle Udhampur M.H ChowkUdhampur42Jal Shakti DepartmentPHE, JammuJal Shakti Sub-Division KatraReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Division KhourJammu44Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division BhaderwahRamban/Ramban49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional Office BardingsBandingra/Sumbal | 63.00 | Jammu | | Department of Home | Department of Home | 39 |
| 41Jal Shakti DepartmentPHE, JammuHydraulic Circle Udhampur M.H ChowkUdhampur42Jal Shakti DepartmentPHE, JammuJal Shakti Sub-Division KatraReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Division KhourJammu44Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division BhaderwahDoda/Doda49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional Office Bandingra/Sumbal | 62.77 | Udhampur/Chenani | | PHE, Jammu | Jal Shakti Department | 40 |
| 42Jal Shakti DepartmentPHE, JammuJal Shakti Sub-Division KatraReasi/Katra43Jal Shakti DepartmentPHE, JammuPHE Sub Division KhourJammu44Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera46Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division RambanRamban/Ramban49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Lal Shakti DepartmentIrrigation & FloodDivisional Office Bandinora/Sumbal | 62.00 | Udhampur | Hydraulic Circle | PHE, Jammu | Jal Shakti Department | 41 |
| 43Jal Shakti DepartmentPHE, JammuPHE Sub Division KhourJammu44Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division RambanRamban/Ramban49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional Office BurldingsBandinora/Sumbal | 61.54 | Reasi/Katra | Jal Shakti Sub-Division | PHE, Jammu | Jal Shakti Department | 42 |
| 44Jal Shakti DepartmentPHE, JammuPHE Divisional Office Building NowsheraRajouri/Nowshera45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division PHE, JammuDoda/Doda49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional OfficeBandinora/Sumbal | 61.54 | Jammu | PHE Sub Division | PHE, Jammu | Jal Shakti Department | 43 |
| 45Jal Shakti DepartmentPHE, JammuSub Divisional Office BhaderwahDoda/Bhaderwah46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division RambanRamban/Ramban49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional OfficeBandinora/Sumbal | 61.54 | Rajouri/Nowshera | PHE Divisional Office | PHE, Jammu | Jal Shakti Department | 44 |
| 46Jal Shakti DepartmentPHE, JammuSub Divisional Lab. BhaderwahDoda/Bhaderwah47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division RambanRamban/Ramban49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional OfficeBandipora/Sumbal | 61.54 | Doda/Bhaderwah | Sub Divisional Office | PHE, Jammu | Jal Shakti Department | 45 |
| 47Jal Shakti DepartmentPHE, JammuSub Division Doda West/Doda EastDoda/Doda48Jal Shakti DepartmentPHE, JammuPHE Sub Division RambanRamban/Ramban49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Jal Shakti DepartmentIrrigation & FloodDivisional OfficeBandipora/Sumbal | 61.54 | Doda/Bhaderwah | Sub Divisional Lab. | PHE, Jammu | Jal Shakti Department | 46 |
| 48Jal Shakti DepartmentPHE, JammuPHE Sub Division RambanRamban/Ramban49Department of HomeDepartment of HomeP/S HarwanSrinagar50Department of HomeDepartment of HomeDPO Premses 09 BurldingsAwantipora51Ial Shakti DepartmentIrrigation & FloodDivisional OfficeBandipora/Sumbal | 61.54 | Doda/Doda | Sub Division Doda | PHE, Jammu | Jal Shakti Department | 47 |
| 49 Department of Home Department of Home P/S Harwan Srinagar 50 Department of Home Department of Home DPO Premses 09 Burldings Awantipora 51 Ial Shakti Department Irrigation & Flood Divisional Office Bandipora/Sumbal | 61.54 | Ramban/Ramban | PHE Sub Division | PHE, Jammu | Jal Shakti Department | 48 |
| 50 Department of Home Department of Home Burldings Awantipora 51 Jal Shakti Department Irrigation & Flood Divisional Office Bandipora/Sumbal | 60.00 | Srinagar | | Department of Home | Department of Home | 49 |
| 51 Jal Shakti Department Irrigation & Flood Divisional Office Bandinora/Sumbal | 60.00 | Awantipora | | Department of Home | Department of Home | 50 |
| Control Kashmir I Mechanical Irrigation 1 | 60.00 | Bandipora/Sumbal | | Irrigation & Flood Control, Kashmir | Jal Shakti Department | 51 |
| 52 Department of Home Fire and Emergency Service Department-UT F&ES Katra | 59.00 | | | Fire and Emergency | Department of Home | 52 |
| 53Jal Shakti DepartmentPHE, JammuPHE Sub Division BanihalRamban/Ramban | 57.69 | Ramban/Ramban | | | Jal Shakti Department | 53 |
| 54 Jal Shakti Department PHE, Jammu Superintending Udhampur | 56.92 | Udhampur | Superintending | PHE, Jammu | Jal Shakti Department | 54 |
| 55 Department of Home Department of Home Police Station Pulwama Pulwama | 55.00 | Pulwama | | Department of Home | Department of Home | 55 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|---|---|------------------------|------------------|
| 56 | Jal Shakti Department | PHE, Jammu | Je quater Basohli | Basohli | 53.85 |
| 57 | Jal Shakti Department | PHE, Jammu | PHE Sub-Division office Bani | Bani | 53.85 |
| 58 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Chowki Choura | Jammu | 53.85 |
| 59 | Jal Shakti Department | PHE, Jammu | Sub Division Kandi-II J.E Qtr. Nagrota | Jammu/Nagrota | 53.85 |
| 60 | Jal Shakti Department | PHE, Jammu | District Lab. Doda | Doda/Doda | 53.85 |
| 61 | Jal Shakti Department | PHE, Jammu | Chowkider Qtr./Hut Chaka Bhaderwah | Doda/Bhaderwah | 53.85 |
| 62 | Department of Home | Department of Home | DPO Pulwaam | Pulwama | 51.00 |
| 63 | Department of Home | Department of Home | PCR /DSP DAR Office Building | Pulwama | 50.00 |
| 64 | Department of Home | Department of Home | PS/Mess Building SHO Residence | Baramu!la Tangmarg | 50.00 |
| 65 | Department of Home | Department of Home | PP Warpura | PD Sopore/Dangerpora | 50.00 |
| 66 | Department of Home | Department of Home | MHC Building | Srinagar | 50.00 |
| 67 | Department of Forest and Ecology | Forest Protection Force | Chowkidar Hut Fatehpora | Ganderbal | 50.00 |
| 68 | Department of Forest and Ecology | PCCF & HoFF | Forest Hut Devigol | Kishtwar/Bunjwah | 50.00 |
| 69 | Department of Forest and Ecology | PCCF & HoFF | Range Office cum Residence (Khellani) | Doda/ Doda | 49.50 |
| 70 | Jal Shakti Department | PHE, Kashmir | WSMP Srinagar | Srinagar / Panthachowk | 48.60 |
| 71 | Jal Shakti Department | PHE, Jammu | Mahore Irrigation Division Dharmari | Reasi/Thuroo | 48.08 |
| 72 | Department of Home | Department of Home | JIC/SOG | Ganderbal | 48.00 |
| 73 | Department of Home | Fire and Emergency Service Department-UT | F&ES Rawalpora | | 48.00 |
| 74 | Jal Shakti Department | PHE, Kashmir | WSMP Srinagar | Budgam / Beerwah | 47.31 |
| 75 | Department of Forest and Ecology | JK Forest Development Corporation Ltd | Head Office at Srinagar Sheik-ul-Alam Complex | Srinagar/Srinagar | 47.00 |
| 76 | Jal Shakti Department | PHE, Jammu | Sub Division Office Vijaypur | Samba/Vijaypur | 46.92 |
| 77 | Jal Shakti Department | PHE, Jammu | Sub Division Office Bari Brahmana | Samba/Bari Brahmana | 46.92 |
| 78 | Jal Shakti Department | PHE, Jammu | Jal Shakti Sub Division Office Building, Reasi | Reasi/Reasi | 46.15 |
| 79 | Jal Shakti Department | PHE, Jammu | JE Residence Quarter/Store at Panthal | Reasi/Katra | 46.15 |
| 80 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Office Building | Rajouri/Nowshera | 46.15 |
| 81 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Office Building | Rajouri/Sunderbani | 46.15 |
| 82 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Office Building | Rajouri/Kalakote | 46.15 |
| 83 | Jal Shakti Department | PHE, Kashmir | Water Works Srinagar | Budgam | 46.15 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|---|---|------------------|
| 84 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisonal office building Irrigation & F | Baramulla/ Baramulla | 45.24 |
| 85 | Department of Home | Department of Home | PS Sopore | PD Sopore Sopure | 45.00 |
| 86 | Department of Home | Fire and Emergency Service Department-UT | F&ES Bari Barhmana | | 45.00 |
| 87 | Department of Forest and Ecology | PCCF & HoFF | BO Hut (Tatarsoo) | Ramban/ Ramban | 45.00 |
| 88 | Department of Forest and Ecology | PCCF & HoFF | BO Hut Trehgam | Kupwara | 45.00 |
| 89 | Department of Home | Department of Home | PS Vilgam | Main Building | 44.44 |
| 90 | Department of Home | Department of Home | PP Chogal | Main Office Building Kupwara/ Handwara | 44.44 |
| 91 | Department of Forest and Ecology | PCCF & HoFF | DCF/ACF Building (Batote) | Ramban/ Batote | 44.00 |
| 92 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut (Gandhri) | Ramban/ Ramban | 43.86 |
| 93 | Department of Home | Fire and Emergency Service Department-UT | F&ES Udhampur Near Police station | | 43.75 |
| 94 | Department of Home | Department of Home | PP Langatt | Main Buildimg Kupwara/ Handwar | 43.44 |
| 95 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Complex Poonch | Poonch | 43.00 |
| 96 | Department of Forest and Ecology | PCCF & HoFF | Rest House (Bhatni) | Ramban/ Ramban | 42.30 |
| 97 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Moori Tehsil Katra | Reasi/Katra | 42.00 |
| 98 | Department of Forest and Ecology | PCCF & HoFF | Resin Hut (Kanga) | Ramban/ Ramban | 42.00 |
| 99 | Jal Shakti Department | PHE, Jammu | District Lab Samba | Samba/Samba | 41.54 |
| 100 | Jal Shakti Department | PHE, Jammu | Sub Division Lab Vijaypur | Samba/Vijaypur | 41.54 |
| 101 | Department of Home | Department of Home | PS ShaheedGanj | Srinagar | 40.00 |
| 102 | Department of Home | Department of Home | P/S Pahalgam | Anantnag | 40.00 |
| 103 | Department of Home | Department of Home | Block B-I&B? | Kulgam | 40.00 |
| 104 | Department of Home | Department of Home | IKAP Ith Security Bn (03) storey main | Budgam | 40.00 |
| 105 | Department of Home | Department of Home | D-Block (24 Quarters) | Udhampur | 40.00 |
| 106 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional building | Baram ulla | 40.00 |
| 107 | Department of Forest and Ecology | JK Forest Development Corporation Ltd | J&K FDCL, Lumber House, Galdni Narwal | Jammu/Bahu | 40.00 |
| 108 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter Passgadh | Kishtwar/ Nagseni | 40.00 |
| 109 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Complex Bhaderwah | Doda | 40.00 |
| 110 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Complex Doda | Doda | 40.00 |
| 111 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Complex Samba | Samba | 40.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|---|------------------------|------------------|
| 112 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff JMIC Bani | Bani | 40.00 |
| 113 | Jal Shakti Department | PHE, Kashmir | WSMP Srinagar | Srinagar / Panthachowk | 39.22 |
| 114 | Department of Forest and Ecology | PCCF & HoFF | BO Hut (Kothi) | Ramban/ Ramban | 38.64 |
| 115 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Dharmari | Reasi/Thuroo | 38.46 |
| 116 | Jal Shakti Department | PHE, Kashmir | WSMP Srinagar | Srinagar / Rainawari | 38.46 |
| 117 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff JMIC Basaholi | Basaholi | 37.00 |
| 118 | Department of Home | Fire and Emergency Service Department-UT | F&ES Hajjin | | 36.00 |
| 119 | Department of Forest and Ecology | PCCF & HoFF | F.R.H. Tote | Reasi/Bhomag | 36.00 |
| 120 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post BMCP (Batote) | Ramban/ Batote | 36.00 |
| 121 | Department of Forest and Ecology | PCCF & HoFF | B.O Quarter singpore at Singpore | Kishtwar/Chatroo | 35.60 |
| 122 | Department of Home | Department of Home | PTWS PCR Complex Srinagar. Adm Block | Srinagar | 35.00 |
| 123 | Department of Home | Department of Home | PTWS PCR Complex Srinagar | Srinagar | 35.00 |
| 124 | Department of Home | Department of Home | Adm block IRP 2nd Battalion fatehpur | Rajouri | 35.00 |
| 125 | Department of Home | Department of Home | GO'S Residence Block at Bn. Hqrs. | Rajouri | 35.00 |
| 126 | Department of Home | Department of Home | Jawan Barrack at Bn. Hqrs | Rajouri | 35.00 |
| 127 | Department of Home | Department of Home | Mess Dinning Hall Bn. Hqrs | Rajouri | 35.00 |
| 128 | Department of Home | Department of Home | Unit stores Block at Bn. Hqrs. | Rajouri | 35.00 |
| 129 | Department of Home | Fire and Emergency Service Department-UT | F&ES Safakadal | | 35.00 |
| 130 | Department of Home | Fire and Emergency Service Department-UT | F&ES Ranawari | | 35.00 |
| 131 | Department of Home | Fire and Emergency Service Department-UT | F&ES Soura | | 35.00 |
| 132 | Department of Home | Sainik Welfare Department,UT of J&K | ZSWO Poonch | | 35.00 |
| 133 | Department of Forest and Ecology | PCCF & HoFF | Range Forest Office, Bani | Kathua/Bani | 35.00 |
| 134 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post Bani | Kathua/Bani | 35.00 |
| 135 | Department of Forest and Ecology | PCCF & HoFF | Divisional Office Building Ramnagar | Udhampur / Ramnagar | 35.00 |
| 136 | Department of Forest and Ecology | PCCF & HoFF | Old DFO Residential quarter | Udhampur / Ramnagar | 35.00 |
| 137 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Pr. District& Sessions Court Ramban | Ramban | 35.00 |
| 138 | Jal Shakti Department | PHE, Jammu | Filtration Plant Ramban | Ramban/Ramban | 34.62 |
| 139 | Jal Shakti Department | PHE, Kashmir | PHE Division Sopore | Baramulla/Dangerpora | 34.62 |
| | | | _ | | |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|---|-------------------|------------------|
| 140 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | 3rd Stage pump House/ shikargah Tral | Pulwama/Tral | 34.12 |
| 141 | Department of Forest and Ecology | PCCF & HoFF | Chowkidar Hut (Patnitop) | Ramban/ Batote | 34.00 |
| 142 | Department of Home | Department of Home | ADM Block JKAP-6th Bn. | Medhar/Ponch | 33.33 |
| 143 | Department of Home | Department of Home | CO residential block | Medhar/Ponch | 33.33 |
| 144 | Department of Home | Department of Home | Dy. CO Residential block | Medhar/Ponch | 33.33 |
| 145 | Department of Home | Department of Home | GO's Block | Medhar/Ponch | 33.33 |
| 146 | Department of Home | Department of Home | Jawan Barrack | Medhar/Ponch | 33.33 |
| 147 | Department of Home | Department of Home | Ration Store recreation hall | Medhar/Ponch | 33.33 |
| 148 | Department of Home | Department of Home | Store Building | Medhar/Ponch | 33.33 |
| 149 | Department of Home | Department of Home | Quarter Guard | Medhar/Ponch | 33.33 |
| 150 | Department of Home | Department of Home | Mess/Dinning Hall | Medhar/Ponch | 33.33 |
| 151 | Jal Shakti Department | PHE, Kashmir | PHE Mechanical Division Srinagar | Budgam/ Bkpora | 33.08 |
| 152 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Bani | Kathua/Bani | 33.00 |
| 153 | Department of Forest and Ecology | PCCF & HoFF | FRH Jai | Doda/ Bhadarwah | 31.95 |
| 154 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut (Bludnpur) | Doda/ Assar | 31.67 |
| 155 | Department of Forest and Ecology | PCCF & HoFF | Front Line Staff building Bhadarwah | Doda/ Bhadarwah | 31.00 |
| 156 | Jal Shakti Department | PHE, Jammu | Jal Shakti (PHE) M Division Kathua | Kathua | 30.77 |
| 157 | Jal Shakti Department | PHE, Jammu | Chowkidar Hut Ramban | Ramban/Ramban | 30.77 |
| 158 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff JMIC Billawar | Billawar | 30.60 |
| 159 | Department of Home | Department of Home | PS Nowhatta/Residential Quarter of SHO | Srinagar | 30.00 |
| 160 | Department of Home | Department of Home | Khanyar | Srinagar | 30.00 |
| 161 | Department of Home | Department of Home | DPO Anantnag | Anantnag | 30.00 |
| 162 | Department of Home | Department of Home | P/S Anantnag | Anantnag | 30.00 |
| 163 | Department of Home | Department of Home | PS Baramulla | Baramulla | 30.00 |
| 164 | Department of Home | Department of Home | CID Hqrs shared building ZPHQ Complex | district | 30.00 |
| 165 | Department of Home | Department of Home | PS City Jammu | Jammu north | 30.00 |
| 166 | Department of Home | Department of Home | Director IIG Kashmir | Srinagar | 30.00 |
| 167 | Department of Home | Department of Home | Zonal Telecom Vlqr Kashmir. | Srinagar | 30.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|---|---------------------------|------------------|
| 168 | Department of Home | Department of Home | Old NGO Block | Kathua | 30.00 |
| 169 | Department of Home | Department of Home | Director Residence (Double Storey) | Udhampur | 30.00 |
| 170 | Department of Home | Department of Home | Shah Block (06 Barracks) | Udhampur | 30.00 |
| 171 | Department of Home | Department of Home | Mani Ram Block (06 B arracks) | Udhampur | 30.00 |
| 172 | Department of Home | Department of Home | Main Block (06 Barracks) | Udhampur | 30.00 |
| 173 | Department of Home | Department of Home | Police Departmental Canteen Store | Srinagar | 30.00 |
| 174 | Department of Home | Fire and Emergency Service Department-UT | F&ES Ramnagar | | 30.00 |
| 175 | Department of Home | Fire and Emergency Service Department-UT | F&ES Ramban/command | | 30.00 |
| 176 | Department of Home | Fire and Emergency Service Department-UT | F&ES hqrs Anantnag | | 30.00 |
| 177 | Department of Home | Fire and Emergency Service Department-UT | F&ES Pahalgam | | 30.00 |
| 178 | Department of Forest and Ecology | Forest Protection Force | Assistant Director's Residence at Sindho | Shopian Shopian | 30.00 |
| 179 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House, Akhnoor | Jammu/Akhnoor | 30.00 |
| 180 | Department of Forest and Ecology | PCCF & HoFF | Range Forest Office Lamberi and R.O | Rajouri/Nowshera | 30.00 |
| 181 | Department of Forest and Ecology | PCCF & HoFF | Inspection Hut Ramnagar | Udhampur / Ramnagar | 30.00 |
| 182 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut Pinger | Udhampur / Ramnagar | 30.00 |
| 183 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter Mandi | Poonch/Haveli | 30.00 |
| 184 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter Salwah | Poonch/ Mendher | 30.00 |
| 185 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut sallar | Anantnag | 30.00 |
| 186 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Budgam | Budgam | 30.00 |
| 187 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Sub-Judge Court Ramnagar | Udhampur | 30.00 |
| 188 | Department of Forest and Ecology | PCCF & HoFF | Forest Checkpost Synthan at Synthan | Kishtwar/Chatroo | 29.88 |
| 189 | Department of Home | Fire and Emergency Service Department-UT | F&ES Bandipora | | 29.40 |
| 190 | Department of Forest and Ecology | PCCF & HoFF | Forest Checkpost Dadpeth at Dadpeth | Kishtwar/Mughalmaida n | 29.00 |
| 191 | Department of Home | Fire and Emergency Service Department-UT | F&ES Naidkhai | | 28.08 |
| 192 | Department of Forest and Ecology | Forest Protection Force | Messes Kitchen Lounge Fatehpora | Ganderbal | 28.07 |
| 193 | Department of Home | Fire and Emergency Service Department-UT | F&ES Babademb | | 28.00 |
| 194 | Department of Forest and Ecology | PCCF & HoFF | Clerical quarters 7 Nos. (Batote) | Ramban/ Batote | 28.00 |
| 195 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Pastuna | Pulwama/ Tral | 28.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|--|-------------------------|------------------|
| 196 | Department of Forest and Ecology | rest PCCF & HoFF Forest Rest House Kishtwar/Paddar | | Kishtwar/Paddar | 27.50 |
| 197 | Department of Forest and Ecology | PCCF & HoFF | Range Office Neeru | Doda/ Bhadarwah | 26.97 |
| 198 | Jal Shakti Department | PHE, Kashmir | Chief Engineer's Office PHE Kashmir | Srinagar/Srinagar North | 26.92 |
| 199 | Department of Home | Department of Home | Family Quarter Block -A (DPL) | Pulwama | 26.00 |
| 200 | Department of Forest and Ecology | Forest Protection Force | Double Storeyed Guard Barrack | Anantnag | 26.00 |
| 201 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House Katra | Reasi/Katra | 26.00 |
| 202 | Department of Forest and Ecology | PCCF & HoFF | R.Office cum Range Qtr. Katra | Reasi/Katra | 26.00 |
| 203 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court | Jammu | 26.00 |
| 204 | Department of Home | Sainik Welfare Department,UT of J&K | ZSWO Udhampur | | 25.80 |
| 205 | Department of Forest and Ecology | Forest Protection Force | Deputy Director's Office | Anantnag | 25.70 |
| 206 | Department of Home | Department of Home | P/S Kothibagh | Srinagar | 25.00 |
| 207 | Department of Home | Department of Home | P/S Rainawari | Srinagar | 25.00 |
| 208 | Department of Home | Department of Home | DTHQ Poonch | Haveli/poonch | 25.00 |
| 209 | Department of Home | Department of Home | "SDRF 2 Bn Gulshan Grund Jammu" | jammu | 25.00 |
| 210 | Department of Home | Fire and Emergency Service Department-UT | F&ES Ramgarh | | 25.00 |
| 211 | Department of Forest and Ecology | JK Forest Development Corporation Ltd | Divisional Office, Extraction Division, | Baramulla/Baramlla | 25.00 |
| 212 | Department of Forest and Ecology | PCCF & HoFF | Mali Hut Thandapani Nursery | Udhampur / Ramnagar | 25.00 |
| 213 | Department of Forest and Ecology | PCCF & HoFF | Community Center Handwara | Kupwara/ Handwara | 25.00 |
| 214 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Pahalgam | Pahalgamm | 25.00 |
| 215 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Pr. District Court Baramulla | Baramulla | 25.00 |
| 216 | Department of Home | Fire and Emergency Service Department-UT | F&ES Batote | | 24.00 |
| 217 | Department of Home | Fire and Emergency Service Department-UT | F&ES Family Qtr | | 24.00 |
| 218 | Department of Forest and Ecology | Wildlife Protection | Regional WLW office Manda | Jammu/Jammu | 24.00 |
| 219 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut at Chapprian | Udhampur / Ramnagar | 24.00 |
| 220 | Department of Forest and Ecology | PCCF & HoFF | Forest Check Post (Patnitop) | Ramban/ Batote | 24.00 |
| 221 | Department of Forest and Ecology | PCCF & HoFF | Guest House at chiagund Verinag | Anantnag | 24.00 |
| 222 | Department of Forest and Ecology | PCCF & HoFF | Range Office Kulgam | Kulgam / Kulgam | 24.00 |
| 223 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Pr. District & Sessions Court Kupwara | Kupwara | 24.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|---|---|---------------------|------------------|
| 224 | Department of Forest and Ecology | PCCF & HoFF | Chowkidar Hut (Khellani) | Doda/ Doda | 23.91 |
| 225 | Department of Forest and Ecology | PCCF & HoFF | Check Post Bumzooa | Anantnag | 23.50 |
| 226 | Jal Shakti Department | PHE, Jammu | PHE Sub- Division- Bani Rest House | Bani | 23.08 |
| 227 | Jal Shakti Department | PHE, Jammu | Sub Division Kandi-III office Sidhra | Jammu/Jammu | 23.08 |
| 228 | Jal Shakti Department | PHE, Jammu | Sub Division R.S. Pura, AEE Office R.S. Pura | Jammu/R.S Pura | 23.08 |
| 229 | Jal Shakti Department | PHE, Jammu | J.E PHE Office R.S. Pura | Jammu/R.S Pura | 23.08 |
| 230 | Jal Shakti Department | PHE, Jammu | Sub division office Mendhar | Poonch/Mendhar | 23.08 |
| 231 | Department of Home | HeadQuarter HomeGuard,SDRF,Civi | | | 23.00 |
| 232 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Lift Irrigation Scheme Marvel Stage 1st | Pulwama/Pampore | 22.40 |
| 233 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Marval Stage 1st Marval | Pulwama/kakapora | 22.40 |
| 234 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Engineering Block | Srinagar | 22.40 |
| 235 | Department of Forest and Ecology | Forest Protection Force | Reporting Centre Keller | Keller Shopian | 22.30 |
| 236 | Department of Home | Fire and Emergency Service Department-UT | F&ES Qazigund | | 22.00 |
| 237 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional office building at Narbal | Budgam /Narbal | 22.00 |
| 238 | Department of Forest and Ecology | PCCF & HoFF | IWDP Building currently used for DFO | Udhampur / Ramnagar | 22.00 |
| 239 | Department of Forest and Ecology | PCCF & HoFF | Old Soil Conservation Hut Batote | Ramban/ Batote | 22.00 |
| 240 | Department of Home | Sainik Welfare Department,UT of J&K | ZSWO Kupwara | | 21.76 |
| 241 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Advocate General (Old Building) | Srinagar | 21.56 |
| 242 | Department of Forest and Ecology | Forest Protection Force | Reporting Centre Manloo | Shopian Shopian | 21.55 |
| 243 | Department of Forest and Ecology | Forest Protection Force | Guard Barrack at Sindho Shirmal | Shopian Shopian | 21.33 |
| 244 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Lift Irrigation Scheme Marvel | Budgam/B.K.Pora | 21.30 |
| 245 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Stage 2nd Suthsoo | Budgam/B.K.Pora | 21.30 |
| 246 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Superintending Engineer Mechanical | Srinagar | 21.00 |
| 247 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Workshop | Baramulla | 21.00 |
| 248 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Bandipora | Bandipora | 21.00 |
| 249 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Tangmarg | Baramulla | 21.00 |
| 250 | Department of Home | Fire and Emergency Service Department-UT | F&ES Rajouri | | 20.52 |
| 1 | | | | | |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--------------------|--|--|--------------------|------------------|
| 252 | Department of Home | Department of Home | Main DPL Anantnag | Anantnag | 20.00 |
| 253 | Department of Home | Iome Department of Home Fammily Quarter 09 Nummber Anantnag | | Anantnag | 20.00 |
| 254 | Department of Home | Department of Home | P/S Srigufwara | Anantnag | 20.00 |
| 255 | Department of Home | Department of Home | O4 Storied SDRF/Aremd Building | Pulwama | 20.00 |
| 256 | Department of Home | Department of Home | GO'S Block | Kulgam | 20.00 |
| 257 | Department of Home | Department of Home | Block-C DPI Kulgam | Kulgam | 20.00 |
| 258 | Department of Home | Department of Home | Block-F DPI Kulgam | Kulgam | 20.00 |
| 259 | Department of Home | Department of Home | Bloek」 DPI. Kulgam | Kulgam | 20.00 |
| 260 | Department of Home | Department of Home | Block-K DPL. Kulgam | Kulgam | 20.00 |
| 261 | Department of Home | Department of Home | P/S Kulgam | Kulgam | 20.00 |
| 262 | Department of Home | Department of Home | P/SQaimch | Kulgam | 20.00 |
| 263 | Department of Home | Department of Home | P/S DH Pora | Kulgam | 20.00 |
| 264 | Department of Home | Department of Home | P/S Qarigund | Kulgam | 20.00 |
| 265 | Department of Home | Department of Home | P/S Devsar | Kulgam | 20.00 |
| 266 | Department of Home | Department of Home | Altemate Accommodation Block"A"SOG | Shopian | 20.00 |
| 267 | Department of Home | Department of Home | Alternate Accommodation Block"B"Mess | Shopian | 20.00 |
| 268 | Department of Home | Department of Home | Aiternalive Accummodation Block | Shopian | 20.00 |
| 269 | Department of Home | Department of Home | PS Awantpora(05 Builuings) | Awantipora | 20.00 |
| 270 | Department of Home | Department of Home | PP Watergam | PD Sopure Watergam | 20.00 |
| 271 | Department of Home | Department of Home | S.SG Lines Srinagar | Srinagar | 20.00 |
| 272 | Department of Home | Department of Home | Indoor Block PTTI V ijaypur | Samba | 20.00 |
| 273 | Department of Home | Department of Home | Security Cetral Pool (wet Canteen building) | Jammu | 20.00 |
| 274 | Department of Home | Department of Home | STWS J&K Sidhra Double Stortl. | Jammu | 20.00 |
| 275 | Department of Home | Department of Home | Indoor Block | Kathua | 20.00 |
| 276 | Department of Home | Department of Home | Shiban Lai Block (02 Barracks) | Udhampur | 20.00 |
| 277 | Department of Home | Department of Home | APHQ/APHQ Kashmir. Balgarden | Srinagar | 20.00 |
| 278 | Department of Home | Department of Home | Adminstrative block-1 jawan Barrack block-2 | Anantnag/Mattan | 20.00 |
| 279 | Department of Home | Department of Home | E Block Barack/5 Story building. | Anantnag/Mattan | 20.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-------------------------------------|---|--|---------------------|------------------|
| 280 | Department of Home | Department of Home | Bn office double story building. | Anantnag/Mattan | 20.00 |
| 281 | Department of Home | Department of Home | "Gulshan Ground jammu" | jammu | 20.00 |
| 282 | Department of Home | Fire and Emergency Service Department-UT | F&ES Samba | | 20.00 |
| 283 | Department of Home | Fire and Emergency Service Department-UT | F&ES Bijbehara | | 20.00 |
| 284 | Department of Home | Fire and Emergency Service Department-UT | F&ES Sallar | | 20.00 |
| 285 | Department of Home | Fire and Emergency Service Department-UT | F&ES Kulgam | | 20.00 |
| 286 | Department of Home | Fire and Emergency Service Department-UT | F&ES Behibagh | | 20.00 |
| 287 | Department of Home | Fire and Emergency Service Department-UT | F&ES Manzgam | | 20.00 |
| 288 | Department of Home | HeadQuarter HomeGuard,SDRF,Civi | Civil Defence Srinagar | | 20.00 |
| 289 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | Divisional Building Batpora Shopian | Shopian/Shopian | 20.00 |
| 290 | Jal Shakti Department | Irrigation & Flood Control, Kashmir | SSD Tangdar Irrigation Residential Quarter at | Kupwara/Karnah | 20.00 |
| 291 | Department of Forest and Ecology | Forest Protection Force | Gaarage Fatehpora | Ganderbal | 20.00 |
| 292 | Department of Forest and Ecology | Wildlife Protection | Range Officer Building Tangmarg | Tangmarg | 20.00 |
| 293 | Department of Forest and Ecology | Wildlife Protection | Staff Quarter Gulmarg | Baramulla | 20.00 |
| 294 | Department of Forest and Ecology | Wildlife Protection | Panyer Staff Quarter | Pulwama/Tral | 20.00 |
| 295 | Department of Forest and Ecology | JK Forest Development Corporation Ltd | Divisional Office, Extraction Division, | Ramban/Ramban | 20.00 |
| 296 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Amroh | Udhampur / Ramnagar | 20.00 |
| 297 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Chirk (Dharmsal) | Samba/Bari Brahmana | 20.00 |
| 298 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut, Gharrian | Chenani | 20.00 |
| 299 | Department of Forest and Ecology | PCCF & HoFF | Forest Inspection Hut, Latti | Latti-Marothi | 20.00 |
| 300 | Department of Forest and Ecology | PCCF & HoFF | Forest Guard Hut, Sira morh | Latti-Marothi | 20.00 |
| 301 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House, Dudu Seri | Latti-Marothi | 20.00 |
| 302 | Department of Forest and Ecology | PCCF & HoFF | Forest Rest House, Chari (Totally damaged) | Latti-Marothi | 20.00 |
| 303 | Department of Forest and Ecology | PCCF & HoFF | Forest Guard Hut, Dudu Seri | Latti-Marothi | 20.00 |
| 304 | Department of Forest and Ecology | PCCF & HoFF | Labour Shed, Dudu Seri | Latti-Marothi | 20.00 |
| 305 | Department of Forest and Ecology | PCCF & HoFF | Range Office | Udhampur | 20.00 |
| 306 | Department of Forest and Ecology | PCCF & HoFF | TSD Udhampur at present with JKFDC | Udhampur | 20.00 |
| 307 | Department of Forest and Ecology | PCCF & HoFF | Old Soil Office | Udhampur | 20.00 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|--|-------------------|------------------|
| 308 | Department of Forest and Ecology | PCCF & HoFF | Guard Hut Gangera | Udhampur | 20.00 |
| 309 | Department of Forest and Ecology | PCCF & HoFF | Van Bhawan Gumat | Jammu | 20.00 |
| 310 | Department of Forest and Ecology | PCCF & HoFF | CF East Circle | Jammu | 20.00 |
| 311 | Department of Forest and Ecology | PCCF & HoFF | CCF Jammu office | Jammu | 20.00 |
| 312 | Department of Forest and Ecology | PCCF & HoFF | DFO Residential Quarter Poonch | Poonch/Haveli | 20.00 |
| 313 | Department of Forest and Ecology | PCCF & HoFF | Range Residential Quarter (ACF) Haveli | Poonch/Haveli | 20.00 |
| 314 | Department of Forest and Ecology | PCCF & HoFF | DFO Office (Complex) | Poonch/Haveli | 20.00 |
| 315 | Department of Forest and Ecology | PCCF & HoFF | Forester Quarter Mendher | Poonch/ Mendher | 20.00 |
| 316 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter Nar Mendher | Poonch/ Mendher | 20.00 |
| 317 | Department of Forest and Ecology | PCCF & HoFF | BO Quarter Samote | Poonch/Surankote | 20.00 |
| 318 | Department of Forest and Ecology | PCCF & HoFF | DFO REsidence, Rajouri | Rajouri | 20.00 |
| 319 | Department of Forest and Ecology | PCCF & HoFF | B.O Hut Bathuni | Rajouri | 20.00 |
| 320 | Department of Forest and Ecology | PCCF & HoFF | Range Office Cum Residence Siraj at Sah | Doda/Doda | 20.00 |
| 321 | Department of Forest and Ecology | PCCF & HoFF | Dining Hall at EAC Handwara | Kupwara/ Handwara | 20.00 |
| 322 | Department of Forest and Ecology | PCCF & HoFF | B O Quarter Pahalgam | Anantnag | 20.00 |
| 323 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Pr. District Sessions Court Complex | Ganderbal | 20.00 |
| 324 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Complex Shopian | Shopian | 20.00 |
| 325 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | ADI Court Sopore | Baramulla | 20.00 |
| 326 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Sub Budge Court Pattan | Baramulla | 20.00 |
| 327 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Munsiff Court Boniyar | Baramulla | 20.00 |
| | | | | | 11965.31 |

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|--|--|---|---|---------------|
| 1 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Complex Udhampur | Udhampur | 1506.94 |
| 2 | Department of Home | Department of Home | DPL Awt09 Buildimgs | Awantipora | 1000.00 |
| 3 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | District Court Complex Srinagar | Srinagar | 620.00 |
| 4 | Department of Home | Department of Home | Crime Hqrs Channi Ramma | Jammu | 400.00 |
| 5 | Jal Shakti Department | PHE, Jammu | PHE Rural Division BC Road Jammu | Jammu/Jammu | 346.15 |
| 6 | Department of Home | Department of Home | DPL Handwara | Adminisatiye Rloch Kupwara/ Handwart | 333.33 |
| 7 | Dept of Law and Parliamentary Affairs | Dept of Law and Parliamentary Affairs | Main Court Complex Pulwama Prichoo Pulwama | Pulwama | 332.16 |
| 8 | Department of Home | Department of Home | Admin. Block.GO resedency quarter Guard | Kupwara | 320.00 |
| 9 | Department of Home | Department of Home | Police Colony Qamerwari Srinagar | Srinagar | 300.00 |
| 10 | Department of Home | Department of Home | CID SB Hgrs Jammu | Jammu | 300.00 |
| 11 | Department of Home | Department of Home | JKAP.14th Bn. Manasbal | Ganderbal | 255.00 |
| 12 | Department of Home | Department of Home | 05 Storey Hostel block Administrative block, | Ganderbal | 255.00 |
| 13 | Department of Home | Department of Home | Security Kashmir Srinagar | Srinagar | 250.00 |
| 14 | Department of Home | Department of Home | APHQ Complex | Jammu | 250.00 |
| 15 | Jal Shakti Department | PHE, Jammu | PHE Division Office Akhnoor | Jammu | 230.77 |
| 16 | Jal Shakti Department | PHE, Jammu | Guest House, Surinsar | Jammu/Nagrota | 230.77 |
| 17 | Department of Home | Department of Home | Indoor Block (Double Block) | Udhampur | 230.00 |
| 18 | Department of Home | Department of Home | PS Kishtwar | Kishtwar | 224.00 |
| 19 | Department of Home | Department of Home | DPL.Bandipora | Bandipora Bandipora | 210.00 |
| 20 | Jal Shakti Department | PHE, Kashmir | WSMP Srinagar | Ganderbal/ Ganderbal | 207.55 |
| 21 | Jal Shakti Department | PHE, Jammu | Rest House Mandi | Ramkot | 207.23 |
| 22 | Department of Home | Department of Home | Administrative Block (Double Storey) | Udhampur | 200.00 |
| 23 | Jal Shakti Department | PHE, Jammu | PHE Division Office Kathua near SP Office Kathua | Kathua | 192.31 |
| 24 | Department of Home | HeadQuarter HomeGuard,SDRF,Civil & | SDRF 2nd Bn Gulshan Ground Jammu | | 187.50 |
| 25 | Department of Home | Department of Home | Police Hospital DPL Jammu | Jammu | 180.00 |
| 26 | Jal Shakti Department | PHE, Jammu | Division Office Building | Samba/Samba | 172.31 |
| 27 | Jal Shakti Department | PHE, Jammu | PHE Divisional Store Rehni Udhampur | Udhampur | 167.38 |
| 28 | Department of Home | Department of Home | Trainees Building mock-I" STC sheere | Baramulla | 160.00 |
| 29 | Department of Home | Department of Home | Block-1 (U-shaped) | jammu | 160.00 |
| 30 | Jal Shakti Department | PHE, Jammu | Barrian Station Udhampur | Udhampur | 153.85 |
| 31 | Jal Shakti Department | PHE, Jammu | Jakhani Station Udhampur | Udhampur | 153.85 |
| 32 | Department of Home | Department of Home | Shergarhi | Srinagar | 150.00 |
| 33 | Department of Home | Department of Home | C)asis Mess (12 Room & 02 VIP suits) | Udhampur | 150.00 |
| 34 | Department of Home | Department of Home | U.SQ 4th, 5th, 6th & 7th (Triple Storey) | Udhampur | 150.00 |
| 35 | Jal Shakti Department | PHE, Jammu | AEE Sub Division Office Ramnagar | Udhampur/Ramnagar | 146.46 |
| 36 | Jal Shakti Department | PHE, Jammu | Office Complex of SE Hydraulic Circle Rajouri | Rajouri | 146.15 |
| 37 | Department of Home | Fire and Emergency Service Department-UT of J&K | F&ES Hqr Srinagar | | 140.00 |
| 38 | Jal Shakti Department | PHE, Jammu | Divisional Office near BSNL exchange office Doda | Doda/Doda | 138.46 |

Category-3 for (>= 100kW and more)

| Sr. No. | Ministry | Department | Sub Department | District | Capacity (kW) |
|---------|-----------------------|--|--|--------------------|---------------|
| 39 | Jal Shakti Department | PHE, Jammu | PHE Sub-Division office Billawar | Billawar | 135.85 |
| 40 | Jal Shakti Department | PHE, Kashmir | WSMP Srinagar | Budgam / Beerwah | 132.92 |
| 41 | Jal Shakti Department | PHE, Jammu | Residentional Quarter of Executive Engineer Kathua | Kathua | 130.77 |
| 42 | Jal Shakti Department | PHE, Jammu | PHE Sub Division Office Gool | Ramban/Gool | 130.77 |
| 43 | Department of Home | Department of Home | ZPHQ-K Complex | Srinagar | 130.00 |
| 44 | Jal Shakti Department | PHE, Jammu | Executive Engineer PHE Udhampu, AEE Udhampur & | Udhampur | 125.54 |
| 45 | Jal Shakti Department | PHE, Jammu | Sub Division Office Samba | Samba/Samba | 119.08 |
| 46 | Department of Home | Department of Home | Diginabal | Ganderbal | 119.00 |
| 47 | Jal Shakti Department | PHE, Jammu | PHE Sub-Division rest house Basohli | Basohli | 117.31 |
| 48 | Department of Home | Department of Home | 04 Storied Adm, Block Building ground flour AROK 1st Floor. | Srinagar | 116.00 |
| 49 | Jal Shakti Department | PHE, Jammu | WSS Dhansar (Station) | Reasi/Katra | 115.38 |
| 50 | Jal Shakti Department | PHE, Jammu | Sub Division Bishnah, near tehsil office Bishnah | Jammu/Bishnah | 115.38 |
| 51 | Jal Shakti Department | PHE, Jammu | Executive Engineer Residence | Doda/Doda | 115.38 |
| 52 | Department of Home | Department of Home | Hqrs. IRP-6th Bn PCR Sgr. | Srinagar | 110.00 |
| 53 | Department of Home | Fire and Emergency Service Department-UT of J&K | F&E Station hqrs Gandhinagar | | 109.50 |
| 54 | Department of Home | Department of Home | Joint Interrogation Centre (JIC)Miran Sahib | Jammu/R.S Pura | 107.00 |
| 55 | Department of Home | Fire and Emergency Service Department-UT of J&K | F&E Station Gangyal/Range workshop | | 106.24 |
| 56 | Jal Shakti Department | PHE, Jammu | Executive Engineer PHE Residence Udhampur | Udhampur | 104.62 |
| 57 | Department of Home | Department of Home | DPO Jammu | Jammu north | 104.00 |
| 58 | Department of Home | Department of Home | Policy Colony, Bemina Srinagar | Srinagar | 100.00 |
| 59 | Department of Home | Department of Home | Adm.Biock (DPO) | Shopian | 100.00 |
| 60 | Department of Home | Department of Home | DPL.sopore | PD Sopot/Soporc | 100.00 |
| 61 | Department of Home | Department of Home | PC"Rafiahad | PD Sopore/Watergam | 100.00 |
| 62 | Department of Home | Department of Home | PC Sopore | PD Sopore/Sopore | 100.00 |
| 63 | Department of Home | Department of Home | DPO Soporc | PD Sopore/ Sopore | 100.00 |
| 64 | Department of Home | Department of Home | Trikuta Hostel Mess 24 Rooms | Udhampur | 100.00 |

Annexure-B of IFB

| Sr. No. | State | Department | Contact Details |
|------------|-------|---|--|
| | | Department of Home | Sh. Munish Sharma 9419182200 Sh. Qazi Irfan 9419484646 homeplg21@gmail.com |
| | J&K | Department of Forest and Ecology | Sh. Imtiyaz 9419037516 plansectionjsd@gmail.com |
| 1 | | Dept. of Law and Parliamentary Affairs | Sh. Khursid Ahmad, Dir(Planning) 7006112305 dirplgforest@gmail.com |
| | | Jal Shakti Department | Sh. Rajesh Talwar 9419141294 Sh. Sunil Sharma, Asstt. Registrar, High Court of J&K 7006827685 Sh. Rasal Singh, Asstt. Registrar, High Court of J&K 9419156181 |

Nodal Officer - Dr. P.R. Dhar, CEO JAKEDA, Mobile: 7006132502, Office No: 1943500433, Email: ceojakeda2@gmail.com

VOLUME 2

DEFINITIONS

<u>&</u>

STANDARD COPY

<u>OF</u>

PURCHASE POWER AGREEMENT (PPA)

INTERPRETATIONS

- 1. Words comprising the singular shall include the plural & vice versa.
- 2. Words indicating one gender include all genders;
- 3. Provisions including the word "agree," "agreed," or "agreement" require the agreement to be recorded in writing;
- 4. An applicable law shall be construed as reference to such applicable law including its amendments or re-enactments from time to time.
- 5. A time of day shall save as otherwise provided in any agreement or document be construed as a reference to Indian Standard Time.
- 6. Different parts of this RFS are to be taken as mutually explanatory and supplementary to each other and if there is any differentiation between or among the parts of this RFS, they shall be interpreted in a harmonious manner so as to give effect to each part.
- 7. The table of contents and any headings or sub headings in the RFS has been inserted for case of reference only & shall not affect the interpretation of this agreement.
- 8. The word "tender" is synonymous with "bid," "RFS", "tenderer," with "bidder," and "tender documents" with "bidding RFS documents," and

"Written" or "in writing" means hand-written, type-written, printed or electronically made, and resulting in a permanent record.

¹. **Definitions**

1.1 The following words and expressions shall have the meanings hereby assigned to them:

1.2Acts / **Codes** shall mean, but not limited to the following, including the latest amendmentsand/or replacements, if any:-

| 1 | .5.1.1 | Electricity Act, 2003 with amendments thereto if any. |
|---|--------|--|
| 1 | .5.1.2 | A.I.E.E Test Codes. |
| 1 | .5.1.3 | American Society of Testing of Materials (ASTMCodes). |
| 1 | .5.1.4 | Relevantstandardsofthe Bureau ofIndian Standards(ISCodes) |
| 1 | .5.1.5 | Arbitration and Conciliation Act, 1996, and Rules made thereunder. |
| 1 | .5.1.6 | Environment (Protection) Act, 1986 and Rules made there under. |
| 1 | .5.1.7 | Other approved standards and/or Rules and Regulations touching the subject matter of the Contract. |

1.3"*Affiliate*" shall mean a Company that, directly or indirectly, controls, or is controlled by, or is under common control with, a Company developing a Project or a Member in a Consortium developing the Project and control means ownership, directly or indirectly, of more than 50% (fifty percent) of the voting shares of such company or right to appoint majority Directors;

1.4 "Applicable Tariff" shall be the quoted Tariff by the selected Solar Power Generators/Project Developers.

1.5 Approved means approval in writing including subsequent written confirmation of previous verbal approval.

1.6'Bill of Quantities' (BOQ) or **'Price Schedules'** or **'Schedules'** means the bill of quantities included in the RFS for the Works.

1.7 "BID" or "PROPOSAL" shall mean the documents submitted by the Bidder towards meeting the techno-commercial and financial qualifying requirements, along with the price bid submitted by the Bidder and submissions during the e-Reverse Auctions, if applicable, as part of its response to the RfS issued by NHPC/NHPC REL

1.8 "Bidder" shall mean Bidding Company submitting the Bid.

1.9 Bidding Company" shall refer to such single company that has submitted the response in accordance with the provisions of this RfS;

1.10 "BID CAPACITY" shall mean contracted capacity of the Solar PV Power

Project(s) with or without Energy Storage Systems (ESS) as proposed by the Bidder.

1.11 **"BUYING ENTITY"** shall mean any Discom/distribution licensee that requires Solar Power to fulfil its RPO under respective RPO regulations / Energy Storage Obligations as per the Government of India's orders or notifications.

1.12 "Capacity Utilization Factor" or "CUF" shall have the same meaning as provided in CERC (Terms and Conditions for Tariff determination from Renewable Energy Sources) Regulations, 2020 as amended from time to time. However for avoidance of any doubt, it is clarified that the CUF shall be calculated on the Capacity at the "Delivery / Inter-connection / Metering Point".

1.13 "Chartered Accountant" shall mean a person practicing in India or a firm whereof all the partners are practicing in India as Chartered Accountants within the meaning of the Chartered Accountants Act, 1949;

1.14 "Company" shall mean a body corporate incorporated in India under the Companies Act, 1956 or the Companies Act, 2013 as applicable;

1.15 "Control" The control shall mean holding more than 50% of paid-up share capital;

1.16'Deleted

1.17 Commercial Operation Date (COD) shall be the actual date of commissioning of the Project as per IEGC-2023 provisions, as indicated on the Commissioning Certificate, upon successful commissioning of the full capacity of the Project or the last part capacity of the Project as the case may be.

1.18"Commissioning of Project" the Project will be considered as commissioned if all equipment as per rated project capacity has been installed and energy has flown into grid;

1.19 "CTU or Central Transmission Utility" shall mean the Central Transmission Utility as defined in sub-section (10) of Section 2 of the Electricity Act 2003;

1.20"Completion (Period)" of Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding is 9 (Nine) months from the date of Signing of PPA.

1.21"**Project Commissioning**" the Project will be considered as commissioned if all equipment as per rated project capacity has been installed and energy has flown into grid;.

1.22"Contractor's Equipment" means all plant, facilities, equipment, machinery, tools, apparatus, appliances or things of every kind required in or for transportation, installation, completion and maintenance of Facilities that are to be provided by the Contractor, but does not include Plant and Equipment, or other things intended to form or forming part of the Facilities.

1.23"Contractor's Representative" means any person nominated by the Contractor and approved by the NHPC/NHPC REL hereof to perform the duties delegated by the Contractor.

1.24"Contractor" means the person(s) whose bid to perform the Contract has been accepted by the NHPC/NHPC REL

1.25 Contract Year" shall mean the period beginning from the Effective Date and ending on the immediately succeeding March 31 and thereafter each period of 12 months beginning on April 1 and ending on March 31 provided that: in the financial year in which the Commencement of Supply Date (CSD) would occur, the Contract Year shall end on the date immediately before the CSD and a new Contract Year shall commence once again from the CSD and end on the immediately succeeding March 31, and thereafter each period of twelve (12) months commencing on April 1 and ending on March 31, and (ii) provided further that the last Contract Year of this Agreement shall end on the last day of the Term of this Agreement

1.26"Day" means calendar day of the Gregorian calendar.

1.27'DISCOM'means 'Concerned Distribution Utility of the State.

1.28'**Drawings', 'Plans'**, shall mean engineering data and drawings submitted by the Contractor with his bid and/or during the progress of the work provided such drawings are acceptable to the Engineer/NHPC/NHPC REL.

1.29"Effective Date" shall be the date of signing of PPA..

1.30"NHPC/NHPC REL" or "Owner" or "NHPC" means NHPC Ltd.and includes the legal successors or permitted assigns of the NHPC/NHPC REL.

1.31"Engineer/Engineer In-Charge" means the person appointed by the NHPC/NHPC REL to perform the duties delegated by the NHPC/NHPC REL.

1.32"Facilities" or "Works" means the Plant and Equipment to be supplied and installed, all works to be executed, all items & things to be supplied/done as well as all the Installation Services including design, fabrication, manufacture, supply, transportation, erection, testing and commissioning together with operation and maintenance works to be carried out by the RTD under the Contract.

1.33 "GOVERNMENT CLIENT ORGANIZATION" shall mean the JAKEDA and/or its departments /Bodies/ /PSU/ organization authorized to sign the PPA with the Roof Top Developer (RTD)

1.34 "Guidelines" shall mean Guidelines by Ministry of New & Renewable Energy (MNRE) vide its Office Memorandum No. 318/17/2024- Grid Connected Rooftop (part-2) dated 03.07.2024 issued Guidelines for Saturation of Government Buildings under PM Surya Ghar: Muft Bijli Yojana including subsequent amendments and clarifications thereto, if any, issued until the last date of bid submission of this RfS.

1.35 "**GRID CODE** Regulations" or "GRID CODE" shall mean the Central Electricity Regulatory Commission (Indian Electricity Grid Code) Regulations, 2023, as amended from time to time.

1.36'Independent External Monitor(s) (IEMs)' means the External Monitor(s) appointed by the NHPC/NHPC REL to oversee the implementation of Integrity Pact.

1.37'Integrity Pact' means the Pact signed between the NHPC/NHPC REL and Contractor committing the persons/officials of both the parties, not to exercise any corrupt influence on any aspect of the Tender/Contract.

1.38"Installation Services" means all those services ancillary to the supply of the Plant and Equipment for the Facilities, to be provided by the RTD under the RfS;

1.39 "Inter-connection point /Delivery/Metering point" shall be the single point, at a locations mentioned in RfS for each project under Connectivity details given by Government Client, in line with applicable regulation/ rules where Solar Roof Top Power is delivered by the RTD from the Project to the Client.

For interconnection with grid and metering, the RTD shall abide by the relevant CERC Regulations, Grid Code and Central Electricity Authority (Installation and Operation of Meters) Regulations, 2006 as applicable, amended and revised from time to time.;

1.40 "Joint Control" shall mean a situation where a company has multiple promoters (but none of the shareholders has more than 50% of voting rights and paid up share capital).

1.41 "Letter of Award" or "LOA" shall mean the letter issued by NHPC Limited/ NHPC REL to the Selected Bidder for award of the Project.

1.42'Metering Point'means the point at which the State Power Utility/Electricity Board measures the quantity of energy supplied to its grid.

1.43 "Month" shall mean a period of thirty (30) days from (and excluding) the date of the event, where specified, else a calendar month;

1.44'NODAL AGENCY' means 'Agency of the Centre/State'.

1.45 *Paid-up* share capital'' shall mean such aggregate amount of money credited as paid-up as is equivalent to the amount received as paid up in respect of shares issued and also includes any amount credited as paid up in respect of shares of the company but does not include any other amount received in respect of such shares, by whatever name called.

1.46 Party: means the NHPC/NHPC REL or the RTD as the context requires and Parties means both of them.

1.47'Person'shall include firms, companies, corporations and associations or bodies (individuals, whether incorporated or not.

1.48'Singular' or 'masculine 'includes' plural'or 'feminine' and vice-versa in their respective context.

1.49"Plant and Equipment" means permanent plant, equipment, machinery, apparatus, materials articles and things of all kinds to be provided and

incorporated in the Facilities by the RTD under the Contract, but does not include Contractor's Equipment.

1.50'Project'shall mean "Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode"

1.51 Project Developer" or Solar Power Developer (SPD) or Rood Top Developer (*RTD*) shall mean Bidding Company participating in the bid and having been selected and allocated a Project capacity by NHPC/NHPC REL (through a competitive bidding process) and supplying power under the PPA. Any reference to the Bidder includes Bidding Company including its successors, executors and permitted assigns and Lead Member of the Company jointly and severally, as the context may require.

1.52'Prudent Utility Practice'means accepted international / Indianpractice(s), standard(s), engineering and operation considerations,taking into account the conditions prevalent at Site including manufacturer's recommendations generally followed in the operation and maintenance of facilities similar to the power plant.

1.53 "RESCO" shall mean Renewable Energy Service Companies.

1.54 "RESCO MODEL" shall mean a business model where the Project Developer sets up a Solar Roof Top PV Power Project on the Building owned by a different entity, by obtaining right to access the Roof/leasing the Roof with the Roof-owning entity on mutually agreed terms and conditions, and enters into the PPA with Roof owner/ DISCOM/ others for supply of Solar power for the term of the PPA. The Project Developer is responsible for construction, commissioning, ownership and operation of the Project for the entire term of the PPA at its own risk and cost.

1.55 "RfS" or "RfS document" or "Bidding Document(S)" or "Tender Documents" shall mean the "Request for Selection" document issued by NHPC/NHPC REL including standard Power Purchase Agreement along with subsequent clarifications and amendments thereof issued.

1.56 "**RTD**" shall mean the Roof Top Developer

1.57 Scheduled Commencement-of-Supply Date (SCSD) in relation to the contracted capacity shall mean the date corresponding to the date of commencement of supply as indicated in the RfS. Scheduled Commencement of Supply Date (SCSD) shall be 09 (Nine) months from the Effective Date of PPA.

1.58"SCC" means the Special Conditions of Contract.

1.59 *"Selected Bidder or Successful Bidder"* shall mean the Bidder selected pursuant to this RfS to set up the Project and supply electrical output as per the terms of PPA;

1.60"Site" means the rooftop spaces and other places upon which the Facilities are to be installed and such other land/water body or places as may be specified in the Contract as forming part of the Site.

1.61'Specification', or Technical Specifications or 'NHPC/NHPC REL's

Requirement'shall mean the Technical specifications forming a part of the RfS and such other schedules as may be mutually agreed upon.

1.62 "STU or State Transmission Utility" shall mean the Board or the Government Company notified by the respective State Government under Sub-Section 1 of Section 39 of the Electricity Act 2003;

1.63'TRANSCO'means 'StateTransmission Company or Corporation'.

1.64 "Ultimate Parent" shall mean a company, which owns not less than fifty one percent (51%) of paid up share capital either directly or indirectly in the Parent and Affiliates.

1.65 'Writing' shall include any manuscript, type written or printed statement, underorover signature of the authorized representative of NHPC/NHPC REL/Contractor and/or seal as the case maybe.

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

STANDARD COPY

OF

PURCHASE POWER AGREEMENT (PPA)

MODEL POWER PURCHASE AGREEMENT (PPA)

BETWEEN

CONTRACTOR AND GOVT.ORGANIZATION, PSU AND OFFICES FOR DESIGN, MANUFACTURE, SUPPLY, ERECTION, TESTING AND COMMISSIONING INCLUDING WARRANTY, OPERATION & MAINTENANCE OF GRID CONNECTED ROOF-TOP SOLAR PHOTOVOLTAIC AND SMALL SOLAR POWER PLANTS IN RESCO MODEL This Power Purchase Agreement (PPA) is executed on ____(date), ___(month), ____(year) at <location>between ______Authorized representative of Purchaser<i.e. Government Organization, PSU and Offices>(detail address), (hereinafter referred to as "Purchaser")ANDM/s (Name of Power Producer) (CIN No. ______), a company incorporated under the Companies Act, 1956/2013 having its registered office at ______(detail address) (hereinafter referred to as "Power Producer" which expression shall, unless repugnant to the meaning or context hereof, be deemed to include its successors and assigns). The Purchaser and Power Producer are each individually referred to as a "Party" and collectively as the "Parties".

WHEREAS:

A. The Power Producer has been notified as successful bidder by <Name of Expert PSU> for "Design, Manufacture, Supply,Erection, Testing and Commissioning including Warranty, Operation & Maintenance of _____ kWp Rooftop Solar PV System at (building name)" (Project) and as per competitive biddingunder RFS No ______ dated _____

B. The Power Producer is engaged in the business of design, supply, erection, testing, commissioning, operating and maintenance power plants, including grid connected rooftop power projects. C. The Power Producer has agreed to install and operate a solar photovoltaic power plant of _____kWp capacity at the Premises after due inspection of the Premises as defined hereinafter and supply the entire Solar Power of the Project to Purchaser on the terms and conditions contained in this Agreement. D. The Purchaser has agreed to purchase the entire Solar Power of the Project on the terms and conditions contained in this Agreement.

NOW THEREFORE in consideration of the mutual promises, conditions and covenants set forth herein, the Parties hereby agree as below:

1. Definitions and Interpretation

1.1 Definitions

In addition to other capitalized terms specifically defined elsewhere in the Agreement or unless the context otherwise requires the following words and phrases shall be defined as follows:

(a) "Actual Monthly Production" means the amount of energy recorded by the Main Metering System during each calendar month of the Term, pursuant to Section 5.2;

(b) "Affiliate" means with respect to any specified Person, any other Person, directly or indirectly controlling, controlled by or under common control with such specified Person)

(c) "Agreement" means this Power Purchase Agreement executed hereof, including the schedules, amendments, modifications and supplements made in writing by the Parties from time to time. (d) "Applicable Law" me (e) ans, with respect to any Person, any constitutional provision, law, statue, rule, regulation, ordinance, treaty, order, decree, judgment, decision, certificate, holding, injunction, registration, license, franchise, permit, authorization, guideline, Governmental Approval, consent or requirement or any Governmental Authority in India having jurisdiction over such Person or its property, enforceable by law or in equity, including the interpretation and administration thereof 1:4 such Governmental Authority.

(f) "Assignment" has the meaning set forth in Section 14.1:

(g) "Business Day" means any day other than Sunday or any other day on

which banks in < name of the State> are required or authorized by Applicable Law to be closed for business:

(h) "Commercial Operation Date" has the meaning set forth in Section 4. 3(b)

(i) "Consents, Clearances and Permits" shall mean all authorization, licenses, approvals, registrations, permits, waivers, privileges, acknowledgements, agreements or concessions required to be obtained From or provided by any concerned authority for the purpose of setting up of the generation facilities and / or supply of power:

(j) "Deemed Generation" has the meaning set forth in Section 5.3 (c)

(k) "Delivery Point" shall be the single point, at a location mutually agreed by the Parties, in line with applicable regulation/ rules where Solar Power is delivered by the Power Producer from the System to the Purchaser.

(I) "Dispute" has the meaning set forth in Section 17.7 (b);

(m) "Disruption Period" has the meaning set forth in Section 5.3 (c) (n) "Distribution Utility" means the local electric distribution owner and operator providing electric distribution and interconnection services to Purchaser at the Premises;

(o) "Due Date" has the meaning set forth in Section 7.4;

(p) "Effective Date" has the meaning set forth in Section 2;

(q) "Estimated Remaining Payments" means as of any date, the estimated remaining Solar Power Payments to be made through the end of the applicable Term, as reasonably determined by the Power Producer in accordance with Section 7.1;

(r) "Expiration Date" means the date on which the Agreement terminates by reason of expiration of the Term.

(s) "Force Majeure Event" has the meaning set forth in Section 11.1

(t) "Governmental Approval" means any approval, consent, franchise, permit, certificate, resolution, concession, license or authorization issued by or on behalf of any applicable Governmental Authority for the purpose of setting up of the Project and / or for sale and purchase of Solar Power of the Project pursuant to the Agreement.

(u) "Governmental Authority" means any central, state, regional, district, town, city or municipal government, whether domestic or foreign, or any department, agency, bureau, or other administrative, regulatory or judicial body of any such government.

(v) "Indemnified Persons" means the Purchaser Indemnified Parties or the Power Producerindemnified Parties, as the context requires. (w) "Insolvency Event" means with respect to a Party, that either

i. Such party has (A) applied for or consented to the appointment of or the taking of possession by a receiver, custodian, trustee, administrator, liquidator on the likes of itself or of all or a substantial part of its assets or business; (B) been unable to pay its debts as suchdebts become due; (C) made a general assignment for the benefit of its creditors, (D)

4 | Page

commenced a voluntary proceeding under any insolvency or bankruptcy law; (E) filed a petition seeking to take advantage of any other law relating to the bankruptcy, insolvency, reorganization, winding up or composition or readjustment of debts; or (F) taken any corporate or other action for the purpose of effecting any of the foregoing; or ii. It is clarified that a dissolution or liquidation will not be an Insolvency Event if such dissolution or liquidation is for the purpose of a merger, consolidation or reorganization, and the resulting company retains credit worthiness similar to the dissolved or liquidated entity and expressly assumes all obligations of the dissolved and liquidated entity under this Agreement and is in a position to the perform them.

(x) "Installation Work" means the construction and installation of the System and the Startup, testing and acceptance (but not the operation and maintenance) thereof; all performed by or for the Power Producer at the Premises. (y) "Invoice Date has the meaning set forth in Section 7.2.

(z) "Losses" means all losses, liabilities, claims, demands, suits, causes of action, judgments, awards, damages, cleanup and remedial obligations, interest, fines, fees, penalties, costs and expenses (including all attorneys' fees and other costs and expenses incurred in defending any such claims or other matters or in asserting or enforcing and indemnity' obligation) (aa) "Main Metering System" means all meter(s) and metering devices owned by the Power Producer and installed at the Delivery point for measuring and recorded the delivery and receipt of energy.

(aa) "Metering Date" means the first Business day of each calendar month subsequent to the month in which the Solar Power is generated by the Power Producer. The billable units shall be equal to the difference between the meter reading on the Metering Date and the meter reading on the previous month's Metering Date.

(bb) "Party" or Parties" has the meaning set forth in the preamble to this Agreement.

(cc) "Performance Ratio" (PR) means the ratio of plant output versus installed plant capacity at any instance with respect to the radiation measured. PR=(Measured output in kW / Installed plant capacity in kW* 1000 W/m2 /Measured radiation intensity in W/m2)

(dd) "Person" means an individual, partnership, corporation, Limited Liability Company, business trust, Joint Stock Company, trust, unincorporated association, joint venture, firm, or other entity, or a Governmental Authority.

(ee) "Power Producer Default" has the meaning set forth in Section 12.1 (a).

(ff) "Power Producer Indemnified" has the meaning set forth in Section 16.2.

(gg) "Premises" means the premises described in Schedule 1 to this Agreement. For the avoidance of doubt, the Premises include, the entirely of any and underlying real property located at the address described in Schedule 1 to this Agreement.

(hh) "Purchase Date" means the date on which title to the System transfers to the Purchaser pursuant to the Purchaser exercising its purchase option under Section 3.2.

(ii) "Purchase Price" means the fee payable by Power Purchaser to the Power Producer under the circumstances described in Section 3.2 (jj) "Purchaser Default" has the meaning set forth in Section 12.2 (a).

(kk) "Purchaser Indemnified Parties" has the meaning set forth in Section 16.1(II) "Representative" has the meaning forth in Section 15.1.

(mm) "Scheduled Complete Date" has the meaning set forth in Section 4.1(g) (nn) "Selectee" means, a new company (i) proposed by the Lenders read with Schedule III hereof and approved .by the Purchaser (ii) or proposed by the Purchaser in accordance. with Schedule III hereof and approved by-the Lenders, for substituting the Power Producer for the residual period of the Agreement by amendment of the Agreement or by execution of a fresh power purchase agreement in-accordance with the terms and conditions contained in the said Schedule. (oo) "Solar Power" means the supply of electrical energy output from the System. (pp) "Solar Power Payment" has the meaning set forth in Section 7.1.

(qq) "System" includes the integrated assembly of photovoltaic panels, mounting, assemblies, inverters, converters, metering, lighting fixtures, transformers, ballasts, disconnects, combiners, switches, wiring devices and wiring, and all other material comprising the Installation Work.

(rr) "System Operations" means the Power Producer's operation; maintenance and repair of the System performed in accordance the requirement herein.

(ss) "Tariff" means the price per kWh set forth in Schedule II hereto.

(tt) "Term" has the meaning set forth in Section 3.1:

1.2 Interpretation

(a) Unless otherwise stated, all references made, in this Agreement to "Sections", "Clauses" and "Schedules" shall refer respectively to Sections, clauses and Schedules of this Agreement. The Schedules to this Agreement form an integral part of this Agreement and shall have effect as though they were expressly set out in the body of this Agreement.

(b) In the Agreement, unless the context otherwise requires (i) words imparting singular connotation shall include plural and vice versa: (ii) the words "include", "includes", and "including" mean include, includes and including "without limitation" and (iii) the words "hereof", "hereto", "herein" and "hereunder" and words of similar import refer to the Agreement as a whole and not to any particular provision of the Agreement.

2. Effective Date

This Agreement shall be effective on the day that falls one Business Day after the date of signing of this Agreement.

3. Terms and Termination

3.1 Term

The term of the Agreement shall commence on the Effective Date and shall continue for twenty five (25) years from the Commercial Operations Date (the "Term"), unless and until terminated earlier pursuant to the provisions of the Agreement. After the Term, the ownership of the System shall be transferred to the Purchaser free of cost.

3.2 Purchase Option/ Purchase Obligation

So long as a Purchaser default shall not have occurred and be continuing, Purchaser has the option to purchase the System by paying the Power Producer the Purchase price as per Schedule III to this Agreement. To exercise its purchase option, the Purchaser shall not less than Ninety (90) days prior to the proposed Purchase Date, provide written notice to the Power, Producer of Purchaser's intent to exercise .its option to purchase the System on such purchase date: In the event Purchaser confirms its intention to exercise the purchase option in writing to the Power Producer, (i) Purchaser shall pay the applicable purchase price to the Power Producer on the Purchase Date, and such payment shall be made in accordance with any written instructions delivered to Purchaser by the Power Producer for payments under the Agreement, and (ii) the Parties shall promptly execute all documents necessary to (A) cause title to the System to pass to Purchaser on the Purchase Date, free and clear of all liens and (B) assign all vendor warranties for the System to Purchaser. Upon execution of the documents and payment of the applicable purchase price in each case as described in the preceding sentence, the agreement shall terminate automatically and the Purchaser shall become the owner of the System. Upon such termination, the Power Producer shall offer' its operations and maintenance ("O&M") services to the Purchaser and the Parties may enter into an O&M agreement in this regard. The terms and conditions of the O&M agreement will be negotiated in good faith between the Parties.

3.3 Conditions of the Agreement prior to installation

In the event that any of the following events or circumstances occur prior to the Commercial Operation Date, the Power Producer may terminate the Agreement, in which case neither Party shall have any liability to the other except for any such liabilities that may have accrued prior to such termination.

(a) There has been a, material adverse change in the rights of Purchaser to occupy the Premises or the Power Producer to install the System at the Premises.

(b) The Power Producer has determined that there are easements, Capacity Cost Recovery (CCRs) or other liens or encumbrances that would materially impair or prevent the installation, operation, maintenance or removal of the System. If any dispute arises before commercial operation date, the same shall be resolved under clause, 17.7 (c)

4 Construction, Installation, Testing and Commissioning of the System.

4.1 Installation Work

a) The Power Producer will cause the Project to be designed, manufactured, supplied, engineered, erected, tested and commissioned, operated & maintained and constructed substantially in accordance with RFS No _____ dated: _____ and the sanction letter issued by <NAME OF EXPERT PSU>. The Power Producer shall provide to the Purchaser a bill of materials listing the major equipment constituting the System. Such bill of materials shall be provided within 30 days of the Commercial Operation Date.

b) The Power Producer shall have access as reasonably permitted by the Purchaser to perform the Installation Work at the Premises in a manner that minimizes inconvenience to and interference with the use of the Premises to the extent commercially practical.

c) It is agreed between the Parties that the Power Producer shall commission the System with a capacity of ___kWp. Power Producer may construct a System of smaller size if it receives only part approval of government subsidies or for any other material commercial reason, as mutually agreed between the Parties in writing. In the event a System of smaller capacity is eventually agreed to be installed, the clauses pertaining to Purchase Price as set out under this Agreement shall be adjusted proportionately as per mutual agreement between the Parties in writing.

d) The Power Producer shall provide and lay the dedicated electrical cables for transmission of Solar Power from the System up to the Delivery Point. Transmission or distribution of Solar Power beyond this point will be the responsibility of the Purchaser. The Delivery Point shall be where the Main Metering System is located.

e) Unless otherwise agreed between the Parties, the Power Producer shall not do (a) chipping of rooftop; or (b) water proofing of roof to be disturbed ;(c) Carry out any other modification of the Premises without the written consent of the Purchaser.

f) The Power Producer shall maintain general cleanliness of area around the Project during construction and operation period of the Project. In case any damages is caused to the equipment / facilities owned by the Purchaser due to the Power Producer, the same shall be made good rectified by the Power Producer at their cost.

g) The Power Producer shall, within fifteen (15) working days of the Effective Date, submit to the Purchaser shop drawings of the Project for approval ("Shop Drawings"). The drawings will have to be approved from the Power Purchaser within 3 working days from the submission of the drawings. If the Purchaser has any objection/recommendation in the Shop Drawings, he shall communicate the same to Power Producer within a period of ten (10) working days of the date of submission of the Shop Drawings. Any delay will extend the Effective Date and such approval shall not be unreasonably withheld Subject to any punchlist items which shall be agreed by the. Purchaser as not being material to completion of the Project, the Power Producer agrees that it shall achieve the completion of the Project/ Commissioning of the Project within the scheduled completion period from the Effective Date ("Scheduled Completion Date"). Purchaser shall ensure that sufficient load is available at the Delivery Point to ensure synchronization and drawl of power from System.

h) If the Power Producer is unable to commence supply of Solar Power to the Purchaser by the Scheduled Completion Date, other than for the reasons specified in Article 11 and 12.2 (Force Majeure or Purchaser Default), the Power Producer or its contractor shall pay to <NAME OF EXPERT PSU> genuine pre-estimated liquidated damages for the delay in such commencement of supply of Solar Power as per the clause of the <NAME OF EXPERT PSU> RFS appended as Schedule VI format as per respective RFS document to this Agreement.

i) The Purchaser shall ensure that all arrangements and infrastructure for receiving Solar Power beyond the Delivery Point are ready on or prior to the Commercial Operation Date and is maintained in such state in accordance with applicable laws through the Term of the Agreement.

j) Power Producer shall fulfill all obligations undertaken by it under this Agreement.

4.2 Approvals and Permits

Each of the Parties shall assist the other Party in obtaining all necessary Government Approvals, third party approvals and permits including but not limited to those listed in Schedule V hereto and any waivers, approvals or releases required pursuant to any applicable CCR.

4.3 System Acceptance Testing

(a) The Power Producer shall give 10 days advance notice to conduct the testing of the Project and shall conduct testing of the Project in the presence of Purchaser's designated representative.

(b) If the results of such testing indicate that the System is capable of generating electric energy (at full rated KWp) for 5 continuous hours using such instruments and meters as have been installed for such purposes, then the Power Producer shall send' a Written notice to Purchaser to that effect, and the date of successful conducting such tests and injection of. Power at Delivery Point shall be the "Commercial Operation Date"

5 System Operations

5.1 The Power Producer as Owner and Operator. The System will be legally and beneficially owned by the Power Producer and will be operated and maintained and, as necessary, repaired by the Power Producer at its sole cost and expense. Replacement of spare parts if any shall be responsibility of Power Producer for the complete period of Agreement. If any repair or maintenance costs incurred by the Power Producer as a result of Purchaser's breach of its obligations, shall be reimbursed in full by Purchaser. Power Producer shall not be responsible for any work done by others on any part of the System/Project authorized by the Purchaser and not authorized in advance by the Power Producer in writing. Power Producer shall not be responsible for any loss, damage, cost or expense arising out of or resulting from improper operation or 'maintenance of the System by Purchaser or anyone instructed to do such work by Purchaser. In the event of a problem with the System, as a result of the Purchaser actions for which Power Producer is not responsible as provided in this Agreement, Purchaser may choose and pay Power Producer for diagnosing and correcting the problem at Power Producer or Power Producer's contractors standard rates.

5.2 Metering

(a) The Power producer shall install the Main Metering System with due certification for the measurement of electrical energy produced by the System.

(b) The meter will be read by Power Producer's personnel on the Metering date. The authorized representative of the Purchaser shall be present at the time of meter reading. Both the Parties shall sign a joint meter reading report. However, in case the Joint meter reading report is not signed in the first three business days of any month due to non-availability of the Purchaser's authorized representative, the report signed by the Power Producer shall be considered as Joint Meter Reading Report. The Parties agree that such Joint meter reading Report shall be final and binding on the Parties.

(c) The Main Metering System at the Delivery Point and any additional meters required by Applicable Law shall be tested, maintained and owned by the Power Producer.

(d) The Power Producer shall connect the Solar output to the existing system of the Purchaser as per the requirements and guidelines of the state DISCOM

(e) The Purchaser may, at its own discretion, install a check meter, at its cost, to verify the measurements of the Main Metering System.

(f) The title to the Solar Power supplied by the Power Producer shall pass to the Purchaser at the Delivery Point.

(g) Power Producer shall be responsible for transformer etc. "if required" & metering as per respective clause of RFS (reference to be quoted).

5.3 System Disruptions

(a) Availability of premises: Purchaser will provide full access of the site to Power Producer for installation, operation and maintenance of solar power plant during the period of Agreement. Power Purchaser will also provide restricted access of the Premises to Power Producer for operation and maintenance of solar power plant.

(b) Purchaser will not provide/construct any structure within its Premises or around its premises which shades the solar panels effecting the generation of the energy during the Agreement period.

(c) Roof Repair and other System Disruptions In the event that (a) the Purchaser repairs the Premises' roof for any reason not directly related to damage, if any, caused by the System, and such repair required the partial or complete temporary disassembly or movement of the System, or (b) any act or omission of Purchaser or Purchaser's employees, Affiliates, agents or subcontractors (collectively, a "Purchaser Act") result in a disruption or outage in System production, and such events attributable to Purchaser (except Force majeure, then, in either case) Purchaser shall (i) pay the Power Producer for all work required by the Power Producer to disassemble or move the System and re-assemble the system after completion of the repair work and (ii) continue to make all payments for the Solar Power during such period of System disruption (the "Disruption Period"). For the purpose of calculating Solar Power Payments and lost revenue for such Disruption Period, Solar Power shall be deemed to have been produced at the average rate. Over the preceding 12 (Twelve) months, or, if the disruption occurs within the first 12 months of operation, the average over such period of operation (deemed generation). Power producer shall inform about the 'disruption or outage in System production, for reasons attributable to purchaser in 'writing with date and time of such occurrences, and Purchaser's liability shall start from the date of intimation for above of disruption or outage in system production, on account of Purchaser.

12 | Page

6 Delivery of Solar Power

6.1 Purchaser Requirement: Purchaser agrees to purchase one hundred percent (100%) of the Solar Power generated by the System and made available by the Power Producer to Purchaser at the Delivery Point during each relevant month of the Term. In the event that the Purchaser is unable to off take 100% of the electricity generated, when it is generated, then Deemed Generation will apply only in case following conditions:

a) In the event Power Producer is generating power more than the available load, and the Purchaser is not able to export or record the excess units generated due to faults in the equipment's of the Purchaser e.g., Net meter Cables, Equipment's etc., which may stop the feeding / record of the Solar Power generated. b) In the event that the Purchaser fails to ensure adequate space for solar equipment to ensure that other structures do not partially or wholly shade any part of the Solar Power Plant and if such shading occurs, the Power Producer may apply for Deemed Generation furnishing the calculation for loss in generation due to such shading supported by the relevant data, which shall be approved by Purchaser within one month of submission failing which the Power Producer shall claim provisional deemed generation till the issue is finally settled.

6.2 Estimated Annual Production

The annual estimate of Solar Power with respect to the System for any given year as determined pursuant to this Section shall be the "Estimated Annual Production". The Estimated Annual Production for each year of the Initial Term is set forth in Schedule IV hereof.

6.3 Suspension of Delivery

Power Producer shall be entitled to suspend delivery of electricity from the System to the Delivery Point for the purpose of maintaining and repairing the System upon giving one week's advance written notice to the Purchaser except in the case 'of emergency repairs. Such suspension of Service shall not constitute a breach of this Agreement provided that the Power Producer shall use commercially reasonable efforts to minimize any 'interruption in service to the Purchaser. However, any preventive maintenance shall be done only during the period when plant is not generating.

7 Tariff and Payments

7.1 Consideration Purchaser shall pay to the Power Producer a monthly payment (the "Solar Power Payment") for the Solar Power generated by the System as per the Metering clause 5.2 (b) above during each calendar month of the Term equal to the actual Monthly Production as recorded in Joint Meter Reading Report for the System for the relevant month multiplied by the Tariff irrespective of (i) whether any or all units of Solar Power has been drawn, consumed or utilized by Purchaser and / or (ii) whether any Solar Power has been injected, whether inadvertently or otherwise, into the grid of the Distribution Utility. The Power Producer will bill the Purchaser for each KWh metered as above at the Delivery Point, at the Tariff prevailing at that point of time. As detailed in Schedule - II, the Tariff will be equal to Rs. ____/kWh levelized tariff as per <NAME OF EXPERT PSU> allocations.The 'year' Considered shall be the financial year which April 1st to 31st March of every year as per <NAME OF EXPERT PSU>Schedule II provides a detailed year on year tariff schedule.

7.2 Invoice

The Power Producer shall invoice Purchaser on the first day of each month (each, an "Invoice Date") commencing on the first Invoice Date to occur after the Commercial Operation Date, for the Solar Power Payment in respect of the immediately preceding month. The last invoice shall -include production only through the Expiration Date of this Agreement.

7.3 The invoice to the purchaser shall include.

(a) The Solar Power calculations for the relevant billing period.

(b) Supporting data, documents and calculations in accordance with this Agreement.

7.4 Time of payment

Purchaser shall pay all amounts due hereunder within 30 days after the date of the receipt of the invoice via email or post ("Due Date").

7.5 Method of Payment

Purchaser shall make all payments under the Agreement by cheque / demand draft/ electronic funds transfer only in immediately available funds to the' account designated by the Power Producer from time to time. All payments made hereunder shall be nonrefundable, subject to the applicable tax deduction at source, and be made free and clear of any other tax, levy, assessment; duties or other charges and not subject to reduction, setoff, or adjustment of any kind. Further, if any taxes and duties are leviable currently or in future, such taxes and duties shall be paid by the Purchaser over and above the solar electricity tariff mentioned in this agreement. Such taxes and duties could include, but not restricted to Electricity Duty, Tax on Sale of Electricity (TOSE). If the Purchaser deducts any tax at source, the Purchaser will issue a tax credit certificates as per law.

7.6 Late Payment Surcharge/ Early Payment Discount

In case payment of any invoice is delayed by the Purchaser beyond its Due Date, a late payment surcharge shall be payable by Purchaser to the Power Producer at the rate of 1.25% per month ("Late Payment Surcharge") calculated on the amount of outstanding payment, calculated on a day to day basis for each day of the delay, compounded on monthly rates. Late Payment Surcharge shall be claimed by the Power Producer, through its subsequent invoice.

7.7 Disputed Payments

In the event that the Purchaser disputes an invoice, it shall give notice of such a dispute within 15 days of receiving the invoice setting out details of the disputed amount. The Purchaser shall pay by the Due Date 100% of any undisputed amount and in case the invoice is disputed, the Purchaser shall pay an amount based on average consumption of last three consecutive undisputed invoices. Amount so recovered shall be subject to final adjustment on resolution of the dispute. Thereafter, the Parties shall discuss and try to resolve the disputed amount within a week of receipt of such notice of dispute. If the Parties resolve the dispute, an appropriate adjustment shall be made in the next invoice. If the dispute has not been resolved by the date of the next invoice the dispute shall be referred to a committee of one member from each of Purchaser and Power Producer, If the dispute is still-not resolved by the next following invoice if shall be 'referred to Arbitration as provided in the present Agreement.

7.8 Change in Law:

(a) For the Purpose of this section 7.8, the term "Change in Law" shall mean the occurrence of any of the following events after the Effective date, resulting into any additional recurring / non-recurring expenditure by the Power Producer or any income to the Power Producer. The enactment, coming into effect, adoption, promulgation, amendment, modification or repeal

(without re-enactment or consolidation) in India, of any Law, including rules and regulations framed pursuant to such Law: or

(i) A change in the interpretation of any Law by any Governmental Authority having the legal power to interpret or apply such Law, or any competent court; or

(ii) The imposition of a requirement, for obtaining any Government Approvals which was not required earlier; or

(iii) a change in the terms and conditions prescribed for obtaining any Government Approvals or the inclusion of any new terms or conditions for Obtaining such Government Approvals; or

(iv) any introduction of any tax made applicable for supply of power by the Power Producer as per the terms of this Agreement. Any benefit due to change in tax on the sale of solar energy shall be passed on to Purchaser.

(v) Any benefit arising due to change in above para (i) to (iv) shall be passed on to the Purchaser. But not include any change in any withholding tax on income or dividends distributed to the shareholders of the Power Producer.

(b) Application and Principles for computing impact of Change in Law:

While determining the consequence of Change in Law under this Article 7.8, the Parties shall have due regard to the Principle that the purpose of compensating the Party affected by such change in Law, is to restore through monthly bill payment, to the extent contemplated in this Article 7.8, the affected Party to the same economic position as if such Change in Law has not occurred and such impact shall be mutually decided in writing.

(c) Solar Power Payment Adjustment Payment on account of Change in Law Subject to provisions mentioned above, the adjustment in Solar Power Payment shall be effective from:

(i) The date of adoption, promulgation, amendment, re-enactment or repeal of the Law or Change in Law; or

(ii) The date of order/judgment of the competent court; of tribunal or Governmental Authority, if the Change in •law is on account of a change in interpretation of Law.

8 General Covenants

8.1 Power Producer's Covenants

The Power Producer covenants and agrees to the following:

(a) Notice of Damage or Emergency: The Power Producer shall (a) promptly notify Purchaser if it becomes aware of any damage to or loss of the use of the System or that could reasonably be expected to adversely affect the System, (b) immediately notify Purchaser once it becomes aware of any event or circumstance that poses an imminent risk to human health, the environment, the System or the Premises.

(b) System Condition: The Power Producer shall take all actions reasonably necessary to ensure that the System is capable of providing Solar Power at a commercially reasonable continuous rate: Subject to there being no Purchaser Default, the Power Producer 'shall provide 24 x 7 offsite / offsite monitoring and maintenance of the System throughout the period of this agreement at no additional cost.

(c) The System shall meet minimum guaranteed generation with Performance Ratio (PR) at the time of commissioning and related Capacity Utilization Factor (CUF) as per the daily normalized irradiance levels of the location during the O&M period. PR shall be minimum of 75% at the time of inspection for initial Project acceptance.

(d) Governmental Approvals: While providing the Installation work, solar Power and System Operations, the Power Producer shall obtain and maintain and secure all Governmental Approval required to be obtained and maintained and secured by the Power Producer and to enable the Power Producer to perform such obligations

(e) The interconnection of the rooftop solar system with the network of the distribution licensee shall be made as per the technical standards for connectivity of distributed generated resources regulation's .as may be notified by the competent authority. The interconnection of the rooftop solar system shall be as per the contracted load and / or respective voltage level applicable to the Purchaser as per the provisions of the guidelines issued by the competent authority.

(f) Health and Safety: The Power Producer shall take all necessary and reasonable safety precautions with respect to providing the installation Work, Solar Power, and System Operations that shall comply with all Applicable Law pertaining-to the health and safety of persons and real and personal property.

8.2 Power Producer's Representatives

During the subsistence of this Agreement, the Power Producer undertakes to respond to all questions, concerns and complaints of the Purchaser regarding the System in a prompt and efficient manner. The Power Producer designates the following individual as its representative pertaining to performance of this Agreement till the Commercial Operation Date:

Name: _____

Telephone: _____

Email: _____

The Power Producer designates the following individuals as its representative and primary point of contact pertaining to performance of this Agreement following the Commercial Operation Date till termination:

Name:

Telephone: _____

Email:

8.3 Purchaser's Covenants

Purchaser covenants and agrees to the following:

(a) Notice of Damage or Emergency: Purchaser shall (a) promptly notify the Power Producer if it becomes aware of any damage to or loss of the use of the System or that could reasonably be expected to adversely affect the System; (b) immediately notify the- Power Producer once it becomes aware of any event or circumstance that poses an imminent risk to human health, the environment, the System or the Premises.

(b) Liens: Purchaser shall not directly or indirectly cause, create, incur, assume or suffer to exist any Liens on or with respect to the System or any interest therein. If Purchaser breaches its obligations under this Clause, it shall immediately notify the Power Producer in writing, and shall promptly cause such Lien to be discharged and released of record without any cost to the Power Producer, and shall indemnify the Power Producer against all costs and expenses (including reasonable attorneys fees and court costs) incurred in discharging and releasing such Lien.

(c) Consents and Approvals: Purchaser shall ensure that any authorizations required of Purchaser under this Agreement, including those required for installation of System at the Premises and to drawl consume Solar Power are provided in a timely manner. The Purchaser shall cooperate with the Power Producer to obtain such approvals, permits, rebates or other 'financial incentives.

(d) Access to Premises Grant of License: Purchaser hereby grants to the Power Producer a license co-terminus with the Term, containing all the rights necessary for the Power Producer to use portions of the Premises for the installation, operation and maintenance of the System pursuant to the terms of this Agreement, including ingress and egress rights to the Premises for the Power Producer and its employees and authorized representatives and access to electrical panels and conduits to interconnect or disconnect the System with the Premises electrical wiring with the consent and approval of the Purchaser's authorized representative identified by the Purchaser. Photo IDs will be provided by the Power Producer. Power Purchaser will assist in availing permissions to the site.

(e) Security: The building which has enhanced security of Solar Power System Purchaser will keep the premises locked. Inspite of these measures, if any damages to the System takes place due to theft or vandalism then the same shall be claimed or reimbursed through insurances by Power producer. In case of theft and vandalism acts, the Purchaser will assist the Power Producer in procedures of filing FIRs, insurance claims and any other related activities. Whenever, the damages to the System occurs (except due to negligence of Power Producer) then the same shall be jointly assessed by both the Parties and a severity level will be decided, which will further decide the duration offered to the Power Producer to correct

the damage, and the and the Power Producer shall he paid the amount on the basis of 'Deemed generation' for such a period. Power Producer shall be entitled to any insurance proceeds received for damages in this clause. Purchaser will not conduct activities on, in or about the Premises that have a reasonable likelihood of causing damage, impairment or otherwise adversely affecting the System. If System is damaged due to any such activity or through any other agency contracted by the Purchaser directly or indirectly, such damage will be borne by the Purchaser.

(f) Regardless of whether Purchaser is owner of the Premises or leases the Premises from a landlord, Purchaser hereby covenants that (a) the Power Producer shall have access to the Premises and System during the Term of this Agreement, and (b) neither Purchaser nor Purchaser's landlord will interferer or handle any of the Power Producer's equipment or the System without written authorization from the Power Producer.

(g) Temporary storage space during installation : Purchaser shall provide sufficient space at the Premises for the temporary storage and. staging of tools, materials and equipment and for the parking of construction crew vehicles and temporary Construction, trailers and facilities reasonably necessary during the Installation Work, System Operations and access for rigging and material handling.

(h) Sunlight Easements: Purchaser will take all reasonable actions as necessary to prevent other building, structures or flora from overshadowing or otherwise blocking access of 'sunlight to the System, including but not limited to-such actions as may be reasonably necessary to obtain a solar access easement for such purpose.

(i) Evacuation — Purchaser shall off take 100% of the Solar Power generated from the Delivery Point, and pay all invoices raised by the Power Producer under this Agreement by the 'Due Date and pay interest on delayed payments, if any, as per this Agreement.

(j) Water - Power Purchaser at zero cost shall arrange Raw Water at a given point as per the requirements of the Power Producer, for periodic cleaning of the solar panels (______ liters - twice a month).

(k) Auxiliary Power — The Purchaser shall provide sufficient auxiliary power to the Power Producer for the maintenance and operation of its system, if available and possible, at the rate Purchaser is paying to the DISCOM. (I) Relocation - If one or more of the Solar Power Plant panels needs to be temporarily moved or its generation suspended, for any other

20 | Page

reason requested by the Power Purchaser, the Power Purchaser will be responsible for preagreed costs on actuals only, arising from moving, disassembling and re-installing/ commissioning the Solar Power Plant, as agreed between the Parties. The Power Producer will be responsible for providing detailed documentary proof of the actual pre-agreed costs borne for such relocation/disassembling. Within 30 days of these satisfactory documents being provided by the Power Producer, the Purchaser shall reimburse these pre-agreed expenses in full, and delayed payment beyond the date mentioned above will attract Late Payment charges as described in Clause 7.6. During any interruption in generation during such relocation, the Purchaser will continue to be billed as per Deemed Generation, during the period of interruption, for the affected Solar Power Plant(s).

9 Representations & Warranties

9.1 Representations and Warranties Relating to Agreement Validity In addition to any other representations and warranties contained in the Agreement, each Party represent and warrants to the other that:

(a) It is duly organized and validly existing and in good standing in the jurisdiction of its incorporation;

(b) It has the full right and authority to enter into, execute, deliver, and perform its obligations under the Agreement.

(c) It has taken all requisite corporate or other action to approve the execution, delivery, and performance of the Agreement;

(d) The Agreement constitutes its legal, valid and binding obligation enforceable against such Party in accordance with its terms;

(e) There is not litigation, action, proceeding or investigation pending or, to the best of its knowledge, threatened before any court or other Governmental Authority by, against, affecting or involving any of it's business or assets that could reasonably be expected to adversely affect its ability to carry out the transactions contemplated herein; and

(f) Its execution and performance of the Agreement and the transactions contemplated hereby do not constitute a breach of any term or provision of, or a default under (i) any contract or agreement to which it or any of its Affiliates is a party or by which it or any of its Affiliates or its or their property is bound, (ii) its organizational documents, or (iii) any Applicable Laws.

10 Taxes and Governmental Fees

10.1 Purchaser obligations

Purchaser shall pay for any taxes, fees or charges imposed or authorized by any Governmental Authority in future (as on date no such taxes/fees/charges are being levied) on sale of the Solar Power to Purchaser pursuant to clause 7. The Power Producer shall notify Purchaser in writing with a detailed statement of such amounts, which shall be invoiced by the Power Producer in the monthly bills and payable by Purchaser. Purchaser shall timely report, make filings for, and pay any and all sales, use, income or other taxes, and any other amounts assessed against it due to its purchase of the Solar Power. This Section10.1 excludes taxes specified in Section 10.2.

10.2 Power Producer Obligations

The Power Producer shall be responsible for all income taxes and any and all franchise fees or similar fees assessed against it due to its ownership of the System. The Power Producer shall not be obligated for any taxes payable by or assessed against Purchaser based on or related to Purchasers overall income or revenues.

11 Force Majeure

11.1 Definition

"Force Majeure Event" means any act or event that prevents the affected Party from performing its obligation in accordance with the Agreement, if such act or event is beyond the reasonable control of the affected Party and such Party had been unable to overcome such act or event with the exercise of due diligence (including the expenditure of reasonable sums). Subject to the foregoing conditions, "Force Majeure Event" shall include without limitation the following acts or events: (i) natural phenomena, such as storms, hurricanes, floods, lightning, volcanic eruptions and earthquakes; (ii) explosions or fires arising from lighting or other causes unrelated to the acts or omissions of the Party seeking to be excused from performance; (iii) acts of war or public disorders, civil disturbances, riots, insurrection, sabotage, epidemic, terrorist acts, or rebellion. A Force Majeure Event shall not be based on 22 | Page

the economic hardship of either Party. In case of any damage because of force majeure event, the System shall be repaired / commissioned at its own cost by the Power Producer.

11.2 Excused Performance:

Except as otherwise specifically provided in theAgreement, neither Party shall be considered in breach of the Agreement or liable for any delay or failure to comply with the Agreement, if and to the extent that such delay or failure is attributable to the occurrence of a Force Majeure Event; provided that the Party claiming relief under this Section 11 shall immediately (i) notify the other Party in writing of the existence of the Force Majeure Event, (ii) exercise all reasonable efforts necessary to minimize delay caused by such Force Majeure Event, (iii) notify the other Party in writing of the cessation or termination of said Force Majeure Event and (iv) provided, however, that Purchaser shall not be excused from making any payments and paying any unpaid amounts due in respect of Solar Power delivered to Purchaser prior to the Force Majeure Event performance interruption.

11.3 Termination as a Consequence of Force Majeure Event

If a Force Majeure Event shall have occurred that has affected the Power Producer's performance of its obligations hereunder and that has continued for a continuous period of one hundred eighty (180) days, then Purchaser shill be entitled to terminate the Agreement and if such Force Majeure Event continues for further ninety (90) days period, the Agreement shall automatically terminate. Upon such termination for a Force Majeure Event, neither Party shall have any liability to the other (other than any such liabilities that have accrued prior to such termination).

12 Default

12.1 Power Producer Defaults and Power Purchaser Remedies

(a) Power Producer Defaults: The following events shall be defaults with respect to the Power Producer (each, a "Power Producer Default").

(i) An Insolvency Event shall have occurred with respect to the Power Producer;

(ii) Failure to achieve Commissioning of the System within the period as per RFS document; and

(iii) The Power Producer breathes any material term of the Agreement and

(A) if such breach can be cured within sixty (60) days after Purchaser's written notice of such breach and the-Power Producer fails to cure the same; or (B) the Power Producer fails to commence and pursue a cure within such sixty (60) days period if a longer cure period is needed.

(b) Purchaser's Remedies:

(i) If a Power Producer Default described in Section 12.1 (a) has occurred and is continuing, in addition to other remedies expressly provided herein, and subject to Section 13, Purchaser shall have a right to deliver a notice of its intention to terminate this Agreement ("Purchaser Preliminary Default Notice"), which shall specify in reasonable detail, the circumstances giving rise to the issue of such notice. (ii) Upon the occurrence and continuation of Power Producer Default and the failure by the Power Producer to cure such default within the applicable cure period specified in this Article; the Purchaser shall be at liberty avail the services of any other firm / successful bidder.

(iii) Following the issue of Purchaser Preliminary Default Notice, it shall be the responsibility of the Parties to discuss as to what steps shall be taken with a view to mitigate the consequences of the relevant Power

Producer's Default having regard to all the circumstances: If the Power Producer Default is not cured within a period of sixty (60) days of the issue of Purchaser Preliminary Default Notice or any other such period mutually agreed upon by the Parties, the Purchaser shall have the right to terminate this Agreement by issuing a Purchaser Termination Notice.

(iv) Upon the delivery of the Purchaser Termination Notice, this Agreement shall stand terminated. The Power Producer shall have the liability tomake payment within sixty (60) days from the date of Purchaser Termination Notice towards compensation to Purchaser equivalent to the difference between the Tariff and the grid rate notified by the relevant Government Authority for that point in time multiplied by the estimated Solar Power generated for a period of two years following the termination, considered on normative capacity utilization factor.

(v) if the Power Producer fails to remove the System from the Premises within one month from me date of termination, the Purchaser shall be entitled to dispose of the System in any manner it deems fit.

(vi) The Power Purchaser may exercise any other remedy it may have at law or equity or under the Agreement. 12.2 Power Purchaser Defaults and Power Producer's Remedies

(a) Purchaser Default : The following events shall be defaults with respect to Purchaser (each, a "Purchaser Default")

(i) An Insolvency Event shall have occurred with respect to Purchaser;

(ii) Purchaser breaches any material term of the Agreement if (A) such breach can be cured within sixty (60) days after the Power Producer's notice of such breach and Purchaser fails to so Cure, or (B) Purchaser fails to commence and pursue said cure within such sixty (60) day period if a longer cure period is needed; and

(iii) Pin-chaser-fails to pay the Power Producer any undisputed amount or, if the amount is disputed, an amount based on average consumption of last three consecutive undisputed invoices to the Power Producer under Section 7.7 of this Agreement within sixty (60) days from the receipt of notice from the Power Producer of such past due amount

(b) Power Producer's Remedies: If a Purchaser Default described in Sections 12.2 (a) has occurred and is continuing, in addition to other remedies expressly provided herein, and subject to Section 13, the Power Producer shall be entitled to terminate this Agreement by serving a fifteen (15) days' notice and upon such termination, (A) the Power Producer shall be entitled to receive from Purchaser the Purchase Price. The Purchase Price payable shall be the Purchase Price Specified in Schedule III that falls on such date. Upon the payment of the Purchase Price, the Power Producer shall cause the title of the System to transfer to the Purchaser and (b) the Power Producer may exercise any other remedy it may have at law or equity or under the Agreement.

13 Limitations of Liability

13.1 Except as expressly provided herein, neither Party shall be liable to the other Party or its Indemnified Persons for any special, punitive, exemplary, indirect, or consequential

damages, losses or damages for lost revenue or lost profits, whether foreseeable or not, arising out of, or in connection with the Agreement.

13.2 Subject to the provisions of the Agreement, the Power Producer shall be solely responsible for the manner in which its obligations under this Agreement are to be performed. All employees and representatives of the Power Producer, or contractors engaged by the Power producer in connection with Power Producer and shall not be deemed to be employees, representatives, contractors of the Purchaser Nothing contained in the Agreement or in any agreement or contract executed by the Power Producer shall be construed to create any contractual relationship between any such employees, representatives or contractors and the Purchaser.

13.3 Notwithstanding any liability. or obligation that may arise under this Agreement, any loss, damage, liability, payment, obligation or expense which is insured or not or for which the Purchaser can claim compensation under any insurance policy, shall not be charged to or payable by the Purchaser

14 Assignment & Novation

14.1 Assignment: Notwithstanding anything contained herein, the Power Producer has the right to assign all or any of its rights under this Agreement (including rights over any assets hereunder), to any third party including, though not restricted to any lender, equipment lessor or other party("Assignment"), with the consent of the Power Purchaser. The Power Purchaser shall not unreasonably withhold such consent. In the event of such assignment, the Purchaser will be able to hold the Power Producer as well as the party to whom the benefits under this contract are assigned, to be jointly and severally responsible for performing the obligations under this contract. Further, in the event of assignment, the Purchaser agrees to make the payments due to the Power Producer under this agreement, directly to the assignee, upon receipt of such notice by the Power Producer. If the Power Producer were to sell the Solar Power Plant, then the new buyer(s) would need to abide by this Agreement. Further, the Power Producer reserves the right to assign whole or part of the assets to lenders/ leasing companies. Purchaser may assign its rights under this Agreement, without the prior consent of Power Producer, to an Affiliate or any successor in interest to Purchaser, whether by way of merger, reorganization or sale of assets (including any sale of a line of business). This Agreement shall insure to the benefit of and be binding upon Purchaser and

its successors or assigns. However, any such actions as intended by the Power Producer under Article 14.1 and Article 14.2 shall be binding on Power Purchaser ,if there are zero material inconsistencies present in the contract provisions during the time of assignment/novation, else, it shall be construed as default in contract and appropriate actions shall be taken as deemed fit.

14.2 Novation:

The Parties agree and acknowledge that the Power Producer may intend to novate the Agreement to a party, and has the right to transfer any or all of its rights and obligations under this Agreement to a party or any other third party ("New Party"), with the consent of the Power Purchaser. The Power Purchaser shall not unreasonably withhold such consent. Upon Novation, the New Party shall automatically and without any further action be entitled to all the same rights and assume the same obligations, under this Agreement, as if it were originally a party to this Agreement. Further, the Purchaser hereby agrees and undertakes that, promptly upon receiving a request from the Power Producer, the Purchaser shall execute such further writings, deeds and/or agreements and take all such further actions as may be necessary for effecting or implementing the transfer of any or all of the Power Producer's rights and/or obligations under this Agreement to the New Party. If the parties agree to do Novation then separate Novation agreement shall be executed.

14.3 Notices

Unless otherwise provided in the Agreement, all notices and communications concerning the Agreement shall be in writing and addressed to the Parties at the addresses set forth below:

Power Producer's address and contact details:

Purchasers address and contact details:

Unless otherwise provided herein, any notice provided for in the Agreement shall be hand delivered, sent by registered post, or by courier delivery, or transmitted by facsimile and shall be deemed delivered to the addressee or its office when received at the address for notice specified above when hand delivered or sent toy courier delivered or sent by courier delivery, upon posting if sent by registered post and upon confirmation of sending when sent by facsimile on the next Business Day.

15. Confidentiality

15.1 Confidentiality obligation

(a) If the Power Producer provides confidential information, including business plans, strategies, financial information, proprietary, patented, licensed, copy righted or trademarked information, and / or technical information regarding the , design, operation and maintenance of the System ("Confidential Information") to Purchaser or, if in the course of performing under the Agreement or negotiating the Agreement Purchaser learns Confidential Information regarding the facilities or plans of the Power Producer, Purchaser shall (a) protect the Confidential information from disclosure to third parties with the same degree of care accorded its own confidential and proprietary information, and (b) refrain from using such Confidential Information, except in the negotiation and performance of the Agreement. Notwithstanding the above, Purchaser may provide such Confidential Information to its officers, directors, manager, employees and Affiliates (collectively "Representatives"), in each case whose access is reasonably necessary for purposes of the Agreement. Each such recipient of confidential information shall be informed by Purchaser of its confidential nature and shall be directed to treat such information confidentially and shall agree to abide by these provisions. Purchaser shall be liable for any breach of this provision by any entity to whom it improperly discloses Confidential Information. All Confidential Information shall remain the property of the Power Producer and shall be returned to. It after Purchaser's need for it has expired or upon the request of the Power Producer.

(b) If the Purchaser provides confidential information, including business plans, strategies, financial information, proprietary, patented, licensed, copy righted or trademarked information, ("Confidential Information") to the Power Producer or, if in the course of

performing under the Agreement or negotiating the Agreement the Power Producer learns Confidential Information regarding the facilities Or plans of the Purchaser, the Power Producer shall (a) protect the Confidential Information from disclosure to third parties with the same degree of care accorded its own confidential and proprietary information, and (b) refrain from using such Confidential Information, except in the negotiation and performance of the Agreement. Notwithstanding the above, the Power Producer may provide such Confidential Information to its officers, directors, managers, employees and Affiliates (collectively, "Representatives"), in each case whose access is reasonably necessary for purposes of the Agreement. Each such recipient of Confidential. Information shall be informed by the Power Producer of its confidential nature and shall be directed to treat such information confidentially and shall agree to abide by these provisions. The Power Producer shall be liable for any breach, of this provision by any entity to whom it improperly discloses Confidential Information. All Confidential Information shall remain the property of the Purchaser and shall be returned to it after the Power Produce's need for it has expired or upon the request of the Purchaser.

15.2 Permitted Disclosures

Notwithstanding any other provision contained herein, neither Party shall be required to hold confidential any information that;

(a) Becomes publicity available other than through the receiving Party.

(b) Is required to be disclosed under Applicable Law or pursuant to a validity issued notice or required filling, but a receiving Party subject to any such requirement shall promptly notify the disclosing Party of such requirement

(c) Is independently developed by the receiving Party ; or

(d) Becomes available to the receiving Party without restriction from a third party under no obligation of confidentiality..

16 Indemnity

16.1 Power Producer's Indemnity

Subject to Section 13, the Power Producer agrees that it shall indemnify and hold harmless Purchaser and its members, officers, employees, students, casual laborers, persons permitted' to run any business or service, such as canteens, stores, photocopy units, banks, 29 | P a g e

post office, courier service, hospital and to any lawful visitors (collectively, the "Purchaser Indemnified Parties") from and against any and all Losses incurred by the Purchaser Indemnified Parties to the extent arising from or out of the following any claim for or arising out of any injury to Or death of any Person or Loss or damage to Property of any Person to the extent arising out of the Power Producer's negligence or willful misconduct. The Power Producer shall not, however, be required to reimburse or indemnify any Purchaser Indemnified Party for any Loss to the extent such Loss is due to the negligence or willful misconduct of any Purchaser indemnified Party.

16.2 Purchaser's Indemnity

Subject to Section 13, Purchaser agrees that it shall indemnity, defend and hold harmless the Power Producer, its permitted successors and assigns and their respective directors, officers, employees, contractors, sub-contractors, and agents (collectively, the "Power Producer indemnified Parties") from and against any and all Losses incurred by the Power Producer Indemnified Parties to the extent arising from or out of any claim for or arising out of any injury to or death or any Person or loss or cl6mage to property of any Person to the extent arising out of Purchaser's negligence or willful misconduct. Purchaser shall not, however, be required to reimburse or indemnify any rowel Producer Indemnified Party for any Loss to the extent such Loss is due to the negligence or willful misconduct of any Power Producer Indemnified

Party.

17. Miscellaneous

17.1 Amendments

This Agreement may only be amended, modified or, supplemented by an instrument in writing executed by duly authorized representatives of the Power Producer and Purchaser. 17.2 Goodwill and Publicity

Neither Party shall use any name, trade name, service mark or trademark of the other Party in any promotional or advertising material without the prior written consent of such other Party. The Parties shall coordinate arid cooperate with each other when making public announcements related to the execution and existence of this Agreement, and each Party shall have the right to promptly review, comment upon and approve any publicity materials, press releases and other public statements by the other Party that refer to, or that describe any aspect of, this Agreement; provided that no such publicity releases or other public statements (except for fillings or other statements or releases as may be required by applicable law) shall be made by either Party without the prior written consent of the other Party. Without limiting the generality of the foregoing and whether or not the consent of the other Party is required or obtained, all public statements must accurately reflect the rights and obligations of the Parties under this Agreement.

17.3 Industry Standards

Except as otherwise set forth herein, for the purpose of the Agreement, the normal standards of performance within the solar photovoltaic power generation Industry in the relevant market shall be the, measure of whether a Party's performance is reasonable and timely. Unless expressly defined herein, words having well-known technical or trade meanings shall be so construed.

17.4 Cumulative Remedies

Except as set forth to the contrary herein, any right or remedy of the Power Producer or Purchaser shall be cumulative and without prejudice to any other right or remedy.

17.5 No Waiver

The failure of the Power Producer or Purchaser to enforce any of the provisions of the Agreement, or the waiver thereof, shall not be construed as a general waiver or relinquishment on its part of any such provision in any other instance or of any other provision in any instance.

17.6 Survival

The obligations under Section 8:1 (d) (Power Producer Covenant), Sections 8.3(d), (e), (f) and (g) (Purchaser Covenants), Section 10 (Taxes and Governmental Fees), Section 13 (Limitation of Liability) Section 12.2 (Notices), Section 15 (Confidentiality), or pursuant to other provisions of this Agreement that, by their nature and context, are intended to survive. Termination of this Agreement shall survive the expiration or termination of this Agreement for any reason.

17.7 Governing Law & Jurisdiction

(a) This Agreement shall be governed by and construed in accordance with the laws of India. The Parties agree that the courts in Delhi shall have jurisdiction over any action or proceeding arising under the Agreement.

(b) In the event of any Dispute, difference of opinion or dispute or claim arising out of or relating to this Agreement or breach, termination or the invalidity thereof, shall firstly be attempted to be resolved by conciliation. Any Dispute that cannot be settled through conciliation procedure shall be referred to arbitration in accordance with the procedure given below. The Parties agree to comply with the awards resulting from arbitration and waive their rights to any form of appeal insofar as such waiver can validly be made. Cost of conciliation shall be equally shared by both the parties.

(c) Arbitration Procedure:

Settlement of Dispute :

If any dispute of any kind whatsoever arises between Purchaser and Power Producer in connection with or arising out of the contract including without prejudice to the generality of the foregoing, any question regarding the existence, validity or termination, the parties shall seek to resolve any such dispute or difference by mutual consent.

If the parties fail to resolve, such a dispute or difference by mutual consent, within 45 days of its arising, then the dispute shall be referred by either party by giving notice to the other party in writing of its intention to refer to arbitration as hereafter provided regarding matter under dispute. No arbitration proceedings will commence unless such notice is given. Any dispute in respect of which a notice of intention to commence arbitration has been given in accordance with Sub Clause , shall be finally settled by arbitration.

In case the Contractor (Power Producer) is a Public Sector Enterprise or a Government Department: In case the Contractor is a Public Sector Enterprise or a Government Department, the dispute shall be referred for resolution in Permanent Machinery for Arbitration (PMA) of the Department of Public Enterprise, Government of India. Such dispute or difference shall be referred by either party for Arbitration to the sole Arbitrator in the Department of Public Enterprises to be nominated by the Secretary to the Government of India incharge of the Department of Public Enterprises. The award of the Arbitrator shall be binding upon the parties to the dispute, provided, however, any party aggrieved by such award may make a further reference for setting aside or revision of the award to the Law Secretary, Department of Legal Affairs, Ministry of Law & Justice, Government of India. Upon such reference the dispute shall be decided by the Law Secretary or the Special Secretary / Additional Secretary, when so authorized by the Law Secretary, whose decision shall bind the Parties finally and conclusive. The Parties to the dispute will share equally the cost of arbitration as intimated by the Arbitrator.

In All Other Cases In all other cases, any dispute submitted by a party to arbitration shall be heard by an arbitration panel composed of three arbitrators, in accordance with the provisions set forth below.

The Purchaser and the Power Producer shall each appoint one arbitrator, and these two arbitrators shall jointly appoint a third arbitrator, who shall chair the arbitration panel. If the two arbitrators do not succeed in appointing a third arbitrator within Thirty (30) days after the later of the two arbitrators has been appointed, the third arbitrator shall, at the request of either party, be appointed by the Appointing Authority for third arbitrator which shall be the President, Institution of Engineers.

If one party fails to appoint its arbitrator within thirty (30) days after the other party has named its arbitrator, the party which has named an arbitrator may request the Appointing Authority to appoint the second arbitrator.

If for any reason an arbitrator is unable to perform its function, the mandate of the Arbitrator shall terminate in accordance with the provisions of applicable laws as mentioned in Clause

_____ (Governing Law) and a substitute shall be appointed in the same manner as the original arbitrator.

Arbitration proceedings shall be conducted with The Arbitration and Conciliation Act, 1996. The venue or arbitration shall be New Delhi.

The decision of a majority of the arbitrators (or of the third arbitrator chairing the arbitration panel, if there is no such majority) shall be final and binding and shall be enforceable in any court of competent jurisdiction as decree of the court. The parties thereby waive any objections to or claims of immunity from such enforcement. The arbitrator(s) shall give reasoned award. Notwithstanding any reference to the arbitration herein, the parties shall continue to perform their respective obligations under the agreement unless they otherwise agree. Cost of arbitration shall be equally shared between the Power Producer and Purchaser.

17.8 Severability

If any term, covenant or condition in the Agreement shall, to arty extent, be invalid or unenforceable in any respect under Applicable Law, the remainder of the Agreement shall not be affected thereby, and each term, covenant or condition of the Agreement shall be valid and enforceable to the fullest extent permitted by Applicable Law and if appropriate, such invalid or unenforceable provision shall be modified or replaced to give effect to the underlying intent of the Parties and to the intended economic benefits of the Parties.

17.9 Successors and Assigns

This Agreement and the rights and obligations under the Agreement shall be binding upon and shall inure to the benefit of the Power Producer and Purchaser and their respective successors and permitted assigns.

17.10 Counterparts

This Agreement may be executed in one or more counterparts, all of which taken together shall constitute one and the same instrument.

17.11 Independent Service Provider

This Agreement is on a principal to principal basis between the parties hereto Nothing contained in tills Agreement shall be construed or deemed to create any association, partnership or joint venture or employer employee relationship or principal-agent relationship in any manner whatsoever between the Parties

17.12 Non-Exclusive Agreement

This Agreement is on non-exclusive basis. The Purchaser is free to engage any other service provider's or may entrust services similar to those provided by the Power Producer under this Agreement to any other person/s, provided the Power Producer's rights under this Agreement are not affected in any manner

17.13 Entire Agreement

This Agreement constitutes the entire agreement between the Parties hereto with respect of the subject matter of this Agreement and supersedes all prior agreements and undertakings; written or oral, with respect to the subject matter hereof except as otherwise expressly provided herein. The Schedules annexed to this Agreement also form a part of this Agreement.

17.14 Insurance

The Power Producer shall maintain at its own costs, throughout the tenure of this Agreement and any extensions thereof all mandatory insurance coverage for adequate amount including but not restricted to comprehensive general liability insurance including theft and vandalism, covering the System and accidental losses, bodily harm, injury, death of all individuals employed/ assigned by the Power Producer to perform the services required under this Agreement.

17.15 Annual accounts reconciliation shall be conducted annually between the Power Producer and Purchaser.

17.16 Entire Agreement

THE PRESIDENT OF INDIA

This Agreement constitutes the entire agreement between the Parties hereto with respect to the subject matter of this Agreement and supersedes all prior agreements and undertakings, written or oral, with respect to the subject matter hereof except as otherwise expressly provided herein. The Schedules annexed to this Agreement also form a part of this Agreement.

IN WITNNESS WHEREOF the Parties have caused the Agreement to be duly executed through their duly authorized representatives as of the date set forth above.

FOR & ON BEHALF OF FOR & ON BEHALF OF

| Signature: |
|---------------|
| Name: |
| Designation: |
| WITNESSES |
| 1) Signature: |
| |

35 | Page

| Name: | Name: |
|---------------|--------------|
| Designation: | Designation: |
| 2) Signature: | 2) Signature |
| Name: | Name |
| Designation: | Designation |
| | |

SCHEDULE-I

Description of the Premises:

| Premises Overview | |
|--------------------|-------------------------------|
| Segment | |
| Type of Roof | |
| Hours of Operation | |
| Security | |
| | |
| | Capacity and Area Requirement |
| Conocity | kWp |
| Capacity | ĸvvp |
| Module Area | sqrmtr (PV System) |
| Number of Building | |
| Roofs | |
| Capacity | |

SCHEDULE II

FEES

Following are the details of the tariff agreed between the parties.

Purchase shall pay power producer for solar power at the rate equal to INR ____/kWh which is the levelized tariff as per <NAME OF EXPERT PSU> allocation. Year wise tariff is as follows;

| | 1 |
|---------|---|
| Year 1 | |
| Year 2 | |
| Year 3 | |
| Year 4 | |
| Year 5 | |
| Year 6 | |
| Year 7 | |
| Year 8 | |
| Year 9 | |
| Year 10 | |

38 | Page

| Year 11 | |
|---------|--|
| | |
| | |
| Year 12 | |
| | |
| | |
| Year 13 | |
| | |
| | |
| Year 14 | |
| | |
| | |
| Year 15 | |
| | |
| | |
| Year 16 | |
| | |
| | |
| Year 17 | |
| | |
| | |
| Year 18 | |
| | |
| | |
| | |
| Year 19 | |
| | |
| | |
| Year 20 | |
| | |
| | |
| Year 21 | |
| | |
| | |
| Year 22 | |
| | |
| | |
| Year 23 | |
| | |
| | |
| Year 24 | |
| | |
| | |
| Year 25 | |
| | |
| | |
| | |

The fees and payment details are provided in detail under clause 7 of this agreement.

SCHEDULE III

The following is the purchase value of the system over a period of 25 years. This may be applicable under the following conditions.

1. The Power Purchaser terminates the PPA before the 25 years PPA Tenure

2. The Power Purchaser wishes to own the Project before the Tenure of the PPA.

3. The Solar Project is relocated or shifted owing to demolition of the Building, damage to the building, change of city plans or any other mason.

The Price reference taken for calculating the total cost of the system is as per RFS rates for the state of ______ has used the CERC guidelines for arriving at the Project cost.

| Year of Term | Calendar Year | Purchase Price in Crores (Rs) |
|----------------|---------------|--------------------------------|
| (End of Year) | | |
| 1 | | |
| 2 | | |
| 3 | | |
| 4 | | |
| 5 | | |
| 6 | | |
| 7 | | |
| 8 | | |
| 9 | | |
| 10 | | |
| 11 | | |
| 12 | | |
| 13 | | |
| 14 | | |
| 15 | | |

40 | Page

| 16 | |
|----|--|
| 17 | |
| 18 | |
| 19 | |
| 20 | |
| 21 | |
| 22 | |
| 23 | |
| 24 | |
| 25 | |

SCHEDULE IV

| | Project : | kW | p Solar On-Grid Power P | roject |
|--------|--------------|---------------|-------------------------|--------------------|
| | Location : | New Delhi | | |
| | | | Design Criteria | |
| | Expected Yea | rly Energy Ge | neration Sheet | |
| End of | Yearly D | egradation | Global incident in | Energy injected |
| Year | 'MWh' (N | Modules & | coll. Plane (GlobInc) | into grid (E_Grid) |
| | Sys | tem) | 'kWh/Sq.mtr' Yearly | 'MWh' Yearly 'A' |

| Client : | Degradation consider in PV | |
|----------|----------------------------|--|
| | system generation data | |
| 1. | 1.0% | |
| 2. | 1.0% | |
| 3. | 1.0% | |
| 4. | 1.0% | |
| 5. | 1.0% | |
| 6. | 1.0% | |
| 7. | 1.0% | |
| 8. | 1.0% | |
| 9. | 1.0% | |

| 1.0% | | |
|------|---|--|
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| 1.0% | | |
| | 1.0% 1.0% | 1.0% 1.0% |

SCHEDULE V

Government approvals

1. To be obtained by the power producer

All approvals including approvals/consents required under local regulations, building codes and approvals required from the distribution utility etc. relating to installation and operation of the system (including the government incentives/subsidies available for the project) and generation and supply of solar power from the Project.

2. To be obtained by Power Purchaser

Any authorizations required of purchaser, including those required for installation of system at the premises. Permissions and coordination with DISCOM or any related organization for NET METERING. Power Purchaser will apply for net metering and bear the cost of net meter only.

| Purchase Price for kWp system @ Rs/- Watt. | | |
|--|--|--|
| Year Salvage Value (Rs.) | | |
| 1st year | | |
| | | |
| 2nd year | | |

REMAINDER OF THE PAGE IS INTENTIONALLY LEFT BLANK

| 3rd year | |
|------------|--|
| | |
| 4th year | |
| | |
| | |
| 5th year | |
| | |
| 6th year | |
| | |
| 7th year | |
| | |
| The summer | |
| Sth year | |
| | |
| 9th year | |
| | |
| 10th year | |
| | |
| 11th year | |
| 11th year | |
| | |
| 12th year | |
| | |
| 13th year | |
| | |
| 14th year | |
| 1712 3 641 | |
| | |
| 15th year | |
| | |
| lóth year | |
| - | |
| | |

| 12th year | |
|-----------|--|
| | |
| 13th year | |
| | |
| 14th year | |
| | |
| 15th year | |
| | |
| lóth year | |
| | |
| 17th year | |
| | |
| 18th year | |
| | |
| 19th year | |
| | |
| 20th year | |
| | |
| 21th year | |
| | |
| 23rd year | |
| | |
| 24th year | |
| | |
| 25th year | |
| | |

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

VOLUME 3

SPECIAL CONDITIONS OF CONTRACTS (SCC) & Technical Specifications

1.0 SCOPE OF WORK

- 1.1 Under this RfS, the Roof top Developer (RTD) shall be required to set up Grid-Connected Solar Roof Top Projects at various Government Institutions attached in the RfS in RESCO Mode, with the primary objective of supplying solar power to Govt. Client Organization, as per the provisions of the RfS and standard Power Purchase Agreement (PPA).
- 1.2 The Projects shall be set up under the RESCO Mode as defined by MNRE and in accordance to the PM Surya Ghar Muft Bijli Yojna Scheme for Government buildings. The scope of Work of the RTD, inter alia, includes all design & engineering, procurement & supply of equipment and materials, testing at manufacturers works, multi level inspections, packing and forwarding, supply, receipt, unloading and storage at site, associated civil works, services, permits, licenses, installation and incidentals, erection, testing, commissioning, Net Metering, obtaining No Objection Certificate (NOC) for grid connectivity (if required) and synchronization with existing Distribution Network Grid for individual Grid Interactive Solar Roof top PV Power projects at various locations.
- 1.3 Obtaining Net-metering and grid connectivity of the Project under this tender would be the responsibility of the bidder in accordance with the prevailing guidelines of the respective State Regulators (JERC/SERC) or Central Electricity Authority (CEA) or the DISCOM in the respective area. The Client could facilitate connectivity and necessary approvals; however, the entire responsibility lies with the RTD only.
- 1.4 The RTD shall take all necessary permits, approvals and licenses, insurance etc., provide training and such other items and services required to complete the scope of work mentioned above.
- 1.5 The RTD shall abide by all Statutory Laws, By Laws, Labour Laws, Factory laws, etc.
- 1.6 The detailed technical specifications & standards for grid-connected solar PV systems/power plants is enclosed in Annexure-1

2.0 SOLAR MODULES AND CONNECTIVITY WITH THE GRID:

2.1 The cells and modules used in the Project shall be sourced only from the models and manufacturers included in the "Approved List of Models and Manufacturers" as published by MNRE and updated as on the date of commissioning of the Project.

- 2.2 Further, the provisions as contained in the O.M. dated 10.03.2021 issued by MNRE on the subject "Approved Models and Manufacturers of Solar Photovoltaic Modules (Requirement of Compulsory Registration) Order, 2019-Implementation-Reg." and its subsequent amendments and clarifications issued until the bid submission deadline, shall be applicable for this RfS. The modules used in the Project under this RfS should have been included in the List-I under the above Order, valid as on the date of invoicing of such modules.
- 2.3 The bidders are advised to take cognizance of the O.M. dated 09.03.2021 issued by MNRE, on the subject of "Imposition of Basic Customs Duty (BCD) on Solar Cells & Modules/Panels", while preparation of their response to this RfS. The above O.M. and its associated orders issued by the Ministry of Finance will be applicable on this tender and imposition of taxes/duties as laid out in the above OM, will not be considered under "Change in Law" under the PPA.

3.0 Connectivity with the Grid

- 3.1 The Project should be designed for interconnection with the grid in accordance with prevailing SERC/JERC/DISCOM regulations in this regard. For interconnection with the grid and metering, the RTD shall abide by applicable Grid Code, Grid Connectivity Standards, Regulations on Communication System for transmission of electricity and other regulations/procedures (as amended from time to time) issued by Appropriate Commissions and Central Electricity Authority (CEA), and any other regulations of the concerned DISCOM where the Project is located. The maximum capacity for interconnection with the grid at a specific voltage level shall be at 415V/ 6.6Kv/11 Kv/33Kv as per Site requirement in line with CERC regulation.
- 3.2 The responsibility of getting connectivity and necessary approvals for grid connection and net-metering shall entirely be with the RTD and shall be at the risk and cost of the RTD. The transmission of power up to, and including at the point(s) of interconnection where metering is done for energy accounting, shall be the responsibility of the RTD at its own cost. The maintenance of transmission system up to and including at the Interconnection Point shall be responsibility of the RTD, to be undertaken entirely at its risk and cost.
- 3.3 Metering arrangement of each Project shall have to be adhered to in line with relevant clauses of the PPA. The entire cost of transmission (if applicable) including cost of construction of line, wheeling charges, Scheduling charges, SOC, MOC, maintenance, losses etc. and any other charges from the Project up to and including at the Interconnection Point will be borne by the RTD.
- 3.4 **Note:** The RTD shall apply for Net-metering approval with the respective DISCOM within 90 days of signing of PPA, failing which, the PPA will be liable to be terminated by the Client Organization, along with encashment of PBG for the respective Project.

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

4.0 CAPACITY UTILIZATION FACTOR (CUF)

- 4.1 The Bidders will declare the annual CUF of the Projects at the time of submission of response to RfS, and the RTD will be allowed to revise the same once within first year of COD. Thereafter, the CUF for the Project shall remain unchanged for the entire term of the PPA. The declared annual CUF shall in no case be less than 14%.
- 4.2 RTD shall maintain generation so as to achieve annual CUF within (+) 10% and (-) 15% of the declared value till the end of 10 years from COD, subject to the annual CUF remaining minimum of 14%, and within (+10%) and (-20%) of the declared value of the annual CUF thereafter till the end of the PPA duration of 25 years. The lower limit will, however, be relaxed by NHPC REL to the extent of non-availability of grid for evacuation which is beyond the control of the RTD. For the first year of operation of the project, the annual CUF shall be calculated based on the first full / part year after CSD of the Project as per terms of PPA of the contracted capacity. Subsequently, the annual CUF will be calculated every year from 1st April of the year to 31st March next year.
- 4.3 Relaxation to the non- availability of grid beyond the control of RTD need to be certified by Govt. Institution/Client. Further, duly signed hindrance register shall be maintained during the time of execution and O&M.

An Illustration to this effect is given below:-

Assuming that the actual date of supply of power of the contracted capacity subsequent to COD of Solar project is 01.10.2025, then the CUF for the part year commencing from 01.10.2025 till 31.03.2026 shall be calculated based on the actual generation metered at the delivery point during the period from 01.10.2025 till 31.03.2026 w.r.t. the contracted capacity. i.e. If 'X' MWh of energy has been metered at the Delivery Point during the period from 01.10.2025 till 31.03.2026 for 'Y' MW of contracted capacity,

CUF = (X MWh/(Y MW*4368)) x100%; where,

[4368=No. of days from 01.10.2025 to 31.03.2026 (182) x No. of Hours in a day (24)]

For the subsequent years commencing from 01.04.2026, the CUF shall be calculated annually based on the following formula:

If 'X' MWh of energy has been metered at the Delivery Point during the period from 1st April of the year to 31st March next year for 'Y' MW of contracted capacity, CUF= (X MWh/(Y MW*8766)) x100%;

5.0 Shortfall in Energy Supply:

Subsequent to Project commissioning, if for any Contract Year, except for the first year after Project commissioning, it is found that the RTD has not been able to supply minimum energy corresponding to the value of annual CUF

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding within the permissible lower limit of CUF declared by the RTD, the RTD shall be levied liquidated damages @50% of the PPA tariff for the shortfall in energy terms, in accordance with terms of the PPA. This compensation shall be applied to the amount of shortfall in energy supply during the year. However, this compensation shall not be applicable in events of Force Majeure identified under PPA affecting supply of power by RTD.

If for any Contract Year, it is found that the RTD has not been able to supply minimum energy corresponding to the value of annual CUF within the permissible lower limit of CUF declared by the RTD, on account of reasons primarily attributable to the RTD, such shortfall shall be dealt as per the applicable provisions of the PPA. However, the shortfall in power supply will not be applicable in events of Force Majeure identified under the PPA, affecting supply of power by the RTD.

The performance criteria as above shall not be applicable for the Contract Year ending on 31st March immediately after the actual Commencement of Supply Date (CSD) of the contracted capacity. For example, if the actual date of supply of power of the contracted capacity subsequent to COD of RE project is 01.10.2025, then the performance criteria stipulated above shall not be applicable for the part year commencing from 01.10.2025 till 31.03.2026 and the same will be effective w.e.f. 01.04.2026.

The detailed list of documents required for verification of energy supply and performance of the Projects will be intimated to the Developers subsequent to commissioning and commencement of supply of power. For each Contract Year, the above data will be required to be submitted by the respective Developers to JAKEDA/its associated departments within 30 days after expiry of the previous Contract Year, for verification of the performance parameters for calculating applicable compensation on account of shortfall.

The reference to the penalty for shortfall to enable NHPC REL to remit the amount to buying utility(ies) and the amount being equal to the damages payable by the buying utility(ies) for not meeting RPO is only a measure of damage. It shall not be construed that the penalty is payable by RTD only if the buying utility(ies) are required to pay damage for not meeting of RPO. Buying utility(ies) shall not be required to prove or establish such payment of damage for not meeting the RPO. For avoidance of doubt, it is clarified here that the above penalty is payable by RTD irrespective of whether the buying utility(ies) are required to pay damage for not meeting the RPO.

6.0 Technical Parameters / Specification

Ministry of New & Renewable Energy (MNRE) vide its Office Memorandum No. 318/17/2024- Grid Connected Rooftop (part-2) dated 03.07.2024 issued Guidelines for Saturation of Government Buildings under PM Surya Ghar: Muft Bijli Yojana. Under the scheme, all Government rooftops under the administrative control of Central Government Ministries/Departments, including autonomous bodies, subordinate offices etc. shall be saturated with rooftop solar to the extent that is technically feasible by 31st December, 2025.

The provisions as contained in the O.M. dated 10.03.2021 issued by MNRE on the subject "Approved Models and Manufacturers of Solar Photovoltaic Modules (Requirement of Compulsory Registration) Order, 2019-Implementation-Reg." and its subsequent amendments and clarifications issued until the bid submission deadline, shall be applicable for this RfS. The modules used in the Project under this RfS should have been included in the List-I under the above Order, valid as on the date of invoicing of such modules.

Detailed Technical Specifications are attached at Annexure-1.

7.0 Commissioning of Projects

Commissioning of the Project shall be carried out by the RTD in line with the procedure as per the PPA document. The Client Organization may authorize any individual or committee or organization to witness and validate the commissioning procedure on site. Commissioning certificates shall be issued by the DISCOM/Client Organization after successful Project commissioning. NHPC REL authorized representative may also be allowed for the inspection/ commissioning if required.

For the issuance of commissioning/completion certificate, the following documents will be deemed to form the completion documents:

- a. At least 15 days prior intimation to JAKEDA and client for witnessing the project inspection/performance.
- b. Detailed checklist for inspection of Roof top SPV power plants as per JAKEDA format. (Annexure-B)
- c. Project completion/ satisfaction certificate from roof top owners/client organization. (Annexure-C).

8.0 Commissioning Schedule and Liquidated Damages Not Amounting to Penalty for Delay in Commissioning

- 8.1 The Scheduled Commissioning Date (SCD) for commissioning of the full capacity of the Project shall be the date as on 09 months from the Effective Date of the PPA.
- 8.2 The maximum time period allowed for commissioning of the full Project Capacity with applicable liquidated damages, shall be limited to the date as on 3 months from the SCD or the extended SCD (if applicable).
- 8.3 In case of delay in commissioning of the Project beyond the SCD until the date as per Clause 8.2 above, as part of the liquidated damages, the total PBG amount for the Project shall be encashed on per-day-basis and proportionate to the balance capacity not commissioned.

For example, in case of a Project of 2 MW capacity, if commissioning of 1 MW capacity is delayed by 18 days beyond the SCD, then the liquidated damages

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding shall be: **PBG amount x (1/2) x (18/90)**. For the purpose of calculations of the liquidated damages, 'month' shall be considered consisting of 30 days.

8.4 In case Commissioning of the Project is delayed beyond the date as per Clause 8.2 above, the PPA capacity shall stand reduced/amended to the Project Capacity commissioned and the PPA for the balance capacity will stand terminated. The total capping of maximum LD may be kept as 5% of the Contract value corresponding to the capacity not commissioned.

9.0 OPERATION & MAINTENANCE (O&M) GUIDELINES TO BE MANDATORILY FOLLOWED BY THE RTD

- 9.1 The RTD shall be responsible for all the required activities for successful O&M of the Rooftop Solar PV system for the PPA term from the date of commissioning of the Project.
- 9.2 Following guidelines shall be followed in this regard. In addition, O & M practices shall be strictly followed
 - 9.2.1 O&M of the Project shall be compliant with grid requirements to achieve committed energy generation.
 - 9.2.2 Deputation of qualified and experienced engineer/ technicians till the O&M period at project site.
 - 9.2.3 Periodic cleaning of solar modules. The modules shall be cleaned with a periodic interval of 15 days or as and when required as per actual site conditions. It's the responsibility of the RTD to get the modules cleaned during O&M Period.
 - 9.2.4 A "hindrance register" shall be maintained to record any issues affecting Project O&M. The same shall be duly signed by both parties of the PPA.
 - 9.2.5 Periodic checks of the Modules, PCUs and BoS shall be carried out as a part of routine, preventive and breakdown maintenance.
 - 9.2.6 Immediate replacement of defective Modules, Invertors/PCUs and other equipment as and when required.
 - 9.2.7 Supply of all spares, consumables and fixtures as required. Such stock shall be maintained for all associated equipment's and materials as per manufacturer's / supplier's recommendations.
 - 9.2.8 All the testing instruments required for Testing, Commissioning and O&M for the healthy operation of the Plant shall be maintained by the RTD. The testing equipment's must be calibrated once in a year from NABL accredited labs and the certificate of calibration must be kept for reference as required.

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

- 9.2.9 If negligence/ mal-operation on part of the RTD's operator results in failure of equipment, such equipment should be repaired/ replaced by the RTD free of cost.
- 9.2.10 Co-ordination with Owner / DISCOM / CEIG as per the requirement for Joint Meter Reading (JMR) Report. The person-in-charge present at site from the RTD's side shall take a joint meter reading in the presence of rooftop owner on a monthly basis.
- 9.2.11 Online Performance Monitoring, controlling, troubleshooting, maintaining of logs & records. A maintenance record register is to be maintained by the operator with effect from Commissioning to record the daily generation, regular maintenance work carried out as well as any preventive and breakdown maintenance along with the date of maintenance, reasons for the breakdown, duration of the breakdown, steps taken to attend the breakdown, etc.
- 9.2.12 For any issues related to operation & maintenance, a toll-free number shall be made available to the rooftop owner to resolve the same within 72 hours.
- 9.2.13 If any jobs covered in O&M Scope as per RfS are not carried out by the RTD during the O&M period, the Client Organization shall take appropriate action as deemed fit. NHPC REL reserves the right to make surprise checks/ inspection visits at its own or through authorized representative to verify the O&M activities being carried out by the RTD. Failure to adhere to above guidelines may result in penal action including debarring from participation in next tender.
- 9.2.14 It is up to RTD and Client to mutually decide on the source and amount of water to be supplied for module cleaning during O&M period.

10.0 Clearances Required from the State Government and Other Local Bodies

Obtaining all clearances, permits, licenses including arrangement of land and connectivity to the Grid and access (if applicable) prior to scheduled date of commencement of supply of power shall be the responsibility of the Generator and the Procurer shall not be responsible in case of delay in obtaining such clearances, permits, licenses etc. The Solar Power Generators are required to obtain necessary clearances and permits as required for setting up the Solar Power Projects, including but not limited to the following:

- (a) No Objection (NOC)/Environmental clearance (if applicable) for the Project.
- (b) Forest Clearance (if applicable) for the land for the Project.
- (c) Approval for water from the concerned authority (if applicable) required for the Project.
- (d) Any other clearances as may be legally required, in order to establish and operate the Project

The above clearances, as applicable for the Project, will be required to be submitted to NHPC REL prior to commissioning of the Project, if sought by NHPC REL. In case of any of the clearances as indicated above being not applicable for the said Project, the RTD shall submit an undertaking in this regard, and it shall be deemed that the RTD has obtained all the necessary clearances for establishing and operating the Project. Any consequences contrary to the above shall be the responsibility of the RTD. The RTD shall also comply with all the laws, regulations, orders and procedures issued by the appropriate authority, applicable for setting up and implementing the Project. The RTD shall be required to follow the applicable rules regarding project registration with the State Nodal Agency in line with the provisions of the applicable policies/regulations of the State where the Projects are being located. It shall be the responsibility of the RTD to remain updated about the applicable charges payable to the SNA under the respective State Solar Policy.

Note: The RTD should apply for all the necessary approvals, permits and clearances not more than 90 days from the Effective Date of PPA (in respect of approvals, permits and clearances those are required for initially establishing the project), which shall be complete in all respects, incorporating the clarifications/changes as required by the concerned authorities. The above timeline shall be adhered to, in order to examine cases where the RTD faces delay in grant of the necessary approvals and permits, for a period substantially greater than the standard period of grant of approval by the respective organizations.

11.0 SERVICE / PROJECT MANAGEMENT (PMC) CHARGES:

- 11.1 The evaluated lowest bidder (L1) shall have to pay Charges @ INR 1485/kW (Indian Rupees One thousand four hundred eighty five only/ kW) + Applicable taxes, to NHPC/NHPC REL towards project management charges (PMC).
- 11.2 The Service Charges will be calculated on the total Project capacity for which PPAs are signed by the RTD. These charges will be payable in two installments as follows:
 - (i) 50% to be paid within 30 days of issuance of LOAs by NHPC/NHPC REL, and,
 - (ii) Remaining 50% to be paid to NHPC/NHPC REL prior to signing of PPA. In case of any reduction/addition in the total Project capacity for which PPAs are executed, the second installment of 50% will be adjusted suitably. No interest shall be payable by NHPC/NHPC REL to RTD in case of refund of excess Service/PMC charges deposited by the RTD. Also, the tax component of the excess Service/PMC charges deposited by the RTD, shall not be refunded.

The due date for payment of PMC Charges shall be the date as intimated by NHPC/NHPC REL, which shall be at the most seven (07) days prior to the signing of PPA. The payment has to be made by the RTD in the form of DD/ Pay Order / NEFT/ RTGS mode to NHPC REL.

Any delay in depositing the said amount to NHPC REL as mentioned above within the stipulated time shall attract delay charges @18% per annum + applicable GST, levied on per day basis, on the amount corresponding to the 1st installment (i.e. 50% of PMC Charges), until (and including) the date of payment of the said amount, which shall not be later than 90 days after issuance of LOA. In case of delay in submission of the 1st

- Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding installment beyond 210 days after issuance of LOA, the awarded capacity shall stand cancelled and BG against EMD will be encashed by NHPC/NHPC REL.
 - 11.3 The 2nd installment shall be paid prior to signing of PPA, and the PPA shall be signed only after deposit of the second installment to NHPC REL.

12.0 SIGNING OF POWER PURCHASE AGREEMENT (PPA)

12.1 Power Purchase Agreement shall be signed between JAKEDA/its associated departments and successful Roof top bidder. A copy of Power Purchase Agreement to be executed between JAKEDA/its associated departments and the Selected Bidder is available at forms provided in the section "Forms and Procedures". The PPA shall be executed within 90 (Ninety) days of the date of issue of Letter of Award, if not extended by JAKEDA/its associated departments.

Separate PPA will be executed between Government client /organization and RTD for each Rooftop plant/ Project site or as per Government client decides.

Subsequent extension in this timeline shall be finalized as mutually agreed by the Client Organization and the RTD. PPA will be executed between Client Organization and selected bidder separately for each Category. The PPA shall be valid for a period of 25 years from date of commissioning of the Project.

Bidder has to make the complete and careful examinations of feasibility of government client Roof top plant/capacity estimation as per the load study of buildings and respective regulation of net metering/net billing before signing of PPA with client.

Any extension of the PPA period beyond 25 years shall be through mutual Agreement between the RTD and Government client /organization.

- 12.2 **Capacity identification and confirmation:** Subsequent to issuance of LoAs by NHPC REL, RTD will be allowed to carry out site visits and due diligence of the buildings offered by the Client Organization, and assess the Project sizing and estimated capacity installation under the cumulative capacity awarded. Within this period, the individual Project size to be set up on each building and the cumulative Project capacity for which PPAs are to be signed, will be finalized by both parties.
- 12.3 The Effective Date of PPA shall be the actual date of signing of PPA.
- 12.4 The PPAs shall be valid for a period of 25 years from the date of Project Commissioning, i.e. the Commercial Operation Date, or COD of the Project. Any extension of the PPA term beyond this shall be carried out through mutual agreement between the RTD and the JAKEDA/its associated departments.

12.5 Limits on capacity modification:

- 12.5.1 Capacity enhancement is also allowed with submission of additional revised BG amount as calculated. Such capacity enhancement will require consent from the concerned Department.
- 12.5.2 All such modifications in the awarded capacity are required to be concluded within 45 days from issuance of LoAs, unless extended by the Beneficiary Organization.
- **Note:** (i) The Performance Bank Guarantee, shall be submitted by the RTD prior to signing of PPA. Before signing of PPA, NHPC REL will verify the shareholding of the Project

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding Company along with a copy of complete documentary evidence. If at this stage, it is found that the documents furnished by the RTDs are false/misleading or misrepresented in any way, then the provisions contained in this RfS will be applicable.

> (ii) In case any extension is given to the Project, the corresponding extension needs to be made in the validity of Perforance Security. The Perforance Security are required to be submitted in the name NHPC REL

> (iii) NHPC REL will issue necessary go-ahead for signing of PPA, upon verification of compliance of pre-requisites by the RTD as brought out above. The PBG along with SFMS of the issued PBG will be submitted to NHPC REL for verification.

(iv) JAKEDA shall facilitate the PPA signing process and will bear no responsibility whatsoever on account of non-availability of indicated buildings and/or non-execution of PPA for any particular building.

(v) The selected Bidder for the Project selected based on this RfS is required to sign PPA with JAKEDA/its associated departments within the timeline as stipulated in Clause 30 of the RfS. In case, JAKEDA/its associated departments offers to execute the PPA with the Selected Bidder and if the Selected Bidder does not submit the requisite documents as per the RfS, or does not meet eligibility criteria upon submission of documents or does not execute the PPA within the stipulated time period, then the Bank Guarantee equivalent to the amount of the EMD shall be encashed by NHPC REL from the Bank Guarantee available with NHPC REL (i.e. EMD or PBG) as liquidated damages not amounting to penalty, the selected Project shall stand cancelled and the selected Bidder expressly waives off its rights and objections, if any, in that respect.

(vi) In addition to the other remedies, this Performance Security can be encashed to recover any damages/dues of the Roof Top Developer in terms of the PPA.

(vii) The PBG of RTDs shall be returned to them, after successful commissioning of their projects as per Terms of PPA, after taking into account any liquidated damages due to delays in commissioning as per relevant Clauses of the RfS.

Annexure 1

TECHNICAL REQUIREMENTS FOR GRID CONNECTED SOLAR PV PROJECTS

TECHNICAL SPECIFICATIONS

1. Solar PV modules:

- i. Solar PV modules capacity to be used should adhere to the Approved List of Models and Manufacturers (ALMM) of Solar Photovoltaic Modules (Requirement for Compulsory Registration) Order 2019 - Implementation issued vide OM NO. 283/54/2018-GRID SOLAR -Part (I) Dated 10th March 2021 and subsequent amendments.
- ii. The PV modules used must qualify to the latest edition of IEC standards or equivalent BIS standards, i.e. IEC 61215/IS14286, IEC 61853-Part I/IS 16170-Part I, IEC 61730 Part-1 &Part 2 and IS 17210(part 1) or IEC 62804 (PID). For the PV modules to be used in a highly corrosive atmosphere throughout their lifetime, they must qualify to IEC 61701/IS 61701. Thin - Film terrestrial photovoltaic (PV) modules must qualify to IS 16077: 2013 / IEC 61646: 2008
- iii. The temperature co-efficient power of the PV module shall be equal to or better than -0.4%/°C for crystalline modules and -0.3 %/°C for thin films modules.
- iv. Solar PV modules of minimum fill factor 75%, to be used.
- v. The rated power of solar PV module shall have maximum tolerance up to +3%.
- vi. The peak-power point current of any supplied module string (series connected modules) shall not vary by +1% from the respective arithmetic means for all modules and/or for all module strings (connected to the same MPPT), as the case may be.
- vii. The peak-power point voltage of any supplied module string (series connected modules) shall not vary by + 2% from the respective arithmetic means for all modules and/or for all module strings (connected to the same MPPT), as the case may be.
- viii. All PV modules should have a nominal power output of >90% at STC during the first 10 years, and >80% during the next 15 years. Further, module shall have nominal power output of >97% during the first year of installation—degradation of the module below 0.7 % per annum. If module(s) fail(s) to exhibit such power output in the prescribed time span, the Contractor will either deliver additional PV Module(s) to compensate the missing power output with no change in area of

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding rooftop used or replace the PV Module(s) with no change in area of rooftop used at Owner's sole option

- ix. Solar PV modules of minimum capacity **500 Wp** to be used.
- x. The PV Module efficiency should be minimum 20%.
- xi. All electrical parameters at STC shall have to be provided.
- xii. The PV modules shall be equipped with IP 65 or better protection level junction box with required numbers of bypass diodes of appropriate rating and appropriately sized output power cable of symmetric length with MC4 or equivalent solar connectors.
- xiii. The bidder should get material warranty from the manufacturer for the Solar Module(s) to be free from the defects and/or failures specified below for a period not less than five (05) years from the date of commissioning.
 - a) Defects and/or failures due to manufacturing.
 - b) Defects and/or failures due to quality of materials.
 - c) Nonconformity to specifications due to faulty manufacturing and/or inspection processes. If the solar Module(s) fails to conform to this warranty, the manufacturer will repair or replace the solar module(s), at the Owners sole option.
- xiv. Modules deployed must use a RF identification tag laminated inside the glass.

The following information must be mentioned in the RFID used on each module:

- a) Name of the manufacturer of the PV module
- b) Name of the manufacturer of Solar Cells.
- c) Month & year of the manufacture (separate for solar cells and modules)
- d) Country of origin (separately for solar cells and module)
- e) I-V curve for the module Wattage, Im, Vm and FF for the module
- f) Unique Serial No and Model No of the module
- g) Date and year of obtaining IEC PV module qualification certificate.
- h) Name of the test lab issuing IEC certificate.
- i) Other relevant information on traceability of solar cells and module as per ISO 9001 and ISO 14001.
- j) Nominal wattage +3%
- k) Brand Name, if applicable.
- xv. Unique Serial No, Model No, Name of Manufacturer, Manufacturing year, Make in India logo and module wattage details should be displayed inside the laminated glass.
- xvi. The modules shall be fitted with water syphoning arrangement wherever necessary to prevent water stagnation on the bottom modules.

2. Inverter/PCU

- i. The Solar Photovoltaic Inverters must comply with the Quality Control Order dated 30.08.2017 for Solar Photovoltaic Inverters and its amendments thereof.
- ii. Inverters/PCU should comply with applicable IEC/equivalent BIS standard for efficiency measurements and environmental tests as per standard codes IEC 61683/IS 61683, IS 16221 (Part 2), IS 16169 and IEC 60068-2(1,2,14,30) /Equivalent BIS Std.
- iii. Maximum Power Point Tracker (MPPT) shall be integrated in the inverter/PCU to maximize energy drawn from the array. Charge controller (if any) / MPPT units environmental testing should qualify IEC 60068-2(1, 2, 14, 30)/Equivalent BIS standard. The junction boxes/enclosures should be IP 65 or better (for outdoor)/ IP 54or better (indoor) and as per IEC 529 Specifications.
- iv. All inverters/PCUs shall be IEC 61000 compliant for electromagnetic compatibility, harmonics, Surge, etc. PCU/inverter should also be DG set interactive. The PCU should also have provision of charge controller in case of systems. The PCU shall have a minimum comprehensive warranty of 10 years.
- v. The PCU/ inverter shall have overloading capacity of minimum 20%.
- vi. Typical technical features of the inverter shall be as follows
 - a. Switching devices: IGBT/MOSFET
 - b. Control: Microprocessor/DSP
 - c. Nominal AC output voltage and frequency: as per CEA/State regulations
 - d. Output frequency: 50 Hz
 - e. Grid Frequency Synchronization range: as per CEA/State Regulations
 - f. Ambient temperature considered: -20°C to 60°C
 - g. Humidity: 95 % Non-condensing
 - h. Protection of Enclosure: IP-54 (Minimum) for indoor and IP-65(Minimum) for outdoor.
 - i. Grid Frequency Tolerance range: as per CEA/State regulations
 - j. Grid Voltage tolerance: as per CEA/State Regulations
 - k. No-load losses: Less than 1% of rated power
 - I. Inverter efficiency (minimum): > 90%
 - m.THD: < 3%
 - n. PF: > 0.9 (lag or lead)
 - o. Should not inject DC power more than 0.5% of full rated output at the interconnection point and comply to IEEE 519.
 - p. The inverter should have the inbuilt facility to communicate system related data through SIM/dongle. The inverter may also be enabled for Wi-Fi based communication.

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

- vii. The output power factor of inverter should be suitable for all voltage ranges or sink of reactive power, inverter should have internal protection arrangement against any sustained fault in feeder line and against the lightning on feeder.
- viii. All the Inverters should contain the following clear and indelible Marking Label & Warning Label as per IS16221 Part II, clause 5. The equipment shall, as a minimum, be permanently marked with:
 - a. The name or trademark of the manufacturer or supplier;
 - b. A model number, name or other means to identify the equipment,
 - c. A serial number, code or other marking allowing identification of manufacturing location and the manufacturing batch or date within a three-month time period.
 - d. Input voltage, type of voltage (a.c. or d.c.), frequency, and maximum continuous current for each input.
 - e. Output voltage, type of voltage (a.c. or d.c.), frequency, maximum continuous current, and for a.c. outputs, either the power or power factor for each output.
 - f. The Ingress Protection (IP) rating
 - ix. Marking shall be located adjacent to each fuse or fuse holder, or on the fuse holder, or in another location provided that it is obvious to which fuse the marking applies, giving the fuse current rating and voltage rating for fuses that may be changed at the installed site.
 - x. In case the consumer is having a $3-\phi$ connection, $3-\phi$ inverter shall be provided by the bidder as per the consumer's requirement and regulations of the State.
- xi. Inverter/PCU shall be capable of complete automatic operation including wakeup, synchronization & shutdown.
- xii. The Inverter should have a provision of remote monitoring of inverter data through sim card. Required website/mobile app platform, where NHPC REL and the user (Consumer) can access the data, should be provided/explained to NHPC REL and consumer while installation. Additionally, if inverter has the facility of inbuilt wi-fi module, that should also be explained to the consumer. On demand, Inverter should also have provision to feed the data to the remote monitoring server using relevant API/ protocols. All the inverter data should be available for monitoring by giving web access. Charges for internet and website/mobile app platform shall be borne by the bidder throughout the contract period.
- xiii. Integration of PV system with Grid & Grid Islanding:
 - a. The output power from SPV would be fed to the inverters/PCU which converts DC produced by SPV array to AC and feeds it into the main electricity grid after synchronization.
 - b. In the event of a power failure on the electric grid, it is required that any independent power-producing inverters attached to the grid turn off in a short

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding period of time. This prevents the DC-to-AC inverters from continuing to feed power into small sections of the grid, known as "islands." Powered islands present a risk to workers who may expect the area to be unpowered, and they may also damage grid-tied equipment. The Rooftop PV system shall be equipped with islanding protection. In addition to disconnection from the grid (due to islanding protection) disconnection due to under and over voltage conditions shall also be provided, if not available in inverter.

c. MCB/MCCB or a manual isolation switch, besides automatic disconnection to grid, would have to be provided at utility end to isolate the grid connection by the utility personnel to carry out any maintenance. This switch shall be locked by the utility personnel.

3. Module Mounting Structure (MMS)

- i. Supply, installation, erection, and acceptance of module mounting structure (MMS) with all necessary accessories, auxiliaries and spare part shall be in the scope of the work.
- ii. The design calculations shall be supplemented with neat sketch and reference to various clauses of technical specification and Indian standards. For MMS design analysis and determination of forces, where computer program (preferably STAAD) is used, the contractor shall submit a write- up on computer program used and its input and output data for review and approval of NHPC REL. It may be ensured that the design has been certified by a recognized Lab/ Institution/chartered Engineer in this regard and submit wind loading calculation sheet to NHPC REL.
- iii. Module mounting structures can be made from three types of materials. They are Hot Dip Galvanized Iron, Aluminium and Hot Dip Galvanized Mild Steel (MS). However, MS will be preferred for raised structure.
- iv. MMS Steel shall be as per latest IS 2062:2011 and galvanization of the mounting structure shall be in compliance of latest IS 4759. MMS Aluminium shall be as per AA6063 T6. For Aluminium structures, necessary protection towards rusting need to be provided either by coating or anodization.
- v. All bolts, nuts, fasteners shall be of stainless steel of grade SS 304 or hot dip galvanized, panel mounting clamps shall be of aluminium and must sustain the adverse climatic conditions. Structural material shall be corrosion resistant and electrolytically compatible with the materials used in the module frame, its fasteners, nuts and bolts.
- vi. Structural material shall be corrosion resistant and electrolytically compatible with the materials used in the module frame and associated fasteners, nuts & bolts.

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

- vii. The module mounting structures should have angle of inclination as per the site conditions to take maximum insolation and complete shadow-free operation during generation hours. However, to accommodate more capacity the angle of inclination may be reduced until the plant meets the specified performance ratio requirements.
- viii. The Mounting structure shall be so designed to withstand the speed for the wind zone of the location where a PV system is proposed to be installed. The PV array structure design shall be appropriate with a factor of safety of minimum 1.5.
 - ix. The upper edge of the module must be covered with wind shield so as to avoid build air ingress below the module. Slight clearance must be provided on both edges (upper & lower) to allow air for cooling.
 - x. Suitable fastening arrangements such as grouting and calming should be provided to secure the installation against the specific wind speed. The bidder shall be fully responsible for any damage to SPV System caused due to high wind velocity within guarantee period as per technical specification.
- xi. The structures shall be designed to allow easy replacement, repairing and cleaning of any module. The array structure shall be so designed that it will occupy minimum space without sacrificing the output from the SPV panels. Necessary testing provision for MMS to be made available at site.
- xii. Adequate spacing shall be provided between two panel frames and rows of panels to facilitate personnel protection, ease of installation, replacement, cleaning of panels and electrical maintenance.
- xiii. The structure shall be designed to withstand operating environmental conditions for a period of minimum 25 years.
- xiv. Module Mounting Structures shall be designed to withstand the extreme weather conditions in the area (as per National Building Code 2016). The Mounting structure shall be so designed to withstand the wind speed of 180 km/hour or cyclonic winds in coastal area. Further structure shall designed for be painted with marine paint.
- xv. Aluminium frames should be avoided for installations in coastal areas.
- xvi. No welding is allowed on the mounting structure.
- xvii. The Bidder is responsible for the waterproofing of the roof disturbed/ pierced for installation of Project throughout the Project installation period and Comprehensive O&M period. The Bidder should immediately take necessary action to repair any damage to the waterproofing.
- xviii. The Rooftop Structures maybe classified in 3 broad categories.

- a. Ballast structure
 - i. For flat RCC Roofs, the mounting structure must be Non-invasive ballast type and any sort of penetration of roof to be avoided.
- ii. The minimum clearance of the structure from the roof level should allow ventilation for cooling, also ease of cleaning and maintenance of panels as well as cleaning of terrace.
- iii. The structures should be suitably loaded with reinforced concrete blocks of appropriate weight with Nominal Mix 1:1.5:3 with 20mm nominal size having compressive strength of 25MPa after 28 days of curing.
- b. Structure with Anchor fastener
 - i. For inclined/ Dome type RCC Roofs, the mounting structure shall be fixed on the roof with proper fasteners/bolts along with chemical grouting and sealing.
- ii. The roof should remain waterproof after erection of the structure.
- c. Metallic roof structure
 - i. The structure design should be as per the slope of the metallic roof.
- ii. Minimum thickness of the structural members shall be 2mm
- iii. Fixing of module mounting structures shall not cause any damage to the roof. The roof should remain waterproof.
- iv. Proper walk way and life line shall be provided.
- v. The inclination angle of structure can be done in two ways:
 - a. With same tilt angle based on the slope of roof to get the maximum output.
 - b. The base of structure should be connected on the Purlin of roof with the proper fastening arrangement.

4. Metering

- i. The specifications net meter/smart meter shall be as per the latest technical specifications issued by the Central Electricity Authority (CEA) and its amendment thereof.
- ii. A Roof Top Solar (RTS) Photo Voltaic (PV) system shall consist of following energy meters:
 - a) Net meter: To record import and export units.
 - b) Generation meter(s): To keep record for total generation of the plant at each termination.
- iii. The installation of meters including CTs & PTs, wherever applicable, shall be carried out by the bidder as per the terms, conditions and procedures laid down by the concerned SERC/DISCOMs.

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

iv. In case access to the generation meter is not available for remote monitoring, another energy meter shall be provided for remote metering.

5. Array Junction Boxes:

- i. The junction boxes are to be provided in the PV array for termination of connecting cables. The Junction Boxes (JBs) shall be made of GRP/FRP/Powder Coated aluminum /cast aluminum alloy with full dust, water & vermin proof arrangement. All wires/cables must be terminated through cable lugs. The JBs shall be such that input & output termination can be made through suitable cable glands. Suitable markings shall be provided on the bus-bars for easy identification and cable ferrules will be fitted at the cable termination points for identification.
- ii. Copper bus bars/terminal blocks housed in the junction box with suitable termination threads Conforming to IP 65 or better standard and IEC 62208 Hinged door with EPDM rubber gasket to prevent water entry, Single /double compression cable glands should be provided.
- iii. Polyamide glands and MC4 Connectors may also be provided. The rating of the junction box shall be suitable with adequate safety factor to interconnect the Solar PV array.
- iv. Suitable markings shall be provided on the bus bar for easy identification and the cable ferrules must be fitted at the cable termination points for identification.
- v. Junction boxes shall be mounted on the MMS such that they are easily accessible and are protected from direct sunlight and harsh weather.

6. DC Distribution Box (DCDB):

- i. May not be required for small plants, if suitable arrangement is available in the inverter.
- ii. DC Distribution Box are to be provided to receive the DC output from the PV array field.
- iii. DCDBs shall be dust & vermin proof conform having IP 65 or better protection, as per site conditions.
- iv. The bus bars are made of EC grade copper of required size. Suitable capacity MCBs/MCCB shall be provided for controlling the DC power output to the inverter along with necessary surge arrestors. MCB shall be used for currents up to 63 Amperes, and MCCB shall be used for currents greater than 63 Amperes

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

7. AC Distribution Box (ACDB):

- i. AC Distribution Panel Board (DPB) shall control the AC power from inverter, and should have necessary surge arrestors, if required. There is interconnection from ACDB to mains at LT Bus bar while in grid tied mode.
- ii. All switches and the circuit breakers, connectors should conform to IEC 60947:2019, part I, II and III/ IS 60947 part I, II and III.
- iii. The isolators, cabling work should be undertaken as part of the project.
- iv. All the Panels shall be metal clad, totally enclosed, rigid, floor mounted, air insulated, cubical type suitable for operation on 1-φ/3-φ, 415 or 230 volts, 50 Hz (or voltage levels as per CEA/State regulations).
- v. The panels shall be designed for minimum expected ambient temperature of 45 degree Celsius, 80 percent humidity and dusty weather.
- vi. All indoor panels will have protection of IP 54 or better, as per site conditions. All outdoor panels will have protection of IP 65 or better, as per site conditions.
- vii. Should conform to Indian Electricity Act and CEA safety regulations (till last amendment).
- viii. All the 415 or 230 volts (or voltage levels as per CEA/State regulations) AC devices / equipment like bus support insulators, circuit breakers, SPDs, Voltage Transformers (VTs) etc., mounted inside the switchgear shall be suitable for continuous operation and satisfactory performance under the following supply conditions.
 - a. Variation in supply voltage: as per CEA/State regulations
 - b. Variation in supply frequency: as per CEA/State regulations
- ix. The inverter output shall have the necessary rated AC surge arrestors, if required and MCB/ MCCB. RCCB shall be used for successful operation of the PV system, if inverter does not have required earth fault/residual current protection.

8. Protections:

The system should be provided with all necessary protections like earthing, Lightning, and Surge Protection, as described below:

i. Earthing Protection

a. The earthing shall be done in accordance with latest Standards.

- b. Each array structure of the PV yard, Low Tension(LT) power system, earthing grid for switchyard, all electrical equipment, inverter, all junction boxes, etc. shall be grounded properly as per IS 3043-2018.
- c. All metal casing/ shielding of the plant shall be thoroughly grounded in accordance with CEA Safety Regulation 2010. In addition, the lightning arrester/masts should also be earthed inside the array field.
- d. Earth resistance should be as low as possible and shall never be higher than 5 ohms.
- e. For 10 KW and above systems, separate three earth pits shall be provided for individual three earthing viz.: DC side earthing, AC side earthing and lightning arrestor earthing.

ii. Lightning Protection

- a. The SPV power plants shall be provided with lightning & over voltage protection, if required. The main aim in this protection shall be to reduce the overvoltage to a tolerable value before it reaches the PV or other sub system components. The source of over voltage can be lightning, atmosphere disturbances etc. Lightning arrestor shall not be installed on the mounting structure.
- b. The entire space occupying the SPV array shall be suitably protected against Lightning by deploying required number of Lightning Arrestors (LAs). Lightning protection should be provided as per NFC17-102:2011/IEC 62305 standard.
- c. The protection against induced high-voltages shall be provided by the use of Metal Oxide Varistors (MOVs)/Franklin Rod type LA/Early streamer type LA.
- d. The current carrying cable from lightning arrestor to the earth pit should have sufficient current carrying capacity according to IEC 62305. According to standard, the minimum requirement for a lightning protection system designed for class of LPS III is a 6 mm2 copper/ 16 mm2 aluminum or GI strip bearing size 25*3 mm thick). Separate pipe for running earth wires of Lightning Arrestor shall be used.

iii. Surge Protection

- a. Internal surge protection, wherever required, shall be provided.
- b. It will consist of three SPD type-II/MOV type surge arrestors connected from +ve and –ve terminals to earth.

9. CABLES

- i. All cables should conform to latest edition of IEC/equivalent BIS Standards along with IEC 60227/IS 694, IEC 60502/IS 1554 standards.
- ii. Cables should be flexible and should have good resistance to heat, cold, water, oil, abrasion etc.
- iii. Armored cable should be used and overall PVC type 'A' pressure extruded insulation or XLPE insulation should be there for UV protection.
- iv. Cables should have Multi Strand, annealed high conductivity copper conductor on DC side and copper/FRLS type Aluminum conductor on AC side. For DC cabling, multi-core cables shall not be used.
- v. Cables should have operating temperature range of -10°C to +80°C and voltage rating of 660/1000V.
- vi. Sizes of cables between array interconnections, array to junction boxes, junction boxes to Inverter etc. shall be so selected to keep the voltage drop less than 2% (DC Cable losses).
- vii. The size of each type of AC cable selected shall be based on minimum voltage drop. However; the maximum drop shall be limited to 2%
- viii. The electric cables for DC systems for rated voltage of 1500 V shall conform to IS 17293:2020.
- ix. All cable/wires are to be routed in a RPVC pipe/ GI cable tray and suitably tagged and marked with proper manner by good quality ferule or by other means so that the cable is easily identified.
- x. All cable trays including covers to be provided.
- xi. The Cable should be so selected that it should be compatible up to the life of the solar PV panels i.e. 25 years.

10. DRAWINGS & MANUALS:

- i. Operation & Maintenance manual/user manual, Engineering and Electrical Drawings shall be supplied along with the power plant.
- ii. The manual shall include complete system details such as array lay out, schematic of the system, inverter details, working principle etc.

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

- iii. The Manual should also include all the Dos & Don'ts of Power Plant along with Graphical Representation with indication of proper methodology for cleaning, Operation and Maintenance etc.
- iv. Step by step maintenance and troubleshooting procedures shall also be given in the manuals.
- v. Bidders should also educate the consumers during their O&M period.

11. TRANSFORMERS (if required)

- i. Dry type appropriate kVA, of transformer Step up along with all protections, switchgears, Vacuum circuit breakers, cables etc. along with required civil work. Dry Type Transformer shall be constructed in accordance to IS: 2026, IS: 11171, Indian Electricity Act 2003, BEE Guideline & CEA notifications and equivalent to any other international standard. Transformers shall be complete & functional in all respects. The other important construction particulars shall be as below.
 - a. The transformers shall be housed in metal protective housing, having a degree of ingress protection. The enclosure shall be provided with suitable hardware (as required).
 - b. The conductors shall be of electrolytic grade copper free from scales & burrs.
 - c. Dry Type Transformer windings shall be of class F insulation or better.
 - d. The core shall be constructed from non-ageing, cold rolled, grain oriented silicon steel laminations.
 - e. The maximum losses for dry type transformer shall not be more than the values specified in latest energy conservation building code (ECBC) of BEE.
 - f. The fittings/accessories, including protection/monitoring device generally required for satisfactory operation of the transformer, are to be provided.
 - g. Suitable rain shed arrangement shall be provided to keep transformer under that arrangement.
 - h. Painting shall be as per employer's requirement (will be finalized during detailed engineering) Type and routine test shall be conducted as per IS11171.
 - i. In case the bidder/contractor has conducted such specified type test(s) within last ten years as on the date of bid opening, he may submit the type test reports to the owner for waiver of conductance of such type test(s). These reports should be for the test conducted on the equipment similar to those proposed to be supplied under this contract and test(s) should have been either conducted at an independent laboratory or should have been

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding witnessed by Owner. In case the bidder is not able to submit report of the type test(s) conducted within last ten years from the date of bid opening, or in case the type test report(s) are not found to be meeting the specification requirements the bidder shall conduct all such tests under this contract at no additional cost to the employer at an independent laboratory or in presence of the Owner's representative and submit the test reports.

12. Data Monitoring:

- i. Remote monitoring of the generation of Solar Power shall be done through appropriate app which shall be provided by bidder. Provision for mobile appbased display shall be made for all the data display with a provision to interact with NHPC REL and Central Ministry /State Psu's/ Government Institutions and other stakeholders.
- ii. Remote Monitoring and data acquisition through Remote Monitoring System software with latest software/hardware configuration and service connectivity for online / real time data monitoring/control complete to be supplied and Comprehensive O&M/control to be ensured by the supplier. Provision for interfacing these data on any other Agency server and portal in future shall be kept. The Contractor must ensure the yearly availability and connectivity of the RMS system to be at least 99%.
- iii. Measurement of Solar PV parameters at Inverter level: These Inverter parameters shall have provision of data logging through Mod Bus (RS-485) protocol.
- iv. Solar Irradiance: An integrating Pyranometer (Class II or better) shall be provided, with the sensor mounted on a Horizontal plane at a shadow free suitable location near solar arrays at one of the Roof Top within premises.
- v. Temperature: Temperature probes for recording the PV Cell temperature shall be provided at one of the modules at shadow free area at one of the Roof Top within premises.
- vi. Regarding the generated power consumption, priority needs to be given for internal consumption first and thereafter any excess power can be exported to the distribution licensee network.
- vii. The bidder is also required to fetch the real time generation data from the solar meter for remote metering and billing. In case the solar meter is not accessible due to DISCOM restrictions, the bidder shall install an additional smart meter just before the solar meter for this purpose within the quoted price.

13. Miscellaneous:

i. Connectivity: The maximum capacity for interconnection with the grid at a specific voltage level shall be as specified in the SERC regulation for Grid connectivity

- Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding and norms of DISCOM and amended from time to time. The Bidder must take approval/NOC from the concerned DISCOM for the connectivity, technical feasibility, and synchronization of Project with DISCOM network and submit the same to NHPC REL before Commissioning Project. Reverse power relay shall be provided by Bidder (if necessary), as per the DISCOM's requirement.
 - ii. Safety measures: Electrical safety of the installation(s) including connectivity with the grid must be taken into account and all the safety rules & regulations applicable as per Electricity Act, 2003 and CEA Safety Regulation 2023 etc. or its latest amendments must be followed.
 - iii. Danger boards should be provided as and where necessary as per IE Act /IE rules as amended up to date.
 - iv. Shadow analysis: The shadow analysis report with the instrument such as Solar Pathfinder or professional shadow analysis software of each site should be provided and the system shall be installed only in shadow free space.
 - v. Firefighting system Portable fire extinguishers/sand buckets shall be provided wherever required as per norms.
 - vi. Bidder should dispose off the packing material, surplus items, unused materials, waste etc. generated during construction, as per applicable Rules and standards. Concrete wastes and dead mortar if any are to be removed and the area is to be cleaned immediately.
 - vii. The bidder shall take fully responsible for safety of manpower, materials and premises. The contractor shall take special care for safe working at height, electrical safety of the installation(s) including connectivity with the grid and follow all the safety rules & regulations applicable as per Electricity Act, 2003, CEA guidelines and other applicable standard practices.
 - viii. The bidder shall provide a display board mentioning the details of the project.
 - ix. Stickers as per NHPC REL requirements e.g. instructions/ guidelines/ danger mark.

14. PERFORMANCE GUARANTEE TEST FOR ROOFTOP SOLAR PV

For determining the Annual CUF, the annual module degradation shall be taken as 0.7% cumulative from 2nd operational year. The duration for which the Generation was interrupted due to non-availability of Grid during solar hours or any other reasons not attributable to Bidder shall not be taken into account for calculation of Annual CUF. The Delivery Point shall be a location mutually agreed by Bidder and NHPC REL in line with applicable policy/ regulation/ rules, where solar power is delivered from the Project to the Procurer, in such a way as to enable measurement of gross energy generated by Project.

Annexure-A

Quality Certification, Standards and Testing for Grid-Connected Solar PV Systems/Power Plants

The applicable codes and standards are as mentioned below:

| Solar PV Modules/Panels | | | | |
|-------------------------------|--|--|--|--|
| Solar PV modules | Description | | | |
| IEC 61215/IS 14286 | Design qualification and type approval for Crystalline | | | |
| | Silicon Terrestrial Photovoltaic (PV) Modules | | | |
| IEC 61701 | Salt Mist corrosion testing of Photovoltaic (PV) Modules | | | |
| IEC 61853-Part1/IS 16170: | Photovoltaic(PV) module performance testing and | | | |
| Part 1 | energy rating-:Irradiance and temperature performance | | | |
| | measurements, and power rating | | | |
| IEC 62176/IS 16664 | Photovoltaic(PV) Modules-Ammonia (NH3) corrosion | | | |
| | testing | | | |
| IS 16077 : 2013 / IEC 61646 : | Thin - Film terrestrial photovoltaic (PV) modules - Design | | | |
| 2008 | qualification and type approval | | | |
| 15004700 4 0 | Photovoltaic(PV) Modules Safety Qualification-Part 1: | | | |
| IEC61730-1,2 | Requirements for construction, Part2:Requirements for | | | |
| | Testing | | | |
| IS 17210 (part 1) or IEC TS | Photovoltaic (PV) modules – Test method for detection of potential induced degradation. IEC 62804-1: Part 1: | | | |
| 62804-1 | | | | |
| Solar PV Inverters | Crystalline Silicon | | | |
| | Safety of power converters for use in photovoltaic power | | | |
| | systems – Part 1: General requirements, and Safety of | | | |
| | power converters for use in photovoltaic power systems | | | |
| IEC 62109 or IS : 16221 | Part 2: Particular requirements for inverters. Safety | | | |
| | compliance (Protection degree IP 65 or better for | | | |
| | outdoor mounting, IP 54 or better for indoor mounting) | | | |
| | Photovoltaic Systems – Power conditioners: Procedure | | | |
| IS/IEC 61683 latest | for | | | |
| (as applicable) | Measuring Efficiency (10%, 25%, 50%, 75% & 90-100% | | | |
| | Loading Conditions) | | | |
| IEC 60068-2 /IEC 62093 (as | Environmental Testing of PV System – Power | | | |
| applicable) | Conditioners and Inverters | | | |
| | Utility-interconnected photovoltaic inverters - Test | | | |
| IEC 62116:2014/ IS16169 | procedure of islanding prevention measures | | | |
| Fuses | | | | |
| IS/IEC 60947 (Part 1, 2 & 3), | General safety requirements for connectors, switches, | | | |
| EN 50521 | circuit breakers (AC/DC): | | | |
| | 1) Low-voltage Switchgear and Control-gear, Part 1: | | | |

| ting up Solar Root Top Plants at Various Departm | ients in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Biddii |
|--|---|
| | General rules |
| | 2) LowVoltage Switchgear and Control-gear, Part 2: |
| | Circuit Breakers |
| | 3) Low-voltage switchgear and Control-gear, Part 3: |
| | Switches, disconnectors switch-disconnectors and fuse- |
| | combination units |
| | 4) EN 50521: Connectors for photovoltaic system- |
| | Safety requirements and tests. |
| | Low-voltage fuses - Part 6: Supplementary requirements |
| IS/IEC 60269-6 | for fuse Links for the protection of solar photovoltaic |
| | energy systems |
| Solar PV Roof Mountin | g Structure |
| IS 2062/IS 4759/ AA6063 T6 | Material for the structure mounting |
| Surge Arrestors | |
| BFC 17-102:2011/ NFC | Lightoning Drotaction Standard |
| 102:2011/ IEC 62305 | Lightening Protection Standard |
| | Electrical installations of buildings - Part 5-53: Selection |
| | and erection of electrical equipment - Isolation, switching |
| IEC 60364-5-53/ IS 15086-5 | and control Low-voltage surge protective devices - Part |
| (SPD) IEC 61643- 11:2011 | 11: Surge protective devices connected to low-voltage |
| | power systems - Requirements and test methods |
| Cables | |
| | General test and measuring method for PVC (Polyvinyl |
| IEC 60227/IS 694, IEC | chloride) insulated cables (for working voltages up to |
| 60502/IS 1554 (Part 1& 2)/ | and including 1100 V, and resistant for outdoor |
| IEC69947 (as applicable) | installation) |
| 10 47000 0000 | Electric Cables for Photovoltaic Systems for Rated |
| IS 17293:2020 | Voltage 1500 V DC |
| Earthing /Lightning | - |
| | IEC 62561-1: Lightning protection system components |
| | (LPSC) - Part: Requirements for connection components |
| | IEC 62561-2: Lightning protection system components |
| IEC 62561/IEC 60634 Series | (LPSC) – Part 2:Requirements for conductors and earth |
| (Chemical earthing) (as | electrodes |
| applicable) | IEC 62561-7: Lightning protection system components |
| | (LPSC) - Part 7:Requirements for earthing enhancing |
| | compounds |
| Junction Boxes | · · |
| | Junction boxes and solar panel terminal boxes shall be |
| | of the thermo-plastic type with IP 65 or better protection |
| | |
| IEC 60529 | for outdoor use, and IP 54 or better protection for indoor |
| IEC 00529 | for outdoor use, and IP 54 or better protection for indoor |

INSPECTION AND MAINTENANCE SCHEDULE

| Component | Activity | Description | Interval | Ву |
|---------------------------------|---|---|---|----------------|
| Module Mounting structure | Inspection | Check whether the MMS are properly in tightened position and check for corrosion. | Quarterly. | Site in charge |
| PV Module | Cleaning | Clean any bird droppings/ dark spots on module | Immediately | Site in charge |
| | Cleaning | Clean PV modules with plain water or mild dishwashing detergent. Do not use brushes, any types of solvents, abrasives, or harsh detergents | As per the site conditions and at least once a week | Site in charge |
| | Inspection (for plants > 100 kWp) | Use infrared camera to inspect for hot spots; bypass diode failure | Annual | Site in charge |
| PV Array | Inspection | Check the PV modules and rack for any damage. Note down location and serial number of damaged modules. | Annual | Site in charge |
| | Inspection | Determine if any new objects, such as vegetation growth, are causing shading of the array and move them if possible. | Quarterly | Site in charge |
| Junction Boxes | Inspection | Inspect electrical boxes for corrosion or intrusion of water or insects. Seal boxes if required. Check position of switches and breakers. Check | Annual | Site in charge |

| | | operation of all protection devices. | | |
|-------------|------------|---|-----------|----------------|
| Wiring | Inspection | Inspect cabling for signs of cracks, defects, lose connections, overheating, arcing, short or open circuits and ground faults. | Annual | Site in charge |
| Spare Parts | Management | Manage inventory of spare parts. to ensure that the amount of power being generated is typical of the conditions. Inspect Inverter housing or shelter for physical Maintenance, if required. | As needed | Site in charge |
| Inverter | Service | Clean or replace any air filters | As needed | Site in charge |
| Instruments | Validation | Spot-check monitoring instruments (pyranometer etc.) with standard instruments to ensure that they are operational and within specifications. | Annual | Site in charge |
| Transformer | Inspection | Inspect transformer oil level, temperature gauges, breather, silica gel, meter, connections etc. | Annual | Site in charge |
| Tracker (if | Inspection | Inspect gears, gear boxes, bearings as | Annual | Site in charge |

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

| present) | | required. | | |
|----------|---------------|---|------------|----------------|
| | Service | Lubricate tracker mounting bearings, gearbox as required | Bi-annual | Site in charge |
| Plant | Monitoring | Daily Operation and Performance Monitoring | Daily | Site in charge |
| Log Book | Documentation | Document all O&M activities in a workbook available to all service personnel | Continuous | Site in charge |

Monthly O & M Report

Month and year:

Name of the Bidder:

LOA No:

Project Capacity:

Address of the site:

Part A

| Component | Activity | Description | Remarks |
|-------------|----------------|------------------------------------|---------|
| Component | | | |
| PV Array | Inspection | Check the PV modules and rack | |
| | | for any damage. | |
| | Inspection | If any new objects, such as | |
| | | vegetation growth etc., are | |
| | | causing shading of the array. | |
| | | Remove if any. | |
| | Vermin Removal | Remove bird nests or vermin | |
| | | from array and rack area. | |
| Junction | Inspection | Inspect electrical boxes for | |
| Boxes | | corrosion, intrusion of water or | |
| | | vermin. • Check position of | |
| | | switches and breakers. • Check | |
| | | status of all protection devices. | |
| Wiring | Inspection | Inspect cabling for signs of | |
| | | cracks, defects, lose | |
| | | connections, corrosion, | |
| | | overheating, arcing, short or | |
| | | open circuits, and ground faults | |
| Inverter | Inspection | Observe instantaneous | |
| | | operational indicators on the | |
| | | faceplate. • Inspect Inverter | |
| | | housing or shelter for any | |
| - | | physical | |
| Inverter | Service | Clean or replace any air filters. | |
| Instruments | Validation | Verify monitoring instruments | |
| | | (pyranometer etc.) with standard | |
| | | instruments to verify their | |
| | | operation within tolerance limits. | |

| Transformer | Inspection | Inspect transformer oil level, | | |
|---------------------|---------------|--|--|--|
| | | temperature gauges, breather, | | |
| | | silica gel, meter, connections etc. | | |
| Plant | Monitoring | Daily Operation and Performance | | |
| | | Monitoring. | | |
| Spare Parts | Management | Manage inventory of spare parts. | | |
| Log Book | Documentation | Maintain daily log records. | | |
| Tracker (if any) | Inspection | Inspect gears, gear boxes, bearings, motors. | | |
| | Service | Lubricate bearings, gear as | | |
| | | required. | | |

Signature of the Authorized signatory of the Contractor

| | Р | art B | | |
|------|-------------------|------------------------|---------------------------------------|---------|
| Date | Generation kWh | Grid outage (hh:mm) | Inverter down period (hh:mm) | Remarks |
| 1 | | | | |
| 2 | | | | |
| 3 | | | | |
| 4 | | | | |
| 5 | | | | |
| 6 | | | | |
| 7 | | | | |
| 8 | | | | |
| 9 | | | | |
| 10 | | | | |
| 11 | | | | |
| 12 | | | | |
| 13 | | | | |
| 14 | | | | |
| 15 | | | | |
| 16 | | | | |
| 17 | | | | |
| 18 | | | | |
| 19 | | | | |
| 20 | | | | |
| 21 | | | | |
| 22 | | | | |
| 23 | | | | |
| 24 | | | | |
| 25 | | | | |
| 26 | | | | |
| 27 | | | | |
| 28 | | | | |
| 29 | | | | |
| 30 | | | | |

| 31 | | |
|----|--|--|
| | | |

Total generation for the month in kWh:

Cumulative generation since commissioning in kWh:

CUF for month in %:

Cumulative CUF since commissioning in %: Date:

Signature of the Authorized signatory of the Contractor

Signature of the Authorized signatory of the custodian of the Building

Annexure-C

| Sample Project Completion Report for Grid-Connected Rooftop | | | | |
|--|--------------|---------------------|----|------|
| Financial year : | | | | |
| Approval No.: | | | | |
| Proposal Title : | | | | |
| Installed by agency : | | | | |
| Project initiated by : | | | | |
| Title of the Project : | | Capacity | | |
| | 1 | Name of the | | |
| Category of the organization / beneficiary: | | contact person: | | |
| Address of contact person : | | | | |
| State : | | District/City : | | |
| Mobile: | | Email: | | |
| Telephone No. : | | Website : | | |
| Other info | | | | |
| Electricity Distribution Company Name : | | | | |
| Electricity consumer account no. as per electricity bill : | | as on Date | | |
| Technology Description | & System I | Design /Specificati | on | |
| Compliance to B | IS/IEC Stand | dards is mandatory) |) | |
| 1. Module | | | | |
| Capacity/Power of each PV | 1. | | 1. | Nos. |

| Module(Wp) : | | | 2. Nos. |
|--|-----------------------------|--------------------------------------|---------|
| Cumulative Capacity of Modules (KWp): | | | |
| Solar cell technology : | | | |
| Module efficiency (in Percentage) : | | | |
| 2. Inverters | | | |
| Type of inverter : | | | |
| Make of inverter : | | | |
| | | | |
| Capacity/Power of each PCU/inverters (VA) : | Capacity / Power Nos. | | |
| Inverter efficiency (Full load) : | | | |
| 3. Metering Arrangement | | | |
| Any other details: | | | |
| Make and Type of Meter : | | | |
| Metering Details: | | | |
| 4. Other informations | | | |
| Units of electricity generated by the solar plant as per meter (in KWh): | | | |
| Monitoring Mechanism : | | | |
| No. of personnel to be trained in O&M | | | |
| Grid connectivity level | | | |
| Grid connectivity level phase | | Grid connectivity level Voltage : | |

VOLUME-4

FORMS & PROCEDURES (FP)

| | TABLE OF CONTENTS | |
|---------|--|--|
| Sl. No. | Description | |
| 1 | Format of Covering Letter | |
| 2 | Bid Security Form (Bank Guarantee) | |
| 3 | Format for POI for Earnest Money Deposit | |
| 4 | Format of Insurance Surety Bond for Earnest Money Deposit | |
| 5 | Performance Security Forms | |
| 6 | Format for POI for Performance Bank Guarantee | |
| 7 | FORMAT OF INSURANCE SURETY BOND FOR PERFORMANCE SECURITY | |
| 8 | Format for Disclosure alongwith Annexure | |
| 9 | CERTIFICATE FOR COMPLIANCE TO ALL PROVISIONS OF RfS DOCUMENT | |
| 10 | DECLARATION REGARDING INSOLVENCY PROCEEDINGS UNDER INSOLVENCY AND BANKRUPTCY CODE 2016 | |
| 11 | Self-Certificate for Cyber Security | |
| 12 | PRICE SCHEDULES | |
| 13 | ECS Form | |
| 14 | Proforma of certificate from the CEO/CFO of the holding company | |
| 15 | Proforma of undertaking | |
| 16 | Proforma of certificate from the CEO/CFO in accordance with item clause 3.2.B.3 of eligibility criteria | |
| 17 | Joint letter of undertaking between the CEO of the holding company / group company or subsidiary company & CEO of the bidder company | |
| 18 | Bidders undertaking (on Letter head of the Company) mentioning the requisite Category for which Bid(s) are submitted | |
| 19 | Undertaking from parent/ holding company | |
| 20 | Parent/ holding company agreement | |
| 21 | Parent/ Holding Company Agreement (In case of Joint Venture/Consortium) | |

1. Format of Covering Letter

(The covering letter should be on the Letter Head of the Bidding Company)

Date: _____

From : ______ [Insert Name and Address of Bidding

Company]

Tel.#:

Fax#:

E-mail address:

То

NHPC/ NHPC REL

NHPC OFFICE COMPLEX,

Sector-33, Faridabad-121003, Haryana, India

Sub: Response to RfS No. for Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various location of JAKEDA and its departments/Bodies in RESCO Mode through Tariff-Based Competitive Bidding [*Insert capacity of projects applied for*].

Dear Sir,

We, the undersigned [Insert name of the 'Bidder'] having read, examined and understood in detail the RfS including Qualification Requirements in particular, terms and conditions of the standard PPA for supply of power for the Term of the PPA to Client Organization, hereby submit our response to RfS.

We confirm that neither we nor any of our Parent Company/ Affiliate/ Group Company has submitted response to RfS other than this response to RfS, directly or indirectly, in response to the aforesaid RfS.

We are submitting application for the development of solar projects in the following category (s):

| Category | Tick applicable category |
|------------|--------------------------|
| Category 1 | |
| Category 2 | |
| Category 3 | |

1. We give our unconditional acceptance to the **RfS**, **No.** ______ and PPA attached thereto, issued by NHPC/ NHPC REL. In token of our acceptance to the RfS & PPA, the same have been digitally signed by us and submitted with the response to RfS. We confirm and undertake that the PPA shall be executed as per the provisions of the RfS and provisions of PPA shall be binding on us. Further, we confirm that the Project(s) shall be commissioned within the schedule stipulated in the RfS of the effective date of PPA for Projects.

2. **Earnest Money Deposit (EMD) : -** (Please read the EMD clause in the RfS carefully before filling)

- 3. We hereby declare that in the event our Project(s) get selected and we are not able to submit Bank Guarantee/POI/Insurance Surety Bond of the requisite value(s) towards PBG, Service charge /PMC Charge for the Projects, within due time as mentioned in this RfS on issue of LoA by NHPC REL/Government Client Organization for the selected Projects and/ or we are not able to sign PPA with Government Client Organization within the timeline as stipulated in the RfS for the selected Projects, NHPC REL shall have the right to encash the EMD submitted by us and return the balance amount (if any) for the value of EMD pertaining to unsuccessful capacity.
- 4. We have submitted our response to RfS strictly as Formats of this RfS, without any deviations, conditions and without mentioning any assumptions or notes in the said Formats. We hereby withdraw any deviation, conditions whether mentioned explicitly or not in our response to this RfS without any cost to NHPC/NHPC REL.

5. Acceptance

We hereby unconditionally and irrevocably agree and accept that the decision made by the NHPC/NHPC REL in respect of any matter regarding or arising out of the RfS shall be binding on us. We hereby expressly waive any and all claims in respect of this process.

6. Familiarity with Relevant Indian Laws & Regulations

We confirm that we have studied the provisions of the relevant Indian laws and regulations as required to enable us to submit this response to RfS and execute the PPA with the Govt. Client Organization or supply of solar power in the event of our selection as Successful Bidder.

7. We are enclosing herewith our response to the RfS with formats duly digitally signed as desired by you in the RfS for your consideration.

- 8. It is confirmed that our response to the RfS is consistent with all the requirements of submission as stated in the RfS and subsequent communications from the NHPC/ NHPC REL.
- 9. The information submitted in our response to the RfS is correct to the best of our knowledge and understanding. We would be solely responsible for any errors or omissions in our response to the RfS.

We confirm that all the terms and conditions of our Bid are valid for a period of upto and including **One Hundred Twenty (120) days** from the technical bid opening date unless extended by us on your request and it shall remain binding upon us and may be accepted by you at any time before the expiration of that period.

10. We hereby understand and confirm that NHPC/NHPC REL reserves its right to verify the documents furnished by us at the time of submission of RfS including availability of the Net Worth to the extent claimed in the RfS with the original documents and bank statements and the shareholding of the Project Company along with a copy of complete documentary evidence supported with originals at any stage from evaluation upto the expiry of PPA.

We understand and confirm that if the aforesaid documents furnished by us are found to be misleading or misrepresenting in any way, NHPC/NHPC REL shall be free to take appropriate action including forfeiture of EMD and blacklisting us for an appropriate period decided by NHPC/NHPC REL.

- 11. We, hereby, declare that only the persons or firms interested in this bid as named here and that no other persons or firms other than those mentioned herein have any interest in this bid or in the PPA to be entered into. We confirm that this bid is made without any connection with any other person, firm or party likewise submitting a bid. We further confirm that this bid is submitted in good faith and without collusion or fraud.
- 12. Contact Person

Details of the contact person are furnished as under:

Name<u>:...</u> Designation<u>:...</u> Company Address PhoneNos.<u>:...</u> Fax Nos. <u>:...</u>

<u>....</u> :...

E-mail address:

13. We confirm that we have understood that we shall approach concerned authorities/departments directly for allotment of land, timelines for availability, possession and connectivity for the Project(s) allotted to us, if any and that NHPC/NHPC REL shall not in any way, directly or indirectly, be responsible and

liable for these matters. We also confirm that we shall be overall responsible to complete all the activities related to Project Development at our own risk and cost.

- 14. We have neither made any statement nor provided any information in this Bid, the best of our knowledge is materially inaccurate or misleading. Further, all the confirmations, declarations and representations made in our Bid are true and accurate. In case this is found to be incorrect after our selection as Successful Bidder, we agree that the same would be treated as a Seller's event of default under PPA, and consequent provisions of PPA shall apply.
- 15. We confirm that we shall sign the Integrity pact with NHPC/ NHPC REL as per Format and submit certificates for compliance as per format of RfS document.
- 16. We hereby undertake that we shall pay the requisite Service / PMC charges prior to signing of PPA as per provision of RfS to NHPC REL in the payment of monthly bills to be made by NHPC REL, as part of payment security mechanism charges under the PPA, to avail the benefits under the payment security mechanism as per PPA.

Dated the_____ day of _____ , 20...

Thanking you,

Yours faithfully,

(Name, Designation and Signature of Person Authorized by the board)

Bid Security Forms Bank Guarantee (Refer ITB Clause 14)

(To be executed on Non-Judicial stamp paper of appropriate value)

Bank Guarantee No.

Date: _____

(Name of Contract)

To: (Name and address of NHPC/ NHPC REL)

WHEREAS (name of Bidder) (hereinafter called "the Bidder") has submitted its bid dated (date of bid) for the performance of the above-named Contract (hereinafter called "the Bid")

KNOW ALL PERSONS by these present that WE (name of Bank) of (address of bank) (hereinafter called "the Bank"), are bound unto NHPC Limited (A Government of India Enterprises) for the sum of: (amount), for which payment well and truly to be made to the said NHPC/NHPC REL, the Bank binds itself, its successors and assigns by these presents.

THE CONDITIONS of this obligation are as follows:

- 1. If the Bidder (a) withdraws or modifies its Bid during the period of bid validity or (b) adopts corrupt or collusive or coercive or fraudulent practices or defaults under the Integrity pact.
- 2. If the Bidder, having been notified of the acceptance of its Bid by the NHPC/ NHPC REL during the period of bid validity
 - a) fails or refuses to sign the PPA when required, or
 - b) fails or refuses to submit the performance security in accordance with the RFS documents.

WE undertake to pay to the NHPC/ NHPC REL up to the above amount upon receipt of its first written demand, without the NHPC/ NHPC REL having to substantiate its demand, provided that in its demand the NHPC/ NHPC REL will mention that the amount claimed by it is due, owing to the occurrence of one or both of the two above-named CONDITIONS, and specifying the occurred condition or conditions.

The Bank declares that this Bank Guarantee is issued by the bank, utilizing the credit limit of M/s-------(Name of Contractor)

This guarantee will remain in force up to and including (date 90 days after the period of bid validity), and any demand in respect thereof must reach the Bank not later than the above date.

Common Seal of the Bank with complete address including Tel/Fax Nos

Staff Authority No. of the officer of the Bank /Signatory

INSTRUCTIONS FOR EXECUTION OF BANK GUARANTEE FOR EARNEST MONEY DEPOSIT

- 1. Bank Guarantee for Earnest Money Deposit should be executed on non-judicial Stamp papers of requisite value in accordance with the stamp Act if applicable to that particular state of Indian Union country of executing Bank, where executed. In case the same is issued by a first class International bank, the law prevalent in the country of execution shall prevail for the purpose of Stamp Duty on the Bank Guarantee. However, in such a case, the Bank Guarantee for Earnest Money Deposit shall be got confirmed by the Bidder through any Indian Scheduled/Nationalized Bank.
- 2. The executing officers of the Bank Guarantee for Earnest Money/Bid Security shall clearly indicate in (block letters) his name, designation, Power of Attorney No. / Signing Power No. as well as telephone/ fax numbers with full correspondence address of the issuing Guarantee etc.
- 3. Each page of the Bank guarantee for Earnest Money Deposit shall be duly signed/initialed by the executing officers and the last page shall be signed in full, indicating the particulars as aforesaid (sub-para 2) under the seal of the Bank.
- 4. Stamp paper shall be purchased in the name of Bank issuing the Bank Guarantee, after the date 'Notice Inviting Tender', not more than six (6) months prior to execution/ issuance of the Bank Guarantee. The name of the purchaser should appear at the back side of stamp paper in the Vendors Stamp. The issuing Bank shall be requested independently for verification/confirmation of the Bank Guarantee issued, non-confirmation of which may lead to rejection of 'Bid Security'.
- 5. Irrevocable, valid and fully enforceable Bank Guarantee in favor of the NHPC/ NHPC REL (Name of NHPC/ NHPC REL) issued by any scheduled bank approved by the Reserve bank of India which is acceptable to the NHPC/ NHPC REL. The Bank Guarantee issued by a Foreign Bank shall be routed through the corresponding branch of such scheduled foreign banks in India or any scheduled Bank, acceptable to the NHPC/ NHPC REL.
- 6. Bank Guarantee for Bid security in original shall be submitted along with the Bid. However, the issuing Bank shall submit an unstamped duplicate copy of Bank Guarantee to the NHPC/ NHPC REL (authority inviting tenders) with a forwarding letter.

Format for POI for Earnest Money Deposit

FORMAT OF PAYMENT ON ORDER INSTRUMENT TO BE ISSUED BY IREDA/REC/PFC (IN LIEU OF BG TOWARDS EMD)

No.

Date

NHPC/ NHPC REL

,

Reg: M/s ______(insert name of the Bidder) – Issuance of Payment on Order Instrument for an amount of Rs.

Dear Sir,

- It is to be noted that M/s. _____ (*insert name of the POI issuing Agency*) ('IREDA/REC/PFC') has sanctioned a non-fund based limit loan of Rs. ___ (Rupees only) to M/s _____ under the Loan Agreement executed on ___ to execute Renewable Energy Projects.
- At the request of M/s ______, on behalf of ______ (insert name of the Bidder), this Payment on Order Instrument (POI) for an amount of Rs. ______ (Rupees(in words)). This Payment on Order Instrument comes into force immediately.

[Insert name of the Bidder].

4. In consideration of the above facts, IREDA/REC/PFC, having its registered office at _____, agrees to make payment for the sum of Rs. ______

lakhs (in words.....) to NHPC/NHPC REL on the following conditions:-

- (a) IREDA/REC/PFC agrees to make payment of the above said amount unconditionally, without demur and without protest within a period of _____ days of receipt of request from NHPC/NHPC REL within the validity period of this letter as specified herein;
- (b) The commitment of IREDA/REC/PFC, under this Payment of Order Instrument will have the same effect as that of the commitment under the Bank Guarantee issued by any Public Sector Bank and shall be enforceable in the same manner as in the case of a Bank Guarantee issued by a Bank and the same shall be irrevocable and shall be honored irrespective of any agreement or its breach between IREDA/REC/PFC or its constituents notwithstanding any dispute that may be raised by the against NHPC/NHPC REL;
- (c) The liability of IREDA/REC/PFC continues to be valid and binding on IREDA/REC/PFC and shall not be terminated, impaired and discharged, by virtue of change in its constitution and specific liability under letter of undertaking shall be binding on its successors or assignors;
- (d) The liability of IREDA/REC/PFC shall continue to be valid and binding on IREDA/REC/PFC and shall not be terminated/ impaired/ discharged by any extension of time or variation and alternation made given or agreed with or without knowledge or consent of the parties (NHPC/NHPC REL and Bidding Party), subject to the however to the maximum extent of amount stated herein and IREDA/REC/PFC is not liable to any interest or costs etc.;
- (e) This Payment of Order Instrument can be invoked either partially or fully, till the date of validity;
- (f) IREDA/REC/PFC agrees that it shall not require any proof in addition to the written demand by NHPC/NHPC REL made in any format within the validity period. IREDA/REC/PFC shall not require NHPC/NHPC REL to justify the invocation of the POI against the SPV/RTD, to make any claim against or any demand against the SPV/RTD or to give any notice to the SPV/RTD;
- (g) The POI shall be the primary obligation of IREDA/REC/PFC and NHPC/NHPC REL shall not be obliged before enforcing the POI to take any action in any court or arbitral proceedings against the SPV/RTD;
- (h) Neither NHPC/NHPC REL is required to justify the invocation of this POI nor shall IREDA/REC/PFC have any recourse against NHPC/NHPC REL in respect of the payment made under letter of undertaking;
- 5. Notwithstanding anything contrary contained anywhere in this POI or in any other

documents, this POI is and shall remain valid up to _____ and

IREDA/REC/PFC shall make payment thereunder only if a written demand or request is raised within the said date and to the maximum extent of Rs.

and IREDA/REC/PFC shall in no case, be liable for any interest, costs, charges and expenses and IREDA's/REC's/PFC's liability in no case will exceed more than the above amount stipulated.

Thanking you,

Yours faithfully

For and on behalf of

M/s. _____ (Name of the POI issuing agency).

()

General Manager (TS)

Copy to:-

M/s.

As per their request

) General Manager (TS)

(

Format of Insurance Surety Bond for Earnest Money Deposit (Refer ITB Clause 14)

(To be executed on Non-Judicial Stamp Paper of Appropriate value)

Insurance Surety Bond No..... Date:

(Name of Contract)

- To: (Name and address of NHPC/ NHPC REL)
- WHEREAS (name of Bidder) (hereinafter called "the Bidder") has submitted its Bid dated (date of bid) for the performance of the above named Contract (hereinafter called "the Bid")
- KNOW ALL PERSONS by these present that We (name of Insurance Company) of (address of Insurance Company) (hereinafter called "the Surety"), are bound unto NHPC Limited (a Govt. of India Enterprises) (hereinafter called "the NHPC/ NHPC REL") for the sum of (amount), for which payment well and truly to be made to the said NHPC/ NHPC REL, the Surety binds itself, its successors and assigns by these presents.

THE CONDITIONS of this obligation are as follows:

- 1. If the Bidder (a) withdraws or modifies its Bid during the period of bid validity, or (b) adopts corrupt or collusive or coercive or fraudulent practices or defaults under Integrity Pact.
- **2**. If the Bidder, having been notified of the acceptance of its Bid by the NHPC/ NHPC REL during the period of bid validity.
 - a) fails or refuses to sign the PPA when required, or
 - **b**) fails or refuses to submit the performance security in accordance with the RFS Documents.

We undertake to pay to the NHPC/ NHPC REL up to the above amount upon receipt of its first written demand, without the NHPC/ NHPC REL having to substantiate its demand, provided that in its demand the NHPC/ NHPC REL will mention that the amount claimed by it is due, owing to the occurrence of one or both of the two above-named CONDITIONS, and specifying the occurred condition or conditions.

The Surety declares that this Insurance Surety Bond is issued by the (name of Insurance Company), as per the applicable rules and regulations of Insurance Regulatory Development Authority of India (IRDAI).

This Insurance Surety Bond will remain in force up to and including (date 90 days after the period of bid validity), and any demand in respect thereof must reach the Surety not later than the above date.

For and on behalf of the Insurance Company

in the capacity of

Common Seal of the Insurance Company with complete address including Tel. Nos. / e-Mail Id.

Staff Authority No. of the officer of the Insurance Company /Signatory

<u>INSTRUCTIONS FOR EXECUTION OF INSURANCE SURETY BOND FOR EANEST</u> <u>MONEY DEPOSIT</u>

- 1. Insurance Surety Bond for Earnest Money Deposit should be executed on non-judicial Stamp papers of requisite value in accordance with the stamp Act if applicable to that particular state of Indian Union country of executing Insurance Company, where executed. In case the same is issued by an International Insurance Company (it should be registered under Insurance Act 1938 or as amended from time to time and approved by the Insurance Regulatory Development Authority of India (IRDAI)) the law prevalent in the country of execution shall prevail for the purpose of Stamp Duty on the Insurance Surety Bond. However, in such a case, the Insurance Surety Bond for Earnest Money Deposit shall be got confirmed by the Bidder through any Indian Scheduled/Nationalized Insurance Company.
- 2. The executing officers of the Insurance Surety Bond for Earnest Money/Bid Security shall clearly indicate in (block letters) his name, designation, Power of Attorney No. / Signing Power No. as well as telephone/ fax numbers with full correspondence address of the issuing Guarantee etc.
- 3. Each page of the Insurance Surety Bond for Earnest Money Deposit shall be duly signed/initialed by the executing officers and the last page shall be signed in full, indicating the particulars as aforesaid (sub-para 2) under the seal of the Insurance Company.
- 4. Stamp paper shall be purchased in the name of Insurance Company issuing the Insurance Surety Bond, after the date 'Notice Inviting Tender', not more than six (6) months prior to execution/ issuance of the Insurance Surety Bond. The name of the purchaser should appear at the back side of stamp paper in the Vendors Stamp. The issuing Insurance Company shall be requested independently for verification/confirmation of the Insurance Surety Bond issued, nonconfirmation of which may lead to rejection of 'Insurance Surety Bond'.

- 6. Irrevocable, valid and fully enforceable Insurance Surety Bond in favor of the NHPC/ NHPC REL (Name of NHPC/ NHPC REL) issued by any Insurance Company registered under Insurance Act 1938 or as amended from time to time and approved by the Insurance Regulatory Development Authority of India (IRDAI) in Indian currency (INR) only is acceptable to the NHPC/ NHPC REL.
- 6. Insurance Surety Bond for Bid security in original shall be submitted along with the Bid. However, the issuing Insurance Company shall submit an unstamped duplicate copy of Insurance Surety Bond directly by registered post (A.D.) to the NHPC/ NHPC REL (authority inviting tenders) with a forwarding letter.

Performance Security Forms

(To be submitted by the RTD)

Bank Guarantee Format for Performance Security

(To be executed on Non-Judicial Stamp Paper of Appropriate value)

Bank Guarantee No..... Date.....

To,

[NHPC/ NHPC REL Name & Address]

Dear Sirs,

In consideration of the ... [NHPC REL]...... (hereinafter referred to as the 'NHPC REL' which expression shall unless repugnant to the context or meaning thereof, include its successors, administrators and assigns) having awarded to M/s[RTD's Name]...... with its Registered/Head Office at (hereinafter referred to as the 'RTD, which expression shall unless repugnant to the context or meaning thereof, include its successors, administrators, executors and assigns), a RFS by issue of NHPC REL Letter of Acceptance No...... dated...... and the same having been acknowledged by the RTD, resulting in a Power Purchase Agreement (PPA) to be entered into with ______[Insert name of the Client Organization], hereinafter referred to as "Client", for purchase of Power [from selected Roof Top Developer (RTD), M/s ______]

As per the terms of the RfS, the ______ [Insert name & address of Bank] hereby agrees unequivocally, irrevocably and unconditionally to pay to NHPC REL at _____ [Insert Name of the Place from the address of the NHPC REL Organization] forthwith on demand in writing from NHPC REL or any Officer authorised by it in this behalf, any amount up to and not exceeding Indian Rupees ______ [Total Value] only, on behalf of M/s ______ [Insert name of the selected Roof Top Developer/Solar Power Developer/Project Company]

We[Name & Address of the Bank].....having Head Office its at.....(hereinafter referred to as the 'Bank', which expression shall, unless repugnant to the context or meaning thereof, include its successors, administrators, executors and assigns) do hereby guarantee and undertake to pay the HPC REL, on demand any and all monies payable by the Contractor to the extent of(*)...... as aforesaid at any time upto recourse or protest and/or without any reference to the RTD. Any such demand made by the NHPC REL on the Bank shall be conclusive and binding notwithstanding any difference between the NHPC REL and the RTD or any dispute pending before any Court, Tribunal, Arbitrator or any other authority. The Bank undertakes not to revoke this guarantee during its currency without previous consent of the NHPC REL and further agrees that the guarantees herein contained shall continue to be enforceable till the NHPC REL discharges this guarantee or till(+)..... [days/month/year] whichever is earlier.

(**)

The NHPC REL shall have the fullest liberty, without affecting in any way the liability of the Bank under this guarantee, from time to time to extend the time for performance of the Contract by the Contractor. The NHPC REL shall have the fullest liberty, without affecting this guarantee, to postpone from time to time the exercise of any powers vested in them or of any right which they might have against the RTD, and to exercise the same at any time in any manner, and either to enforce or to forbear to enforce any covenants, contained or implied, in the Contract between the NHPC REL and the Contractor or any other course or remedy or security available to the NHPC REL. The Bank shall not be released of its obligations under these presents by any exercise by the NHPC REL of its liberty with reference to the matters aforesaid or any of them or by reason of any other act or forbearance or other acts of omission or commission on the part of the NHPC REL or any other indulgence shown by the NHPC REL or by any other matter or thing whatsoever which under the law would, but for this provision have the effect of relieving the Bank.

The Bank declare that this Bank Guarantee is issued by the bank, utilizing the credit limit of M/s --------- (Name of RTD) also agrees that the NHPC REL at its option shall be entitled to enforce this Guarantee against the Bank as a principal debtor, in the first instance without proceeding against the RTD and notwithstanding any security or other guarantee the NHPC REL may have in relation to the RTD's liabilities.

- i) Our liability under this Bank Guarantee shall not exceed _____(*)_____.
- ii) This Bank Guarantee shall be valid upto _____(+)_____.
- iii) We are liable to pay the guaranteed amount or any part thereof under this Bank Guarantee only and only if NHPC REL serves upon Bank a written claim or demand on or before ----- (@)_____

WITNESS (Signature) (Name) (Name).

(Official Address) (Designation with Bank Stamp)/with staff Authority no.

Complete Address of the Bank with Tele-Fax

Notes :

1. (*) This sum shall be as per Clause 31.1 of ITB of this RfS.

(@) This date will be 12(Twelve) months beyond the Commissioning Date of the Project.

(+) This date will be 9 (Nine) months beyond the Scheduled Commencement of Supply Date of the Project.

- 2. Bank Guarantee should be executed on appropriate stamp paper of requisite value; such stamp paper should be purchased in the name of Issuing Bank, not more than six (6) months prior to execution / issuance of Bank Guarantee. The name of the purchaser should appear at the back side of stamp paper in the Vendors Stamp. Bank guarantee should contain rubber stamp of the authorized signatory of the bankindicating the name, designation and signature/ power of attorney number as well as telephone/ fax numbers with full correspondence address of the Bank.
 - 3. Bank Guarantee is required to be submitted directly to the NHPC REL by the issuing bank (on behalf of RTD) The RTD can submit an advance copy of Bank Guarantee to the NHPC REL.
 - 4. The issuing bank shall write the name of bank's controlling branch/ Head Office along with contact details like telephone/ fax and full correspondence address in order to get the confirmation of BG from that branch/ Head office, if so required.

Format for POI for Performance Bank Guarantee

FORMAT OF PAYMENT ON ORDER INSTRUMENT (POI) TO BE ISSUED BY IREDA/REC/PFC (IN LIEU OF PBG)

No.

,

Date

NHPC REL

Reg: M/s ______ (insert name of the PPA signing entity) (Project No.

(insert project ID issued by NHPC REL) – Issuance of Payment on Order Instrument for an amount of Rs.

Dear Sir,

- It is to be noted that M/s. _____ (insert name of the POI issuing Agency) ('IREDA/REC/PFC') has sanctioned a non-fund based limit loan of Rs. ___(Rupees _____only) to M/s _____ under the Loan Agreement executed on ____ to execute Renewable Energy Projects.
- 2. At the request of M/s ______, this Payment on Order Instrument (POI) for an amount of Rs._____(Rupees(in words)). This Payment on Order Instrument comes into force immediately.
- 3. In consideration of the ------ [Insert name of the RTD] (hereinafter referred to as

Selected Roof Top Developer) submitting the response to RfS inter alia for selection of Categories_____ kW, at [Insert name of the place] under RfS for (insert name of the RfS), for supply of power there from on long term basis, in response to the RfS dated_____ issued by NHPC REL (hereinafter referred to as NHPC REL) and NHPC REL considering such response to the RfS of ____ [insert the name of the selected Roof Top Developer] (which expression shall unless repugnant to the context or meaning thereof include its executers, administrators, successors and assignees) and selecting the Project of the Roof Top Developer (RTD) and issuing Letter of Award No to Name (Insert of selected Roof Top Developer) as per terms of RfS and the same having been accepted by the selected RTD resulting in a Power Purchase Agreement (PPA) to be entered into, for purchase of Power [from selected Roof Top Developer. As per the terms of the RfS, the [insert name & address of IREDA/PFC/REC] hereby agrees unequivocally, irrevocably and unconditionally to pay to NHPC REL at [Insert Name of the Place from the address of the NHPC REL] forthwith on demand in writing from NHPC REL or any Officer authorized by it in this behalf, any amount up to and not exceeding Rupees-[Total Value] only, on behalf of M/s [Insert name of the selected Roof Top

Developer Project Company].

4. In consideration of the above facts, IREDA/REC/PFC, having its registered office at ______, agrees to make payment for the sum of Rs. ______

lakhs (in words.....) to NHPC REL on the following conditions:-

- (a) IREDA/REC/PFC agrees to make payment of the above said amount unconditionally, without demur and without protest within a period of _____ days of receipt of request from NHPC REL within the validity period of this letter as specified herein;
- (b) The commitment of IREDA/REC/PFC, under this Payment of Order Instrument will have the same effect as that of the commitment under the Bank Guarantee issued by any Public Sector Bank and shall be enforceable in the same manner as in the case of a Bank Guarantee issued by a Bank and the same shall be irrevocable and shall be honored irrespective of any agreement or its breach between IREDA/REC/PFC or its constituents notwithstanding any dispute that may be raised by the against NHPC REL;
- (c) The liability of IREDA/REC/PFC continues to be valid and binding on IREDA/REC/PFC and shall not be terminated, impaired and discharged, by virtue of change in its constitution and specific liability under letter of undertaking shall be binding on its successors or assignors;
- (d) The liability of IREDA/REC/PFC shall continue to be valid and binding on IREDA/REC/PFC and shall not be terminated/ impaired/ discharged by any extension of time or variation and alternation made given or agreed with or without knowledge or consent of the parties (NHPC REL and Bidding Party), subject to the however to the maximum extent of amount stated herein and IREDA/REC/PFC is not liable to any interest or costs etc.;
- (e)This Payment of Order Instrument can be invoked either partially or fully, till the date of validity;
- (f) IREDA/REC/PFC agrees that it shall not require any proof in addition to the written demand by NHPC REL made in any format within the validity period. IREDA/REC/PFC shall not require NHPC REL to justify the invocation of the POI against the SPV/RTD, to make any claim against or any demand against the RTD or to give any notice to the RTD;
- (g) The POI shall be the primary obligation of IREDA/REC/PFC and NHPC REL shall not be obliged before enforcing the POI to take any action in any court or arbitral proceedings against the RTD;
- (h) Neither NHPC REL is required to justify the invocation of this POI nor shall IREDA/REC/PFC have any recourse against NHPC REL in respect of the payment made under letter of undertaking;
- 5. Notwithstanding anything contrary contained anywhere in this POI or in any other documents, this POI is and shall remain valid upto and IREDA/REC/PFC shall make payment thereunder only if a written demand or request is raised within the said date and to the maximum extent of Rs.....and IREDA/REC/PFC shall in no case, be liable for any interest, costs, charges and expenses and IREDA's/REC's/PFC's liability in no case will exceed more than the above amount stipulated.

Thanking you,

Yours faithfully

| | For and on behalf of |
|----------------------|-----------------------------------|
| M/s | |
| | (name of the POI issuing agency). |
| | () |
| | General Manager (TS) |
| Copy to:- | |
| M/s. | |
| As per their request | |

() General Manager (TS)

FORMAT OF INSURANCE SURETY BOND FOR PERFORMANCE SECURITY

(To be executed on Non-Judicial Stamp Paper of Appropriate value)

Insurance Surety Bond No..... Date.....

To,

[NHPC REL's Name & Address]

Dear Sirs,

- Wehaving its Head Office at.....(hereinafter referred to as the 'Surety, which expression shall, unless repugnant to the context or meaning thereof, include its successors, administrators, executors and assigns) do hereby guarantee and undertake to pay the NHPC REL, on demand any and all monies payable by the RTD to the extent of(*)..... aforesaid time as at any upto contest, recourse or protest and/or without any reference to the RTD. Any such demand made by the NHPC REL on the Insurance Company shall be conclusive and binding notwithstanding any difference between the NHPC REL and the RTD or any dispute pending before any Court, Tribunal, Arbitrator or any other authority. The Surety undertakes not to revoke this guarantee during its currency without previous consent of the NHPC REL and further agrees that the guarantees herein contained shall continue to be enforceable till the NHPC REL discharges this guarantee or till(+)...... [days/month/year] whichever is earlier.
- The NHPC REL shall have the fullest liberty, without affecting in any way the liability of the Insurance Company under this guarantee, from time to time to extend the time for performance of the Contract by the RTD. The NHPC REL shall have the fullest liberty, without affecting this guarantee, to postpone from time to time the exercise of any powers vested in them or of any right which they might have against the RTD, and to exercise the same at any time in any manner, and either to enforce or to forbear to enforce any covenants, contained or implied, in the Contract between the NHPC REL and the RTD or any other course or remedy or security available to the NHPC REL.

The Insurance Company shall not be released of its obligations under these presents by any exercise by the NHPC REL of its liberty with reference to the matters aforesaid or any of them or by reason of any other act or forbearance or other acts of omission or commission on the part of the NHPC REL or any other indulgence shown by the NHPC REL or by any other matter or thing whatsoever which under the law would, but for this provision have the effect of relieving the Insurance Company.

- The Surety declares that this Insurance Surety Bond is issued by the(Name of Insurance Company), as per the applicable rules and regulations of Insurance Regulatory Development Authority of India (IRDAI) and also agrees that the NHPC REL at its option shall be entitled to enforce this Guarantee against the Insurance Company as a principal debtor, in the first instance without proceeding against the RTD and notwithstanding any security or other guarantee the NHPC REL may have in relation to the RTD's liabilities.
- i) Our liability under this Insurance Surety Bond shall not exceed _____(*)____.
- ii) This Insurance Surety Bond shall be valid upto _____(+)____.
- iii) We are liable to pay the guaranteed amount or any part thereof under this Insurance Surety Bond only and only if NHPC REL serves upon Surety a written claim or demand on or before _____(@)_____

Dated thisday of20_....at....

WITNESS

.....

(Signature)

(Name)

(Signature).....

(Name).....

(Official Address)

.....

(Designation with Stamp of Insurance Company)/with staff Authority no.

Complete Address of the Insurance Company with Telephone No. and e-Mail Id.

Notes :

1. (*) This sum shall be as per Clause 31.1 of RfS.

(@) This date will be 12(Twelve) months beyond the Commissioning Date of the Project.

(+) This date will be 9 (Nine) months beyond the Scheduled Commencement of Supply Date of the Project.

2. Insurance Surety Bond should be executed on appropriate stamp paper of requisite value, such stamp paper should be purchased in the name of Issuing Insurance Company, not more than six (6) months prior to execution / issuance of Insurance Surety Bond. The name of the

purchaser should appear at the back side of stamp paper in the Vendors Stamp. Insurance Surety Bond should contain rubber stamp of the authorized signatory of the Insurance Company indicating the name, designation and signature/ power of attorney number as well as telephone numbers / e-Mail Id with full correspondence address of the Insurance Company.

- In case the same is issued by an International Insurance Company (it should be registered under Insurance Act 1938 or as amended from time to time and approved by the Insurance Regulatory Development Authority of India (IRDAI)), the law prevalent in the country of execution shall prevail for the purpose of Stamp Duty on the Insurance Surety Bond. However, in such a case, the Insurance Surety Bond shall be got confirmed through any Indian Scheduled/Nationalized Insurance Company.
- 3. Insurance Surety Bond is required to be submitted directly to the NHPC REL by the issuing Insurance Company (on behalf of RTD) under registered post (A.D.). The RTD can submit an advance copy of Insurance Surety Bond to the NHPC REL.
- 4. The issuing Insurance Company shall write the name of Insurance Company's controlling branch/ Head Office along with contact details like telephone no., e-Mail Id and full correspondence address in order to get the confirmation of Surety Bond from that branch/ Head office, if so required.

{Ref:ITB CI.34}

Format for Disclosure

(To be submitted on the Letter Head of the Bidding Company)

DISCLOSURE

Ref. No.

Date:

From: (Insert name and address of Bidding Company)

Tel. #: Fax#:

E-mail address#

То

(Enter Address of NHPC/NHPC REL)

Sub: Response to RfS No..... dated..... for.....

Dear Sir/ Madam,

We hereby declare and confirm that only we are participating in the RfS Selection process for the RfS No. and that our Parent, Affiliate or Ultimate Parent or any Group Company with which we have direct or indirect relationship are not separately participating in this selection process.

We further declare and confirm that in terms of the definitions of the RfS, M/s (enter name of the Promoter/Promoters) is/are our Promoter(s), and has/have a direct/indirect Control in the bidding company as per the Companies Act 2013. No other entity has a direct/indirect control in the bidding company except the entity(ies) mentioned above.

We further declare that the above statement is true & correct. We undertake that if at any stage it is found to be incorrect, in addition to actions applicable under the RfS/PPA including but not limited to cancellation of our response to this RfS and LoA/PPA as applicable, we, i.e. M/s (enter name of the bidding company), including our Parent, Ultimate Parent, and our Affiliates shall be suspended/debarred from participating in any of the upcoming tenders issued by NHPC/ NHPC REL for a period upto 2 years from the date of default as notified by NHPC/ NHPC REL.

We also understand that the above is in addition to the penal consequences that may follow from the relevant laws for the time being in force.

We further declare that we have read the provisions of Clause 34 of ITB of the RfS, and are complying with the requirements as per the referred OM dated 23.02.2023 except SI.17 of the OM, including subsequent amendments and clarifications thereto. Accordingly, we are also enclosing necessary certificates (Annexure to this format) in support of the above compliance under the RfS. We understand that in case of us being selected under this RfS, any of the above certificates is found false, NHPC/NHPC REL shall take appropriate action as deemed necessary.

We further declare that we are fully aware of the binding provisions of the ALMM Order and the Lists(s) thereunder, while quoting the tariff in RfS for (Enter the name of the RfS).

We further understand that the List-I (Solar PV Modules) of ALMM Order, Annexure-I of the OM, issued by MNRE on 10th March, 2021 will be updated by MNRE from time to time. We also understand that the Modules to be procured for this project, shall be from the List-I of the ALMM Order applicable on the date of invoicing of such modules.

We also further understand and accept that we shall be liable for penal action, including but not limited to blacklisting and invocation of Performance Bank Guarantee, if we are found not complying with the provisions of ALMM Order, including those mentioned above.

Dated the day of , 20 .

Thanking you,

Yours faithfully,

Name, Designation, Seal and Signature of Authorized Person in whose name Power of Attorney/ Board Resolution/ Declaration.

Annexure to Format for Disclosure

DECLARATION

RESTRICTION ON PROCUREMENT FROM CERTAIN COUNTRIES: MoF OM No 7/10/2021- PPD-(1) dated 23.02.2023

(To be submitted on the Letter Head of the Bidding Company)

Ref. No. Date:

From: (Insert name and address of Bidding Company)

Tel#:

Fax#:

E-mail address#

То

NHPC/ NHPC REL,

NHPC OFFICE COMPLEX,

Sector-33, Faridabad-121003,

Haryana, India.

Sub: Response to RfS No dated for the tender for

Dear Sir/ Madam,

This is with reference to attached order No. OM No. F.7/10/2021-PPD(1) dated 23.02.2023 issued by Department of Expenditure, MoF, Govt of India.

We are hereby submitting the following declaration in this regard:

.....

"I have read the clause regarding restrictions on procurement from a bidder of a country which shares a land border with India; I certify that this bidder is not from such a country or, if from such a country, has been registered with the Competent Authority. I hereby certify that this bidder fulfils all requirements in this regard and is eligible to be considered. Where applicable, evidence of valid registration by the Competent Authority shall be attached."

We further declare that the above statement is true & correct. We are aware that if at any stage it is found to be incorrect, our response to the tender will be rejected.

Dated the day of , 20....

Thanking you,

We remain,

Yours faithfully,

Name, Designation, Seal and Signature of Authorized Person in whose name Power of Attorney/ Board Resolution/ Declaration.

Encl: OM dated 23.07.2020, as referred above.

CERTIFICATE FOR COMPLIANCE TO ALL PROVISIONS OF RfS DOCUMENT

(Certificate of Compliance to All Provisions of RfS Document / "NIL" Deviation Certificate)

[To be submitted offline in a separate sealed envelope along with EMD and others]

Bidder's Name and Address :

To,

General Manager (Civil Contracts-I)

NHPC Limited, Room No.218,

2nd Floor, Jyoti Sadan

NHPC OFFICE COMPLEX,

Sector-33, Faridabad-121003,

Haryana, India

Dear Sir,

1. With reference to our Bid dated for "Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various location of JAKEDA and its departments/Bodies in RESCO Mode through Tariff-Based Competitive Bidding, RfS

No. ______, we hereby confirm that we have read the provisions of RfS document along with its subsequent Amendment(s) / Clarification(s) / Addenda/Errata and further confirm that our Bids (i.e. both Technical Bid and Financial Bid) are strictly in conformity with the provisions of the RfS Document including its Amendments / Clarifications / Errata / Revisions thereof and we have **not taken any deviation** to any of the provisions of the aforesaid RfS document.

- Further, we agree that the entire work for development of Solar Power Project(s) shall be performed as per the provisions of RfS document including Technical Requirements as per the RfS document.
- 3. We confirm that any deviation/ variation / additional conditions to the provisions of RfS Document read in conjunction with its Amendment(s)/Clarification(s)/ Addenda / Errata found anywhere in our Technical Bid and Financial Bid, implicit or explicit, save those pertaining to rebates offered, stands unconditionally withdrawn, without any cost implication whatsoever to the NHPC/NHPC REL, failing which the EMD be forfeited.
- 4. We hereby confirm that our Technical Bid does not contain any Financial/Price content entry. However if anything is contained, we shall be solely responsible for any implication.

5. Further, we confirm that our Financial Bid does not contain any matter in respect of Technical and / or Commercial aspects other than the details specifically sought in the Financial Bid.

Date :

| Signature of Authorized sig | natory |
|--------------------------------|--------|
| (Name of Authorized Signatory) | |

Place

| (Designation) | |
|---------------|--|
|---------------|--|

(Company Seal)

DECLARATION REGARDING INSOLVENCY PROCEEDINGS UNDER INSOLVENCY AND BANKRUPTCY CODE 2016

Subject: Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding.

It is declared that "no insolvency proceeding is admitted by the Adjudicating Authority against (name of the bidder / partner of JV / Parent Holding Company / Sub-Contractor) under the IBC 2016".

(To be submitted on letter head duly signed by the authorized representative of bidder/each partner of JV/Consortium/ Parent Company/Sub-contractor)

{REFER Clause 37 OF ITB}

Self-Certificate for Cyber Security

We declare that cyber security/ safety of the equipment/ process to be supplied / services to be rendered as safe to connect.

Note:

- 1. All the items/components/services specified under the Contract shall be supplied in compliance to cyber security policies/guidelines of the buyer organisation/GoI guidelines..
- 2. We shall strictly follow the order from Ministry of Power (Government of India) Circular No.25-11/6/2018-PG Dated 02.07.2020 and Circular No. 12/34/2020-T&R dated 08.06.2021 (along with its ammendments issued from time to time) during procurement of any equipments / hardware to be used in power supply system, network and/or associated IT / OT/ IoT infrastructure.

(Seal & Signature of Bidder)

Date:

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

PRICE SCHEDULES

PRICE SCHEDULE FOR CATEGORY-1

NAME OF WORK- Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding.

Tender Reference No.- NH/CCW/CC-I/CO-361/PR61324/7

Tender ID- 2025_NHPC_842440_1

CATEGORY-1

(This BOQ template must not be modified/ replaced by the bidder and the same should be uploaded after filling the relevent columns, else the bidder is liable to be rejected for this tender. Bidders are allowed to enter the Bidder Name and Values only) Notes:

1. Indicative Tariff in Rupees upto two decimal place per kWh (unit).

2. The Price quoted in this schedule shall be inclusive of all Taxes, Duties, Levies, Charges etc.

3. The bidders will have to submit bids quoting a fixed levelised tariff for entire project duration of 25 years.

| Name of the Bidder/ Bidding Firm / Company : | | | |
|--|-----------------------------------|--------------------------------|------------------------------|
| S.No. | Description (1) | Indicative Tariff (Rs./kWh) | Indicative Tariff (in Words) |
| 1 | Category-1 for (>= 5kW to < 20kW) | | |

PRICE SCHEDULE FOR CATEGORY-2

NAME OF WORK- Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding.

Tender Reference No.- NH/CCW/CC-I/CO-361/PR61324/7

Tender ID- 2025_NHPC_842440_1

CATEGORY-2

(This BOQ template must not be modified/ replaced by the bidder and the same should be uploaded after filling the relevent columns, else the bidder is liable to be rejected for this tender. Bidders are allowed to enter the Bidder Name and Values only) Notes:

1. Indicative Tariff in Rupees upto two decimal place per kWh (unit).

2. The Price quoted in this schedule shall be inclusive of all Taxes, Duties, Levies, Charges etc.

3. The bidders will have to submit bids quoting a fixed levelised tariff for entire project duration of 25 years.

| Name of the Bidder/ Bidding Firm / Company : | | | |
|--|-------------------------------------|--------------------------------|------------------------------|
| S.No. | Description (1) | Indicative Tariff (Rs./kWh) | Indicative Tariff (in Words) |
| 1 | Category-2 for (>= 20kW to < 100kW) | | |

PRICE SCHEDULE FOR CATEGORY-3

| & Parliamentary Affa | election of Roof Top Developer for Setting up Sola airs, Jal Shakti, Irrigation & Flood Control, For RESCO Mode on Tariff-Based Competitive Biddin | est, Ecology & Environme | • · · · |
|--|--|--|------------------------------|
| Tender Reference No | • NH/CCW/CC-I/CO-361/PR61324/7 | Tender ID- 2 | 025_NHPC_842440_1 |
| CATEGORY-3 | | | |
| else the bidder is liable Notes: 1. Indicative Tariff in F 2. The Price quoted in t | ust not be modified/ replaced by the bidder and the to be rejected for this tender. Bidders are allowed to Rupees upto two decimal place per kWh (unit). this schedule shall be inclusive of all Taxes, Duties, e to submit bids quoting a fixed levelised tariff for en | enter the Bidder Name and Levies , Charges etc. | Values only) |
| Name of the Bidder/ Bidding Firm / Company : | | | |
| S.No. | Description (1) | Indicative Tariff (Rs./kWh) | Indicative Tariff (in Words) |
| 1 | Category-3 for (>= 100kW and above) | | |

ECS FORM

ELECTRONIC CLEARING SERVICE (CREDIT CLEARING) (PAYMENT TO PARTIES THROUGH CREDIT CLEARING MECHANISM)

Г

| | | | N | 0. : | | | | | | |
|----|---|----|---|------|------|------|------|------|------|------|
| 1. | BIDDER'S NAME | : | - | | | | | | | |
| | Address | : | | | | | | | | |
| | | | | | | | | | | |
| | Phone/Mobile No. | : | | | | | | | | |
| 2. | PERTICULARS OF BANK ACCOUNT | : | | | | | | | | |
| A. | BANK NAME | : | | | | | | | | |
| B. | BRANCH NAME | | | | | | | | | |
| | Address | : | | | | | | | | |
| | Telephone No. | : | | | | | | | | |
| C. | IFSC code of the Bank | :[| | | | | | | | |
| | (For payments through RTGS) | | | | | | | | | |
| D. | ACCOUNT TYPE | : | | | | | | | | |
| | (S.B. Account/Current Account or Cash Credit with Code 10/11/13) | | | | | | | | | |
| E. | ACCOUNT NUMBER | : | | | | | | | | |
| | (As appearing on the Cheque Book) | | | | | | | | | |

I hereby declared that the particulars given above are correct and complete. If the transaction is delayed or not effected at all for reasons of incomplete or incorrect information, I would not hold the user Company responsible.

(.....)

Signature of the Bidder

Certified that the particulars furnished above are correct as per our records.

(Bank's Stamp)

| (| |) |
|---|---------------------|------|
| (| |) |
| (| • • • • • • • • • • | |

Signature of the Authorised Official from the Bank

Date:

Date:

12. (a) PROFORMA OF CERTIFICATE FROM THE CEO/CFO OF THE HOLDING COMPANY.

То

General Manager (Civil Contracts-I), NHPC Ltd., Room No.218, 2nd Floor, Jyoti Sadan, NHPC OFFICE COMPLEX, Sector-33, Faridabad-121003, Haryana, India Tele No :+91 (129) 2270596

Subject: Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding.

TENDER ID:

Dear Sirs,

- **3.0** I further, certify that the figures in the unaudited unconsolidated financial statements are true and correct and same have been duly reflected in the audited consolidated financial statements and / or Annual Report of the HoldingCompany.

| | Your | S | fa | ithfully | |
|--------------------------|-------|--------|-----|----------|----------|
| | (Sign | ature) | | | |
| Date : | (Name | | | | |
| &Designation) Place : | (Name | of | the | Holding | Company) |
| | | | | | |

Setting up Solar Roof Top Plants at Various Departments in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

(Seal of Holding Company)

Note :

*Strike off whichever is notapplicable.

12. (b) PROFORMA OF UNDERTAKING

(TO BE FURNISHED ON NON-JUDICIAL STAMP PAPER OF APPROPRIATE VALUE)

(To be executed by the Holding Company Supported by Board Resolution and submitted by the Bidder alongwith the Bid, Applicable to the Bidder for meeting the stipulated Financial Qualifying Requirement as per clause 3.2.B of Eligibility Criteria)

То

General Manager (Civil Contracts-I), NHPC Ltd., Room No.218, 2nd Floor, Jyoti Sadan, NHPC OFFICE COMPLEX, Sector-33, Faridabad-121003, Haryana, India Tele No :+91 (129) 2270596 Subject: Selection of Reef Ten Developer for

Subject: Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

Dear Sirs,

1.0 We, M/s..... declare that we are the holding company of M/s (Name of the Bidder) and have controlling interest therein.

- 3.0 This undertaking is irrevocable and unconditional, and shall remain in force till the successful execution and performance of the entire contract and/or till it is discharged by Employer.

4.0 We are herewith enclosing a copy of the Board Resolution in support of this undertaking.

Yours faithfully

(Signature of Authorised Signatory on behalf of the Holding Company)

Witness :

| (1) | (Name &Designation) |
|-----|------------------------------|
| | (Name of the HoldingCompany) |
| (2) | (Seal of the HoldingCompany) |
| | |

Date:

Place:

12. (c) PROFORMA OF CERTIFICATE FROM THE CEO/CFO IN ACCORDANCE WITH ITEM CLAUSE 3.2.B OF ELIGIBILITY CRITERIA

То

General Manager (Civil Contracts-I), NHPC Ltd., Room No.218, 2nd Floor, Jyoti Sadan, NHPC OFFICE COMPLEX, Sector-33, Faridabad-121003, Haryana, India Tele No :+91 (129) 2270596

Subject: Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding.

Dear Sirs,

| I, Mr (CEO/ CFO of the Comp | any) of | (Name | of |
|---|----------------------|------------|--------------|
| theCompany) Declare that the financial results | of M/s | (Nameof | |
| theBidder)are under audit as on | | (Date of | applications |
| for bid) and the Certificate from the practicing | | | |
| Chartered Accountant certifying the financial param | neters is also not a | vailable. | |
| | Yours | faithfully | |
| | (Signature) | | |
| Date : | (Name &Designati | on) | |
| Place : | (Name of the Com | pany) | |
| (Sea | ll of Company) | | ••••• |
| | | | |

Note : *Strike off if not applicable.

(d) JOINT LETTER OF UNDERTAKING BETWEEN THE CEO OF 12. THE HOLDING COMPANY / GROUP COMPANY OR SUBSIDIARY COMPANY & CEO OF THE BIDDER COMPANY IN ACCORDANCE WITH CLAUSE NO. 3.2.A OF ELIGIBILITY CRITERIA / QUALIFYING REQUIREMENTS.

(To be submitted by Bidder alongwith the bid)

То

General Manager (Civil Contracts-I), NHPC Ltd., Room No.218, 2nd Floor, Jyoti Sadan, NHPC OFFICE COMPLEX, Sector-33, Faridabad-121003, Haryana, India Tele No :+91 (129) 2270596

Subject: " Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding."

Dear Sirs,

I,Mr./Mrs(CEO of the Holding Company/Group Company)* declare thatM/s. (Name of the Holding Company/Group Company)* is theHolding Company/Group Company* of M/s. (Name of the Bidder Company) I hereby confirm that we meet qualifying requirement specified at Clause 3.2.A.1* / 3.2.A.2 * / **3.2.A.3 of Eligibility Requirement** prior to the last date of submission of bid. Furthermore, I confirm to extend full support for complete performance of the contract in case of award.

OR

I. Mr./Mrs (CEO of the Subsidiary Company)* declare that M/s.....(Name of theS ubsidiary company) is the Subsidiary Company of M/s. (Name of the Bidding Company) I hereby confirm that we meet qualifying requirement specified at Clause 3.2.A.1* / 3.2.A.2 */ 3.2.A.3*of Eligibility Requirement prior to the last date of submission of application for enlistment. Further more, I confirm to extend full support for complete performance of the contract in case of award.

1.0

| Yours faithfully | Yours faithfully |
|----------------------------------|------------------------------------|
| (Signature) | (Signature) |
| (Name& Designation) | (Name& |
| | Designation) |
| (Name of the Subsidiary Company) | (Name of the Holding Company/Group |
| | Company) |
| (Seal of Subsidiary Company) | (Seal of Holding Company) |

Date:

Place:

*Strike off whichever is not applicable.

(a) Bidders undertaking (on Letter head of the Company) mentioning the requisite Category for which Rfs(s) are submitted

То

General Manager (Civil Contracts-I), NHPC Ltd., Room No.218, 2nd Floor, Jyoti Sadan, NHPC OFFICE COMPLEX, Sector-33, Faridabad-121003, Haryana, India Tele No :+91 (129) 2270596

Subject: Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

Dear Sirs,

We, M/s..... declare that we have submitted the bid(s) for the following zone(s):

| Sl. No. | Category | Tentative Capacity (kW) | Tick the Applicable Category |
|---------|-------------------------------------|----------------------------|------------------------------------|
| 1 | Category-1 for (>= 5kW to < 20kW) | 9005.37 | |
| 2 | Category-2 for (>= 20kW to < 100kW) | 11965.31 | |
| 3 | Category-3 for (>= 100kW and above) | 13701.91 | |

Yours faithfully

(Signature of Authorised Signatory on behalf of the Company)

UNDERTAKING FROM PARENT/ HOLDING COMPANY

(On the letter head of Parent/Holding Company)

No:

Date:....

Name & address of the Employer

Sub: "Selection of Roof Top Developer for Setting up Solar Roof Top Plants at Various Departments of Law, Justice & Parliamentary Affairs, Jal Shakti, Irrigation & Flood Control, Forest, Ecology & Environment and Home in State of J&K through JAKEDA in RESCO Mode on Tariff-Based Competitive Bidding

Tender ID:

Dear Sirs,

- In case the Bidder, M/s(Name of Subsidiary company) gets qualified and awarded the work, We do hereby undertake:
- i) to enter into a separate agreement with the Employer as per the Employer's approved format included in the Bid documents.
 - We do hereby also confirm that we are not participating either as a sole Bidder or in another way against the above Invitation for Bids.

Yours faithfully,

For & on behalf of M/s. (Name & Address of the Parent/Holding Company)

(Office Seal)

Station: Date:

Note: 1. This Letter of undertaking should be on the letterhead of the Parent/Holding Company and should be signed by a person competent and having the Power of Attorney to bind the Parent/Holding Company. Power of Attorney in favour of this person to do so together with the authority of its executant be enclosed with this Letter of Undertaking.

2. In line with ITB clause if bidder is qualifying on the strength of Subsidiary company / Group Company under the same Apex/Parent company, then a commitment/ undertaking from such Subsidiary / Group Company for assured support and unhindered access to its assets and resources to the Bidder is also required to be submitted. This Letter of undertaking should be on the letterhead of the Subsidiary company/ Group Company and should be signed by a person competent and having the Power of Attorney to bind the Subsidiary company/ Group Company. Power of Attorney in favour of this person to do so together with the authority of its executant be enclosed with this Letter of Undertaking

PARENT/ HOLDING COMPANY AGREEMENT

THIS AGREEMENT IS MADE on this the..... th day of between the NHPC, a company incorporated under the laws of India and having its Registered Office at Faridabad India (hereinafter referred to as the 'Employer' which expression shall unless repugnant to the subject or context or meaning thereof include its successors, administrators, executors and assigns) of the one part; and(name of Parent/Holding Company) a company organized and existing under the laws of..... and having its Principal Office at (Hereinafter referred to as the "Parent/Holding Company" which expression shall unless repugnant to the subject or context or meaning thereof include its respective successors, administrators, executors and assigns) of the other part;

And whereas, in consideration of the aforesaid commitment, the Parent/Holding Company hereby enters into this agreement with the Employer for providing full support for technical and financial requirements to the Contractor and be responsible and liable for successful performance and completion of the works described in the said Contract on the following terms and conditions:

NOW THEREFORE THE PARTIES HERETO HEREBY AGREE AND THIS AGREEMENT WITNESSETH AS FOLLOWS:

- 1. In this Agreement except where the context otherwise requires, the following expressions shall have the meaning hereinafter respectively assigned to them:
- 2. "Contract" shall mean the Contract dated entered into between the Employer and the Contractor for the execution of the Work described as above.
- 3.(a) The Parent/Holding Company hereby agrees to the Employer to ensure due and faithful performance of the obligations and liabilities by the Contractor under the Contract and remain responsible to irrevocably and unconditionally provide full technical and financial support to the Contractor for completion of the works covered under the Contract. The provisions of Contract shall mutatis-mutandis apply to the Parent/Holding Company.
- 3. (b) In the event of breach and/ or failure on the part of the Contractor to perform or fulfill any of its obligations and liabilities under the Contract, the Employer may at its discretion call upon the Parent/Holding Company and the Parent/Holding Company shall be obliged to execute and perform or cause to be executed and performed and to satisfy the obligations and liabilities of the Contractor under the Contract in accordance with the terms and conditions thereof without prejudice to any other right or remedy, besides encashing the Bank guarantee(s).

- 3(c) The Parent/Holding Company shall indemnify and keep indemnified and harmless the Employer at all times against any loss, damage, cost charge and expense whatsoever that may be suffered or incurred by or caused to the Employer on account of such breach.
- 3(d) It shall not be necessary for the Employer to proceed against the Contractor before proceeding against the Parent/Holding Company and the Parent/Holding Company shall be liable to fulfill its obligations and liabilities hereunder notwithstanding the Employer having undertaken any proceedings and/or obtaining any security from the Contractor for the performance of its obligations under the Contract. In order to give affect to this Agreement, the Employer may at its option be entitled to act as if the Parent/Holding Company was the Contractor for successful completion of the works.

In the event, qualification of the Bidder/ Subsidiary Company is considered on the credentials of another Subsidiary/ Group Company under the same apex 'Parent Company' and due to any reason whatsoever, 'Parent Company' or any other group company wants to divest its investment in the direct or indirect subsidiary(ies) as a result of which any of these companies may not remain subsidiary(ies) of the 'Parent Company' then the 'parent Company' undertakes to ensure the performance of the works by arranging the required inputs in case of failure of Applicant subsidiary Company.

- 3(e) However, where the Bidder disputes the occurrence of a breach under the Contract and if such dispute is referred to Settlement of Disputes under General Conditions for the Contract Clause 6.0, the Contractor is obliged to carry on the works under the Contract. In case, during the pendency of the dispute in arbitration, the Contractor does not carry on the work satisfactorily, the Employer by notice to the Parent/Holding Company shall be entitled to invoke this Agreement, as if a breach had occurred for the purpose of Clause 3(b) hereinabove.
- 4. It is agreed that the obligations undertaken by the Parent/Holding Company hereunder shall be performed by it notwithstanding any difference or dispute between the Employer and the Contractor pending before any court, tribunal, arbitration or any other authority or forum.
- 5. This Agreement shall come into force and effect upon the Commencement Date of the Contract and shall remain in force and effective till the date of expiry of the Defects Liability Period by the Employer pursuant to the Conditions for the Contract.
- 6. This guarantee is in addition to and without prejudice to the securities offered by and on behalf of Contractor to the Employer and all rights and remedies in respect thereof be reserved. This guarantee shall be a continuing guarantee and be in force notwithstanding discharge of Contractor by operation of any law or insolvency/ bankruptcy / winding up/dissolution of the Contractor.
- 7. The Employer shall have the full liberty from time to time to vary any of the terms and conditions of the Contract by mutual agreement between the Employer and the Contractor and to extend time for performance thereunder by the Contractor or any other party thereto in accordance with the terms of the Contract and / or to postpone for any time and from time to time any of the powers exercisable by the Employer against the Contractor and either to enforce or forebear from enforcing any of the terms and conditions of the Contract and/or the securities available to the Employer from the Contractor and the Parent/Holding Company shall not be released from its obligations and liabilities under this Agreement in any manner whatsoever by any exercise by the Employer of the liberty and / or the rights with reference to the matters as aforesaid or by reason of time being given to the Contractor or any other act

of forbearance, waiver or omission on the part of Employer or any indulgence by the Employer to the Contractor or of any other matter or thing whatsoever which under the law relating to sureties would but for this provision have the effect of releasing the Parent/ Holding Company from its obligations and liabilities hereunder.

- 8. Nothing contained in this Agreement shall be construed or interpreted in any way as modifying or amending or relieving in any manner whatsoever the Contractor from their obligations under the Contract.
- 9. This Agreement shall be interpreted and be governed under the Law of India.
- 10. Any dispute or difference which may arise between the parties out of or in connection with this Agreement and which the Parties are unable to settle amicably shall be settled by reference to arbitration as per General/ Particular / Special Conditions of Contract. The venue of arbitration shall be New Delhi, India and the arbitration shall be conducted in accordance with the provisions of Arbitration and Conciliation Act 1996. The courts of Faridabad/Chandigarh shall have exclusive jurisdiction provided however that any award made in such arbitration shall be enforceable in any court of competent jurisdiction.
 - IN WITNESS WHEREOF THE PARTIES HERETO HAVE PUT THEIR HANDS HEREUNTO ON THE..... DAY MONTH ANDYEAR FIRST ABOVE WRITTEN AT FARIDABAD.
 - For and on behalf of the For and on behalf of the Parent/ Holding Company Employer (through authorized (through duly authorized representative) representative)

Witnesses:

Witnesses:

1 2.